UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Amendment No. 2 to FORM S-4

REGISTRATION STATEMENT

Under

The Securities Act of 1933

BANK OF AMERICA CORPORATION

(Exact Name of Registrant as Specified in its Charter)

Delaware (State or other jurisdiction of incorporation)

6021 (Primary Standard Industrial Classification Code Number) 56-0906609 (I.R.S. Employer Identification Number)

Bank of America Corporate Center 100 N. Tryon Street Charlotte, North Carolina 28255 (704) 386-8486

(Address, including Zip Code, and Telephone Number, including Area Code, of Registrant's Principal Executive Offices)

Paul J. Polking, Esq.
Executive Vice President and General Counsel
Bank of America Corporation
Bank of America Corporate Center
Charlotte, North Carolina 28255
(704) 386-5724

(Name, Address, including Zip Code, and Telephone Number, including Area Code, of Agent for Service)

With copies to:

Gary A. Spiess, Esq.
Executive Vice President,
General Counsel and Secretary
FleetBoston Financial Corporation
100 Federal Street
Boston, Massachusetts 02110
(617) 434-2870

Edward D. Herlihy, Esq. Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 (212) 403-1000

John C. Murphy, Jr., Esq. Cleary, Gottlieb, Steen & Hamilton 2000 Pennsylvania Avenue, NW Washington, DC 20006 (202) 974-1500

Approximate date of commencement of the proposed sale of the securities to the public: As soon as practicable after this Registration Statement becomes effective and upon completion of the merger described in the enclosed joint proxy statement/prospectus.

If the securities being registered on this Form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box.

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act of 1933, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with section 8(a) of the Securities Act of 1933 or until the registration statement shall become effective on such dates as the Commission, acting pursuant to said section 8(a), may determine.

Information contained herein is subject to completion or amendment. A registration statement relating to these securities has been filed with the Securities and Exchange Commission. These securities may not be sold nor may offers to buy be accepted prior to the time the registration statement becomes effective. This joint proxy statement/prospectus shall not constitute an offer to sell or the solicitation of any offer to buy nor shall there be any sale of these securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction.





MERGER PROPOSED—YOUR VOTE IS VERY IMPORTANT

The boards of directors of Bank of America Corporation and FleetBoston Financial Corporation each have unanimously approved a strategic merger that will combine our two franchises to create the first banking institution with a truly national scope, with an unrivaled presence in America's growth and wealth markets. After completion of the merger, we expect that current Bank of America stockholders will, as a group, own approximately 72% of the combined company and FleetBoston stockholders will, as a group, own approximately 28% of the combined company.

	Bank of America Common Stock	FleetBoston Common Stock	Implied Value per Share of FleetBoston Common Stock
At October 24, 2003	\$81.86	\$31.80	\$45.46
At [], 2004	\$[]	\$[]	\$[]

Since announcement of the merger, Bank of America common stock has generally traded below the closing price on the trading day preceding the announcement of the merger, and the market price of Bank of America common stock may fluctuate up or down prior to the merger, which will result in corresponding fluctuations in the implied value per share of FleetBoston common stock. The merger agreement does not include a price-based termination right or provisions that would limit the impact of increases or decreases in the market price of Bank of America common stock. You should obtain current market quotations for the shares of both companies.

We expect that the merger will generally be tax free to FleetBoston stockholders, except for taxes on cash received by FleetBoston stockholders instead of receiving fractions of shares of Bank of America common stock.

We cannot complete the merger unless the stockholders of both of our companies approve it. Each of us will hold a special meeting of our stockholders to vote on this merger proposal. **Your vote is important.** Whether or not you plan to attend your special stockholders' meeting, please take the time to vote your shares in accordance with the instructions contained in this document. Your failure to vote will have the same effect as voting against the merger. The places, dates and times of the special meetings are as follows:

For Bank of America stockholders:

the Bank of America board of directors unanimously recommends that
Bank of America stockholders vote
FOR
adoption of the merger agreement.

For FleetBoston stockholders:

the FleetBoston board of directors unanimously recommends that FleetBoston stockholders vote FOR approval of the merger agreement.

This document describes the special meetings, the merger, the documents related to the merger and certain other matters Please carefully read this entire document, including "Risk Factors Relating to the Merger" beginning on page 19 for a discussion of the risks relating to the merger. You also can obtain information about our companies from documents that we have filed with the Securities and Exchange Commission.

KENNETH D. LEWIS Chairman, President and Chief Executive Officer Bank of America Corporation CHARLES K. GIFFORD Chairman and Chief Executive Officer FleetBoston Financial Corporation

Bank of America common stock is quoted on the NYSE under the symbol "BAC." FleetBoston common stock is quoted on the NYSE under the symbol "FBF."

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved the Bank of America common stock to be issued under this joint proxy statement/prospectus or determined if this joint proxy statement/prospectus is accurate or adequate.

Any representation to the contrary is a criminal offense.

The date of this joint proxy statement/prospectus is [stockholders on or about [1, 2004.

], 2004, and it is first being mailed or otherwise delivered to Bank of America stockholders and FleetBoston

BANK OF AMERICA CORPORATION

Bank of America Corporate Center Charlotte, North Carolina 28255

NOTICE OF SPECIAL MEETING OF STOCKHOLDERS

Bank of America Corporation will hold a special meeting of stockholders in the Palmetto Ballroom of the International Trade Center, 200 North College Street, Charlotte, North Carolina, at 10:00 a.m., local time, on March 17, 2004 to consider and vote upon the following matters:

- a proposal to adopt the Agreement and Plan of Merger, dated as of October 27, 2003, by and between Bank of America Corporation and FleetBoston Financial Corporation, pursuant to which FleetBoston will merge with and into Bank of America;
- a proposal to adopt the Amended and Restated Bank of America 2003 Key Associate Stock Plan;
- a proposal to adopt an amendment to the Bank of America amended and restated certificate of incorporation, to increase the number of authorized shares of Bank of America common stock from 5 billion to 7.5 billion; and
- a proposal to approve the adjournment of the special meeting, if necessary, to solicit additional proxies, in the event that there are not sufficient votes at the time of the special meeting to approve the proposals.

The Bank of America board of directors has fixed the close of business on January 26, 2004 as the record date for the Bank of America special meeting. Only Bank of America stockholders of record at that time are entitled to notice of, and to vote at, the Bank of America special meeting, or any adjournment or postponement of the Bank of America special meeting. Holders of the Bank of America common stock, 7% Cumulative Redeemable Preferred Stock, Series B, which we refer to as the Series B Preferred Stock, and ESOP Convertible Preferred Stock, Series C, which we refer to as the ESOP Preferred Stock, vote together without regard to class and will be entitled to vote at the special meeting. A complete list of the Bank of America stockholders entitled to vote at the Bank of America special meeting will be made available for inspection by any Bank of America stockholder for ten days prior to the Bank of America special meeting at the principal executive offices of Bank of America, and at the time and place of the Bank of America special meeting. Each share of Bank of America common stock and Series B Preferred Stock is entitled to one vote, and each share of ESOP Preferred Stock is entitled to vote at the affirmative vote of a majority of the votes represented by the outstanding shares of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock entitled to vote at the special meeting, voting together without regard to class. Adoption of the Amended Stock Plan requires the affirmative vote of a majority of the votes cast at the special meeting by the holders of the Bank of America common stock, Series B Preferred Stock, voting together without regard to class.

Whether or not you plan to attend the special meeting, please submit your proxy with voting instructions. Please vote as soon as possible by accessing the Internet site listed on the Bank of America proxy card, by calling the toll-free number listed on the Bank of America proxy card, or by submitting your proxy card by mail. To submit your proxy by mail, please complete, sign, date and return the accompanying proxy card in the enclosed self-addressed, stamped envelope. This will not prevent you from voting in person, but it will help to secure a quorum and avoid added solicitation costs. Any holder of Bank of America common stock, Series B Preferred Stock or ESOP Preferred Stock who is present at the Bank of America special meeting may vote in person instead of by proxy, thereby canceling any previous proxy. In any event, a proxy may be revoked in writing at any time before the Bank of America special meeting in the manner described in the accompanying joint proxy statement/prospectus.

The Bank of America board of directors approved, by the unanimous vote of the directors present, the merger agreement and unanimously recommends that Bank of America stockholders vote "FOR" adoption of the merger agreement and the other proposals. Each of the proposals is independent, and is not contingent on approval by stockholders, of the other proposals.

BY ORDER OF THE BOARD OF DIRECTORS,

/s/ Ra	chel R. Cummings		
	R. Cummings ate Secretary		
[], 2004		

YOUR VOTE IS IMPORTANT. PLEASE COMPLETE, SIGN, DATE AND RETURN YOUR PROXY CARD WHETHER OR NOT YOU PLAN TO ATTEND THE SPECIAL MEETING.

FLEETBOSTON FINANCIAL CORPORATION

100 Federal Street Boston, Massachusetts 02110

NOTICE OF SPECIAL MEETING OF STOCKHOLDERS

FleetBoston Financial Corporation will hold a special meeting of stockholders in the Auditorium on the ground floor of The Federal Reserve Bank of Boston, 600 Atlantic Avenue, Boston, Massachusetts, at 10:00 a.m., local time, on March 17, 2004 to consider and vote upon the following matters:

- a proposal to approve the Agreement and Plan of Merger, dated as of October 27, 2003, by and between Bank of America Corporation and FleetBoston Financial Corporation, pursuant to which FleetBoston will merge with and into Bank of America; and
- a proposal to approve the adjournment of the special meeting, if necessary, to solicit additional proxies, in the event that there are not sufficient votes at the time of the special meeting to approve the proposal.

The FleetBoston board of directors has fixed the close of business on January 26, 2004 as the record date for the FleetBoston special meeting. Only FleetBoston stockholders of record at that time are entitled to notice of, and to vote at, the FleetBoston special meeting, or any adjournment or postponement of the special meeting. A complete list of FleetBoston stockholders entitled to vote at the special meeting will be made available for inspection by any FleetBoston stockholder at the time and place of the FleetBoston special meeting. In order for the merger agreement to be approved, the holders of a majority of the outstanding shares of FleetBoston common stock entitled to vote thereon must vote in favor of approval of the merger agreement.

Whether or not you plan to attend the special meeting, please submit your proxy with voting instructions. Please vote as soon as possible by accessing the Internet site listed on the FleetBoston proxy card, by calling the toll-free number listed on the FleetBoston proxy card, or by submitting your proxy card by mail. To submit your proxy by mail, please complete, sign, date and return the accompanying proxy card in the enclosed self-addressed, stamped envelope. This will not prevent you from voting in person, but it will help to secure a quorum and avoid added solicitation costs. Any holder of FleetBoston common stock who is present at the FleetBoston special meeting may vote in person instead of by proxy, thereby canceling any previous proxy. In any event, a proxy may be revoked in writing at any time before the FleetBoston special meeting in the manner described in the accompanying joint proxy statement/prospectus.

The FleetBoston board of directors has unanimously approved the merger agreement and unanimously recommends that FleetBoston stockholders vote "FOR" approval of the merger agreement.

/s/	Gary A. S	Spiess			
	ry A. Spies cretary	s			
Г		1 2004			

BY ORDER OF THE BOARD OF DIRECTORS,

YOUR VOTE IS IMPORTANT. PLEASE COMPLETE, SIGN, DATE AND RETURN YOUR PROXY CARD, WHETHER OR NOT YOU PLAN TO ATTEND THE SPECIAL MEETING.

REFERENCES TO ADDITIONAL INFORMATION

This document incorporates important business and financial information about Bank of America and FleetBoston from documents that are not included in or delivered with this document. You can obtain documents incorporated by reference in this document, other than certain exhibits to those documents, by requesting them in writing or by telephone from the appropriate company at the following addresses:

Bank of America Corporation

Bank of America Corporate Center Charlotte, North Carolina 28255 Attention: Investor Relations Telephone: (704) 386-5681 FleetBoston Financial Corporation

100 Federal Street
Boston, Massachusetts 02110
Mail Stop MA DE 10032F
Attention: Investor Relations
Telephone: (617) 434-7858

You will not be charged for any of these documents that you request. Bank of America stockholders and FleetBoston stockholders requesting documents should do so by March 10, 2004 in order to receive them before the special meetings.

See "Where You Can Find More Information" on page 111.

TABLE OF CONTENTS

	Page
QUESTIONS AND ANSWERS ABOUT VOTING PROCEDURES FOR THE SPECIAL MEETINGS	1
SUMMARY	3
SELECTED CONSOLIDATED HISTORICAL FINANCIAL DATA OF BANK OF AMERICA	13
SELECTED CONSOLIDATED HISTORICAL FINANCIAL DATA OF FLEETBOSTON	14
SELECTED CONSOLIDATED UNAUDITED PRO FORMA FINANCIAL INFORMATION	15
COMPARATIVE PER SHARE DATA	17
RISK FACTORS RELATING TO THE MERGER	19
CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS	21
THE BANK OF AMERICA SPECIAL MEETING	22
Matters to Be Considered Proxies Solicitation of Proxies Householding Record Date Voting Rights and Vote Required Recommendations of the Bank of America Board of Directors Attending the Meeting	22 22 23 23 23 24 24 24
THE FLEETBOSTON SPECIAL MEETING	26
Matters to Be Considered Proxies Solicitation of Proxies Record Date Voting Rights and Vote Required Recommendation of the FleetBoston Board of Directors Attending the Meeting	26 26 27 27 27 27 28 28
INFORMATION ABOUT THE COMPANIES	29
THE MERGER	30
General Structure Background of the Merger Bank of America's Reasons for the Merger; Recommendation of the Merger by the Bank of America Board of Directors FleetBoston's Reasons for the Merger; Recommendation of the Merger by the FleetBoston Board of Directors Analyses of Financial Advisors Opinion of Goldman Sachs Opinion of Morgan Stanley FleetBoston Bank of America	30 30 31 33 35 38 38 47 48 50
Board of Directors and Management of Bank of America following Completion of the Merger	53

i

Distribution of Bank of America Shares Fractional Shares Public Trading Markets Bank of America Dividends Appraisal Rights of Dissenting Stockholders	54 55 55 55 55
Regulatory Approvals Required for the Merger FleetBoston's Directors and Officers Have Financial Interests in the Merger THE MERGER AGREEMENT	59 61 66
Terms of the Merger Treatment of FleetBoston Stock Options and Other Equity-Based Awards Closing and Effective Time of the Merger Representations, Warranties, Covenants and Agreements Declaration and Payment of Dividends Agreement Not to Solicit Other Offers Expenses and Fees Conditions to Complete the Merger Amendment, Waiver and Termination of the Merger Agreement Resales of Bank of America Stock by Affiliates Employee Benefit Matters	66 66 67 67 69 69 70 71 71 72 72
The Stock Options Purpose of the Stock Option Agreements Exercise; Expiration Rights Under the Stock Option Agreements	73 73 73 73 75
ACCOUNTING TREATMENT	76
MATERIAL FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER	76
DESCRIPTION OF BANK OF AMERICA CAPITAL STOCK Description of Common Stock Description of Preferred Stock	78 78 79
COMPARISON OF STOCKHOLDERS' RIGHTS	84
General Preferred Stock Selected Provisions in the Restated Articles of Incorporation of FleetBoston	84 91 91
ADOPTION OF BANK OF AMERICA'S AMENDED STOCK PLAN	93
Number of Shares Types of Awards Administration Eligibility Awards of Stock Options and Stock Appreciation Rights Awards of Restricted Stock Shares and Restricted Stock Units Stock Plan Benefits Table 2002 Associates Stock Option Plan Take Ownership!	93 94 94 95 95 96 97 98
Barnett Employee Stock Option Plan	98

	Page
Internal Revenue Code Section 162(m)	98
Withholding for Payment of Taxes	99
Changes in Capitalization and Similar Changes	100
Changes in Control	100
Amendment and Termination of the Plan	100
No Repricings	100
Federal Income Tax Treatment	100
AMENDMENT TO THE BANK OF AMERICA AMENDED AND RESTATED CERTIFICATE OF INCORPORATION	102
COMPARATIVE MARKET PRICES AND DIVIDENDS	103
UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION	104
NOTES TO THE UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION	108
LEGAL MATTERS	113
<u>EXPERTS</u>	113
OTHER MATTERS	113
Bank of America 2004 Annual Meeting Stockholder Proposals	113
FleetBoston 2004 Annual Meeting Stockholder Proposals	114
ANNUAL MEETINGS	114
WHERE YOU CAN FIND MORE INFORMATION	114

APPENDICES:

	Page
APPENDIX A	
Agreement and Plan of Merger, dated as of October 27, 2003, by and between FleetBoston Financial Corporation and Bank of America Corporation	A-1
APPENDIX B	
Stock Option Agreement, dated as of October 27, 2003, between FleetBoston Financial Corporation, as issuer, and Bank of America Corporation, as grantee	B-1
APPENDIX C	
Stock Option Agreement, dated as of October 27, 2003, between Bank of America Corporation, as issuer, and FleetBoston Financial Corporation, as grantee	C-1
APPENDIX D	
Opinion of Goldman, Sachs & Co.	D-1
APPENDIX E	
Opinion of Morgan Stanley & Co. Incorporated	E-1
APPENDIX F	
Chapter 7-1.1-74 of the Rhode Island Business Corporation Act	F-1
APPENDIX G	
Section 262 of the Delaware General Corporation Law	G-1
APPENDIX H	
Bank of America's 2003 Key Associate Stock Plan, as Amended and Restated	H-1

QUESTIONS AND ANSWERS ABOUT VOTING PROCEDURES FOR THE SPECIAL MEETINGS

Q: What matters will be considered at the special meetings?

A: At the Bank of America special meeting, Bank of America stockholders will be asked to vote in favor of adopting the merger agreement. Bank of America stockholders will also be asked to vote in favor of adopting Bank of America's Amended Stock Plan, as well as a proposal to increase the number of authorized shares of Bank of America common stock. Each of the proposals is independent, and is not contingent on approval by stockholders, of the other proposals. At the FleetBoston special meeting, FleetBoston stockholders will be asked to vote in favor of adopting the merger agreement.

Q: What do I need to do now?

A: After you have carefully read this document and have decided how you wish to vote your shares, please vote by accessing the Internet site listed on your proxy card, by calling the toll-free number listed on your proxy card or by submitting your proxy card by mail. To submit your proxy card by mail, you must complete, sign, date and mail your proxy card in the enclosed postage paid return envelope as soon as possible. This will enable your shares to be represented and voted at the Bank of America special meeting or the FleetBoston special meeting.

Q: Why is my vote important?

A: The failure of a Bank of America or a FleetBoston stockholder to vote, by proxy or in person, will have the same effect as a vote against the merger and, for Bank of America stockholders, the proposal to increase the number of authorized shares of Bank of America common stock. In the case of Bank of America, the merger and the proposal to increase the number of authorized shares of Bank of America common stock must be approved by the holders of a majority of the votes represented by the outstanding shares of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock entitled to vote at its special meeting, voting together without regard to class. Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock, voting together without regard to class. In the case of FleetBoston, the merger must be approved by the holders of a majority of the outstanding shares of FleetBoston common stock entitled to vote at its special meeting.

Q: If my shares are held in street name by my broker, will my broker automatically vote my shares for me?

A: No. Your broker cannot vote your shares without instructions from you. You should instruct your broker as to how to vote your shares, following the directions your broker provides to you. Please check the voting form used by your broker.

Q: What if I abstain from voting or fail to instruct my broker?

A: If you abstain from voting or fail to instruct your broker to vote your shares and the broker submits an unvoted proxy, the resulting "broker non-vote" will be counted toward a quorum at your special meeting, but it will have the same effect as a vote against the merger. In the case of the Bank of America special meeting, abstentions and broker non-votes also will have the same effect as votes cast against the proposal to increase the number of authorized shares of Bank of America common stock. Each of the proposals is independent, and is not conditioned on approval by stockholders, of the other proposals. Abstentions from voting, as well as broker non-votes, are not treated as votes cast and, therefore, will have no effect on the proposal to approve Bank of America's Amended Stock Plan.

Q: Can I attend the special meeting and vote my shares in person?

A: Yes. All stockholders, including stockholders of record and stockholders who hold their shares through banks, brokers, nominees or any other holder of record, are invited to attend their respective special meeting. Stockholders of record can vote in person at the special meeting. If you are not a stockholder of record, you must obtain a proxy, executed in your favor, from the record holder of your shares, such as a

broker, bank or other nominee, to be able to vote in person at the special meeting. If you plan to attend the special meeting, you must hold your shares in your own name or have a letter from the record holder of your shares confirming your ownership, and you must bring a form of personal photo identification with you in order to be admitted. We reserve the right to refuse admittance to anyone without proper proof of share ownership and without proper photo identification.

Q: Can I change my vote?

A: Yes. You may revoke your signed proxy card at any time before it is voted by signing and returning a proxy card with a later date, delivering a written revocation letter to, in the case of Bank of America, Rachel R. Cummings, Corporate Secretary, or in the case of FleetBoston, Gary A. Spiess, Secretary, or by attending the appropriate special meeting in person, notifying Ms. Cummings or Mr. Spiess, as the case may be, and voting by ballot at the special meeting. Ms. Cummings's mailing address is Bank of America Corporation, NC1-007-56-11, 100 North Tryon Street, Charlotte, North Carolina 28255, and Mr. Spiess's mailing address is FleetBoston Financial Corporation, 100 Federal Street, MA DE 10026A, Boston, Massachusetts 02110. If you have voted your shares by telephone or through the Internet, you may revoke your prior telephone or Internet vote by recording a different vote using telephone or Internet voting, or by signing and returning a proxy card dated as of a date that is later than your last telephone or Internet vote.

Any stockholder entitled to vote in person at the appropriate special meeting may vote in person whether or not a proxy has been previously given, but the mere presence (without notifying Ms. Cummings or Mr. Spiess, as the case may be) of a stockholder at the special meeting will not constitute revocation of a previously given proxy.

Q: If I am a FleetBoston stockholder, should I send in my FleetBoston stock certificates now?

A: No. You should NOT send in your FleetBoston stock certificates at this time. We will send you instructions for exchanging FleetBoston stock certificates for shares of Bank of America stock. Unless FleetBoston stockholders specifically request to receive Bank of America stock certificates, the shares of Bank of America stock they receive in the merger will be issued in book-entry form. Bank of America stockholders do not need to exchange or take any other action regarding their Bank of America stock certificates in connection with the merger.

Q: When do you expect to complete the merger?

A: We expect to complete the merger in the first half of 2004. However, we cannot assure you when or if the merger will occur. We must first obtain the approvals of our stockholders at the special meetings and the necessary regulatory approvals.

Q: Whom should I call with questions?

A: Bank of America stockholders should call the Bank of America Investor Relations Department at (704) 386-5681 with any questions about the merger and related transactions.

FleetBoston stockholders should call the FleetBoston Investor Relations Department at (617) 434-7858 with any questions about the merger and related transactions.

SUMMARY

This summary highlights selected information from this document. It does not contain all of the information that is important to you. We urge you to carefully read the entire document and the other documents to which we refer in order to fully understand the merger and the related transactions. See "Where You Can Find More Information" on page 111. Each item in this summary refers to the page of this document on which that subject is discussed in more detail.

In the Merger FleetBoston Common Stockholders Will Have the Right to Receive 0.5553 of a Share of Bank of America Common Stock Per Share of FleetBoston Common Stock (page 29)

We are proposing the merger of FleetBoston with and into Bank of America. If the merger is completed, FleetBoston stockholders other than FleetBoston and Bank of America will have the right to receive 0.5553 of a share of Bank of America common stock for each share of FleetBoston common stock held immediately prior to the merger. We refer to this 0.5553-to-1 ratio as the exchange ratio. Upon completion of the merger, current holders of Bank of America common stock will, as a group, own approximately 72% of the outstanding common stock of the combined company and current holders of FleetBoston common stock will, as a group, own approximately 28% of the outstanding common stock of the combined company. Bank of America will not issue any fractional shares of Bank of America common stock in the merger. FleetBoston stockholders who would otherwise be entitled to a fractional share of Bank of America common stock will instead receive an amount in cash based on the average of the closing sale prices of Bank of America common stock on the five full trading days immediately prior to the date on which the merger is completed.

Example: If you hold 110 shares of FleetBoston common stock, you will receive 61 shares of Bank of America common stock and a cash payment instead of the 0.083 of a share of Bank of America common stock that you otherwise would have received (i.e., 110 shares x 0.5553 = 61.083 shares).

Tax-Free Transaction to FleetBoston Stockholders (page 74)

The merger has been structured to qualify as a reorganization for U.S. federal income tax purposes. Accordingly, holders of FleetBoston stock generally will not recognize any gain or loss for U.S. federal income tax purposes on the exchange of their FleetBoston stock for Bank of America stock in the merger, except for any gain or loss that may result from the receipt of cash instead of a fractional share of Bank of America common stock, and Bank of America and FleetBoston will not recognize any gain or loss for U.S. federal income tax purpose in the merger. It is a condition to our respective obligations to complete the merger that Bank of America receive the opinion of Cleary, Gottlieb, Steen & Hamilton, and FleetBoston receive the opinion of Wachtell, Lipton, Rosen & Katz, in each case, that the merger qualifies as a reorganization for U.S. federal income tax purposes. In connection with the filing of the registration statement, Cleary, Gottlieb, Steen & Hamilton, counsel to Bank of America, and Wachtell, Lipton, Rosen & Katz, counsel to FleetBoston, have delivered to Bank of America and FleetBoston, respectively, their opinions that the merger will qualify as a "reorganization" for United States federal income tax purposes.

The federal income tax consequences described above may not apply to all holders of FleetBoston stock. Your tax consequences will depend on your individual situation. Accordingly, we strongly urge you to consult your tax advisor for a full understanding of the particular tax consequences of the merger to you.

Comparative Market Prices and Share Information (page 100)

Bank of America common stock is quoted on the NYSE under the symbol "BAC." FleetBoston common stock is quoted on the NYSE under the symbol "FBF." The following table shows the closing sale prices of Bank of America common stock and FleetBoston common stock as reported on the NYSE on October 24, 2003, the last trading day before we announced the merger, and on [], 2004, the last practicable trading day before the distribution of this document. This table also shows the implied value of the merger consideration proposed

for each share of FleetBoston common stock, which we calculated by multiplying the closing price of Bank of America common stock on those dates by 0.5553, the exchange ratio

	Bank of America Common Stock	FleetBoston Common Stock	One Share of FleetBoston Common Stock
At October 24, 2003	\$ 81.86	\$ 31.80	\$ 45.46
At [], 2004	\$ []	\$ []	\$ []

Because the 0.5553 exchange ratio is fixed and will not be adjusted as a result of changes in the market price of Bank of America common stock, the implied value of the merger consideration will fluctuate with the market price of Bank of America common stock. The merger agreement does not include a price-based termination right or provisions that would limit the impact of increases or decreases in the market price of Bank of America common stock. You should obtain current market quotations for the shares of both companies from a newspaper, the internet or your broker. In addition, set forth below is a table showing the implied value of the merger consideration to FleetBoston stockholders based on a range of hypothetical Bank of America common stock prices. This table is for illustrative purposes only, and the actual prices at which shares of Bank of America common stock may trade between the date hereof and closing and thereafter may be above or below the range set forth below.

Bank of America Common Stock Hypothetical Value	Implied Value of One Share of FleetBoston Common Stock
\$65	\$36.09
\$70	\$38.87
\$75	\$41.65
\$80	\$44.42
\$85	\$47.20
\$90	\$49.98

Bank of America anticipates that it will repurchase approximately 67 million shares through 2004 and 23 million shares in 2005, net of option exercises. Actual share repurchases may be more or less than anticipated due to various factors including capital requirements, market conditions and legal considerations affecting the amount and timing of repurchase activity.

Goldman Sachs Provided an Opinion to the Bank of America Board of Directors that the Exchange Ratio was Fair from a Financial Point of View to Bank of America (page 38)

In connection with the merger, Bank of America retained Goldman, Sachs & Co., sometimes referred to as Goldman Sachs, and Banc of America Securities, LLC as financial advisors. In deciding to approve the merger, the Bank of America board of directors considered the oral opinion of Goldman Sachs provided to the Bank of America board of directors on October 26, 2003, subsequently confirmed in writing, that, as of the date of the opinion and based upon and subject to the considerations described in its opinion and other matters as Goldman Sachs considered relevant, the exchange ratio was fair to Bank of America from a financial point of view. A copy of the opinion is attached to this document as Appendix D. Bank of America stockholders should read the opinion completely and carefully to understand the assumptions made, matters considered and limitations on the review undertaken by Goldman Sachs in providing the opinion. The opinion of Goldman Sachs will not reflect any developments that may occur or may have occurred after the date of its opinion and prior to the completion of the merger. Bank of America does not currently expect that it will request an updated opinion from Goldman Sachs. Bank of America agreed to pay a transaction fee of up to \$25 million to Goldman Sachs for its services in

connection with the merger. The principal portion of this fee is payable upon completion of the merger. The Goldman Sachs opinion is not a recommendation as to how any stockholder of Bank of America should vote with respect to the merger or any other matter. Banc of America Securities did not provide an opinion to the Bank of America board and did not receive fees in connection with the transaction.

Morgan Stanley Provided an Opinion to the FleetBoston Board of Directors that the Exchange Ratio was Fair from a Financial Point of View to Holders of FleetBoston Common Stock (page 47)

In deciding to approve the merger, the FleetBoston board of directors considered the opinion of its financial advisor, Morgan Stanley & Co. Incorporated, sometimes referred to as Morgan Stanley, provided to the FleetBoston board of directors on October 26, 2003, that as of the date of the opinion, and based on and subject to the considerations in its opinion, the exchange ratio in the merger agreement was fair from a financial point of view to holders of FleetBoston common stock. A copy of the opinion is attached to this document as Appendix E. FleetBoston stockholders should read the opinion completely and carefully to understand the assumptions made, matters considered and limitations of the review undertaken by Morgan Stanley in providing its opinion. The opinion of Morgan Stanley will not reflect any developments that may occur or may have occurred after the date of its opinion and prior to the completion of the merger. FleetBoston does not currently expect that it will request an updated opinion from Morgan Stanley. FleetBoston agreed to pay Morgan Stanley a transaction fee of up to \$25 million in connection with the merger. The principal portion of such fee is payable upon completion of the merger. The Morgan Stanley opinion is not a recommendation as to how any stockholder of FleetBoston should vote with respect to the merger or any other matter.

Bank of America Currently Expects to Continue to Pay a Quarterly Cash Dividend of at Least \$0.80 per Share (page 53)

During 2002, Bank of America paid cash dividends on its common stock totaling \$2.44 per share. Bank of America currently expects to continue to pay a quarterly dividend of at least \$0.80 per share of common stock. Based on the 0.5553 exchange ratio and this annual dividend rate, holders of FleetBoston common stock would experience an anticipated annual dividend rate increase of approximately 27% (from \$1.40 to \$1.78 per share of FleetBoston common stock equivalent). Although there is no present plan or intention to decrease this dividend, the Bank of America board of directors may, subject to applicable law, change this dividend amount at any time, and Bank of America's ability to pay dividends on its common stock is subject to various legal and regulatory limitations.

Our Reasons for the Merger (pages 33 and 35)

Our companies are proposing to merge because, among other things, we believe that both sets of stockholders will benefit from the facts that:

- · the merger will create the first banking institution with a truly national scope, with an unrivaled presence in America's growth and wealth markets;
- the merger would establish a new Bank of America that will serve approximately 33 million consumer relationships, with leading market shares throughout the Northeast, Southeast, Midwest, Southwest and West regions of the United States; and
- the resulting banking franchise will be America's best, with unrivaled retail distribution in the most attractive markets.

The Bank of America Board of Directors Recommends that Bank of America Stockholders Vote "FOR" Adoption of the Merger Agreement, the Increase in the Number of Authorized Shares and the Amended Stock Plan (page 23)

The Bank of America board of directors believes that the merger is in the best interests of Bank of America and its stockholders and has unanimously approved the merger agreement. The Bank of America board of

directors unanimously recommends that Bank of America stockholders vote "FOR" adoption of the merger agreement. The Bank of America board of directors has also unanimously approved the proposals to increase the number of authorized shares of Bank of America common stock and to adopt the Amended Stock Plan. The Bank of America board of directors unanimously recommends that Bank of America stockholders vote "FOR" the proposals to increase the number of authorized shares of Bank of America common stock and to adopt the Amended Stock Plan.

The FleetBoston Board of Directors Recommends that FleetBoston Stockholders Vote "FOR" Approval of the Merger Agreement (page 27)

The FleetBoston board of directors believes that the merger is in the best interests of FleetBoston and its stockholders and has unanimously approved the merger agreement. The FleetBoston board of directors unanimously recommends that FleetBoston stockholders vote "FOR" approval of the merger agreement.

FleetBoston's Directors and Officers Have Financial Interests in the Merger That Differ From Your Interests (page 59)

FleetBoston's executive officers and directors have economic interests in the merger that are different from, or in addition to, their interests as FleetBoston stockholders. The FleetBoston board of directors considered these interests in its decision to approve the merger agreement. Several executive officers of FleetBoston have existing agreements that provide severance benefits in the case of qualifying terminations of employment following a change in control of FleetBoston. As further described on page 59, in the event that a qualifying termination of employment were to occur immediately following the completion of the merger, the amount of the cash severance payable under the existing FleetBoston agreements is currently estimated to be \$0 for Mr. Gifford (due to the reduction described on page 59), \$7,850,000 for Mr. McQuade, \$7,400,000 for Mr. Sarles, \$0 for Mr. Warner (due to the reduction described on page 59) and \$5,025,000 for Mr. Moynihan, although at completion of the merger the right of each of Messrs. McQuade, Sarles and Moynihan to this cash severance will be replaced by the right to receive a deferred amount under his new employment agreement with Bank of America. As further described on page 59, in the event that qualifying terminations of employment were to occur immediately following the completion of the merger, the aggregate amount of the cash severance payable under the existing FleetBoston agreements with the ten other FleetBoston executive officers is currently estimated to be approximately \$30.2 million.

Four of FleetBoston's executive officers — Eugene McQuade, Brian Moynihan, Jay Sarles and Bradford Warner — have entered into employment agreements with Bank of America that provide for their employment after completion of the merger. Under the new employment agreements with Bank of America, the annual base salary payable to each of Messrs. McQuade, Sarles, Warner and Moynihan will be \$1,250,000, \$850,000, \$700,000 and \$600,000, respectively. Some of FleetBoston's compensation and benefits plans, including its equity award plans, provide for the payment or accelerated vesting of rights or benefits upon a transaction such as the merger. Also, Messrs. Gifford and McQuade will join the board of directors of the combined company, with Mr. Gifford serving as its Chairman.

FleetBoston officers and directors who serve on the combined company's board of directors are expected to be compensated for their services in that capacity in accordance with Bank of America's standard director compensation policy. While subject to modification by the board of directors, currently this policy provides for an annual retainer of \$100,000, payable in cash and restricted stock, and an attendance fee of \$1,500 for each meeting of the board or committee of the board, in each case for non-employee directors. In addition, non-employee directors will be eligible to receive restricted stock under Bank of America's Directors' Stock Plan based on a total award value determined from time to time by the Bank of America board of directors. Non-employee directors who chair certain committees also receive a committee chairperson annual retainer

(currently \$20,000 for the chairpersons of the asset quality review committee, audit committee, compensation committee and executive committee, and \$10,000 for the chairperson of the corporate governance committee). Consistent with Bank of America's standard director compensation policy, it is anticipated that non-employee directors of FleetBoston who become directors of the combined company will receive remuneration for the period served as directors between the merger date and the next annual meeting of stockholders, and employee directors will receive no additional compensation for their service as directors.

Holders of Bank of America Preferred Stock Have Appraisal Rights (page 53)

Bank of America is incorporated in Delaware. Under Delaware law, holders of Bank of America common stock do not have any right to dissent from the merger or to a related court determination, in a proceeding known as an appraisal, of the fair value of their shares in connection with the merger. However, holders of Bank of America Series B Preferred Stock and ESOP Preferred Stock are entitled to appraisal rights in connection with the merger.

Holders of FleetBoston Series VII Preferred Stock Have Appraisal Rights (page 53)

FleetBoston is incorporated in Rhode Island. Under Rhode Island law, holders of FleetBoston common stock do not have any right to dissent from the merger or to a related court determination, in a proceeding known as an appraisal, of the fair value of their shares in connection with the merger. However, holders of FleetBoston Series VII Fixed/Adjustable Rate Cumulative Preferred Stock are entitled to appraisal rights in connection with the merger.

Information about the Companies (page 28)

Bank of America Corporation

Bank of America Corporation is a Delaware corporation, a bank holding company and a financial holding company under U.S. federal law. Through its banking and various nonbanking subsidiaries, Bank of America provides a diversified range of banking and nonbanking financial services and products, primarily throughout the Mid-Atlantic (Maryland, Virginia and the District of Columbia), the Midwest (Illinois, Iowa, Kansas and Missouri), the Southeast (Florida, Georgia, North Carolina, South Carolina and Tennessee), the Southwest (Arizona, Arkansas, New Mexico, Oklahoma and Texas), the Northwest (Oregon, Idaho and Washington) and the West (California and Nevada) regions of the United States and in selected international markets. As of September 30, 2003, Bank of America had total consolidated assets of approximately \$737.1 billion, total consolidated deposits of approximately \$408.5 billion and total consolidated stockholders' equity of approximately \$50.4 billion. The principal executive offices of Bank of America are located in the Bank of America Corporate Center, 100 N. Tryon Street, Charlotte, North Carolina 28255, and our telephone number is (704) 386-8486.

FleetBoston Financial Corporation

FleetBoston Financial Corporation is a Rhode Island corporation and a diversified financial services company offering a comprehensive array of financial solutions to its customers. FleetBoston's three major domestic business lines are Personal Financial Services, Regional Commercial Financial Services and Investment Management, and National Commercial Financial Services. FleetBoston's other lines of business are International Banking and Capital Markets. As of September 30, 2003, FleetBoston had total consolidated assets of approximately \$196.4 billion, total consolidated deposits of approximately \$132.5 billion and total consolidated stockholders' equity of approximately \$17.6 billion. The principal executive offices of FleetBoston are located at 100 Federal Street, Boston, Massachusetts 02110, and our telephone number is (617) 434-2200.

Conditions That Must Be Satisfied or Waived for the Merger to Occur (page 68)

Currently, we expect to complete the merger in the first half of 2004. However, as more fully described in this document and in the merger agreement, the completion of the merger depends on a number of conditions being satisfied or, where legally permissible, waived. These conditions include, among others, approval by the stockholders of each company, the receipt of all required regulatory approvals, such as approval by the Board of Governors of the Federal Reserve System, and the receipt of legal opinions by each company regarding the tax treatment of the merger.

We cannot be certain when, or if, the conditions to the merger will be satisfied or waived, or that the merger will be completed.

Termination of the Merger Agreement (page 68)

We may agree to terminate the merger agreement before completing the merger, even after stockholder approval, as long as the termination is approved by each of our boards of directors.

In addition, either of us may decide to terminate the merger agreement, even after stockholder approval, if certain conditions have not been met, such as obtaining the necessary regulatory approvals or the other company's material breach of a representation, warranty or covenant. Either of us may terminate the merger agreement if the merger has not been completed by October 27, 2004, unless the reason the merger has not been completed by that date is a breach of the merger agreement by the company seeking to terminate the merger agreement. If either of us fails to obtain stockholder approval of the merger, we may not terminate the merger agreement before October 27, 2004 as long as both of us continue to make specified efforts to restructure the transaction for the purpose of presenting it to the stockholders again for approval.

Board of Directors and Management of Bank of America following Completion of the Merger (page 50)

After the merger, the board of directors of the combined company will have a total of nineteen members, twelve of whom will be individuals who are currently Bank of America directors and seven of whom will be individuals who are currently FleetBoston directors or officers. Charles K. Gifford will be the Chairman of the Board of the combined company. Kenneth D. Lewis will be the Chief Executive Officer, Eugene M. McQuade will be the President and James H. Hance, Jr. will be the Vice Chairman and Chief Financial Officer.

Following completion of the merger, executive officers of the combined company will comprise an eleven-member Risk and Capital Committee, chaired by the Chief Executive Officer, which will be composed of seven current Bank of America executives and four current FleetBoston executives.

The Rights of FleetBoston Stockholders will be Governed by Delaware Law and by New Governing Documents after the Merger (page 82)

The rights of FleetBoston stockholders will change as a result of the merger due to differences in Bank of America's and FleetBoston's governing documents and due to the fact that the companies are incorporated in different states (FleetBoston in Rhode Island and Bank of America in Delaware). Bank of America's stockholders' rights will not change as a result of the merger. This document contains descriptions of stockholder rights under each of the Bank of America and FleetBoston governing documents and applicable state law, and describes the material differences between them, including the following:

• Amendments to the Bank of America amended and restated certificate of incorporation and amendments by stockholders to the Bank of America bylaws require an affirmative vote of a majority of its outstanding voting shares, while amendments to the FleetBoston restated articles of incorporation and

bylaws relating to various matters (such as the number of directors and business combinations with significant shareholders) require a supermajority vote of such shares:

- Bank of America's stockholder notice procedures require that a stockholder's director nominations must be received by the Secretary of Bank of America not later than the close of business on the 75th day nor earlier than the close of business on the 120th day prior to the first anniversary of the date Bank of America commenced mailing its proxy materials for the preceding year's annual meeting. FleetBoston's stockholder notice procedures require that a stockholder's director nominations must be received by the Secretary of FleetBoston not less than 30 days prior to the date of the meeting of FleetBoston stockholders;
- Bank of America does not have a classified board of directors, whereas FleetBoston's board of directors is divided into three classes, with each class being elected for a three year term;
- Directors of Bank of America may be removed by shareholders with or without cause, whereas directors of FleetBoston may only be removed for cause by a vote of holders of 80% or more of the outstanding shares; and
- The Bank of America board of directors has adopted a policy of requiring stockholder approval for severance agreements with senior executives that provide severance benefits in amounts exceeding two times the sum of any senior executive's base salary and bonus, as well as a policy of requiring stockholder approval prior to the adoption of any stockholder rights plan. The FleetBoston board of directors has not adopted any specific policies requiring stockholder approval prior to these actions by the board.

FleetBoston Granted a Stock Option to Bank of America (page 70)

To induce Bank of America to enter into the merger agreement, FleetBoston granted Bank of America an option to purchase up to 209,496,275 shares of FleetBoston common stock at a price per share of \$31.80; however, in no case may Bank of America acquire more than 19.9% of the outstanding shares of FleetBoston common stock under this stock option agreement. Bank of America cannot exercise the option unless the merger is not completed and specified triggering events occur. These events generally relate to business combinations or acquisition transactions involving FleetBoston and a third party. We do not know of any event that has occurred as of the date of this document that would allow Bank of America to exercise the option.

The option could have the effect of discouraging a company from trying to acquire FleetBoston prior to completion of the merger or termination of the merger agreement. Upon the occurrence of certain triggering events, FleetBoston may be required to repurchase the option and any shares of FleetBoston common stock purchased under the option at a predetermined price, or Bank of America may choose to surrender the option to FleetBoston for a cash payment of \$1.8948 billion. In no event will the total profit received by Bank of America with respect to this option exceed \$2.3685 billion.

The FleetBoston stock option agreement is attached to this document as Appendix B.

Bank of America Granted a Stock Option to FleetBoston (page 70)

To induce FleetBoston to enter into the merger agreement, Bank of America granted FleetBoston an option to purchase up to 297,374,945 shares of Bank of America common stock at a price per share of \$81.86; however, in no case may FleetBoston acquire more than 19.9% of the outstanding shares of Bank of America common stock under this stock option agreement. FleetBoston cannot exercise the option unless the merger is not completed and specified triggering events occur. These events generally relate to business combinations or

acquisition transactions involving Bank of America and a third party. We do not know of any event that has occurred as of the date of this document that would allow FleetBoston to exercise the option.

The option could have the effect of discouraging a company from trying to acquire Bank of America prior to completion of the merger or termination of the merger agreement. Upon the occurrence of certain triggering events, Bank of America may be required to repurchase the option and any shares of Bank of America common stock purchased under the option at a predetermined price, or FleetBoston may choose to surrender the option to Bank of America for a cash payment of \$1.8948 billion. In no event will the total profit received by FleetBoston with respect to the option exceed \$2.3685 billion.

The Bank of America stock option agreement is attached to this document as Appendix C.

Regulatory Approvals Required for the Merger (page 56)

We have agreed to use our reasonable best efforts to obtain all regulatory approvals required to complete the transactions contemplated by the merger agreement. These approvals include approval from the Federal Reserve Board and various federal, state and foreign regulatory authorities. Bank of America and FleetBoston have completed, or will complete, the filing of applications and notifications to obtain the required regulatory approvals.

Although we do not know of any reason why we cannot obtain these regulatory approvals in a timely manner, we cannot be certain when or if we will obtain them.

Bank of America will Hold its Special Meeting on March 17, 2004 (page 21)

The Bank of America special meeting will be held on March 17, 2004, at 10:00 a.m., local time, in the Palmetto Ballroom of the International Trade Center, 200 North College Street, Charlotte, North Carolina. At the Bank of America special meeting, Bank of America stockholders will be asked to:

- · adopt the merger agreement;
- · adopt Bank of America's Amended Stock Plan;
- approve the proposal to increase the number of authorized shares of Bank of America common stock from 5 billion to 7.5 billion; and
- approve the adjournment of the special meeting, if necessary, to solicit additional proxies, in the event that there are not sufficient votes at the time of the special meeting to approve the proposals.

Record Date. Only holders of record at the close of business on January 26, 2004 will be entitled to vote at the Bank of America special meeting. Holders of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock vote together without regard to class. Each share of Bank of America common stock and Series B Preferred Stock is entitled to one vote, and each share of ESOP Preferred Stock is entitled to two votes. As of the record date of January 26, 2004, there were 1,448,613,889 shares of Bank of America common stock, 7,776 shares of Series B Preferred Stock, and 1,261,824 shares of ESOP Preferred Stock entitled to vote at the Bank of America special meeting.

Required Vote. Adoption of the merger agreement and approval of the proposal to increase the number of authorized shares of Bank of America common stock each requires the affirmative vote of a majority of the votes represented by the outstanding shares of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock entitled to vote, voting together without regard to class. Because the required vote is based on the affirmative vote of a majority of votes outstanding, your failure to vote, including a broker non-vote or an

abstention, will have the same effect as a vote against the merger and the proposal to increase the number of authorized shares of Bank of America common stock. Each of the proposals is independent, and is not contingent on approval by stockholders, of the other proposals.

Adoption of Bank of America's Amended Stock Plan requires the affirmative vote of a majority of the votes cast at the special meeting by the holders of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock, voting together without regard to class. Because the required vote is based on the affirmative vote of the majority of votes cast, your failure to vote, including a broker non-vote or an abstention, will not be treated as a vote cast and, therefore, will have no effect on the proposal to approve Bank of America's Amended Stock Plan.

If you abstain from voting or fail to instruct your broker to vote your shares and the broker submits an unvoted proxy, the resulting abstention or "broker non-vote" will be counted toward a quorum at the special meeting.

As of the Bank of	America record date, directors and executive officers of Bank o	of America and their affiliates had the right to v	ote [] shares of Bank of America
common stock, or []% of the outstanding Bank of America common stock entitled	d to be voted at the Bank of America special m	eeting; [] shares of Series B
Preferred Stock, or []% of the outstanding Series B Preferred Stock; and [] shares of ESOP Preferred Stock, or []	% of the outstanding shares of ESOP
Preferred Stock. At that da	ate, directors and executive officers of FleetBoston and their aff	filiates, including FleetBoston, had the right to	vote [] shares of Bank of
America common stock en	ntitled to be voted at the Bank of America special meeting, or []% of the outstanding Bank of America	common stock; [] shares of
Series B Preferred Stock,	or []% of the outstanding Series B Preferred Stock; and	[] shares of ESOP Preferred Stock, or []% of the outstanding shares of
ESOP Preferred Stock. W.	hile it is currently expected that Bank of America's directors ar	nd executive officers will vote their shares of B	ank of America common stock in favor
of the merger, Bank of An	nerica's directors and executive officers have not entered into a	any agreements with respect to the voting of the	eir shares.

FleetBoston will Hold its Special Meeting on March 17, 2004 (page 25)

The FleetBoston special meeting will be held on March 17, 2004, at 10:00 a.m., local time, in the Auditorium on the ground floor of The Federal Reserve Bank of Boston, 600 Atlantic Avenue, Boston, Massachusetts. At the FleetBoston special meeting, FleetBoston stockholders will be asked to:

- · approve the merger agreement; and
- approve the adjournment of the special meeting, if necessary, to solicit additional proxies, in the event that there are not sufficient votes at the time of the special
 meeting to approve the merger agreement.

Record Date. Only common stockholders of record at the close of business on January 26, 2004 will be entitled to vote at the FleetBoston special meeting. Each share of FleetBoston common stock is entitled to one vote. As of the record date of January 26, 2004, there were [] shares of FleetBoston common stock entitled to vote at the FleetBoston special meeting.

Required Vote. To approve the merger agreement, the holders of a majority of the outstanding shares of FleetBoston common stock entitled to vote must vote in favor of approving the merger agreement. Because approval is based on the affirmative vote of a majority of shares outstanding, a FleetBoston stockholder's failure to vote, a broker non-vote or an abstention will have the same effect as a vote against the merger.

As of the FleetBoston record date, directors and executive officers of FleetBoston and their affiliates had the right to vote [] shares of FleetBoston common stock, or []% of the outstanding FleetBoston common

stock entitled to be voted at the FleetBoston special meeting. At that date, directors and executive officers of Bank of America and their affiliates, including Bank of America, had the right to vote [] shares of FleetBoston common stock entitled to be voted at the FleetBoston special meeting, or []% of the outstanding FleetBoston common stock. While it is currently expected that FleetBoston's directors and executive officers will vote their shares of FleetBoston common stock in favor of the merger, FleetBoston's directors and executive officers have not entered into any agreements with respect to the voting of their shares.

Comparative Market Prices and Dividends

Bank of America common stock and FleetBoston common stock are listed on the NYSE, among other stock exchanges. The following table sets forth the high and low closing prices of shares of Bank of America common stock and FleetBoston common stock as reported on the NYSE, and the quarterly cash dividends paid per share in the periods indicated.

		Bank of America Common Stock			FleetBoston Common Stock		
	High	Low	Dividend	High	Low	Dividend	
2001							
First Quarter	\$ 55.47	\$ 46.75	\$ 0.56	\$43.64	\$ 34.39	\$ 0.33	
Second Quarter	61.94	49.59	0.56	42.28	36.06	0.33	
Third Quarter	65.00	51.00	0.56	39.73	32.65	0.33	
Fourth Quarter	64.99	52.15	0.60	38.64	31.45	0.33	
2002							
First Quarter	69.18	58.85	0.60	37.21	31.10	0.35	
Second Quarter	76.90	67.45	0.60	36.60	30.70	0.35	
Third Quarter	71.94	57.90	0.60	31.75	18.75	0.35	
Fourth Quarter	71.42	54.15	0.64	27.49	17.75	0.35	
2003							
First Quarter	72.48	65.63	0.64	27.64	21.98	0.35	
Second Quarter	79.89	68.00	0.64	31.15	24.55	0.35	
Third Quarter	83.53	74.87	0.80	31.54	29.35	0.35	
Fourth Quarter	82.50	72.85	0.80	43.65	31.18	0.35	

Through [], 2004

Bank of America stockholders and FleetBoston stockholders are advised to obtain current market quotations for Bank of America common stock and FleetBoston common stock. The market price of Bank of America common stock and FleetBoston common stock will fluctuate between the date of this joint proxy statement/prospectus and the completion of the merger. No assurance can be given concerning the market price of Bank of America common stock or FleetBoston common stock before or after completion of the merger.

SELECTED CONSOLIDATED HISTORICAL FINANCIAL DATA OF BANK OF AMERICA

Set forth below are highlights from Bank of America's consolidated financial data as of and for the years ended December 31, 2002 through 1998 and Bank of America's unaudited consolidated financial data as of and for the nine months ended September 30, 2003 and 2002. The results of operations for the nine months ended September 30, 2003 are not necessarily indicative of the results of operations for the full year or any other interim period. Bank of America management prepared the unaudited information on the same basis as it prepared Bank of America's audited consolidated financial statements. In the opinion of Bank of America management, this information reflects all adjustments, consisting of only normal recurring adjustments, necessary for a fair presentation of this data for those dates. You should read this information in conjunction with Bank of America's consolidated financial statements and related notes included in Bank of America's Annual Report on Form 10-K for the year ended December 31, 2002, and Bank of America's Quarterly Report on Form 10-Q for the quarter ended September 30, 2003, which are incorporated by reference in this document and from which this information is derived. See "Where You Can Find More Information" on page 111.

Bank of America — Summary of Selected Financial Data⁽¹⁾

	Years Ended December 31,				Nine Months Ended September 30,		
(Dollars in millions, except	2002	2001	2000	1999	1998	2003	2002
per share information)							
Income statement							
Net interest income	\$ 20,923	\$ 20,290	\$ 18,349	\$ 18,127	\$ 18,298	\$ 15,878	\$ 15,549
Noninterest income	13,571	14,348	14,582	14,179	12,189	12,379	10,141
Total revenue	34,494	34,638	32,931	32,306	30,487	28,257	25,690
Provision for credit losses	3,697	4,287	2,535	1,820	2,920	2,256	2,532
Gains on sales of securities	630	475	25	240	1,017	802	326
Noninterest expense	18,436	20,709	18,633	18,511	20,536	14,845	13,604
Income before income taxes	12,991	10,117	11,788	12,215	8,048	11,958	9,880
Income tax expense	3,742	3,325	4,271	4,333	2,883	3,874	3,245
Net income	9,249	6,792	7,517	7,882	5,165	8,084	6,635
Average common shares issued and outstanding (in thousands)	1,520,042	1,594,957	1,646,398	1,726,006	1,732,057	1,494,369	1,526,946
Average diluted common shares issued and outstanding (in thousands)	1,565,467	1,625,654	1,664,929	1,760,058	1,775,760	1,523,523	1,573,203
Performance ratios							
Return on average assets	1.40%	1.05%	1.12%	1.28%	0.88%	1.43%	1.36%
Return on average common shareholders' equity	19.44	13.96	15.96	16.93	11.56	21.85	18.71
Total equity to total assets (period end)	7.62	7.80	7.42	7.02	7.44	6.84	7.31
Total average equity to total average assets	7.19	7.49	7.02	7.55	7.67	6.53	7.29
Dividend payout ratio	40.07	53.44	45.02	40.54	50.18	38.53	41.37
Per common share data							
Earnings	\$ 6.08	\$ 4.26	\$ 4.56	\$ 4.56	\$ 2.97	\$ 5.41	\$ 4.34
Diluted earnings	5.91	4.18	4.52	4.48	2.90	5.31	4.22
Cash dividends paid	2.44	2.28	2.06	1.85	1.59	2.08	1.80
Book value	33.49	31.07	29.47	26.44	26.60	33.83	32.07
Average balance sheet				<u> </u>			
Total loans and leases	\$ 336,819	\$ 365,447	\$ 392,622	\$ 362,783	\$ 347,840	\$ 351,119	\$ 334,703
Total assets	662,401	649,547	671,573	616,838	584,487	758,140	651,257
Total deposits	371,479	362,653	353,294	341,748	345,485	401,985	368,142
Long-term debt (2)	66,045	69,622	70,293	57,574	49,969	67,702	66,161
Common shareholders' equity	47,552	48,609	47,057	46,527	44,467	49,455	47,396
Total shareholders' equity	47,613	48,678	47,132	46,601	44,829	49,512	47,457
Tour shareholders equity							
Capital ratios							
Risk-based capital:							
Tier 1	8.22%	8.30%	7.50%	7.35%	7.06%	8.25%	8.13%
Total	12.43	12.67	11.04	10.88	10.94	12.17	12.38
Leverage	6.29	6.56	6.12	6.26	6.22	5.96	6.35
Market price per share of common stock							
Closing	\$ 69.57	\$ 62.95	\$ 45.88	\$ 50.19	\$ 60.13	\$ 78.04	\$ 63.80
High closing	76.90	65.00	59.25	75.50	87.94	83.53	76.90
Low closing	54.15	46.75	38.00	48.00	48.06	65.63	57.90
LOW Closing	57.15	40.75	30.00	70.00	70.00	05.05	57.90

⁽¹⁾ As a result of the adoption of SFAS 142 on January 1, 2002, Bank of America no longer amortizes goodwill. Goodwill amortization expense was \$662, \$635, \$635 and \$633 in 2001, 2000, 1999 and 1998, respectively.

⁽²⁾ Includes long-term debt related to trust preferred securities.

SELECTED CONSOLIDATED HISTORICAL FINANCIAL DATA OF FLEETBOSTON

Set forth below are highlights from FleetBoston's consolidated financial data as of and for the years ended December 31, 2002 through 1998 and FleetBoston's unaudited consolidated financial data as of and for the nine months ended September 30, 2003 and 2002. Discontinued operations reported in FleetBoston's historical consolidated statements of income have been excluded. The results of operations for the nine months ended September 30, 2003 are not necessarily indicative of the results of operations for the full year or any other interim period. The unaudited information was prepared on the same basis as FleetBoston's audited consolidated financial statements. In the opinion of FleetBoston management, this information reflects all adjustments, consisting of only normal recurring adjustments, necessary for a fair presentation of this data for those dates. You should read this information in conjunction with FleetBoston's consolidated financial statements and related notes included in FleetBoston's Annual Report on Form 10-K for the year ended December 31, 2002, and FleetBoston's Quarterly Report on Form 10-Q for the quarter ended September 30, 2003, which are incorporated by reference in this document and from which this information is derived. See "Where You Can Find More Information" on page 111.

FleetBoston — Summary of Selected Financial Data(1)

	Years Ended December 31,			Nine Months Ended September 30,			
(Dollars in millions, except	2002	2001	2000	1999	1998	2003	2002
per share information)							
Income statement							
Net interest income	\$ 6,420	\$ 7,287	\$ 7,756	\$ 7,973	\$ 7,555	\$ 4,736	\$ 4,869
Noninterest income	5,036	4,555	7,559	6,091	5,303	3,698	3,725
Total revenue	11,456	11,842	15,315	14,064	12,858	8,434	8,594
Provision for credit losses	2,760	2,324	1,290	1,056	907	830	2,010
Noninterest expense	6,404	7,977	8,100	9,089	7,454	4,778	4,738
Income from continuing operations before income taxes	2,292	1,541	5,925	3,919	4,498	2,826	1,846
Income tax expense	768	573	2,353	1,538	1,693	1,003	619
Income from continuing operations	1,524	968	3,572	2,381	2,805	1,823	1,227
Average common shares issued and outstanding (in thousands)	1,045,336	1,074,204	1,081,377	1,095,740	1,094,700	1,047,454	1,045,092
Average diluted common shares issued and outstanding (in							
thousands)	1,048,734	1,083,676	1,098,740	1,121,488	1,119,721	1,050,655	1,049,208
Performance ratios							
Return on average assets (2)	0.82%	0.48%	1.66%	1.12%	1.41%	1.26%	0.88%
Return on average common shareholders' equity	8.84	4.96	20.12	13.88	17.73	14.23	9.44
Total equity to total assets (period end)	8.84	8.64	8.84	7.97	8.01	8.96	9.01
Total average equity to total average assets	9.12	9.25	8.10	7.83	8.12	8.88	9.15
Dividend payout ratio	125.00	159.52	34.36	50.23	40.32	59.32	120.69
Per common share data							
Earnings from continuing operations	\$ 1.44	\$ 0.88	\$ 3.27	\$ 2.12	\$ 2.51	\$ 1.73	\$ 1.16
Diluted earnings from continuing operations	1.44	0.87	3.22	2.07	2.45	1.72	1.16
Cash dividends paid	1.40	1.32	1.20	1.08	0.98	1.05	1.05
Book value	15.78	16.61	17.31	15.92	14.78	16.46	15.84
Average balance sheet							
Total loans and leases	\$ 119,932	\$ 128,792	\$ 138,609	\$ 138,571	\$ 130,098	\$ 123,283	\$ 120,320
Total assets	189,866	208,931	223,238	223,238	201,042	194,506	190,887
Total deposits	120,488	124,489	130,381	138,152	134,146	128,724	120,157
Long-term debt	22,658	27,945	31,191	26,198	12,957	18,827	23,306
Common shareholders' equity	17,046	18,949	17,562	16,788	15,480	16,995	17,186
Total shareholders' equity	17,316	19,330	18,134	17,479	16,319	17,266	17,457
							
Capital ratios							
Risk-based capital:							
Tier 1	8.24%	7.37%	8.08%	7.15%	7.56%	8.64%	8.24%
Total	11.72	10.95	11.87	11.44	11.65	11.79	11.77
Leverage	8.27	7.50	8.01	6.85	7.30	8.38	8.34
Market price per share of common stock							
Closing	\$ 24.30	\$ 36.50	\$ 37.56	\$ 34.81	\$ 44.69	\$ 30.15	\$ 20.33
High closing	37.21	43.64	43.00	46.75	45.00	31.54	37.21
Low closing	17.75	31.45	25.25	33.81	31.69	21.98	18.75

⁽¹⁾ As a result of the adoption of SFAS 142 on January 1, 2002, FleetBoston no longer amortizes goodwill. Goodwill amortization expense was \$301, \$292, \$269 and \$209 in 2001, 2000, 1999 and 1998, respectively.

SELECTED CONSOLIDATED UNAUDITED PRO FORMA FINANCIAL INFORMATION

The following table shows information about our financial condition and results of operations, including per share data and financial ratios, after giving effect to the merger. This information is called pro forma financial information in this document. The table sets forth the information as if the merger had become effective on September 30, 2003, with respect to financial condition data, and on January 1, 2002, with respect to results of operations data. This pro forma financial information assumes that the merger is accounted for using the purchase method of accounting and represents a current estimate based on available information of the combined company's results of operations. See "Accounting Treatment" on page 73. The pro forma financial information includes adjustments to record the assets and liabilities of FleetBoston at their estimated fair values and is subject to further adjustment as additional information becomes available and as additional analyses are performed. The pro forma financial statements do not currently include any amount related to the estimated \$800 million after-tax (\$1.27 billion pre-tax) merger related costs that will be incurred to combine the operations of Bank of America and FleetBoston. The estimated merger related costs will result from actions taken with respect to both Bank of America and FleetBoston operations, facilities and associates. The charges will be recorded based on the nature and timing of these integration actions. See the Notes to the Unaudited Pro Forma Condensed Combined Financial Information for a further discussion of the treatment of integration charges. This table should be read in conjunction with, and is qualified in its entirety by, the historical financial statements, including the notes thereto, of Bank of America and FleetBoston incorporated by reference in this document and the more detailed pro forma financial information, including the notes thereto, appearing elsewhere in this document. See "Where You Can Find More Information" on page 111 an

The pro forma financial information, while helpful in illustrating the financial characteristics of the combined company under one set of assumptions, does not reflect the impact of possible revenue enhancements, expense efficiencies, asset dispositions and share repurchases, among other factors, that may result as a consequence of the merger and, accordingly, does not attempt to predict or suggest future results. It also does not necessarily reflect what the historical results of the combined company would have been had our companies been combined during these periods.

SELECTED CONSOLIDATED UNAUDITED PRO FORMA FINANCIAL INFORMATION

		Twelve Months Ended December 31, 2002		Nine Months Ended September 30, 2003	
(Dollars in millions, except per share information)					
Income statement					
Net interest income	\$	27,548	\$	20,771	
Noninterest income		18,601		15,973	
Total revenue		46,149		36,744	
Provision for credit losses		6,457		3,086	
Gains on sales of securities		636		906	
Noninterest expense		25,347		19,961	
Income from continuing operations before income taxes		14,981		14,603	
Income tax expense		4,399		4,810	
Income from continuing operations		10,582		9,793	
Average common shares issued and outstanding (in thousands)	2	,100,497		2,076,046	
Average diluted common shares issued and outstanding					
(in thousands)	2	,147,810		2,106,977	
Performance ratios(1)					
Return on average assets		n/m		1.33%	
Return on average common shareholders' equity		n/m		13.72	
Total equity to total assets (period end)		n/m		10.03	
Total average equity to total average assets		n/m		9.71	
Dividend payout ratio		48.51%		44.16	
Per common share data ⁽¹⁾					
Earnings	\$	5.03	\$	4.71	
Diluted earnings		4.93		4.65	
Cash dividends paid		2.44		2.08	
Book value		n/m		46.50	
Average balance sheet(1)					
Total loans and leases		n/m	\$	474,538	
Total assets		n/m		983,481	
Total deposits		n/m		531,004	
Long-term debt ⁽²⁾		n/m		87,606	
Common shareholders' equity		n/m		95,174	
Total shareholders' equity		n/m		95,502	
Capital Ratios ⁽¹⁾					
Risk-based capital:					
Tier 1		n/m		7.75%	
Total		n/m		11.51	
Leverage		n/m		5.97	

⁽¹⁾ Average balance sheet amounts and capital and other ratios as of December 31, 2002 are not meaningful (n/m) as purchase accounting adjustments were calculated as of September 30, 2003.

⁽²⁾ Includes long-term debt related to trust preferred securities.

COMPARATIVE PER SHARE DATA

The following table sets forth for Bank of America common stock and FleetBoston common stock certain historical, pro forma and pro forma-equivalent per share financial information. The pro forma and pro forma-equivalent per share information gives effect to the merger as if the merger had been effective on the dates presented, in the case of the book value data, and as if the merger had become effective on January 1, 2002, in the case of the income from continuing operations and dividends paid data. The pro forma data in the tables assume that the merger is accounted for using the purchase method of accounting and represents a current estimate based on available information of the combined company's results of operations. See "Accounting Treatment" on page 73. The pro forma financial adjustments record the assets and liabilities of FleetBoston at their estimated fair values and are subject to adjustment as additional information becomes available and as additional analyses are performed. The information in the following table is based on, and should be read together with, the historical financial information that we have presented in our prior filings with the Securities and Exchange Commission and the pro forma financial information that appears elsewhere in this document. See "Where You Can Find More Information" on page 111 and "Unaudited Pro Forma Condensed Combined Financial Information" on page 101.

The pro forma information, while helpful in illustrating the financial characteristics of the combined company under one set of assumptions, does not reflect the impact of possible revenue enhancements, expense efficiencies, asset dispositions and share repurchases, among other factors, that may result as a consequence of the merger and, accordingly, does not attempt to predict or suggest future results. It also does not necessarily reflect what the historical results of the combined company would have been had our companies been combined during these periods. Upon completion of the merger, the operating results of FleetBoston will be reflected in the consolidated financial statements of Bank of America on a prospective basis.

Comparative Per Share Date

		Comparative Per Share Data				
	Bank of America Historical	FleetBoston Historical	Pro Forma Combined	Per Equivalent FleetBoston Share ⁽¹⁾		
Income from continuing operations for the twelve months ended December 31, 2002						
Basic	\$ 6.08	\$ 1.44	\$ 5.03	\$ 2.79		
Diluted	5.91	1.44	4.93	2.74		
Income from continuing operations for the nine months ended September 30, 2003						
Basic	5.41	1.73	4.71	2.62		
Diluted	5.31	1.72	4.65	2.58		
Cash Dividends Paid						
For the twelve months ended December 31, 2002	2.44	1.40	2.44	1.35		
For the nine months ended September 30, 2003	2.08	1.05	2.08	1.16		
Book Value ⁽²⁾						
As of December 31, 2002	33.49	15.78	n/m	n/m		
As of September 30, 2003	33.83	16.46	46.50	25.82		

⁽¹⁾ Reflects FleetBoston shares at the exchange ratio after giving effect to the pro forma adjustments. See "Unaudited Pro Forma Condensed Combined Financial Information" on page 101.

⁽²⁾ Book value as of December 31, 2002 is not meaningful (n/m) as purchase accounting adjustments were calculated as of September 30, 2003.

RECENT DEVELOPMENTS

Bank of America Corporation

On January 15, 2004, Bank of America announced fourth quarter earnings of \$2.73 billion, or \$1.83 per share (diluted) compared to \$2.61 billion, or \$1.69 per share (diluted), earned in the fourth quarter of 2002. For the full year, Bank of America earned \$10.8 billion, or \$7.13 per share (diluted) compared to \$9.25 billion, or \$5.91 per share (diluted) in 2002.

Bank of America's Tier I Capital Ratio was 7.85% at December 31, 2003 down from 8.25% at September 30, 2003. The decrease was due to increased repurchases of common stock related to favorable market conditions. During the fourth quarter, Bank of America repurchased 56 million shares offset, in part, by the issuance of 7 million shares related to the exercise of stock options.

On January 28, 2004, Bank of America announced that its board of directors had approved the repurchase of up to 90 million shares of common stock, authorizing management to spend up to \$9 billion within 18 months. This repurchase program is intended to be implemented through purchases made from time to time either in the open market or through private transactions, including accelerated buyback programs.

FleetBoston Financial Corporation

On January 15, 2004, FleetBoston announced fourth quarter earnings of \$732 million, or \$.68 per share (diluted) compared to \$261 million, or \$.24 per share (diluted), earned in the fourth quarter of 2002. For the full year, FleetBoston earned \$2.6 billion, or \$2.45 per share (diluted) compared to \$1.2 billion, or \$1.12 per share (diluted) in 2002.

FleetBoston's Tier I Capital Ratio was 8.91% at December 31, 2003 up from 8.64% at September 30, 2003.

RISK FACTORS RELATING TO THE MERGER

In addition to the other information contained in or incorporated by reference into this document, you should carefully consider the following risk factors in deciding whether to vote in favor of the merger.

The implied market value of the shares of Bank of America common stock to be received by holders of shares of FleetBoston common stock in the merger will fluctuate

Upon completion of the merger, each share of FleetBoston common stock will be exchanged for the right to receive 0.5553 of a share of Bank of America common stock. There will be no adjustment to the exchange ratio for changes in the market price of either shares of FleetBoston common stock or shares of Bank of America common stock and the merger agreement does not include a price-based termination right. Accordingly, the market value of the shares of Bank of America common stock that holders of shares of FleetBoston common stock will become entitled to receive upon completion of the merger will depend on the market value of the shares of Bank of America common stock at the time of completion of the merger, and could vary significantly from the market value on the date of this document or the date of the FleetBoston special meeting. The market value of the shares of Bank of America common stock to be received in the merger will also continue to fluctuate after completion of the merger. For historical market prices of shares of Bank of America common stock, please refer to "Comparative Market Prices and Dividends" on page 100. You should obtain current market quotations for shares of Bank of America common stock and for shares of FleetBoston common stock.

The fairness opinions obtained by Bank of America and FleetBoston from Goldman Sachs and Morgan Stanley, respectively, will not reflect changes in circumstances between the signing of the agreement and the merger

Bank of America and FleetBoston have not obtained updated opinions as of the date of this document from Goldman Sachs and Morgan Stanley, Bank of America's and FleetBoston's respective financial advisors. Changes in the operations and prospects of Bank of America or FleetBoston, general market and economic conditions and other factors which may be beyond the control of Bank of America and FleetBoston, and on which the fairness opinions were based, may alter the value of Bank of America or FleetBoston or the prices of shares of Bank of America common stock and shares of FleetBoston common stock by the time the merger is completed. The opinions do not speak as of the time the merger will be completed or as of any date other than the dates of such opinions. For a description of the opinions that Bank of America and FleetBoston received from their respective financial advisors, please refer to "The Merger – Opinion of Goldman Sachs" on page 37 and "The Merger – Opinion of Morgan Stanley" on page 44. For a description of the other factors considered by Bank of America's board of directors in determining to approve the merger, please refer to "The Merger – Bank of America's Reasons for the Merger" on page 32. For a description of the other factors considered by FleetBoston's board of directors in determining to approve the merger, please refer to "The Merger – FleetBoston's Reasons for the Merger" on page 34.

The combined company may fail to realize all of the anticipated benefits of the merger

The merger is expected to generate after-tax cost savings and expense reductions of \$1.1 billion when fully phased-in. The expense reductions are intended to be achieved by eliminating duplicative technology operations and redundant staff, reductions in business units, facility consolidations and purchasing efficiencies. The combined company may fail to realize some or all of the anticipated cost savings and other benefits of the transaction as a result of, among other things, unanticipated costs, deterioration in the U.S. economy and other factors. In addition, the integration of FleetBoston's business and operations with those of Bank of America, including systems conversions, may take longer than anticipated, may be more costly than anticipated and may have unanticipated adverse results relating to FleetBoston's or Bank of America's existing businesses or customer base.

The merger agreement limits FleetBoston's ability to pursue alternatives to the merger

The merger agreement contains "no shop" provisions that, subject to limited exceptions, limit FleetBoston's ability to discuss, facilitate or commit to competing third-party proposals to acquire all or a significant part of the company. In addition, FleetBoston has granted to Bank of America an option to acquire up to approximately 209.5 million shares of FleetBoston common stock under the FleetBoston stock option agreement. These provisions might discourage a potential competing acquiror that might have an interest in acquiring all or a significant part of FleetBoston from considering or proposing that acquisition even if it were prepared to pay consideration with a higher per share market price than that proposed in the merger, or might result in a potential competing acquiror proposing to pay a lower per share price to acquire FleetBoston than it might otherwise have proposed to pay.

The 10 percent national deposit cap may restrict the combined company's ability to make additional bank acquisitions in the U.S.

Federal banking law contains provisions that limit the ability of the Federal Reserve Board to approve an application by a bank holding company to acquire banks located outside of the acquiror's home state without regard to state law if the acquisition would result in the combined company holding more than 10 percent of the deposits held by insured depository institutions in the United States. The combined company's home state will be North Carolina. While the percentage of the combined company's deposits will change from time to time as deposits flow into and between institutions and the total level of deposits in the United States changes, we expect that the combined company will hold a percentage of deposits that approaches the 10 percent limit. While this limit does not impede the combined company's ability to grow by attracting additional deposits from new or existing customers, by acquiring thrifts or by other transactions that do not involve the acquisition of a bank located outside of North Carolina, unless it is repealed or modified, or unless a transaction can be structured appropriately to avoid its application, the national deposit cap will restrict the ability of the combined company to acquire additional banks located outside of North Carolina that would result in holding deposits in excess of the 10 percent limit. Repeal or modification of the national deposit cap would require an act of Congress, and it is the responsibility of the Federal Reserve Board as part of its application approval process to determine if an acquisition would or would not surpass the 10 percent national limit.

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This document contains or incorporates by reference a number of forward-looking statements regarding the financial condition, results of operations, earnings outlook, business and prospects of Bank of America, FleetBoston and the potential combined company and may include statements for the period following the completion of the merger. You can find many of these statements by looking for words such as "plan," "believe," "expect," "intend," "anticipate," "estimate," "project," "potential," "possible" or other similar expressions.

The forward-looking statements involve certain risks and uncertainties. The ability of either Bank of America or FleetBoston to predict results or the actual effects of its plans and strategies, or those of the combined company, is inherently uncertain. Accordingly, actual results may differ materially from anticipated results. Some of the factors that may cause actual results or earnings to differ materially from those contemplated by the forward-looking statements include, but are not limited to, those discussed above under "Risk Factors Relating to the Merger" on page 19, as well as the following:

- · projected business increases following process changes and other investments are lower than expected;
- · competitive pressure among financial services companies increases significantly;
- general economic conditions are less favorable than expected;
- · political conditions and related actions by the United States military abroad may adversely affect either company's businesses and economic conditions as a whole;
- changes in the interest rate environment reduce interest margins and impact funding sources;
- changes in foreign exchange rates increase risk to foreign currency exposure;
- changes in market rates and prices may adversely impact the value of financial products and assets;
- · legislation or regulatory environments, requirements or changes adversely affect businesses in which either company is engaged;
- · litigation liabilities, including costs, expenses, settlements and judgments, may adversely affect either company or its businesses; and
- decisions to downsize, sell or close units or otherwise change the business mix of either company.

Because these forward-looking statements are subject to assumptions and uncertainties, actual results may differ materially from those expressed or implied by these forward-looking statements. Bank of America stockholders and FleetBoston stockholders are cautioned not to place undue reliance on these statements, which speak only as of the date of this document or the date of any document incorporated by reference in this document.

All subsequent written and oral forward-looking statements concerning the merger or other matters addressed in this document and attributable to Bank of America or FleetBoston or any person acting on their behalf are expressly qualified in their entirety by the cautionary statements contained or referred to in this section. Except to the extent required by applicable law or regulation, Bank of America and FleetBoston undertake no obligation to update these forward-looking statements to reflect events or circumstances after the date of this document or to reflect the occurrence of unanticipated events.

THE BANK OF AMERICA SPECIAL MEETING

This section contains information from Bank of America for Bank of America stockholders about the special meeting of Bank of America stockholders that has been called to consider and adopt the merger agreement.

Together with this document, we are also sending you a notice of the Bank of America special meeting and a form of proxy that is solicited by the Bank of America board of directors. The Bank of America special meeting will be held on March 17, 2004, at 10:00 a.m., local time, in the Palmetto Ballroom of the International Trade Center, 200 North College Street, Charlotte, North Carolina.

Matters to Be Considered

The purpose of the Bank of America special meeting is to vote upon the following matters:

- a proposal to adopt the merger agreement;
- a proposal to adopt Bank of America's Amended Stock Plan;
- a proposal to adopt an amendment to increase the number of authorized shares of Bank of America common stock from 5 billion to 7.5 billion;

You also will be asked to vote on a proposal to approve the adjournment of the Bank of America special meeting, if necessary, to solicit additional proxies, in the event that there are not sufficient votes at the time of the special meeting to approve the proposals.

Each copy of this document mailed to Bank of America stockholders is accompanied by a form of proxy with instructions for voting by mail, by telephone or through the Internet. If voting by mail, you should complete and return the proxy card accompanying this document in order to ensure that your vote is counted at the Bank of America special meeting, or at any adjournment or postponement of the Bank of America special meeting, regardless of whether you plan to attend the Bank of America special meeting. You may also vote your shares by telephone or through the Internet. Information and applicable deadlines for voting by telephone or through the Internet are set forth in the enclosed proxy card instructions.

You may revoke your signed proxy card at any time before it is voted by signing and returning a proxy card with a later date, delivering a written revocation letter to Bank of America's Corporate Secretary, or by attending the Bank of America special meeting in person, notifying the Corporate Secretary, and voting by ballot at the Bank of America special meeting. If you have voted your shares by telephone or through the Internet, you may revoke your prior telephone or Internet vote by recording a different vote, or by signing and returning a proxy card dated as of a date that is later than your last telephone or Internet vote.

Any stockholder entitled to vote in person at the Bank of America special meeting may vote in person whether or not a proxy has been previously given, but the mere presence (without notifying the Corporate Secretary) of a stockholder at the Bank of America special meeting will not constitute revocation of a previously given proxy.

Written notices of revocation and other communications regarding the revocation of your proxy should be addressed to:

Bank of America Corporation NC1-007-56-11 100 North Tryon Street Charlotte, North Carolina 28255 Attention: Rachel R. Cummings

Corporate Secretary

If your shares are held in street name by a bank, broker or other nominee, you should follow the instructions of your bank, broker or other nominee regarding the revocation of proxies.

All shares represented by valid proxies that we receive through this solicitation, and that are not revoked, will be voted in accordance with the instructions on the proxy card. If you make no specification on your proxy card as to how you want your shares voted before signing and returning it, your proxy will be voted "FOR" adoption of the merger agreement, adoption of Bank of America's Amended Stock Plan, approval of the proposal to increase the number of authorized shares of Bank of America common stock and approval of the proposal to adjourn the special meeting, if necessary, to solicit additional proxies in the event that there are not sufficient votes at the time of the special meeting to approve the proposals. According to the Bank of America bylaws, business to be conducted at a special meeting of stockholders may only be brought before the meeting pursuant to Bank of America's notice of meeting. Accordingly, no matters other than the matters described in this joint proxy statement/prospectus will be presented for action at the Bank of America special meeting or at any adjournment or postponement of the Bank of America special meeting.

Bank of America stockholders should NOT send Bank of America stock certificates with their proxy cards. If the merger is completed, Bank of America stockholders will not need to exchange their current Bank of America stock certificates.

Solicitation of Proxies

We will bear the entire cost of soliciting proxies from you. In addition to solicitation of proxies by mail, we will request that banks, brokers and other nominees send proxies and proxy material to the beneficial owners of Bank of America common stock and secure their voting instructions, if necessary. We will reimburse the banks, brokers and other nominees for their reasonable expenses in taking those actions. We also have made arrangements with Georgeson Shareholder Communications, Inc. to assist us in soliciting proxies and have agreed to pay them \$20,000 plus reasonable expenses for these services. If necessary, we also may use several of our regular employees, who will not be specially compensated, to solicit proxies from Bank of America stockholders, either personally or by telephone, facsimile, letter or other electronic means.

Householding

Unless it has received contrary instructions, Bank of America may send a single copy of this joint proxy statement/prospectus to any household at which two or more Bank of America stockholders reside if Bank of America believes the stockholders are members of the same family. Each stockholder in the household will continue to receive a separate proxy card. This process, known as "householding," reduces the volume of duplicate information received at your household and helps to reduce Bank of America's expenses.

If you would like to receive your own joint proxy statement/prospectus, follow the instructions described below. Similarly, if you share an address with another stockholder and together both of you would like to receive only a single joint proxy statement/prospectus, follow these instructions:

If your shares are registered in your own name, please contact our transfer agent, Mellon Investor Services, and inform them of your request by calling them at (800) 642-9855 or writing to them at P.O. Box 3315, South Hackensack, New Jersey 07606-1915.

If a bank, broker or other nominee holds your shares, please contact your bank, broker or other nominee directly.

Record Date

The Bank of America board of directors has fixed the close of business on January 26, 2004 as the record date for determining the Bank of America stockholders entitled to receive notice of and to vote at the Bank of America special meeting. At that time, 1,448,613,889 shares of Bank of America common stock, 7,776 shares of Series B Preferred Stock and 1,261,824 shares of ESOP Preferred Stock were outstanding.

Voting Rights and Vote Required

The presence, in person or by properly executed proxy, of the holders of a majority of the votes represented by the aggregate of all of the outstanding shares of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock is necessary to constitute a quorum at the Bank of America special meeting. You are entitled to one vote for each share of Bank of America common stock and Series B Preferred Stock you held as of the record date, and two votes for each share of ESOP Preferred Stock you held as of the record date. Abstentions and broker non-votes will be counted solely for the purpose of determining whether a quorum is present. When we refer to broker non-votes, we are referring to unvoted proxies submitted by brokers, who are not able to vote on a proposal absent instructions from the applicable beneficial owner.

Adoption of the merger agreement and approval of the proposal to increase the number of authorized shares of Bank of America common stock each requires the affirmative vote of a majority of the votes represented by the outstanding shares of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock entitled to vote, voting together without regard to class. Because the required vote is based on the affirmative vote of a majority of the votes outstanding, your failure to vote, a broker non-vote or an abstention will have the same effect as a vote against the merger and the proposal to increase the number of authorized shares of Bank of America common stock. Each of the proposals is independent, and is not contingent on approval by stockholders, of the other proposals.

Adoption of Bank of America's Amended Stock Plan requires the affirmative vote of a majority of the votes cast at the special meeting by the holders of the Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock, voting together without regard to class. Because the required vote is based on the affirmative vote of a majority of the votes cast, your failure to vote, a broker non-vote or an abstention will not be treated as a vote cast and, therefore, will have no effect on the proposal to approve Bank of America's Amended Stock Plan.

The Bank of America board of directors urges Bank of America stockholders to: complete, date and sign the accompanying proxy card and return it promptly in the enclosed postage-paid envelope if voting by mail; call the toll-free number listed in the proxy card instructions if voting by telephone; or access the Internet site listed in the proxy card instructions if voting through the Internet.

As of the record date:

- Directors and executive officers of Bank of America and their affiliates had the right to vote [] shares of Bank of America common stock, or []% of the Bank of America common stock outstanding on that date; [] shares of Series B Preferred Stock, or []% of the outstanding Series B Preferred Stock; and [] shares of ESOP Preferred Stock, or []% of the outstanding shares of ESOP Preferred Stock.
- Directors and executive officers of FleetBoston and their affiliates, including FleetBoston (excluding the shares subject to the Bank of America stock option described in "The Stock Option Agreements" on page 64), had the right to vote [] shares of Bank of America common stock, or []% of the Bank of America common stock outstanding on that date; [] shares of Series B Preferred Stock, or []% of the outstanding Series B Preferred Stock; and [] shares of ESOP Preferred Stock, or []% of the outstanding shares of ESOP Preferred Stock.

Recommendations of the Bank of America Board of Directors

The Bank of America board of directors has unanimously approved the merger agreement and the transactions it contemplates. The Bank of America board of directors determined that the merger agreement and the transactions it contemplates are advisable and in the best interests of Bank of America and its stockholders,

and unanimously recommends that you vote "FOR" adoption of the merger agreement. See "The Merger — Bank of America's Reasons for the Merger; Recommendation of the Merger by the Bank of America Board of Directors" on page 32 for a more detailed discussion of the Bank of America board of directors' recommendation.

The Bank of America board of directors also has unanimously approved the proposals to increase the authorized number of shares of Bank of America common stock and adopt the Amended Stock Plan. The Bank of America board of directors determined that the proposals are advisable and in the best interests of Bank of America and its stockholders. The Bank of America board of directors unanimously recommends that you vote "FOR" the proposals to increase the authorized number of shares of Bank of America common stock and to adopt the Amended Stock Plan.

Attending the Meeting

All Bank of America stockholders, including stockholders of record and stockholders who hold their shares through banks, brokers, nominees or any other holder of record, are invited to attend the Bank of America special meeting. Stockholders of record can vote in person at the special meeting. If you are not a stockholder of record, you must obtain a proxy, executed in your favor, from the record holder of your shares, such as a broker, bank or other nominee, to be able to vote in person at the special meeting. If you plan to attend the special meeting, you must hold your shares in your own name or have a letter from the record holder of your shares confirming your ownership, and you must bring a form of personal photo identification with you in order to be admitted. We reserve the right to refuse admittance to anyone without proper proof of share ownership and without proper photo identification.

THE FLEETBOSTON SPECIAL MEETING

This section contains information from FleetBoston for FleetBoston stockholders about the special meeting of FleetBoston stockholders that has been called to consider and approve the merger agreement.

Together with this document, we are also sending you a notice of the FleetBoston special meeting and a form of proxy that is solicited by the FleetBoston board of directors. The FleetBoston special meeting will be held on March 17, 2004 at 10:00 a.m., local time, in the Auditorium on the ground floor of The Federal Reserve Bank of Boston, 600 Atlantic Avenue, Boston, Massachusetts.

Matters to Be Considered

The purpose of the FleetBoston special meeting is to vote on a proposal for approval of the merger agreement.

You also will be asked to vote upon a proposal to approve the adjournment of the special meeting, if necessary, to solicit additional proxies, in the event that there are not sufficient votes at the time of the special meeting to approve the merger agreement.

Proxies

Each copy of this document mailed to FleetBoston common stockholders is accompanied by a form of proxy with instructions for voting by mail, by telephone or through the Internet. If voting by mail, you should complete and return the proxy card accompanying this document to ensure that your vote is counted at the FleetBoston special meeting, or at any adjournment or postponement of the FleetBoston special meeting, regardless of whether you plan to attend the FleetBoston special meeting. You may also vote your shares by telephone or through the Internet. Information and applicable deadlines for voting by telephone or through the Internet are set forth in the enclosed proxy card instructions.

You may revoke your signed proxy card at any time before it is voted by signing and returning a proxy card with a later date, delivering a written revocation letter to FleetBoston's Secretary, or by attending the FleetBoston special meeting in person, notifying the Secretary, and voting by ballot at the FleetBoston special meeting. If you have voted your shares by telephone or through the Internet, you may revoke your prior telephone or Internet vote by recording a different vote, or by signing and returning a proxy card dated as of a date that is later than your last telephone or Internet vote.

Any stockholder entitled to vote in person at the FleetBoston special meeting may vote in person whether or not a proxy has been previously given, but the mere presence (without notifying the Secretary) of a stockholder at the FleetBoston special meeting will not constitute revocation of a previously given proxy.

Written notices of revocation and other communications about revoking your proxy should be addressed to:

FleetBoston Financial Corporation 100 Federal Street MA DE 10026A Boston, Massachusetts 02110 Attention: Gary A. Spiess Secretary

If your shares are held in street name by a broker, bank or other nominee, you should follow the instructions of your broker, bank or other nominee regarding the revocation of proxies.

All shares represented by valid proxies that we receive through this solicitation, and that are not revoked, will be voted in accordance with your instructions on the proxy card. If you make no specification on your proxy

card as to how you want your shares voted before signing and returning it, your proxy will be voted "FOR" approval of the merger agreement and "FOR" approval of the proposal to adjourn the special meeting, if necessary, to solicit additional proxies in the event that there are not sufficient votes at the time of the special meeting to approve the merger agreement. According to the FleetBoston bylaws, business to be conducted at a special meeting of stockholders may only be brought before the meeting pursuant to FleetBoston's notice of meeting. Accordingly, no matters other than the matters described in this joint proxy statement/prospectus will be presented for action at the FleetBoston special meeting or at any adjournment or postponement of the FleetBoston special meeting.

FleetBoston stockholders should NOT send FleetBoston stock certificates with their proxy cards. FleetBoston stockholders will be mailed a transmittal form with instructions on how to exchange their FleetBoston stock certificates for Bank of America stock certificates and cash instead of fractional shares, if applicable.

Solicitation of Proxies

We will bear the entire cost of soliciting proxies from you. In addition to solicitation of proxies by mail, we will request that banks, brokers, and other record holders send proxies and proxy material to the beneficial owners of FleetBoston common stock and secure their voting instructions, if necessary. We will reimburse the record holders for their reasonable expenses in taking those actions. We have also made arrangements with D.F. King & Co., Inc. to assist us in soliciting proxies and have agreed to pay them \$10,000 plus reasonable expenses for these services. If necessary, we may use several of our regular employees, who will not be specially compensated, to solicit proxies from FleetBoston stockholders, either personally or by telephone, facsimile, letter or other electronic means.

Record Date

The FleetBoston board of directors has fixed the close of business on January 26, 2004 as the record date for determining the FleetBoston stockholders entitled to receive notice of and/or vote at the FleetBoston special meeting. At that time, [] shares of FleetBoston common stock were outstanding.

Voting Rights and Vote Required

The presence, in person or by properly executed proxy, of the holders of a majority of the outstanding shares of FleetBoston common stock is necessary to constitute a quorum at the FleetBoston special meeting. Abstentions and broker non-votes will be counted for the purpose of determining whether a quorum is present. When we refer to broker non-votes, we are referring to unvoted proxies submitted by brokers, who are not able to vote on the merger agreement absent instructions from the applicable beneficial owner.

Approval of the merger agreement requires the affirmative vote of the holders of a majority of the outstanding shares of FleetBoston common stock entitled to vote at the FleetBoston special meeting. You are entitled to one vote for each share of FleetBoston common stock you held as of the record date. Holders of shares of FleetBoston preferred stock are not entitled to vote at the FleetBoston special meeting.

Because the affirmative vote of the holders of a majority of the outstanding shares of FleetBoston common stock entitled to vote at the FleetBoston special meeting is needed for us to proceed with the merger, the failure to vote by proxy or in person will have the same effect as a vote against the merger. Abstentions and broker non-votes also will have the same effect as a vote against the merger. Accordingly, the FleetBoston board of directors urges FleetBoston stockholders to: complete, date, and sign the accompanying proxy card and return it promptly in the enclosed postage-paid envelope if voting by mail; call the toll-free number listed in the proxy card instructions if voting by telephone; or access the Internet site listed in the proxy card instructions if voting through the Internet.

As of the record date:

- Directors and executive officers of FleetBoston and their affiliates had the right to vote [] shares of FleetBoston common stock, or []% of the outstanding FleetBoston common stock at that date.
- Directors and executive officers of Bank of America and their affiliates, including Bank of America (excluding the shares subject to the FleetBoston stock option described in "The Stock Option Agreements" on page 70), had the right to vote [] shares of FleetBoston common stock, or []% of the outstanding FleetBoston common stock on that date.

Recommendation of the FleetBoston Board of Directors

The FleetBoston board of directors has unanimously approved the merger agreement and the transactions it contemplates. The FleetBoston board of directors determined that the merger agreement and the transactions it contemplates are advisable and in the best interests of FleetBoston and its stockholders and unanimously recommends that you vote "FOR" approval of the merger agreement. See "The Merger — FleetBoston's Reasons for the Merger; Recommendation of the Merger by the FleetBoston Board of Directors" on page 34 for a more detailed discussion of the FleetBoston board of directors' recommendation.

Attending the Meeting

All FleetBoston stockholders, including stockholders of record and stockholders who hold their shares through banks, brokers, nominees or any other holder of record, are invited to attend the FleetBoston special meeting. Stockholders of record can vote in person at the special meeting. If you are not a stockholder of record, you must obtain a proxy executed in your favor, from the record holder of your shares, such as a broker, bank or other nominee, to be able to vote in person at the special meeting. If you plan to attend the special meeting, you must hold your shares in your own name or have a letter from the record holder of your shares confirming your ownership and you must bring a form of personal photo identification with you in order to be admitted. We reserve the right to refuse admittance to anyone without proper proof of share ownership and without proper photo identification.

INFORMATION ABOUT THE COMPANIES

Bank of America Corporation Bank of America Corporate Center 100 N. Tryon Street Charlotte, North Carolina 28255 (704) 386-8486

Bank of America Corporation is a Delaware corporation, a bank holding company and a financial holding company under federal law. Through its banking and various nonbanking subsidiaries, Bank of America provides a diversified range of banking and nonbanking financial services and products, primarily throughout the Mid-Atlantic (Maryland, Virginia and the District of Columbia), the Midwest (Illinois, Iowa, Kansas and Missouri), the Southeast (Florida, Georgia, North Carolina, South Carolina and Tennessee), the Southwest (Arizona, Arkansas, New Mexico, Oklahoma and Texas), the Northwest (Oregon, Idaho and Washington) and the West (California and Nevada) regions of the United States and in selected international markets. As of September 30, 2003, Bank of America had total consolidated assets of approximately \$737.1 billion, total consolidated deposits of approximately \$408.5 billion, and total consolidated stockholders' equity of approximately \$50.4 billion. The principal executive offices of Bank of America are located in the Bank of America Corporate Center, 100 N. Tryon Street, Charlotte, North Carolina 28255, and its telephone number is (704) 386-8486.

Additional information about Bank of America and its subsidiaries is included in documents incorporated by reference in this joint proxy statement/prospectus. See "Where You Can Find More Information" on page 111.

FleetBoston Financial Corporation 100 Federal Street Boston, Massachusetts 02110 (617) 434-2200

FleetBoston is a Rhode Island corporation and a diversified financial services company offering a comprehensive array of financial solutions to its customers. FleetBoston's three major domestic business lines are Personal Financial Services, Regional Commercial Financial Services and Investment Management, and National Commercial Financial Services. FleetBoston's other lines of business are International Banking and Capital Markets. As of September 30, 2003, FleetBoston had total consolidated assets of approximately \$196.4 billion, total consolidated deposits of approximately \$132.5 billion and total consolidated stockholders' equity of approximately \$17.6 billion. The principal executive offices of FleetBoston are located at 100 Federal Street, Boston, Massachusetts 02110, and its telephone number is (617) 434-2200.

On August 28, 2003, FleetBoston announced its plan to acquire Progress Financial Corporation. As of September 30, 2003, Progress had total consolidated assets of approximately \$1.2 billion, total consolidated deposits of approximately \$746.6 million and total consolidated stockholders' equity of approximately \$64.3 million. Based on the exchange ratio formula in the merger agreement with Progress and FleetBoston's recent common stock price of \$[] as of January [], 2004, completion of the Progress merger will require the issuance of less than [] million shares of FleetBoston common stock. FleetBoston expects to complete its acquisition of Progress before completing the merger with Bank of America.

Additional information about FleetBoston and its subsidiaries is included in documents incorporated by reference in this joint proxy statement/prospectus. See "Where You Can Find More Information" on page 111.

THE MERGER

The following discussion contains material information pertaining to the merger. This discussion is subject, and qualified in its entirety by reference, to the merger agreement, stock option agreements and financial advisor opinions attached as Appendices to this document. We encourage you to read and review those documents as well as the discussion in this document.

General

The next sections of this document, "The Merger Agreement" on pages 63 through 69 and "The Stock Option Agreements" on pages 70 through 73, have additional and more detailed information regarding the legal documents that govern the merger, including information about the conditions to completion of the merger and the provisions for terminating or amending the merger agreement.

We are furnishing this document to Bank of America stockholders and FleetBoston stockholders in connection with the solicitation of proxies by the boards of directors of Bank of America and FleetBoston for use at their respective special meetings of stockholders and any adjournment or postponement of their respective special meetings.

Structure

The merger agreement provides for the merger of FleetBoston with and into Bank of America. Bank of America will be the surviving corporation upon completion of the merger.

Upon completion of the merger, FleetBoston stockholders other than FleetBoston or Bank of America will have the right to receive 0.5553 of a share of Bank of America common stock for each share of FleetBoston common stock that they hold immediately prior to the merger. If the number of shares of common stock of Bank of America or FleetBoston changes before the merger is completed because of a reorganization, recapitalization, reclassification, stock dividend, stock split, reverse stock split, or other similar event, then an appropriate and proportionate adjustment will be made to the exchange ratio. FleetBoston stockholders will receive cash instead of any fractional shares of Bank of America common stock that would have otherwise been issued at the completion of the merger. Each share of each series of FleetBoston preferred stock issued and outstanding immediately prior to the completion of the merger, except for specified shares of FleetBoston preferred stock held by FleetBoston and Bank of America, will be converted into a share of a series of Bank of America preferred stock with the same terms (to the fullest extent possible) as the corresponding FleetBoston preferred stock

Upon completion of the merger, holders of Bank of America common stock immediately prior to the merger will own approximately 72%, and holders of FleetBoston common stock immediately prior to the merger will own approximately 28%, of the outstanding Bank of America common stock.

Bank of America will account for the merger as a purchase for financial reporting purposes. The merger has been structured to qualify as a "reorganization" within the meaning of Section 368(a) of the Internal Revenue Code for federal income tax purposes, and it is a condition to our respective obligations to complete the merger that Bank of America and FleetBoston each receive a legal opinion to that effect. Bank of America and FleetBoston may alter the method of effecting the combination of the companies if requested by either company and consented to by the other company (such consent not to be unreasonably withheld). However, the change will not be made if it alters or changes the number or type of shares of Bank of America stock into which shares of FleetBoston stock will be converted, adversely affects the tax treatment of FleetBoston stockholders pursuant to the merger, or materially impedes or delays completion of the merger.

Background of the Merger

The FleetBoston board of directors has periodically discussed and reviewed FleetBoston's business, strategic direction, performance and prospects in the context of developments in the financial services industry and the competitive landscape in the markets in which FleetBoston operates and elsewhere. The FleetBoston board of directors has also at times discussed with senior management various potential strategic alternatives involving possible acquisitions or business combinations that could complement and enhance the company's competitive strengths and strategic position. Also, senior management of FleetBoston has, from time to time, had informal discussions with representatives of other financial institutions regarding industry trends and issues and exploratory discussions of the potential benefits and issues arising from possible combinations among various industry players including FleetBoston.

Bank of America's board of directors and senior management also regularly review the financial services industry environment, including the trend towards consolidation in the industry, and periodically discuss ways in which to enhance the company's competitive position. Senior management of Bank of America have, over time, considered the possibility of acquisitions and strategic combinations with a variety of financial institutions and the potential strategic fit with such institutions based on their lines of businesses, their management and employee cultures and their geographic locations.

During 2003, senior management of FleetBoston continued to have informal discussions with various financial institutions concerning possible strategic opportunities. Included were discussions with senior management of Bank of America. Over time, these discussions increased the interest of FleetBoston senior management in the strategic benefits of a possible transaction with Bank of America, based on their assessment of FleetBoston's strategies, competitive position and prospects, Bank of America's and its senior management's track record and the apparent fit between the people, business models and cultures of the two institutions. During this period, FleetBoston senior management periodically apprised the FleetBoston board of directors of its informal discussions, including those with Bank of America. Although no decision was made to pursue any business combination at that time, the FleetBoston board of directors determined that exploratory discussions should continue.

Discussions continued periodically, and during September and early October 2003, Messrs. Gifford (Chairman and Chief Executive Officer of FleetBoston), McQuade (President and Chief Operating Officer of FleetBoston) and Sarles (Vice Chairman and Chief Administrative Officer of FleetBoston) spoke with Messrs. Lewis (Chairman, President and Chief Executive Officer of Bank of America) and Hance (Vice Chairman and Chief Financial Officer of Bank of America) on several occasions, during which the participants began to discuss more specifically the potential benefits of a transaction and some of the key issues in a possible transaction, including governance and key management arrangements for a combined company and the importance of continued commitment to New England communities and other key FleetBoston constituencies, as well as a broad range of possible values potentially to be offered to FleetBoston stockholders. Senior management of FleetBoston reported to the FleetBoston board of directors regarding the status of discussions, and although no conclusion was reached that FleetBoston should pursue any particular strategic alternative, Mr. Gifford was authorized to engage in further discussions. During this period, senior management of Bank of America reported to the Executive Committee of its board of directors on the status of discussions with FleetBoston, and the Executive Committee authorized Mr. Lewis to have further conversations with FleetBoston.

On October 21, 2003, senior management of FleetBoston reviewed with the FleetBoston board of directors its most recent discussions regarding possible strategic alternatives for FleetBoston. Based on this review of the most recent discussions with Bank of America and others, the FleetBoston board of directors authorized FleetBoston management to pursue a potential business combination transaction with Bank of America. Senior management of FleetBoston so advised Wachtell, Lipton, Rosen & Katz, its outside counsel, and arrangements were discussed to facilitate mutual due diligence and management discussions between the parties. Senior management of FleetBoston also informed Bank of America senior management of the outcome of the most recent meeting of the FleetBoston board of directors, and advised Bank of America that FleetBoston was

prepared to engage in a discussion with respect to the consideration to be received by FleetBoston stockholders in a business combination. On October 22, 2003, the Bank of America board of directors authorized Bank of America management to pursue, along with Cleary, Gottlieb, Steen & Hamilton, its outside counsel, negotiations with FleetBoston

Also on October 22, 2003, Bank of America and FleetBoston entered into a confidentiality agreement and continued discussions among senior management relating to the key terms of a transaction. The terms discussed included a stock-for-stock merger with a fixed exchange ratio. After further discussion, Mr. Lewis and Mr. Gifford determined that, assuming satisfactory conclusion of due diligence and negotiation of the terms of a definitive agreement, they were each prepared to present to their respective boards of directors a proposed merger with an exchange ratio that reflected an approximately \$45 per share valuation for each outstanding share of FleetBoston common stock. This determination, arrived at through arms' length negotiations, took into account, among other things, the impact of FleetBoston's unique franchise, the companies' respective contributions to the business, operations, financial condition and prospects of the combined company, anticipated cost savings and the merger consideration on the combined company's earnings and prospects. Other terms discussed included strong commitments by the combined company to FleetBoston representation on the board of directors and in senior management, as well as commitments to employment, charitable giving and corporate presence in Boston, Massachusetts and New England, as well as other FleetBoston communities. At the conclusion of these discussions, Messrs. Lewis and Gifford reached a general understanding on these key terms (which terms were reflected in the merger agreement as ultimately executed, with the final exchange ratio being calculated based on Bank of America common stock price at the close of trading that day) subject to continuing due diligence, legal documentation and board approval, and agreed that the two companies would work expeditiously towards finalizing a transaction.

The parties and their counsel met in New York City early on Friday, October 24, 2003 for mutual confidential due diligence and management discussions. Counsel for the parties continued discussions regarding the legal documentation for the transaction. The FleetBoston board of directors met telephonically on Friday afternoon and was apprised of the status of discussions, anticipated pricing and the progress of due diligence investigations. Later that day, senior managements of the companies reached an understanding as to the 0.5553 exchange ratio, which reflected a \$45 per share valuation for shares of FleetBoston common stock based on the October 22, 2003 closing price for Bank of America common stock. FleetBoston recognized that the fixed exchange ratio by its nature would not be adjusted to compensate for possible declines in Bank of America's stock price and that the terms of the merger agreement would not include any price-based termination right or "collar" provisions, and determined not to pursue such provisions in the context of the merger discussions in view of the factors discussed under "—FleetBoston's Reasons for the Merger; Recommendation of the Merger by the FleetBoston Board of Directors" on page 34. FleetBoston and Bank of America asked their outside financial advisors to be prepared to discuss the transaction with the respective board of directors on Sunday, October 26, 2003.

Due diligence continued throughout the weekend as the parties and their counsel continued to discuss the terms of the definitive merger agreement and other related agreements. The parties also negotiated the terms of post-closing employment arrangements with Messrs. McQuade, Moynihan, Sarles and Warner of FleetBoston and commenced drafting agreements reflecting those terms.

At mid-afternoon on Sunday, October 26, 2003, the Bank of America board of directors met with members of Bank of America senior management and Bank of America's outside financial advisor, Goldman Sachs. Bank of America senior management reviewed with the Bank of America board of directors information regarding Bank of America, FleetBoston and the terms of the proposed transaction. Bank of America senior management apprised the Bank of America board of directors of its due diligence investigations of FleetBoston. Goldman Sachs reviewed with the Bank of America board of directors additional information, including financial information regarding the two companies and the transaction. Goldman Sachs rendered to the Bank of America board of directors its oral opinion (subsequently confirmed in writing) that, as of the date of its opinion and based

upon and subject to the considerations described in its opinion and other matters as Goldman Sachs considered relevant, the proposed exchange ratio was fair, from a financial point of view, to Bank of America. Bank of America's general counsel discussed with the board of directors the legal standards applicable to its decisions and actions with respect to the proposed transaction and reviewed the legal terms of the proposed merger. Following review and discussion among the members of the Bank of America board of directors, the 14 members of the board of directors in attendance (three directors being unable to attend the meeting) voted unanimously to approve the merger with FleetBoston. At a subsequent meeting, the full Bank of America board of directors, with all directors present, unanimously reaffirmed and ratified the actions taken at the October 26, 2003 meeting.

Late in the afternoon on Sunday, October 26, 2003, the FleetBoston board of directors met with members of FleetBoston's senior management and FleetBoston's outside legal and financial advisors. FleetBoston senior management reviewed with the FleetBoston board of directors information regarding Bank of America, FleetBoston and the terms of the proposed transaction. FleetBoston senior management apprised the FleetBoston board of directors of its due diligence investigations of Bank of America. Morgan Stanley reviewed its financial analysis regarding the proposed merger with the FleetBoston board of directors and rendered to the FleetBoston board of directors its oral opinion (subsequently confirmed in writing) that, as of the date of its opinion and based on and subject to the considerations in its opinion, the proposed exchange ratio was fair, from a financial point of view, to holders of FleetBoston common stock. Representatives of Wachtell, Lipton, Rosen & Katz discussed with the board of directors the legal standards applicable to its decisions and actions with respect to the proposed transaction and reviewed the legal terms of the proposed merger. Management then reviewed with the FleetBoston board of directors human resources and employment matters arising in connection with the transaction. Following review and discussion among the members of the FleetBoston board of directors, the board of directors voted unanimously to approve the merger with Bank of America.

Following approval of each board of directors, the parties and their counsel continued to finalize and document the legal terms of the definitive agreements for the transaction, and later the merger agreement, the stock option agreements and the employment agreements were executed by the parties to the agreements. The transaction was announced in a joint press release early on the morning of October 27, 2003.

Bank of America's Reasons for the Merger; Recommendation of the Merger by the Bank of America Board of Directors

The Bank of America board of directors expects the merger to create the first banking institution with a truly national scope, with an unrivaled presence in America's growth and wealth markets. The merger would establish a new Bank of America that will serve approximately 33 million consumer relationships, with leading market shares throughout the Northeast, Southeast, Midwest, Southwest and West regions of the United States. The Bank of America board of directors also expects the combined company to feature the most extensive and convenient delivery network through more than 5,600 retail banking offices, more than 16,500 ATMs, leading online and electronic bill pay services, and 24-hour telephone banking.

The Bank of America board of directors consulted with Bank of America management as well as financial and legal advisors and determined that the merger is in the best interests of Bank of America and Bank of America stockholders. In reaching its conclusion to approve the merger agreement, the Bank of America board considered the following factors as generally supporting its decision to enter into the merger agreement:

- its understanding of Bank of America's business, operations, financial condition, earnings and prospects and of FleetBoston's business, operations, financial condition, earnings and prospects, including FleetBoston's unique franchise in the regions in which it operates;
- its understanding of the current and prospective environment in which Bank of America and FleetBoston operate, including national and local economic conditions, the competitive environment for financial institutions generally and continuing consolidation in the financial services industry, and the likely effect of these factors on Bank of America in light of, and in absence of, the proposed transaction;

- the review by the Bank of America board of directors with its legal and financial advisors of the structure of the merger and the financial and other terms of the
 merger and stock option agreements, including the exchange ratio, the expectation of Bank of America's legal advisors that the merger will qualify as a transaction of
 a type that is generally tax-free to stockholders for U.S. federal income tax purposes and, based on the exchange ratio and assuming continuation of Bank of
 America's current per share dividend rate of \$0.80 per quarter, an anticipated annual dividend rate increase of approximately 27% for holders of FleetBoston
 common stock (from \$1.40 to \$1.78 annually per share of FleetBoston common stock equivalent);
- the fact that the complementary nature of the respective customer bases, business products and skills of Bank of America and FleetBoston could result in
 opportunities to obtain synergies as products are cross-marketed and distributed over broader customer bases and best practices are compared and applied across
 businesses;
- the fact that FleetBoston's and Bank of America's branch franchises operate principally in different geographic markets, which the Bank of America board of
 directors believed to present a desirable strategic opportunity for expansion of its existing presence and market share. In particular, the Bank of America board of
 directors considered that the resulting institution's branch network and franchise would extend across geographies that include some of the most affluent and
 populous regions in the country:
- the potential expense saving opportunities to be achieved by eliminating duplicative technology and operations functions, facility consolidations, staffing reductions and purchasing efficiencies, currently estimated by management to be approximately \$1.1 billion per year after-tax when fully phased in after the merger, as well as the potential incremental revenue opportunities, currently estimated by management to be approximately \$195 million per year by 2005, in each case based on a review by management of the business and operations of Bank of America and FleetBoston including an assessment of the two companies' computer systems, personnel, premises and service contracts to determine where redundancies exist, and management's experience in the integration of businesses in connection with transactions similar to the merger;
- the fact that application of such potential expense savings and other transaction-related assumptions and adjustments (including assumed repurchases of 67 million shares of Bank of America common stock through 2004 and 23 million shares in 2005) to the combined net income forecasts for Bank of America and FleetBoston made by various brokerage firms and published as consensus estimates by First Call would result in projected net income for the combined company of \$12.643 billion in 2004 (reflecting six months' income contribution from FleetBoston) and, applying an assumed 9% annual growth rate to the 2004 First Call consensus estimates, \$16.060 billion in 2005, or \$7.10 and \$7.97 per share of Bank of America common stock in 2004 and 2005, respectively, representing approximately 2% dilution and approximately 1% accretion in earnings per share for such periods when compared to the estimates for Bank of America common stock without giving effect to the transaction:
- · the scale, scope, strength and diversity of operations, product lines and delivery systems that could be achieved by combining Bank of America and FleetBoston;
- the reports of Bank of America management and the financial presentation by Goldman Sachs to Bank of America's board of directors concerning the operations, financial condition and prospects of FleetBoston and the expected financial impact of the merger on the combined company, including pro forma assets, earnings, deposits and, in the case of the reports of Bank of America's management, the regulatory capital ratios;
- the proposed board and management arrangements which would position the combined company with strong leadership and experienced operating management;
- the employment arrangements with key FleetBoston senior executives which would help assure the continuity of management, the likelihood of a successful integration and the successful operation of the combined company;

- the likelihood that the regulatory approvals needed to complete the transaction will be obtained;
- the historical and current market prices of Bank of America common stock and FleetBoston common stock, as well as the financial analyses prepared by Goldman Sachs; and
- the opinion delivered to the Bank of America board of directors by Goldman Sachs to the effect that, as of the date of the opinion and based upon and subject to the considerations described in its opinion and other matters as Goldman Sachs considered relevant, the exchange ratio was fair, from a financial point of view, to Bank of America

The Bank of America board of directors also considered potential risks associated with the merger in connection with its deliberations of the proposed transaction, including the challenges of integrating FleetBoston's businesses, operations and workforce with those of Bank of America, the need to obtain FleetBoston stockholder, Bank of America stockholder and regulatory approvals in order to complete the transaction, the risks associated with achieving the anticipated cost savings, and the need and ability to retain key FleetBoston management.

The Bank of America board of directors also considered the fact that the combined institution is expected to hold aggregate deposits approaching the 10% national acquisition limit under U.S. federal law, and that Bank of America's ability to make further acquisitions in the United States may be restricted as a result.

The Bank of America board of directors was aware of the allocation of seven of nineteen (or 37%) of the seats on the board of the combined company to FleetBoston directors, which is greater than the pro forma percentage ownership of the combined company by FleetBoston common shareholders (approximately 28%) and the percentage contributions of FleetBoston to the combined company based on various financial metrics, including total assets and estimated 2004 net income (in each case approximately 21%). The Bank of America board of directors understood that board participation at this level was a condition to the willingness of FleetBoston to enter into the merger agreement and deemed it appropriate given the strategic importance of the transaction to Bank of America.

The Bank of America board of directors considered all of these factors as a whole and, on balance, concluded that they supported a favorable determination to enter into the merger agreement.

The foregoing discussion of the information and factors considered by the Bank of America board of directors is not exhaustive, but includes all material factors considered by the Bank of America board of directors. In view of the wide variety of factors considered by the Bank of America board of directors in connection with its evaluation of the merger and the complexity of these matters, the Bank of America board of directors did not consider it practical to, nor did it attempt to, quantify, rank or otherwise assign relative weights to the specific factors that it considered in reaching its decision. The Bank of America board of directors evaluated the factors described above and reached a consensus that the merger was advisable and in the best interests of Bank of America and its stockholders. In considering the factors described above, individual members of the Bank of America board of directors may have given different weights to different factors.

The Bank of America board of directors unanimously approved the merger agreement and unanimously recommends that Bank of America stockholders vote "FOR" adoption of the merger agreement.

FleetBoston's Reasons for the Merger; Recommendation of the Merger by the FleetBoston Board of Directors

The FleetBoston board of directors expects the merger to create the first banking institution with a truly national scope, with an unrivaled presence in America's growth and wealth markets. The merger would establish a new combined company that will serve approximately 33 million consumer relationships, with leading market shares throughout the Northeast, Southeast, Midwest, Southwest and West regions of the United States. The

FleetBoston board of directors also expects the combined company to feature the most extensive and convenient delivery network through more than 5,600 retail banking offices, more than 16,500 ATMs, leading online and electronic bill pay services, and 24-hour telephone banking.

The FleetBoston board of directors consulted with FleetBoston management as well as financial and legal advisors and determined that the merger is in the best interests of FleetBoston and FleetBoston stockholders. In reaching its conclusion to approve the merger agreement, the FleetBoston board considered the following factors as generally supporting its decision to enter into the merger agreement:

- its understanding of FleetBoston's business, operations, financial condition, earnings and prospects and of Bank of America's business, operations, financial condition, earnings and prospects;
- its understanding of the current and prospective environment in which FleetBoston and Bank of America operate, including national and local economic conditions, the competitive environment for financial institutions generally, the trend toward consolidation in the financial services industry, and the likely effect of these factors on FleetBoston in light of, and in absence of, the proposed transaction;
- the fact that, based on the closing price of Bank of America common stock on October 24, 2003, the value of the per share merger consideration to be received by FleetBoston stockholders in the merger represented a premium of approximately 43% over the closing price of FleetBoston common stock on the NYSE on October 24, 2003, and a premium of approximately 45% over the weighted average closing price of FleetBoston common stock on the NYSE for the thirty trading days prior to the approval of the transaction by the FleetBoston board;
- the review by the FleetBoston board of directors with its legal and financial advisors of the structure of the merger and the financial and other terms of the merger and stock option agreements, including the exchange ratio, the expectation of FleetBoston's legal advisors that the merger will qualify as a transaction of a type that is generally tax-free to stockholders for U.S. federal income tax purposes and, based on the exchange ratio and assuming continuation of Bank of America's current per share dividend rate of \$0.80 per quarter, an anticipated annual dividend rate increase of approximately 27% for holders of FleetBoston common stock (from \$1.40 to \$1.78 annually per share of FleetBoston common stock equivalent);
- its belief in Bank of America's serious commitment to Boston and New England as well as FleetBoston's other communities and other constituencies, as evidenced by the employment, charitable giving, board composition and business location provisions of the merger agreement;
- the fact that the complementary nature of the respective customer bases, business products and skills of FleetBoston and Bank of America could result in
 opportunities to obtain synergies as products are cross-marketed and distributed over broader customer bases and best practices are compared and applied across
 businesses:
- the fact that FleetBoston's and Bank of America's branch franchises operate principally in different geographic markets, which the FleetBoston board of directors believed to present a desirable strategic opportunity for expansion of its existing presence and market share. In particular, the FleetBoston board of directors considered that the resulting institution's branch network and franchise would extend across geographies that include some of the most affluent and populous regions in the country;
- the potential cost saving opportunities to be achieved by eliminating duplicative technology and operations functions, facility consolidations, staffing reductions and purchasing efficiencies, currently estimated to be approximately \$1.1 billion per year after-tax when fully phased in after the merger, as well as the potential incremental revenue opportunities, currently estimated to be approximately \$195 million per year by 2005, in each case based on a review of the business and operations of Bank of America and FleetBoston including an assessment of the two companies' computer systems, personnel, premises and service contracts to determine where redundancies exist, and experience in the integration of businesses in connection with transactions similar to the merger;

- the fact that application of such potential expense savings and other transaction-related assumptions and adjustments (including assumed repurchases of 67 million shares of Bank of America common stock through 2004 and 23 million shares in 2005) to the combined net income forecasts for Bank of America and FleetBoston made by various brokerage firms and published as consensus estimates by First Call would result in projected net income for the combined company of \$12.643 billion in 2004 (reflecting six months' income contribution from FleetBoston) and, applying an assumed 9% annual growth rate to the 2004 First Call Consensus estimates, \$16.060 billion in 2005, or \$7.10 and \$7.97 per share of Bank of America common stock in 2004 and 2005, respectively, representing approximately 2% dilution and approximately 1% accretion in earnings per share for such periods when compared to the estimates for Bank of America common stock without giving effect to the transaction:
- · the scale, scope, strength and diversity of operations, product lines and delivery systems that could be achieved by combining FleetBoston and Bank of America;
- the reports of FleetBoston management and the financial presentation by Morgan Stanley to FleetBoston's board of directors concerning the operations, financial condition and prospects of Bank of America and the expected financial impact of the merger on the combined company, including pro forma assets, earnings, deposits and, in the case of the reports of FleetBoston's management, the regulatory capital ratios;
- the proposed board and management arrangements of the combined company which would position the combined company with strong leadership and experienced
 operating management;
- the employment arrangements with key FleetBoston senior executives which would help assure the continuity of management, the likelihood of a successful integration and the successful operation of the combined company;
- the likelihood that the regulatory approvals needed to complete the transaction will be obtained;
- the historical and current market prices of Bank of America common stock and FleetBoston common stock, as well as comparative valuation analyses for the two
 companies prepared by Morgan Stanley; and
- the opinion delivered to the FleetBoston board of directors by Morgan Stanley to the effect that, as of the date of the opinion and based upon and subject to the considerations in its opinion, the exchange ratio was fair, from a financial point of view, to holders of FleetBoston common stock.

The FleetBoston board of directors also considered potential risks associated with the merger in connection with its deliberations of the proposed transaction, including the challenges of integrating FleetBoston's businesses, operations and workforce with those of Bank of America, the need to obtain Bank of America stockholder, FleetBoston stockholder and regulatory approvals in order to complete the transaction, the risks associated with achieving the anticipated cost savings, the different litigation, regulatory and reputational risks inherent in the varied businesses in which Bank of America is active, and the need and ability to retain key FleetBoston management.

The FleetBoston board of directors also considered the fact that the combined institution is expected to hold aggregate deposits approaching the 10% national acquisition limit under U.S. federal law, and that the combined institution's ability to make further acquisitions in the United States may be restricted as a result. In addition, the board considered the interest that FleetBoston executive officers and directors have with respect to the merger in addition to their interests as holders of FleetBoston common stock and the risk that these interests might influence their decision with respect to the merger. See "The Merger — FleetBoston's Directors and Officers Have Financial Interests in the Merger" on page 61.

The FleetBoston board also considered that the fixed exchange ratio, by its nature, would not adjust upwards to compensate for declines, or downwards to compensate for increases, in Bank of America's stock price prior to completion of the merger, and that the terms of the merger agreement did not include stock-price-based

termination rights or "collar" provisions that would be triggered by a decrease in the value of the merger consideration implied by the Bank of America stock price. The FleetBoston board determined that this structure was appropriate and the risk acceptable in view of:

- its focus on the relative intrinsic values and financial performances of the merger partners and the percentage of the combined company to be owned by former FleetBoston stockholders, in addition to the market premium at the time of signing;
- its recognition that price-based termination rights and collar provisions are not the norm in merger agreements between major banking institutions; the merger
 agreement's inclusion of other structural protections such as the ability to terminate the merger agreement in the event of a material adverse change at Bank of
 America;
- the reciprocal nature of collars, meaning that a collar would limit the ability of FleetBoston stockholders to participate in the potential upside in Bank of America's stock price prior to completion of the merger; and
- the structure of most collars providing only limited protection against downward price declines, and only through the date on which the merger is completed.

The FleetBoston board of directors considered all of these factors as a whole and, on balance, concluded that they supported a favorable determination to enter into the merger agreement.

The foregoing discussion of the information and factors considered by the FleetBoston board of directors is not exhaustive, but includes all material factors considered by the FleetBoston board of directors in connection with its evaluation of the merger and the complexity of these matters, the FleetBoston board of directors did not consider it practical to, nor did it attempt to, quantify, rank or otherwise assign relative weights to the specific factors that it considered in reaching its decision. The FleetBoston board of directors evaluated the factors described above, including asking questions of FleetBoston management and FleetBoston legal and financial advisors, and reached consensus that the merger was in the best interests of FleetBoston and FleetBoston stockholders. In considering the factors described above, individual members of the FleetBoston board of directors may have given different weights to different factors.

The FleetBoston board of directors determined that the merger, the merger agreement and the transactions contemplated by the merger agreement are in the best interests of FleetBoston and its stockholders. Accordingly, the FleetBoston board of directors unanimously approved the merger agreement and unanimously recommends that FleetBoston stockholders vote "FOR" adoption of the merger agreement.

Analyses of Financial Advisors

FleetBoston engaged Morgan Stanley to act as its financial advisor and Bank of America engaged Goldman Sachs and Banc of America Securities to act as its financial advisors in connection with the merger based on their experience and expertise. Morgan Stanley, Goldman Sachs and Banc of America Securities are internationally recognized investment banking firms that have substantial experience in transactions similar to the merger. Banc of America Securities is a subsidiary of Bank of America.

Opinion of Goldman Sachs

On October 26, 2003, Goldman Sachs, financial advisor to Bank of America, delivered its oral opinion to the Bank of America board of directors, subsequently confirmed in writing, that, as of the date of the opinion and based upon and subject to the considerations described in its opinion and other matters as Goldman Sachs considered relevant, the exchange ratio of 0.5553 of a share of Bank of America common stock to be issued in respect of each share of FleetBoston common stock pursuant to the merger agreement was fair to Bank of America from a financial point of view.

The full text of the written opinion of Goldman Sachs, dated October 27, 2003, which sets forth, among other things, the assumptions made, procedures followed, matters considered and limitations on the review undertaken by Goldman Sachs in connection with the opinion, is attached to this document as Appendix D. Holders of Bank of America common stock are urged to, and should, read this opinion carefully and in its entirety. Goldman Sachs provided its opinion for the information and assistance of the Bank of America board of directors in connection with its consideration of the transaction contemplated by the merger agreement. The Goldman Sachs opinion is not a recommendation as to how any holder of Bank of America common stock should vote with respect to the merger or any related matter. The opinion of Goldman Sachs will not reflect any developments that may occur or may have occurred after the date of its opinion and prior to the completion of the merger. Bank of America does not currently expect that it will request an updated opinion from Goldman Sachs.

In connection with its opinion, Goldman Sachs reviewed, among other things:

- the merger agreement;
- annual reports to stockholders and annual reports on Form 10-K of Bank of America and FleetBoston for the five fiscal years ended December 31, 2002;
- interim reports to stockholders and quarterly reports on Form 10-Q of Bank of America and FleetBoston;
- · other communications from Bank of America and FleetBoston to their respective stockholders;
- First Call research analyst estimates for each of Bank of America and FleetBoston approved by the management of Bank of America for use in connection with the opinion (First Call is a database owned and operated by Thomson Financial, which contains broker-sourced research, earnings estimates, equity and fixed income ownership information, insider trading information, and corporate news releases);
- · internal financial analyses and forecasts for Bank of America prepared by the management of Bank of America;
- · internal financial analyses and forecasts for FleetBoston prepared by the management of FleetBoston; and
- financial analyses and forecasts for the combined company on a pro forma basis prepared by the management of Bank of America, including cost savings and
 operating synergies projected by the management of Bank of America to result from the merger.

Goldman Sachs also held discussions with members of the senior management of each of Bank of America and FleetBoston regarding their assessment of the strategic rationale for, and the potential benefits of, the merger and the past and current business operations, financial condition and future prospects of Bank of America and FleetBoston.

In addition, Goldman Sachs:

- · reviewed the reported price and trading activity for FleetBoston common stock and Bank of America common stock;
- compared financial and stock market information for Bank of America and FleetBoston with similar information for other selected companies, the securities of which
 are publicly traded;
- · reviewed the financial terms of selected recent business combinations in the banking industry specifically and in other industries generally; and
- performed other studies and analyses as Goldman Sachs considered appropriate.

Goldman Sachs relied upon the accuracy and completeness of all of the financial, accounting, legal, tax and other information discussed with or reviewed by it and assumed the accuracy and completeness of this information for purposes of rendering its opinion. In that regard, Goldman Sachs assumed, with the consent of

the Bank of America board of directors, that the forecasts reviewed by Goldman Sachs were reasonably prepared on a basis reflecting the best currently available estimates and judgments of each of Bank of America and FleetBoston, as the case may be. Goldman Sachs also assumed that all governmental, regulatory or other consents and approvals necessary for the completion of the merger would be obtained without any adverse effect on Bank of America or FleetBoston or on the expected benefits of the transaction in any way material to its analysis. Goldman Sachs is not an expert in the evaluation of loan and lease portfolios for purposes of assessing the adequacy of the allowances for losses with respect to such portfolios and, accordingly, Goldman Sachs assumed that the allowances for losses are in the aggregate adequate to cover those losses. In addition, Goldman Sachs did not review individual credit files nor did it make an independent evaluation or appraisal of the assets and liabilities (including any derivative or off-balance-sheet assets and liabilities) of Bank of America or FleetBoston or any of their respective subsidiaries, and Goldman Sachs was not furnished with any such evaluation or appraisal. Goldman Sachs 'opinion does not address Bank of America's underlying business decision to engage in the merger. In addition, Goldman Sachs is not expressing any opinion as to the prices at which the shares of Bank of America common stock or FleetBoston common stock will trade at any time.

The following is a summary of the material financial analyses presented by Goldman Sachs to the Bank of America board of directors in connection with providing its opinion to the Bank of America board of directors. The following summary, however, does not purport to be a complete description of the financial analyses performed by Goldman Sachs. The order of analyses described does not represent relative importance or weight given to the analyses performed by Goldman Sachs. Some of the summaries of the financial analyses include information presented in tabular format. The tables must be read together with the full text of each summary and alone are not a complete description of Goldman Sachs' financial analyses. Except as otherwise noted, the following quantitative information, to the extent that it is based on market data, is based on market data as it existed on or before October 24, 2003, and is not necessarily indicative of current market conditions.

Summary of Selected Analyses. The following is a brief overview of some of the financial analyses performed by Goldman Sachs. Goldman Sachs did not present a separate overview of its analyses to the Bank of America board of directors, as each individual analysis was discussed in detail with the Bank of America board of directors. This overview does not include the underlying assumptions or descriptions of each analysis or all other analyses performed by Goldman Sachs in connection with the delivery of its opinion. Goldman Sachs did not attribute any particular weight, or equal weight, to any factor or analysis considered by it. In order to fully understand this overview and the Goldman Sachs analyses, please read the description of each analysis that follows below.

Implied Premium and Multiple Analysis

Goldman Sachs determined the offer price implied by the exchange ratio and Bank of America's stock price and analyzed this implied price as a percentage of FleetBoston's core deposits and a multiple to FleetBoston's latest twelve months' earnings per share, estimated 2004 GAAP earnings per share and estimated 2005 GAAP earnings per share, both with and without anticipated synergies, and to stated and tangible book value.

Historical Market Performance Analysis Goldman Sachs reviewed the historical trading prices of Bank of America and FleetBoston common stock and an index of other large banking companies and determined that, over the periods reviewed, FleetBoston common stock had generally outperformed Bank of America common stock and the index, and Bank of America common stock had generally outperformed the index.

Historical Exchange Ratio Analysis Goldman Sachs reviewed the ratios of the market prices of FleetBoston and Bank of America common stock over various periods preceding the announcement of the merger (ranging from the closing price on October 24, 2003 to the average for the one-year period preceding the announcement) and compared these ratios to the exchange ratio.

Public Market Comparison

large and medium-to-large capitalization U.S. banking companies.

Goldman Sachs reviewed information about selected banking mergers and acquisitions valued at more than \$5 billion since 1997, including multiples of merger consideration to the target's stated book value, tangible book value, latest twelve months' earnings per share and current-year I/B/E/S median estimated earnings per share, as well as the premiums represented by the merger consideration to core deposits and target stock price six days prior to announcement of the merger.

Pro Forma Merger Analysis

Goldman Sachs prepared illustrative pro forma analyses of the potential financial impact of the merger using a variety of assumptions described in more detail below, and estimated that the proposed transaction would be approximately 2.3% dilutive to GAAP earnings per share and approximately 1.0% dilutive to cash earnings per share in 2004 and approximately 0.6% accretive to GAAP earnings per share and approximately 2.7% accretive to cash earnings per share in 2005.

Contribution Analysis

Goldman Sachs estimated the pro forma share ownership in the combined company of FleetBoston shareholders and compared this to the

Using a variety of assumptions described in more detail below, Goldman Sachs derived a 2006 net present value per share of Bank of America common stock on a standalone basis and pro forma for the proposed merger.

as of October 24, 2003, common equity, tangible common equity, total assets and estimated net income for 2003, 2004 and 2005.

expected percentage contribution of FleetBoston to the combined company in a variety of financial measures, including market capitalization

Goldman Sachs compared financial and market information for Bank of America and FleetBoston to corresponding information for several

Implied Premium and Multiple Analysis. Based upon the exchange ratio and the \$81.03 closing market price of Bank of America common stock on October 22, 2003, Goldman Sachs calculated that each share of FleetBoston common stock would be converted into Bank of America common stock with an implied market value of \$45.00, representing a premium of 40.7% to the closing price per share of FleetBoston common stock of \$31.99 on October 22, 2003 and that the implied aggregate consideration of approximately \$47.4 billion represented a core deposit premium of 33.6%, based on core deposits of approximately \$103.1 billion. Goldman Sachs also analyzed the multiple of the implied offer price to:

- earnings per share (EPS) for the latest twelve months ended September 30, 2003;
- estimated earnings per share for 2004 and 2005, based on First Call mean estimates for 2004 and a long-term earnings growth rate of 9%, without any cost savings;
- estimated earnings per share for 2004 and 2005, based on First Call mean estimates for 2004 and a long-term earnings growth rate of 9%, with assumed fully phased-in after-tax cost savings of \$1.1 billion; and
- stated and tangible book value per share as of September 30, 2003.

The results of these analyses are summarized as follows:

Price/EPS — No Synergies	EPS	Price/EPS
Latest Twelve Months (9/30/2003)	\$ 2.00	22.5x
2004E	\$ 2.75	16.4x
2005E	\$ 3.00	15.0x
Price/EPS — Fully Phased-In Synergies	6 290	11 0
2004E	\$ 3.80	11.8x
2005E	\$ 4.14	10.9x
Price/Book Value	Book Value	Price/Book Value
Stated	\$ 16.46	2.7x
Tangible	\$ 12.13	3.7x

Historical Market Performance Analysis. Goldman Sachs reviewed and compared the historical daily trading prices of Bank of America common stock, FleetBoston common stock and a Bank Composite Index composed of common stock of Citigroup Inc., Wells Fargo & Company, J.P. Morgan Chase & Co., Wachovia Corporation, U.S. Bancorp, Bank One Corporation, Fifth Third Bancorp, BB&T Corporation, National City Corporation and SunTrust Banks, Inc. for the period from October 24, 2002 to October 24, 2003. The analysis indicated that for the period from October 24, 2002 to October 24, 2003, FleetBoston common stock generally outperformed both Bank of America common stock and the above index, and Bank of America common stock generally outperformed the Bank Composite Index or performed comparably with the Bank Composite Index.

Historical Exchange Ratio Analysis. Goldman Sachs reviewed the implied historical exchange ratios determined by dividing the closing price per share of FleetBoston common stock by the closing price per share of Bank of America common stock over the one-year period from October 24, 2002 through October 24, 2003. In addition, Goldman Sachs calculated the average of these historical daily exchange ratios for the one-week, three- month, six-month and one-year periods ended October 24, 2003. The following table presents the results of these calculations.

Period Average	Exchange Ratio
Merger	0.5553
October 24, 2003	0.3885
One-Week Average	0.3954
Three-Month Average	0.3889
Six-Month Average	0.3841
One-Year Average	0.3718

Public Market Comparison. Goldman Sachs reviewed and compared certain financial information for Bank of America and FleetBoston to corresponding financial information, ratios and public market multiples for the following large-capitalization and medium-to-large capitalization publicly traded corporations in the banking industry:

Large-Capitalization

- · Citigroup Inc.
- · Wells Fargo & Company
- J.P. Morgan Chase & Co.
- · Wachovia Corporation
- U.S. Bancorp
- Bank One Corporation
- · Fifth Third Bancorp

Medium-to-Large Capitalization

- BB&T Corporation
- · National City Corporation
- · SunTrust Banks, Inc.
- The PNC Financial Services Group, Inc.
- KeyCorp

Although none of the selected companies is directly comparable to Bank of America or FleetBoston, the companies included were chosen because they are publicly traded companies with operations that for purposes of analysis may be considered similar to certain operations of Bank of America and FleetBoston.

Goldman Sachs calculated and compared various financial multiples and ratios based on publicly available financial data as of June 30, 2003, information it obtained from SEC filings and Institutional Brokers Estimate System, or I/B/E/S estimates. I/B/E/S is a database owned and operated by Thomson Financial, which contains estimated and actual earnings, cash flows, dividends, sales, and pre-tax income data for companies in the United States, Europe, Asia and emerging markets. The multiples and ratios of Bank of America were calculated using the closing price of Bank of America common stock on October 24, 2003. The multiples and ratios of FleetBoston were calculated using the closing price of FleetBoston common stock on October 24, 2003. The multiples and ratios for each of the selected companies were based on the most recent publicly available information. With respect to the selected companies, Goldman Sachs calculated:

- the multiple of price to 2003 and 2004 I/B/E/S median estimated earnings;
- the ratio of price to 2004 I/B/E/S median estimated earnings, as a multiple of I/B/E/S median estimated earnings growth;
- the multiple of price to stated and tangible book value per share;
- · premium to core deposits;
- the ratio of tangible common equity to tangible assets; and
- dividend yield.

The following table presents the results of this analysis:

		Large-Capitalization Selected Banks		Medium-to-Large Capitalization Selected Banks		
	Range	Median	Range	Median	Bank of America	FleetBoston
Price/2003E Earnings	11.2x - 18.7x	13.9x	10.2x - 13.5x	13.1x	11.6x	13.1x
Price/2004E Earnings	11.2x – 16.6x	12.5x	10.8x - 12.6x	12.1x	11.3x	11.6x
2004E P/E to Growth	1.0x - 1.3x	1.2x	1.2x - 1.7x	1.4x	1.1x	1.3x
Price/Book (Stated)	1.6x - 3.9x	2.6x	1.6x - 2.1x	1.9x	2.5x	1.9x
Price/Book (Tangible)	2.0x - 4.4x	3.7x	2.0x - 3.4x	2.4x	3.3x	2.6x
Premium to Core Deposits	20.1% - 54.2%	29.7%	13.4% - 26.7%	18.5%	26.9%	13.9%
Tangible Common Equity/Tangible Assets	4.5% - 8.6%	6.0%	5.9% - 8.0%	6.6%	5.2%	6.7%
Dividend Yield	2.0% - 3.9%	3.1%	2.8% - 4.5%	3.9%	3.9%	4.4%

Selected Transactions Analysis. Goldman Sachs analyzed certain information relating to the following selected transactions in the banking industry since January 1, 1997 with values exceeding \$5 billion:

- First Bank System, Inc./U.S. Bancorp
- · NationsBank Corporation/Barnett Banks, Inc.
- · First Union Corporation/CoreStates Financial Corp.
- National City Corporation/First of America Bank Corporation
- Travelers Group, Inc./Citicorp
- NationsBank Corporation/BankAmerica Corporation
- Bank One Corporation/First Chicago NBD Corporation
- Norwest Corporation/Wells Fargo & Company
- Star Banc Corporation/Firstar Corporation
- SunTrust Banks, Inc./Crestar Financial Corporation
- Deutsche Bank AG/Bankers Trust Corporation
- · Fleet Financial Group, Inc./BankBoston Corporation

- Firstar Corporation/Mercantile Bancorporation
- HSBC Holdings plc/Republic New York Corporation
- · AmSouth Bancorporation/First American Corporation
- The Chase Manhattan Corporation/J.P. Morgan & Co.
- · FleetBoston/Summit Bancorp
- Firstar Corporation/U.S. Bancorp
- First Union Corporation/Wachovia Corporation

The following table compares information derived by Goldman Sachs with respect to the ranges, averages and medians relating to the implied value received by stockholders in the second-named merger partner (target) for these selected transactions:

	Selected Transactions		
	Range	Average	Median
Multiple of implied value of equity consideration received by target stockholders to stated book value	2.1x - 5.4x	3.5x	3.4x
Multiple of implied value of equity consideration received by target stockholders to tangible book value	2.4x - 6.0x	4.1x	4.0x
Multiple of implied value of equity consideration per share received by target stockholders to latest twelve months earnings per share	13.4x - 28.8x	21.5x	21.5x
Multiple of implied value of equity consideration per share received by target stockholders to current-year estimated earnings per share based on I/B/E/S median estimates	12.1x - 24.7x	17.1x	17.1x
Premium to core deposits	16.3% - 53.8%	33.8%	33.7%
Premium of implied offer value to target stock price six days prior to announcement	0.7% – 44.6%	25.4%	27.2%

Pro Forma Merger Analysis. Goldman Sachs prepared illustrative pro forma analyses of the potential financial impact of the merger using First Call mean 2004 earnings per share estimates for Bank of America and FleetBoston and a long-term earnings per share growth rate of 9% to arrive at 2005 earnings per share estimates for both companies and assuming:

- completion of the merger on June 30, 2004;
- expense savings of 25% of FleetBoston's non-interest expense base, or \$1.1 billion on an after-tax basis, partially phased in for 2004 and fully phased in for 2005 (and Goldman Sachs noted that the 25% figure was equal to the median cost savings, as a percentage of the target companies' trailing 12-month total noninterest expense, projected at public announcement for U.S. bank and thrift mergers valued at greater than \$5 billion since 1997);
- \$260 million of additional after-tax earnings from business model changes, partially offset by use of excess free cash flow for incremental share repurchases and associated opportunity cost of share repurchases of \$65 million;
- · additional after-tax identifiable intangible amortization of \$382 million; and
- dividend payout ratio increases to 46% beginning in 2004.

For each of the years 2004 and 2005, Goldman Sachs compared the projected earnings per share of Bank of America common stock, on a standalone basis, to the projected pro forma earnings per share of Bank of America common stock after the merger. Based on these analyses, the proposed transaction would be approximately 2.3% dilutive to GAAP earnings per share and approximately 1.0% dilutive to cash earnings per share in 2004, and approximately 0.6% accretive to GAAP earnings per share and approximately 2.7% accretive to cash earnings per share in 2005.

Contribution Analysis. Based upon the exchange ratio, Goldman Sachs calculated that holders of Bank of America common stock and holders of FleetBoston common stock would own 72.9% and 27.1%, respectively, of the outstanding common stock of the combined company. Goldman Sachs reviewed the relative contributions of each of Bank of America and FleetBoston to the combined company on a pro forma basis in terms of market capitalization as of October 24, 2003, common equity as of September 30, 2003, tangible common equity as of September 30, 2003, total assets as of September 30, 2003 and 2003, 2004 and 2005 estimated net income based upon First Call mean estimates. The following table compares the pro forma ownership of the Bank of America and FleetBoston stockholders in the combined company with each company's respective contribution to each element of the analysis:

	Pro Forma Ownership of Bank of America Stockholders	Pro Forma Ownership of FleetBoston Stockholders
	72.9%	27.1%
	Bank of America Contribution to Combined Company	FleetBoston Contribution to Combined Company
Market capitalization as of October 24, 2003	79.1%	20.9%
Common Equity	74.4%	25.6%
Tangible Common Equity	74.8%	25.2%
Total Assets	79.0%	21.0%
2003E Net Income	80.9%	19.1%
2004E Net Income	79.2%	20.8%
2005E Net Income	79.2%	20.8%

Net Present Value Comparison. Based upon First Call estimates of 2004 earnings per share for Bank of America and FleetBoston, earnings growth rates of 9%, a dividend payout ratio of 46%, a terminal value exit multiple of 11.6x 2006 estimated earnings per share, and a discount rate of 9%, Goldman Sachs derived a net present value of \$86.43 per share of Bank of America common stock on a standalone basis in 2006. Using assumptions similar to those used for the Pro Forma Merger Analysis described above, Goldman Sachs derived a net present value of \$86.87 per share of Bank of America common stock on a pro forma basis.

The preparation of a fairness opinion is a complex process and is not necessarily susceptible to partial analysis or summary description. Selecting portions of the analyses or of the summary described above, without considering the analyses as a whole, could create an incomplete view of the processes underlying Goldman Sachs' analyses and opinion. In arriving at its fairness determination, Goldman Sachs considered the results of all the analyses and did not attribute any particular weight to any factor or analysis considered by it; rather, Goldman Sachs made its determination as to fairness on the basis of its experience and professional judgment after considering the results of all the analyses. No company or transaction used in the above analyses as a comparison is directly comparable to Bank of America, FleetBoston or the contemplated transaction.

Goldman Sachs prepared these analyses for purposes of providing an opinion to the Bank of America board of directors as to the fairness, from a financial point of view, to Bank of America of the exchange ratio of 0.5553 of a share of Bank of America common stock to be issued in respect of each share of FleetBoston common stock pursuant to the merger agreement. These analyses do not purport to be appraisals or to necessarily reflect the prices at which businesses or securities actually may be sold. Analyses based upon forecasts of future results are

not necessarily indicative of actual future results, which may be significantly more or less favorable than suggested by these analyses. Because these analyses are inherently subject to uncertainty and are based upon numerous factors or events beyond the control of the parties or their respective advisors, none of Bank of America, FleetBoston or Goldman Sachs assumes responsibility if future results are materially different from those forecast.

As described above, the opinion of Goldman Sachs to the Bank of America board of directors was one of many factors taken into consideration by the Bank of America board in making its determination to approve the merger agreement. The foregoing summary does not purport to be a complete description of the analyses performed by Goldman Sachs

Goldman Sachs, as part of its investment banking business, is continually engaged in performing financial analyses with respect to businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, competitive biddings, secondary distributions of listed and unlisted securities, private placements and other transactions as well as for estate, corporate and other purposes. Goldman Sachs has acted as financial advisor to Bank of America in connection with the merger.

In addition, as described in its fairness opinion to the Bank of America board of directors, Goldman Sachs has provided certain investment banking services to Bank of America from time to time, including having acted as co-manager on several note and capital securities offerings, having acted as manager of a public offering of Bank of America common stock in November 2001, and having acted as financial advisor in connection with the restructuring of Bank of America's EquiCredit home equity business in April 2002, and Goldman Sachs also participates in certain joint ventures from time to time with respect to certain investment arrangements. For the past two years, Goldman Sachs has received fees for investment banking and financial advisory services provided to Bank of America and its affiliates (excluding fees described below in connection with the merger) of approximately \$24.6 million, in the aggregate. Goldman Sachs also has provided investment banking services to FleetBoston from time to time, including having acted as co-manager or lead-manager on several note and capital securities offerings, and having acted as financial advisor to FleetBoston in connection with the acquisitions of Summit Bancorp in March 2001 and the asset management businesses of Liberty Financial Companies in November 2001, and Goldman Sachs participates in a joint venture with FleetBoston with respect to certain loan arrangements. For the past two years, Goldman Sachs has received fees for investment banking and financial advisory services provided to FleetBoston and its affiliates of approximately \$1.6 million, in the aggregate. Goldman Sachs also may provide investment banking services to Bank of America or FleetBoston in the future, in connection with which it may receive compensation.

In addition, Goldman Sachs is a full-service securities firm engaged, either directly or through its affiliates, in securities trading, investment management, financial planning and benefits counseling, and financing and brokerage activities for both companies and individuals. In the ordinary course of their trading, investment management, financing and brokerage activities, Goldman Sachs and its affiliates may actively trade the debt and equity securities of Bank of America and FleetBoston (or related derivative securities) for their own account and for the accounts of their customers and may at any time hold long and short positions of such securities.

Bank of America selected Goldman Sachs as its financial advisor because Goldman Sachs is an internationally recognized investment banking firm that has substantial experience in transactions similar to the merger. Pursuant to a letter agreement, dated October 25, 2003, Bank of America engaged Goldman Sachs to act as its financial advisor in connection with a potential transaction with FleetBoston. Pursuant to this letter agreement, Bank of America agreed to pay Goldman Sachs a transaction fee of \$25 million. Of this amount, a fee of \$3 million was payable upon execution of the engagement letter, a fee of \$5 million was payable upon delivery of its opinion and a fee of \$17 million is payable upon completion of the merger. Bank of America has also agreed to reimburse Goldman Sachs for its reasonable out-of-pocket expenses, including attorneys' fees and disbursements, and to indemnify Goldman Sachs against certain liabilities, including certain liabilities under the federal securities laws.

Opinion of Morgan Stanley

Pursuant to the terms of an engagement letter, Morgan Stanley was engaged to provide financial advisory services to FleetBoston in connection with the merger. FleetBoston selected Morgan Stanley to act as its financial advisor based on Morgan Stanley's qualifications, expertise and reputation, as well as its knowledge of the business and affairs of FleetBoston. On October 26, 2003, Morgan Stanley delivered its oral opinion, subsequently confirmed in writing as of the same date, to the FleetBoston board of directors that, subject to and based on the factors considered in its opinion, the exchange ratio pursuant to the merger agreement was fair, from a financial point of view, to the holders of shares of FleetBoston common stock.

The full text of Morgan Stanley's written opinion, dated as of October 26, 2003, which sets forth, among other things, the assumptions made, procedures followed, matters considered and limitations on the review undertaken by Morgan Stanley in rendering its opinion, is attached to this joint proxy statement/prospectus as Appendix E. Holders of FleetBoston common stock are urged to, and should, read this opinion carefully and in its entirety. Morgan Stanley's opinion, directed to the board of directors of FleetBoston, addresses only the fairness, from a financial point of view, of the exchange ratio pursuant to the merger agreement to the holders of shares of FleetBoston common stock, and does not address any other aspect of the acquisition or constitute a recommendation to any FleetBoston stockholder as to how to vote at the FleetBoston special meeting. This summary is qualified in its entirety by reference to the full text of the opinion. The opinion of Morgan Stanley will not reflect any developments that may occur or may have occurred after the date of its opinion and prior to the completion of the merger. In addition, FleetBoston does not currently expect that it will request an updated opinion from Morgan Stanley.

In arriving at its opinion, Morgan Stanley, among other things:

- reviewed certain publicly available financial statements and other information of FleetBoston and Bank of America, respectively;
- reviewed certain internal financial statements and other financial and operating data concerning FleetBoston and Bank of America prepared by the managements of FleetBoston and Bank of America, respectively, including among other things, limited financial forecasts and profit plans for each company;
- discussed the past and current operations and financial condition and the prospects of FleetBoston and Bank of America, including information relating to certain strategic, financial and operational benefits anticipated from the merger, with senior executives of FleetBoston and Bank of America, respectively;
- · reviewed the pro forma impact of the merger on Bank of America's earnings per share, consolidated capitalization and financial ratios;
- · reviewed the reported prices and trading activity of FleetBoston common stock and Bank of America common stock;
- compared the financial performance of FleetBoston and Bank of America and the prices and trading activity of FleetBoston common stock and Bank of America
 common stock with that of certain other comparable publicly traded companies and their securities;
- reviewed the financial terms, to the extent publicly available, of certain precedent transactions;
- · participated in discussions among representatives of FleetBoston and Bank of America and their financial and legal advisors;
- · reviewed the draft merger agreement and certain related documents; and
- considered other factors and performed other analyses as Morgan Stanley deemed appropriate.

Morgan Stanley assumed and relied upon, without independent verification, the accuracy and completeness of the information reviewed by it for the purposes of its opinion. With respect to financial forecasts, profit plans and information regarding certain strategic, financial and operational benefits anticipated from the merger,

Morgan Stanley assumed that they were reasonably prepared on bases reflecting the best available estimates and judgments of the future financial performance of FleetBoston and Bank of America. Morgan Stanley assumed that the merger will be completed in accordance with the terms of the merger agreement, including, among other things, that the merger will be treated as a tax-free reorganization, pursuant to the Internal Revenue Code. Morgan Stanley did not make any independent valuation or appraisal of the assets or liabilities of FleetBoston and Bank of America, nor was Morgan Stanley furnished with any such appraisals, and Morgan Stanley did not make any independent examination of the loan loss reserves or examine any individual loan credit files of Bank of America or FleetBoston. Morgan Stanley's opinion was necessarily based on financial, economic, market and other conditions as in effect on, and the information made available to it as of, the date of its opinion.

The following is a summary of the material financial analyses performed by Morgan Stanley, in connection with preparing its oral opinion and its written opinion letter, dated October 26, 2003. Some of these summaries of financial analyses include information presented in tabular format. In order to understand fully the financial analysis used by Morgan Stanley, the tables must be read together with the text of each summary. The tables alone do not constitute a complete description of the financial analyses.

FleetBoston

Summary of Selected Analyses. Morgan Stanley performed several analyses to evaluate the exchange ratio, including, among others, analyses of comparable companies, dividend discount, implied acquisition value and precedent transactions. The table below summarizes the range of values for a share of FleetBoston common stock implied by selected Morgan Stanley analyses compared to the market price per share of FleetBoston common stock as of October 24, 2003 and the implied value of the merger consideration per share as of the same date. This summary does not include the underlying assumptions or descriptions of each analysis or all other analyses performed by Morgan Stanley in connection with the delivery of its opinion. Morgan Stanley did not attribute any particular weight, or equal weight, to any factor or analysis considered by it. In order to fully understand this summary, please read the description of each analysis that follows below:

As of October 24, 2003:

	Price Range Implied by Analysis per FleetBoston share	Market Price per FleetBoston share	Implied Value of Merger Consideration per FleetBoston share
Analysis			
Comparable Company	\$30-33	\$31.80	\$45.46
Dividend Discount	31-36	31.80	45.46
Implied Acquisition Value	41-46	31.80	45.46
Precedent Transaction	34-47	31.80	45.46

Comparable Company Analysis. As part of its analysis, Morgan Stanley compared financial information, including price-to-earnings ratios, of FleetBoston with companies that share certain characteristics with FleetBoston. This group, which we refer to in this document as the FleetBoston Peer Group, included:

- Wachovia Corporation
- Bank One Corporation
- U.S. Bancorp
- Fifth Third Bancorp
- BB&T Corporation
- National City Corporation
- · SunTrust Banks, Inc.
- The PNC Financial Services Group, Inc.
- KeyCorp

Morgan Stanley compared, for the FleetBoston Peer Group, the FleetBoston Peer Group excluding Fifth Third Bancorp and FleetBoston, the market price to 2004 GAAP earnings per share estimate from I/B/E/S, available as of October 24, 2003.

The following table reflects these multiples:

	GAAP Earnings
FleetBoston (as of October 24, 2003)	11.6x
FleetBoston Peer Group Median	12.3
FleetBoston Peer Group Median excluding Fifth Third Bancorp	12.2

Price/2004E

Applying a range of FleetBoston Peer Group price-to-2004 GAAP earnings per share multiples to FleetBoston's 2004 I/B/E/S earnings per share estimate implied a range of values for FleetBoston common stock of approximately \$30 to \$33 per share. Morgan Stanley noted that the closing price of FleetBoston common stock, as of October 24, 2003, was \$31.80 per share. Based on the merger exchange ratio of 0.5553 and the closing price of Bank of America common stock of \$81.86 as of October 24, 2003, the implied value of the merger consideration was \$45.46 per share of FleetBoston common stock.

Dividend Discount Analysis. Morgan Stanley performed a dividend discount analysis to determine a range of implied present values of FleetBoston common stock, assuming FleetBoston continued to operate as a stand-alone entity. The range was determined by adding:

- the present value of an estimated future dividend stream for FleetBoston over the five-year period from 2004 through 2008, and
- the present value of an estimated "terminal value" of FleetBoston common stock at the end of the year 2008.

Morgan Stanley made the following assumptions in performing its analysis:

- a constant tangible common equity/tangible assets ratio of .0665 for a projected dividend stream,
- earnings projections were based on I/B/E/S estimates with an assumed earnings growth rate of 9% for 2005 through 2008, consistent with the I/B/E/S long-term growth estimate for FleetBoston,
- "terminal value" of FleetBoston common stock at the end of the period was determined by applying two different price-to-earnings multiples (11x and 12x) to year 2008 projected earnings, and
- the dividend stream and terminal values were discounted to present values using a range of discount rates from 12% to 14%.

Based on the above assumptions, this analysis implied a fully diluted stand-alone value of FleetBoston common stock ranging from approximately \$31 to \$36 per share. As described above under "Comparable Company Analysis," as of October 24, 2003, the implied value of the merger consideration was \$45.46 per share of FleetBoston common stock.

Implied Acquisition Value. Morgan Stanley performed an analysis of the potential implied acquisition valuation of FleetBoston assuming that the net present value of assumed cost savings was added to the implied fully diluted stand-alone value of FleetBoston common stock as described above under "Dividend Discount Analysis." Assuming pre-tax cost savings of 25% of FleetBoston's non-interest expense base, 3% annual growth in expenses and an estimated pre-tax restructuring charge of \$1.3 billion, and using an assumed discount rate of 13% and a terminal multiple of 11.5x (the above discount rate and terminal multiple represent the midpoint of the respective ranges), Morgan Stanley estimated the implied acquisition value of FleetBoston common stock ranged from approximately \$41 to \$46 per share. As described above under "Comparable Company Analysis," as of October 24, 2003, the implied value of the merger consideration was \$45.46 per share of FleetBoston common stock.

Precedent Transaction Analysis. Morgan Stanley performed an analysis of the precedent transactions announced since September 1, 2000 with transaction values of over \$10 billion by commercial bank holding companies that shared characteristics with the merger to compare the premium to market price one day prior to announcement of the transaction and projected earnings indicated by the merger consideration to those multiples indicated for the precedent transactions. For each transaction, the acquired company's estimated earnings per share was based on I/B/E/S earnings per share estimates prior to announcement of the transaction.

The three transactions constituting the precedent transactions were:

Acquiror	Target
First Union	Wachovia
Firstar	U.S. Bancorp
Chase Manhattan	J.P. Morgan

The following table reflects the results of the analysis:

	Earnings	(1 day prior)*
FleetBoston/Bank of America merger	17.0x	43%
Range of precedent transactions	12.9-17.1x	7-21%

Drice/Forward

The foregoing FleetBoston multiples indicated by the merger consideration were based on the closing prices of Bank of America and FleetBoston common stock as of October 24, 2003, and FleetBoston's 2004 I/B/E/S earnings per share estimate as of October 24, 2003.

Using a range of price to estimated forward earnings per share multiples of 13x to 17x indicated by the precedent transactions resulted in a value of FleetBoston common stock ranging from approximately \$36 to \$47 per share. Using a premium to market (one day prior to announcement) range of 7-21% indicated by the precedent transactions resulted in a value of FleetBoston common stock ranging from approximately \$34 to \$38 per share. As described above under "Comparable Company Analysis," as of October 24, 2003, the implied value of the merger consideration was \$45.46 per share of FleetBoston common stock.

No company or transaction used in the comparable company and precedent transaction analyses is identical to FleetBoston or the merger. Accordingly, an analysis of the results of the foregoing necessarily involves complex considerations and judgments concerning financial and operating characteristics of FleetBoston and other factors that could affect the public trading value of the companies to which they are being compared. Mathematical analysis (such as determining the average or median) is not in itself a meaningful method of using comparable transaction data or comparable company data.

Bank of America

Trading Multiples. As part of its analysis, Morgan Stanley compared certain financial information of Bank of America with companies that share certain characteristics with Bank of America. This group, which we refer to in this document as the Bank of America Peer Group, included:

- · Citigroup, Inc.
- · Wells Fargo & Company
- JP Morgan Chase & Co.

^{*} Two days prior to public announcement of the transaction by the parties in the case of the Chase Manhattan/J.P. Morgan transaction, which was one day prior to the first date that the news of such transaction was publicly reported.

- Bank One Corporation
- Wachovia Corporation
- · U.S. Bancorp

Morgan Stanley compared, the market price to 2004 GAAP earnings per share estimate from I/B/E/S, available as of October 24, 2003 for the Bank of America Peer Group and Bank of America.

Based on the Bank of America Peer Group trading range, Morgan Stanley applied a P/E range of 11-12x to the Bank of America 2004 I/B/E/S earnings per share estimate. The implied range of values for Bank of America common stock, derived from the analysis, was \$80 to \$87 per share, compared to the closing price of Bank of America common stock of \$81.86 as of October 24, 2003.

No company or transaction used in the comparable company analysis is identical to Bank of America or the merger. Accordingly, an analysis of the results of the foregoing necessarily involves complex considerations and judgments concerning financial and operating characteristics of Bank of America and other factors that could affect the public trading value of the companies to which they are being compared. Mathematical analysis (such as determining the average or median) is not in itself a meaningful method of using comparable transaction data or comparable company data.

Dividend Discount Analysis. Morgan Stanley performed a dividend discount analysis to determine a range of present values of Bank of America common stock, excluding the effects of the merger. The range was determined by adding:

- the present value of an estimated future dividend stream for Bank of America over the five-year period from 2004 through 2008, and
- the present value of the "terminal value" of Bank of America common stock at the end of the year 2008.

Morgan Stanley made the following assumptions in performing its analysis:

- a constant tangible common equity/tangible assets ratio of 5.2% for a projected dividend stream,
- earnings projections based on I/B/E/S estimates for 2004, with an assumed earnings growth rate of 9% for the years 2005 through 2008, consistent with the I/B/E/S long-term growth estimate for Bank of America,
- "terminal value" of Bank of America common stock at the end of the period determined by applying a price-to-earnings multiple range (11x to 12x) to year 2008 projected earnings, and
- the dividend stream and terminal values discounted to present values using a range of discount rates of 12% to 14%.

Based on the above assumptions, the fully diluted stand-alone value of Bank of America common stock ranged from approximately \$81 to \$93 per share. As of October 24, 2003, the market price per share of Bank of America common stock was \$81.86.

Sensitivity Analysis to Dividend Discount Analysis. Using the same methodology as described above under "Dividend Discount Analysis," Morgan Stanley performed a dividend discount analysis assuming no earnings per share growth in 2004 or 2005 from the 2003 earnings per share estimate, with annual earnings per share growth of 9% for each year from 2005 through 2008. All other assumptions were consistent with assumptions described above under "Dividend Discount Analysis." Based on these assumptions, the fully diluted stand-alone value of Bank of America common stock ranged from approximately \$74 to \$86 per share.

Pro Forma Merger Analysis. Morgan Stanley analyzed the financial impact of the merger on the estimated earnings per share for Bank of America and FleetBoston common stock, as well as the estimated dividend per share for the current holders of FleetBoston common stock, using the estimated cost savings expected by Bank of

America management to result from the merger as well as Bank of America earnings estimates for 2004 through 2005 based on Bank of America I/B/E/S estimates for 2004 and an assumed earnings growth rate of 9% for 2005. This analysis indicated that the merger would be approximately 27% accretive to the estimated dividend per share for the current holders of FleetBoston common stock, and with cost savings and revenue enhancements, the merger would be approximately 2.3% dilutive to the current estimate of Bank of America's 2004 GAAP earnings per share and approximately 0.5% accretive to Bank of America's 2005 earnings per share as well as 43% and 47% accretive to the current estimate of FleetBoston's 2004 earnings per share and FleetBoston's 2005 earnings per share respectively.

Contribution Analysis. Morgan Stanley compared FleetBoston and Bank of America stockholders' respective percentage ownership of the combined company to FleetBoston and Bank of America's respective contribution (and the implied ownership based on such contribution) to the combined company using I/B/E/S estimates of 2004 net income and tangible book value as of September 30, 2003. As a result of its analysis, Morgan Stanley noted that FleetBoston contributed 21% of the fully diluted market capitalization, 25% of the tangible book value and 21% of the 2004 estimated net income to the implied company. In addition, Morgan Stanley noted that the merger exchange ratio implied that FleetBoston stockholders would hold 28% of the pro forma ownership of the combined company.

Morgan Stanley performed a variety of financial and comparative analyses for the purpose of rendering its opinion. The preparation of a fairness opinion is a complex process and is not necessarily susceptible to a partial analysis or summary description. In arriving at its opinion, Morgan Stanley considered the results of all of its analyses as a whole and did not attribute any particular weight to any particular analysis or factor considered by it. Furthermore, Morgan Stanley believes that selecting any portion of its analyses, without considering all of its analyses, would create an incomplete view of the process underlying its analyses and its opinion. In addition, Morgan Stanley may have given various factors more or less weight than other factors and may have deemed various assumptions more or less probable than other assumptions, so that the ranges of valuations resulting from any particular analysis described above should not be taken to be Morgan Stanley's view of the actual value of FleetBoston or Bank of America.

In performing its analyses, Morgan Stanley made numerous assumptions with respect to industry performance, general business and economic conditions and other matters, many of which are beyond the control of FleetBoston or Bank of America. Any estimates contained in the analyses performed by Morgan Stanley are not necessarily indicative of future results or actual values, which may be significantly more or less favorable than those suggested by such estimates. Such analyses were prepared solely as a part of Morgan Stanley's analysis of the fairness, from a financial point of view, of the exchange ratio to the holders of shares of FleetBoston common stock pursuant to the merger agreement and were provided to the FleetBoston board of directors in connection with the delivery of the Morgan Stanley opinion. The analyses do not purport to be appraisals of value or to reflect the prices at which FleetBoston or Bank of America might actually trade. Further, the analyses should not be viewed as the sole determining factors of the value of FleetBoston. In addition, as described above, the Morgan Stanley opinion was one of many factors taken into consideration by the FleetBoston board of directors in making its determination to approve the merger. The exchange ratio pursuant to the merger agreement was determined through arm's-length negotiations between FleetBoston and Bank of America and was approved by the FleetBoston board of directors. Morgan Stanley did not recommend any specific consideration to the FleetBoston board of directors or advise that any given consideration constituted the only appropriate consideration for the acquisition.

Morgan Stanley is an internationally recognized investment banking and advisory firm. Morgan Stanley, as part of its investment banking business, is continuously engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, competitive biddings, secondary distributions of listed and unlisted securities, private placements and valuations for corporate, estate and other purposes. In the past, as described in its fairness opinion to the FleetBoston board of directors, Morgan Stanley and its affiliates have provided financial advisory and financing services for FleetBoston and Bank of

America and have received fees for the rendering of these services. Morgan Stanley received approximately \$8.1 million in such fees from FleetBoston (excluding fees in connection with the merger as described below) and approximately \$9.0 million in such fees from Bank of America within the last two years. In the ordinary course of its business, Morgan Stanley and its affiliates may, from time to time, trade in the securities and indebtedness of FleetBoston or Bank of America for their own accounts or the account of investment funds and other clients under the management of Morgan Stanley and for the account of its customers and, accordingly, may at any time hold a long or short position in such securities or indebtedness for any such account. In addition, Morgan Stanley and its affiliates may, from time to time, act as a counterparty to either FleetBoston or Bank of America and have received compensation for such activities.

Pursuant to an engagement agreement with FleetBoston, Morgan Stanley was formally retained to provide financial advisory services and a financial fairness opinion in connection with the acquisition, and FleetBoston agreed to pay Morgan Stanley a transaction fee of up to \$25 million in connection with the merger. FleetBoston also agreed to reimburse Morgan Stanley for expenses incurred by Morgan Stanley in performing its services. In addition, FleetBoston also has agreed to indemnify Morgan Stanley and its affiliates, their respective directors, officers, agents and employees and each person, if any, controlling Morgan Stanley or any of its affiliates against certain liabilities and expenses, including liabilities under the federal securities laws, related to or arising out of Morgan Stanley's engagement and any related transactions.

Board of Directors and Management of Bank of America following Completion of the Merger

Board of Directors of Bank of America. After the merger, the Bank of America board of directors will consist of nineteen members, twelve of whom will be individuals who are currently Bank of America directors and seven of whom will be individuals who are currently FleetBoston directors or officers. Continuing for a period of at least two years following completion of the merger, Bank of America will maintain the twelve to seven ratio of Bank of America directors to former FleetBoston directors. During that time, any vacancy among the twelve Bank of America directors will be filled by nominees proposed to the nominating committee by a majority of the remaining Bank of America directors, and any vacancy among the seven former FleetBoston directors or officers will be filled by nominees proposed to the nominating committee by a majority of the remaining former FleetBoston directors. The Bank of America board of directors currently has seventeen members.

Chairman and Executive Officers of Bank of America. Charles K. Gifford, currently Chairman and Chief Executive Officer of FleetBoston, will be Chairman of the combined company. Kenneth D. Lewis, currently Chairman and Chief Executive Officer of Bank of America, will be Chief Executive Officer of the combined company.

In addition, we expect that the following individuals will hold responsibilities in the combined company as follows:

- Amy Woods Brinkley, Chief Risk Executive
- Edward J. Brown III, President of Global Corporate and Investment Banking
- · Barbara J. Desoer, President of Consumer Products
- · James H. Hance, Jr., Vice Chairman and Chief Financial Officer
- · Liam E. McGee, President of Consumer Banking
- · Brian T. Moynihan, President of Wealth Management
- Eugene M. McQuade, President of the combined company, whose responsibilities will also include Technology & Operations, Global Treasury Services, Principal Investing, Latin America, management services and international retail banking

- H. Jay Sarles, Vice Chairman and special adviser to the Chief Executive Officer
- R. Eugene Taylor, President of Commercial Banking
- · Bradford H. Warner, President of Small Business and Premier Banking

Ms. Brinkley and Ms. Desoer, as well as Messrs. Brown, Hance, McGee and Taylor, are current employees of Bank of America. Messrs. Moynihan, McQuade, Sarles and Warner are current employees of FleetBoston.

Information about the current Bank of America directors and executive officers can be found in Bank of America's proxy statement dated March 27, 2003. Information about the current FleetBoston directors and executive officers can be found in FleetBoston's proxy statement dated March 17, 2003. Bank of America's and FleetBoston's Annual Reports on Form 10-K for the year ended December 31, 2002 are incorporated by reference in this joint proxy statement/prospectus. See "Where You Can Find More Information" on page 103.

For more information see "The Merger — FleetBoston's Directors and Officers Have Financial Interests in the Merger" on page 59.

Distribution of Bank of America Shares

When the merger is completed, holders of FleetBoston common stock other than FleetBoston and Bank of America will have the right to receive 0.5553 of a share of Bank of America common stock for each share of FleetBoston common stock they own. No fractional shares of Bank of America common stock will be issued in the merger. Each share of each series of FleetBoston preferred stock issued and outstanding immediately prior to the completion of the merger, except for specified shares of FleetBoston preferred stock held by FleetBoston and Bank of America, will be converted into a share of a series of Bank of America preferred stock with the same terms (to the fullest extent possible) as the corresponding FleetBoston preferred stock.

An exchange agent will mail to each holder of a FleetBoston common stock certificate a letter of transmittal and instructions for surrendering FleetBoston stock certificates in exchange for statements indicating book-entry ownership of Bank of America common stock. However, if a holder of a FleetBoston common stock certificate makes a special request, Bank of America will issue to the requesting holder a Bank of America stock certificate. When you deliver your FleetBoston stock certificates to the exchange agent along with a properly executed letter of transmittal and any other required documents, your FleetBoston stock certificates will be cancelled and you will receive statements indicating book-entry ownership of Bank of America common stock, or, if requested, stock certificates representing the number of full shares of Bank of America common stock to which you are entitled under the merger agreement. You will receive payment in cash, without interest, instead of any fractional shares of Bank of America common stock which would have been otherwise issuable to you as a result of the merger.

Holders of FleetBoston common stock should not submit their FleetBoston stock certificates for exchange until they receive the transmittal instructions and a form of letter of transmittal from the exchange agent.

Until you exchange your FleetBoston stock certificates for shares of Bank of America stock, you will not receive any dividends or other distributions in respect of shares of Bank of America stock. Once you exchange your FleetBoston stock certificates for shares of Bank of America stock, you will receive, without interest, any dividends or distributions with a record date after the effective time of the merger and payable with respect to your shares of Bank of America common stock, as well as any dividends with respect to FleetBoston common stock declared before the effective time of the merger but unpaid.

If your FleetBoston stock certificate has been lost, stolen or destroyed, you may receive shares of Bank of America common stock upon the making of an affidavit of that fact. Bank of America may require you to post a bond in a reasonable amount as an indemnity against any claim that may be made against Bank of America with respect to the lost, stolen or destroyed FleetBoston stock certificate.

After completion of the merger, there will be no further transfers on the stock transfer books of FleetBoston, except as required to settle trades executed prior to completion of the merger.

There is no need for Bank of America stockholders to submit their Bank of America stock certificates to Bank of America, FleetBoston, the exchange agent or to any other person in connection with the merger or otherwise take any action as a result of the completion of the merger.

Fractional Shares

Bank of America will not issue any fractional shares of Bank of America common stock. Instead, a FleetBoston stockholder who would otherwise have received a fraction of a share of Bank of America common stock will receive an amount of cash equal to the fraction of a share of Bank of America common stock to which the holder would otherwise be entitled multiplied by the average of the closing prices of Bank of America common stock for the five full trading days immediately preceding the completion of the merger as reported on the NYSE Composite Transactions Reporting System.

Public Trading Markets

Bank of America common stock trades on the NYSE and on the Pacific Exchange under the symbol "BAC." Bank of America common stock is also listed on the London Stock Exchange, and certain shares are listed on the Tokyo Stock Exchange. FleetBoston common stock trades on the NYSE and on the Boston Stock Exchange under the symbol "FBF." Upon completion of the merger, FleetBoston common stock will be delisted from the NYSE and the Boston Stock Exchange and deregistered under the Securities Exchange Act of 1934. The newly issued Bank of America common stock and preferred stock issuable pursuant to the merger agreement will be listed on the NYSE. In the case of the Bank of America 6.75% Perpetual Preferred Stock, depositary shares each representing a one-fifth interest in a share of the Bank of America 6.75% Perpetual Preferred Stock will be listed on the NYSE.

The shares of Bank of America common stock to be issued in connection with the merger will be freely transferable under the Securities Act of 1933, except for shares issued to any stockholder who may be deemed to be an affiliate of FleetBoston, as discussed in "The Merger Agreement — Resales of Bank of America Stock by Affiliates" on page 69.

As reported on the NYSE, the closing sale price per share of Bank of America common stock on October 24, 2003 was \$81.86. As reported on the NYSE, the closing sale price per share of FleetBoston common stock on October 24, 2003 was \$31.80. Based on the Bank of America closing sale price per share and the exchange ratio, the implied per share value of FleetBoston common stock was \$45.46 as of that date. The closing sale price per share of Bank of America common stock on the NYSE on []. The closing sale price per share of FleetBoston common stock on the NYSE on [] 2004, the last practicable trading day before the date of this document, was \$[]. The implied per share value of FleetBoston common stock was \$[] as of that date based on the Bank of America closing sale price per share on that date and the exchange ratio. The implied value of one share of FleetBoston common stock as of these dates was calculated by multiplying Bank of America's closing sale price per share by 0.5553, the exchange ratio. Because the stock price of both companies will fluctuate, you should obtain current market quotations for the shares.

Bank of America Dividends

During 2002, Bank of America paid cash dividends on its common stock totaling \$2.44 per share. Bank of America currently expects to continue to pay a quarterly dividend of at least \$0.80 per share of common stock. Based on the 0.5553 exchange ratio and this annual dividend rate, holders of FleetBoston common stock would

experience an anticipated annual dividend rate increase of approximately 27% (from \$1.40 to \$1.78 per share of FleetBoston common stock equivalent). Although there is no present plan or intention to decrease this dividend, the Bank of America board of directors may, subject to applicable law, change this dividend amount at any time, and Bank of America's ability to pay dividends on its common stock is subject to various legal and regulatory limitations.

Bank of America currently pays a quarterly dividend of \$0.80 per share of common stock, and, although there is no present plan or intention to decrease this dividend, the Bank of America board of directors may, subject to applicable law, change this dividend amount at any time. During the first half of 2003, FleetBoston paid cash dividends totaling \$0.70 per share of common stock, and Bank of America paid cash dividends totaling \$1.28 per share of common stock.

Bank of America stockholders will be entitled to receive dividends when and if declared by the Bank of America board of directors out of funds legally available for dividends. The Bank of America board of directors periodically will consider the payment of dividends, taking into account Bank of America's financial condition and level of net income, Bank of America's future prospects, economic conditions, industry practices and other factors, including applicable banking laws and regulations.

Under the merger agreement, FleetBoston has agreed that, until the merger is completed, it will not pay regular quarterly cash dividends in excess of \$0.35 per share of FleetBoston common stock. FleetBoston and Bank of America also have agreed to coordinate the declaration of dividends so that holders of FleetBoston common stock will not receive two dividends, or fail to receive one dividend, for any quarter with respect to their FleetBoston common stock and any Bank of America common stock any holder receives in the merger.

Appraisal Rights of Dissenting Stockholders

Appraisal rights are statutory rights that enable stockholders to dissent from an extraordinary transaction, such as a merger, and to demand that the corporation pay the fair value for their shares as determined by a court in a judicial proceeding instead of receiving the consideration offered to stockholders in connection with the extraordinary transaction. Appraisal rights are not available in all circumstances, and exceptions to these rights are provided under the laws of Delaware and Rhode Island, which are the states of incorporation of Bank of America and FleetBoston, respectively. As a result of these exceptions, the holders of Bank of America common stock, as well as the holders of FleetBoston common stock and Series VI 6.75% Perpetual Preferred Stock, which we refer to as the FleetBoston Series VI Stock, are not entitled to appraisal rights in the merger.

The holders of Bank of America Series B Preferred Stock and ESOP Preferred Stock, as well as the holders of FleetBoston Series VII Fixed/Adjustable Rate Cumulative Preferred Stock, which we refer to as the FleetBoston Series VII Stock, however, are entitled to appraisal rights in the merger.

Delaware General Corporation Law. In order to exercise appraisal rights, a holder of Bank of America Series B Preferred Stock or ESOP Preferred Stock must demand and perfect the rights in accordance with Section 262 of the Delaware General Corporation Law. Generally, in order to exercise these appraisal rights, the holders of Bank of America Series B Preferred Stock and ESOP Preferred Stock must not vote in favor of adoption of the merger agreement and must make a written demand for appraisal before completion of the merger.

The following is intended as a brief summary of the material provisions of the Delaware statutory procedures required to be followed in order to dissent from the merger and perfect appraisal rights. This summary, however, is not a complete statement of all applicable requirements and is qualified in its entirety by reference to Section 262 of the Delaware General Corporation Law, the full text of which appears in Appendix G of this document.

Each step must be taken in strict compliance with the applicable provisions of Section 262 in order for holders of Bank of America Series B Preferred Stock and ESOP Preferred Stock to perfect their appraisal rights.

Section 262 requires that stockholders be notified that appraisal rights will be available not less than 20 days before the special meeting to vote on the merger. A copy of Section 262 must be included with the notice. This document constitutes Bank of America's notice to the holders of the Bank of America Series B Preferred Stock and ESOP Preferred Stock of the availability of appraisal rights in connection with the merger in compliance with the requirements of Section 262. If you are a holder of Bank of America Series B Preferred Stock or ESOP Preferred Stock and wish to consider exercising your appraisal rights, you should carefully review the text of Section 262 contained in Appendix G since failure timely and properly to comply with the requirements of Section 262 will result in the loss of your appraisal rights under Delaware law.

Holders of Bank of America Series B Preferred Stock or ESOP Preferred Stock who elect to demand appraisal of their shares must:

- · deliver to Bank of America a written demand for appraisal of the shares before the vote with respect to the merger is taken; and
- · continuously hold the shares of record through the completion of the merger.

If a holder of Bank of America Series B Preferred Stock or ESOP Preferred Stock fails to comply with these conditions and the merger is completed, the holder will have no appraisal rights with respect to the shares.

All demands for appraisal should be addressed to the Corporate Secretary of Bank of America at Bank of America Corporation, NC1-007-56-11, 100 North Tryon Street, Charlotte, North Carolina 28255, and must be delivered before the vote on the merger is taken at the Bank of America special meeting. All demands for appraisal must be executed by, or on behalf of, the record holder of the shares for which appraisal rights are being exercised. The demand must reasonably inform Bank of America of the identity of the holder and the intention of the holder to demand appraisal of his, her or its shares.

Within 10 days after the effective date of the merger, Bank of America must give written notice that the merger has become effective to each holder who has properly filed a written demand for appraisal. At any time within 60 days after the effective date, any holder who has demanded appraisal has the right to withdraw the demand for appraisal. Within 120 days after the effective date, either Bank of America or any holder who has complied with the requirements of Section 262 may file a petition in the Delaware Court of Chancery demanding a determination of the fair value of the shares held by all holders entitled to appraisal. Bank of America has no obligation or intention to file such a petition in the event there are dissenting stockholders. Accordingly, the failure of a holder to file such a petition within the period specified could nullify the holder's previously written demand for appraisal.

If a petition for appraisal is duly filed by a holder and a copy of the petition is delivered to Bank of America, Bank of America will then be obligated, within 20 days after receiving service of a copy of the petition, to provide the Chancery Court with a duly verified list containing the names and addresses of all holders who have demanded an appraisal of their shares. After notice to dissenting holders, the Chancery Court is empowered to conduct a hearing upon the petition, and to determine those holders who have complied with Section 262 and who have become entitled to the appraisal rights provided thereby. The Chancery Court may require the holders who have demanded payment for their shares to submit their stock certificates to the Register in Chancery for notation thereon of the pendency of the appraisal proceedings; and if any stockholder fails to comply with that direction, the Chancery Court may dismiss the proceedings as to that holder.

After determination of the holders entitled to appraisal of their shares, the Chancery Court will appraise the shares, determining their fair value exclusive of any element of value arising from the accomplishment or expectation of the merger, together with a fair rate of interest. When the value is determined, the Chancery Court will direct the payment of that value, with interest on that amount accrued during the pendency of the proceeding, if the Chancery Court so determines, to the holders entitled to receive the same, upon surrender by those holders of the certificates representing those shares.

The Court of Chancery may impose costs of the appraisal proceeding upon Bank of America and the holders participating in the appraisal proceeding as the Court of Chancery deems equitable in the circumstances. Upon the application of a holder, the Chancery Court may order all or a portion of the expenses incurred by any holder in connection with the appraisal proceeding, including reasonable attorneys' fees and the fees and expenses of experts, to be charged pro rata against the value of all shares entitled to appraisal. Any holder who had demanded appraisal rights will not, after the effective date, be entitled to vote shares subject to that demand for any purpose or to receive payments of dividends or any other distribution with respect to those shares, other than with respect to payment as of a record date prior to the effective date. If no petition for appraisal is filed within 120 days after the effective date of the merger, or if the holder delivers a written withdrawal of his or her demand for appraisal and an acceptance of the merger within 60 days after the effective date of the merger, then the right of that holder to appraisal will cease. Any withdrawal of a demand for appraisal made more than 60 days after the effective date of the merger may only be made with the written approval of Bank of America as the surviving corporation. However, no appraisal proceeding in the Chancery Court may be dismissed as to any stockholder without the approval of the court.

Rhode Island Business Corporation Act. In order to exercise appraisal rights, a holder of shares of FleetBoston Series VII Stock who files with FleetBoston a written objection to the merger may elect to receive payment of the fair value of those shares in cash in accordance with Title 7, Chapter 1.1, Section 74 of the Rhode Island Business Corporation Act.

Any stockholder contemplating the exercise of the right to dissent from the merger and exercise appraisal rights should review carefully the provisions of Section 74 reprinted as Appendix F to this joint proxy statement/prospectus. A summary of the principal steps to be taken if the right to dissent is to be exercised is set forth below. This summary should be read in conjunction with, and is qualified in its entirety by reference to, the full text of Section 74.

Each step must be taken in strict compliance with the applicable provisions of Section 74 in order for holders of shares of FleetBoston Series VII Stock to perfect their dissenters' rights.

Under the Rhode Island Business Corporation Act, a holder of FleetBoston Series VII Stock who dissents from the merger may, in certain circumstances and subject to certain limitations described below, demand the fair value of his, her or its shares. However, in order to receive the fair value of those shares, the dissenting stockholder must take certain steps to perfect his or her dissenter's rights.

In particular, the dissenting stockholder must:

- · file with FleetBoston, prior to or at the special meeting, a written objection to the merger, and
- within 10 days of the special meeting (provided the merger is approved by the required vote), make a written demand on Bank of America as the surviving corporation for payment of the fair value of his or her shares as of the day prior to the special meeting, excluding any appreciation or depreciation in anticipation of the merger.

Stockholders will not receive any notice with respect to the expiration of the 10-day period. Any stockholder failing to make demand within the 10-day period will be bound by the terms of the merger. Any stockholder making the demand is thereafter only entitled to payment and not entitled to vote or exercise any other rights of a stockholder. No demand may be withdrawn unless FleetBoston or, following the merger, Bank of America, consents to the withdrawal.

Within 20 days after demanding payment for his, her or its shares, each stockholder demanding payment must submit the certificate or certificates representing his, her or its shares to the corporation for notation on the certificate that the demand has been made. The failure to do so, at the option of the corporation, would terminate appraisal rights unless a court of competent jurisdiction directs otherwise for good and sufficient cause shown.

If, within 30 days after the effective date of the merger, the dissenting stockholder and Bank of America agree upon the fair value of the dissenting stockholder's shares, Bank of America must make payment for those shares within 90 days after the effective date of the merger, upon the surrender of the dissenting stockholder's stock certificates representing his, her or its shares. Upon payment of the agreed value, the dissenting stockholder will cease to have any interest in the shares.

If, within 30 days after the effective date of the merger, a dissenting stockholder and Bank of America do not agree on the fair value of the dissenting stockholder's shares, then Bank of America, within 30 days after receipt of a written request from any dissenting shareholder given within 60 days after the effective date of the merger, will, or at its election at any time within the period of 60 days may, file a petition in any court of competent jurisdiction in the county in Rhode Island where the registered office of the corporation is located demanding that the court determine the fair value of the dissenting stockholder's shares. If Bank of America does not have a registered office in Rhode Island, then the petition will be filed in the county where the registered office of FleetBoston was last located. If Bank of America fails to file this petition, any dissenting stockholder may do so in Bank of America's name.

The judgment will only be payable upon and concurrently with the surrender to Bank of America of the stock certificates representing the shares. Upon payment of the judgment, the stockholder will cease to have any interest in the shares.

The right of a dissenting stockholder to be paid the fair value of his or her shares as provided above would cease if FleetBoston abandons the merger.

Regulatory Approvals Required for the Merger

We have agreed to use our reasonable best efforts to obtain all regulatory approvals required to complete the transactions contemplated by the merger agreement. These approvals include approval from the Federal Reserve Board and various federal, state and foreign regulatory authorities. Bank of America and FleetBoston have completed, or will complete, the filing of applications and notifications to obtain the required regulatory approvals.

Federal Reserve Board. The merger is subject to approval by the Federal Reserve Board pursuant to Sections 3 and 4 of the Bank Holding Company Act of 1956. On November 14, 2003, Bank of America filed the required application and notification with the Federal Reserve Board for approval of the merger.

The Federal Reserve Board is prohibited from approving any transaction under the applicable statutes that (1) would result in a monopoly, (2) would be in furtherance of any combination or conspiracy to monopolize or to attempt to monopolize the business of banking in any part of the United States, or (3) may have the effect in any section of the United States of substantially lessening competition, tending to create a monopoly or resulting in a restraint of trade, unless the Federal Reserve Board finds that the anti-competitive effects of the transaction are clearly outweighed in the public interest by the probable effect of the transaction in meeting the convenience and needs of the communities to be served. The Federal Reserve Board may approve an interstate acquisition without regard to state law, if, after completion, the combined entity would control 10% or less of the total deposits of insured depository institutions in the United States.

In addition, in reviewing a transaction under the applicable statutes, the Federal Reserve Board will consider the financial and managerial resources of the companies and their subsidiary banks and the convenience and needs of the communities to be served as well as the companies' effectiveness in combating money-laundering activities. In connection with its review, the Federal Reserve Board will provide an opportunity for public comment on the application for the merger, and is authorized to hold a public meeting or other proceeding if it determines that would be appropriate.

Under the Community Reinvestment Act of 1977, which we refer to as the CRA, the Federal Reserve Board must take into account the record of performance of each of Bank of America and FleetBoston in meeting the credit needs of the entire community, including low- and moderate-income neighborhoods, served by each company and their subsidiaries. All of Bank of America's and FleetBoston's subsidiary depository institutions (each, a "Bank") that are subject to the CRA have either an outstanding or satisfactory CRA rating with the appropriate federal regulator. None of the Banks received any negative comments from its respective federal regulator in its last CRA examination relating to those ratings that were material and remain unresolved.

Bank of America's and FleetBoston's rights to exercise their respective options under the stock option agreements are also subject to the prior approval of the Federal Reserve Board, to the extent that the exercise of their respective options under the stock option agreements would result in Bank of America or FleetBoston, as the case may be, owning more than 5% of the outstanding shares of FleetBoston common stock or Bank of America common stock, respectively. In considering whether to approve Bank of America's and FleetBoston's right to exercise its respective option, including its respective right to purchase more than 5% of the outstanding shares of FleetBoston common stock or Bank of America common stock, as the case may be, the Federal Reserve Board would generally apply the same statutory criteria it will apply to its consideration of the merger.

Department of Justice, Federal Trade Commission and Other Antitrust Authorities Bank of America's acquisition of FleetBoston's U.S. nonbanking business is subject to review by the Antitrust Division of the United States Department of Justice, which we refer to as the Antitrust Division, or the United States Federal Trade Commission, which we refer to as the FTC, to determine whether it complies with applicable antitrust law. Under the provisions of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, and its related rules, the merger cannot be completed until both Bank of America and FleetBoston file notification of the proposed transaction with the Antitrust Division and the FTC and the specified waiting periods have expired or been terminated. On November 17, 2003, Bank of America and FleetBoston filed their pre-merger notification and report forms pursuant to the Hart-Scott-Rodino Act. On December 17, 2003, the FTC granted early termination of the waiting period under the Hart-Scott-Rodino Act with respect to Bank of America's acquisition of FleetBoston's U.S. nonbanking businesses.

At any time before or after the acquisition is completed, the Antitrust Division or the FTC could take action under the antitrust laws as it deems necessary or desirable in the public interest, including seeking to enjoin the acquisition or seeking divestiture of substantial assets of Bank of America or FleetBoston or their subsidiaries. Private parties also may seek to take legal action under the antitrust laws under some circumstances. Based upon an examination of information available relating to the businesses in which the companies are engaged, Bank of America and FleetBoston believe that the completion of the merger will not violate U.S. antitrust laws. However, Bank of America and FleetBoston can give no assurance that a challenge to the merger on antitrust grounds will not be made, or, if such a challenge is made, that Bank of America and FleetBoston will prevail.

In addition, the merger may be reviewed by the state attorneys general in the various states in which Bank of America and FleetBoston operate. While Bank of America and FleetBoston believe there are substantial arguments to the contrary, these authorities may claim that there is authority, under the applicable state and federal antitrust laws and regulations, to investigate and/or disapprove the merger under the circumstances and based upon the review set forth in applicable state laws and regulations. There can be no assurance that one or more state attorneys general will not attempt to file an antitrust action to challenge the merger.

Additional State Regulatory Authorities. The change in control of FleetBoston's subsidiaries that act as insurance underwriters or agencies is subject to the receipt of necessary approvals from, or notice to, various U.S. state insurance regulatory authorities. Applications or notifications are being made to the insurance departments of those states. Applications or notifications also have been filed with various state banking and other authorities in connection with the merger.

Other Requisite Approvals, Notices and Consents. Approvals also will be required from certain other non-bank regulatory authorities in connection with the changes, as a result of the merger, in the ownership of certain businesses that are controlled by FleetBoston. Ownership changes regarding registered broker-dealers controlled by FleetBoston are subject to review by the various regulatory and self-regulatory organizations, including the Securities and Exchange Commission, the NYSE and the National Association of Securities Dealers

Certain Foreign Approvals. Approvals also will be required from, and notices must be submitted to, foreign bank, securities and competition regulatory authorities in connection with the merger and the change in ownership of certain businesses that are controlled by FleetBoston abroad. Bank of America and FleetBoston have filed and submitted, or will shortly file and submit, all applications and notices required to be submitted to obtain these approvals and provide these notices.

Timing. We cannot assure you that all of the regulatory approvals described above will be obtained, and, if obtained, we cannot assure you as to the date of any approvals or the absence of any litigation challenging such approvals. Likewise, we cannot assure you that the Antitrust Division, the FTC or any state attorney general will not attempt to challenge the merger on antitrust grounds, and, if such a challenge is made, we cannot assure you as to its result.

Pursuant to the Bank Holding Company Act and the Bank Merger Act, a transaction approved by the Federal Reserve Board may not be completed until 30 days after approval is received, during which time the Antitrust Division may challenge the merger on antitrust grounds. The commencement of an antitrust action would stay the effectiveness of an approval unless a court specifically ordered otherwise. With the approval of the Federal Reserve Board and the concurrence of the Antitrust Division, the waiting period may be reduced to no less than 15 days.

Bank of America and FleetBoston believe that the merger does not raise substantial antitrust or other significant regulatory concerns and that they will be able to obtain all requisite regulatory approvals on a timely basis without the imposition of any condition that would have a material adverse effect on Bank of America or FleetBoston.

We are not aware of any material governmental approvals or actions that are required for completion of the merger other than those described above. It is presently contemplated that if any such additional governmental approvals or actions are required, those approvals or actions will be sought. There can be no assurance, however, that any additional approvals or actions will be obtained.

FleetBoston's Directors and Officers Have Financial Interests in the Merger

Some of the members of FleetBoston management and the FleetBoston board of directors have interests in the merger that are in addition to, or different from, their interests as FleetBoston stockholders generally.

- Several executive officers of FleetBoston have existing agreements with FleetBoston that provide severance benefits in connection with qualifying terminations following a change in control of FleetBoston.
- Four of FleetBoston's executive officers Messrs. McQuade, Moynihan, Sarles and Warner have entered into new agreements with Bank of America that provide for their employment with Bank of America following completion of the merger. These four executive officers will also serve on the 11-member Risk and Capital Committee of the combined company, which will be chaired by Kenneth Lewis, Bank of America's chief executive officer.
- Some of FleetBoston's compensation and benefits plans, including the FleetBoston equity award plans, provide for the accelerated vesting of rights or benefits upon a change in control such as the merger.
- Following the merger, Messrs. Gifford and McQuade will join the board of directors of the combined company, and Mr. Gifford will serve as Chairman of the board of directors of the combined company.

Existing FleetBoston Agreements.

Several of the executive officers of FleetBoston, including Messrs. Gifford, McQuade, Sarles, Moynihan and Warner, have previously entered into agreements with FleetBoston that contain change in control severance provisions. The merger will be a change in control for purposes of these agreements. Under these agreements, if, during the two-year period following a change in control, the employment of a covered executive is terminated other than for "cause" or disability, or by the covered executive for "good reason" (which includes a termination by the executive for any reason during the thirty day-period following the first anniversary of the change in control), the covered executive will be entitled to receive a pro-rata annual bonus through the date of termination, based on the average of the annual bonuses payable to the executive during the three years prior to the year in which the change in control occurs (Mr. Robert C. Lamb, Jr. and Ms. M. Anne Szostak have agreements that provide for a minimum annual bonus of \$750,000 and \$1 million, respectively), and three or two times (depending on the covered executive) the sum of the covered executive's base salary and average annual bonus. In the event that a qualifying termination of employment were to occur immediately following the completion of the merger, the amount of the cash severance payable under the existing FleetBoston agreements is currently estimated to be \$0 for Mr. Gifford (due to the reduction described below), \$7,850,000 for Mr. McOuade, \$7,400,000 for Mr. Sarles, \$0 for Mr. Warner (due to the reduction described below) and \$5,025,000 for Mr. Moynihan, although at completion of the merger the right of each of Messrs. McQuade, Sarles and Moynihan to this cash severance will be replaced by the right to receive a deferred amount under his new employment agreement with Bank of America. See "New Employment Agreements with Bank of America" on page 60. In the event that qualifying terminations of employment were to occur immediately following the completion of the merger, the aggregate amount of the cash severance payable under the existing FleetBoston agreements with the ten other FleetBoston executive officers is currently estimated to be approximately \$30.2 million. In addition, the covered executive will be entitled to three or two years (depending on the covered executive) of (1) additional age and service credit under FleetBoston's tax-qualified and supplemental defined benefit retirement plans, (2) employer contributions under FleetBoston's taxqualified and supplemental defined contribution retirement plans and (3) continued welfare benefit coverage.

In the case of Messrs. Gifford and Warner, any cash severance amounts under the existing FleetBoston agreements will be reduced by the cash severance amounts that these executives were entitled to receive under their change in control agreements with the former BankBoston Corporation as a result of the merger between Fleet Financial Group, Inc. and BankBoston Corporation on October 1, 1999. Ms. Szostak is also party to an employment agreement that provides for severance benefits upon a qualifying termination of employment, but she is not entitled to duplicate benefits.

If any amounts or benefits received under the existing agreements or otherwise are subject to the excise tax imposed under section 4999 of the Internal Revenue Code, an additional payment will be made to restore the executive to the after-tax position that he or she would have been in if the excise tax had not been imposed, as long as the net after-tax benefit to the executive of making that additional payment is at least \$50,000.

New Employment Agreements with Bank of America.

In connection with the execution of the merger agreement, Bank of America entered into employment agreements with each of Messrs. McQuade, Sarles, Warner and Moynihan to be effective upon completion of the merger. Each agreement contemplates a three-year employment period commencing upon completion of the merger. During the employment period:

- Mr. McQuade will serve as President of Bank of America with the following business lines reporting to him: Administrative Services, International, Global Treasury Management, Principal Investing, Technology and Operations.
- Mr. Sarles will serve as Vice Chairman, special advisor to the chief executive officer and co-head of merger integration.

- Mr. Warner will serve as President of Bank of America's premier and small business banking operations.
- Mr. Moynihan will serve as President of Bank of America's wealth management business.

During the employment period, each executive will report directly to Bank of America's chief executive officer, and Mr. McQuade will also be a member of Bank of America's board of directors. The annual base salary payable to each of the executive officers is \$1,250,000 for Mr. McQuade, \$850,000 for Mr. Sarles, \$700,000 for Mr. Warner and \$600,000 for Mr. Moynihan. During the employment period, each executive will be eligible to receive an annual bonus and annual equity awards commensurate with his position and on a comparable basis to similarly situated executives of Bank of America. Generally, the executives will participate in the benefit plans of Bank of America on the same basis as similarly situated executives of Bank of America, except that the FleetBoston supplemental executive retirement plan and fringe benefits will continue to be provided.

Under the new employment agreements, the cash severance amounts payable under the executive's existing agreements with FleetBoston will be deferred until the executive's employment with Bank of America is terminated, with half of that amount converted into Bank of America restricted common stock units and the other half converted into a deferred compensation account with interest credited at the prior month's one year constant maturity treasury rate as determined each month by the Federal Reserve. The deferred cash severance amount is currently estimated to be \$7,850,000 for Mr. McQuade, \$7,400,000 for Mr. Sarles, \$0 for Mr. Warner and \$5,025,000 for Mr. Moynihan.

The new employment agreements provide that, upon the termination of an executive's employment with Bank of America other than for "cause," death or "disability", or if the executive terminates employment for "good reason," the executive will be entitled to receive any unpaid base salary, a pro-rata annual bonus (based on the highest bonus earned in the three years prior to the date of termination), and two times the sum of the executive's base salary and highest annual bonus. In addition, any equity incentive awards granted to the executive following the date of the completion of the merger will vest to the extent they would have vested in accordance with their terms during the two-year period following the date of termination, and the executive will be entitled to receive the retirement and welfare benefits that were payable upon a termination by the company other than for "cause" or by the executive for "good reason" pursuant to the terms of the existing agreements with FleetBoston. See "Existing FleetBoston Agreements" on page 60.

If any amounts or benefits received under the new agreements or otherwise are subject to the excise tax imposed under Section 4999 of the Internal Revenue Code, an additional payment will be made to restore the executive to the after-tax position that he would have been in if the excise tax had not been imposed, as long as the net after-tax benefit to the executive of making such additional payment is at least \$50,000.

Other FleetBoston Benefit Plans.

Under FleetBoston's Deferred Compensation Plan 2, upon a change in control, participants become vested under the plan, and the annual applicable interest rate equivalents with respect to deferred amounts become fixed, so that generally they are not less than the applicable rates as in effect prior to the change in control. Under the FleetBoston Management Incentive Bonus Plan, eligible executives are entitled to a pro-rata bonus for the year in which the change in control occurs, subject to reduction or offset in the event an eligible executive is entitled to a duplicate bonus in respect of the same period of service under any other FleetBoston plan or agreement. The merger will be a change in control for purposes of these plans. In addition, Bank of America has agreed to continue the FleetBoston Supplemental Executive Retirement Plan for the benefit of the FleetBoston employees participating as of the completion of the merger.

Equity-Based Awards.

The merger agreement provides that, upon completion of the merger, each outstanding and unexercised stock option or stock appreciation right to acquire shares of FleetBoston common stock (or, in the case of stock appreciation rights, receive a cash payment based on the price of FleetBoston common stock), and each share of FleetBoston restricted common stock and restricted stock unit award in respect of FleetBoston common stock granted under FleetBoston's stock plans will cease to represent the right to acquire or receive these shares of FleetBoston common stock or these payments, and will be converted into a right to acquire or receive the number of shares of (or, in the case of stock appreciation rights, a payment based on the price of) Bank of America common stock equal to the number of shares of FleetBoston common stock covered by the FleetBoston equity award times the exchange ratio, with the per share exercise price of each converted stock option and stock appreciation right equaling its pre-merger per share exercise price divided by the exchange ratio. See "The Merger Agreement — Treatment of FleetBoston Stock Options and Other Equity-Based Awards" on page 63.

Under the equity-based plans of FleetBoston and related award agreements, stock options to acquire FleetBoston common stock and FleetBoston restricted common stock and restricted stock unit awards in respect of FleetBoston common stock granted before the announcement of the merger, including those held by FleetBoston's executive officers, will fully vest or become free of restrictions and be deemed earned in full upon completion of the merger. Any FleetBoston equity-based awards granted to FleetBoston employees after the announcement of the merger, including any held by FleetBoston's executive officers, will not vest or become free of restrictions upon completion of the merger; however, these awards will vest upon certain qualifying terminations of employment following completion of the merger, including a termination by the combined company without cause or by the employee for good reason.

Assuming the merger is completed on April 1, 2004, the number of unvested stock options to acquire shares of FleetBoston common stock, and the number of shares of FleetBoston restricted stock or common stock underlying restricted stock unit awards held by the following FleetBoston executive officers that will become fully vested or free of restrictions and be deemed earned in full upon the completion of the merger is as follows: (1) Mr. Gifford 486,726; (2) Mr. McQuade 350,787; (3) Mr. Sarles 351,048; and (4) Mr. Warner 220,857; and (5) Mr. Moynihan 219,665.

Assuming the merger is completed on April 1, 2004, the number of unvested stock options to acquire shares of FleetBoston common stock and the number of shares of FleetBoston restricted stock or common stock underlying restricted stock unit awards held by the other executive officers of FleetBoston as a group that will become fully vested or free of restrictions and be deemed earned in full upon the completion of the merger is 1,542,598. The non-employee directors of FleetBoston do not hold any stock options or other equity-based awards that will vest or become free of restrictions as a result of the completion of the merger.

FleetBoston Director Representation.

As described above under "The Merger — Board of Directors and Management of Bank of America following Completion of the Merger" on page 50, upon completion of the merger, seven of the current FleetBoston directors and officers will join the Bank of America board of directors, which will have a total of nineteen directors. This seven to twelve ratio of FleetBoston to Bank of America directors will be maintained for at least two years following completion of the merger, and any vacancy among the seven former FleetBoston directors during this period will be filled by nominees proposed to the nominating committee by a majority of the remaining former FleetBoston directors. The FleetBoston directors on the combined company board will include Mr. Gifford, who will serve as the chairman of the board, Mr. McQuade and five other current FleetBoston directors who will be selected prior to completion of the merger. FleetBoston officers and directors who serve on the combined company's board of directors are expected to be compensated for their services in that capacity in accordance with Bank of America's standard director compensation policy. Consistent with this policy, it is anticipated that non-employee members of the FleetBoston board of directors who join the Bank of America

board will receive remuneration for the period served as directors for the period between the merger date and the next annual meeting of stockholders, and employee members of the FleetBoston board of directors who join the Bank of America board will receive no additional compensation for their service as directors.

Protection of FleetBoston Directors and Officers Against Claims.

Bank of America has agreed to indemnify and hold harmless each present and former director, officer and employee of FleetBoston from liability and expenses for matters arising at or prior to the completion of the merger to the fullest extent provided by applicable law, the FleetBoston restated articles of incorporation and the FleetBoston bylaws, and to the same extent that they would have been indemnified as a FleetBoston director, officer and employee or otherwise under indemnification agreements they may have had with FleetBoston. Bank of America also has agreed that it will maintain FleetBoston's current policy of directors' and officers' liability insurance coverage for the benefit of FleetBoston directors and officers for six years following completion of the merger.

THE MERGER AGREEMENT

The following describes certain aspects of the merger, including material provisions of the merger agreement. The following description of the merger agreement is subject to, and qualified in its entirety by reference to, the merger agreement, which is attached to this document as Appendix A and is incorporated by reference in this document. We urge you to read the merger agreement carefully and in its entirety.

Terms of the Merger

The merger agreement provides for the merger of FleetBoston with and into Bank of America. Bank of America will be the surviving corporation in the merger. Each share of FleetBoston common stock issued and outstanding immediately prior to the completion of the merger, except for specified shares of FleetBoston common stock held by FleetBoston and Bank of America, will be converted into the right to receive 0.5553 of a share of Bank of America common stock. Each share of each series of FleetBoston preferred stock issued and outstanding immediately prior to the completion of the merger, except for specified shares of FleetBoston preferred stock held by FleetBoston and Bank of America, will be converted into a share of a series of Bank of America preferred stock with the same terms (to the fullest extent possible) as the corresponding FleetBoston preferred stock.

Bank of America will not issue any fractional shares of Bank of America common stock in the merger. Instead, a FleetBoston stockholder who otherwise would have received a fraction of a share of Bank of America common stock will receive an amount in cash rounded to the nearest cent. This cash amount will be determined by multiplying the fraction of a share of Bank of America common stock to which the holder would otherwise be entitled by the average of the closing sale prices of Bank of America common stock on the NYSE Composite Transactions Tape as reported by *The Wall Street Journal* for the five full trading days immediately preceding the date of the effective time of the merger.

The merger agreement provides that upon completion of the merger, the Bank of America board of directors will have ninteen members, twelve of whom will be individuals who are currently Bank of America directors and seven of whom will be individuals who are currently FleetBoston directors or officers.

Treatment of FleetBoston Stock Options and Other Equity-Based Awards

The merger agreement provides that, upon completion of the merger, each outstanding and unexercised stock option or stock appreciation right to acquire shares of FleetBoston common stock (or, in the case of stock appreciation rights, receive a cash payment based on the price of FleetBoston common stock) and each share of FleetBoston restricted common stock and restricted stock unit awards in respect of FleetBoston common stock granted under the FleetBoston stock plans will cease to represent the right to acquire or receive shares of (or, in the case of stock appreciation rights, a payment based on the price of) FleetBoston common stock and will be converted into, and become a right, to acquire or receive the number of shares of (or, in the case of stock appreciation rights, a payment based on the price of) Bank of America common stock equal to the number of shares of FleetBoston common stock covered by the option, stock appreciation right or restricted stock or restricted stock unit award times the exchange ratio, with the exercise price of each converted stock option and stock appreciation right per share of Bank of America common stock equaling the per share exercise price of the FleetBoston stock option or stock appreciation right divided by the exchange ratio. FleetBoston's employee stock purchase plan will be terminated shortly before the merger is completed.

Bank of America has agreed to assume FleetBoston's obligations with respect to the FleetBoston stock options, stock appreciation rights and other equity-based awards that are converted into Bank of America stock options, stock appreciation rights or other equity-based awards as described above and otherwise in accordance with the terms of the plans and award agreements under which they have been granted. Bank of America has agreed to reserve additional shares of Bank of America common stock to satisfy its obligations under the

converted stock options, stock appreciation rights and other equity-based awards and file a registration statement with the SEC on an appropriate form to the extent necessary to register Bank of America common stock subject to the converted stock options, stock appreciation rights and other equity-based awards.

Closing and Effective Time of the Merger

The merger will be completed only if all of the following occur:

- the merger agreement is adopted by Bank of America stockholders,
- · the merger agreement is approved by FleetBoston stockholders,
- · we obtain all required governmental and regulatory consents and approvals, and
- · all other conditions to the merger discussed in this document and the merger agreement are either satisfied or waived.

The merger will become effective when articles of merger are filed with the Secretary of State of Rhode Island and a certificate of merger is filed with the Secretary of State of the State of Delaware. In the merger agreement, we have agreed to cause the completion of the merger to occur no later than the fifth business day following the satisfaction or waiver of the last of the conditions specified in the merger agreement, or on another mutually agreed date. It currently is anticipated that the effective time of the merger will occur during the first half of 2004, but we cannot guarantee when or if the merger will be completed. The Bank of America amended and restated certificate of incorporation and the Bank of America bylaws as in effect immediately prior to the effective time will be the Bank of America amended and restated certificate of incorporation and the Bank of America bylaws upon completion of the merger.

Representations, Warranties, Covenants and Agreements

The merger agreement contains generally reciprocal customary representations and warranties of FleetBoston and Bank of America relating to their respective businesses. With the exception of certain representations that must be true and correct in all material respects, no representation will be deemed untrue or incorrect as a consequence of the existence or absence of any fact or event unless that fact or event has had or is reasonably likely to have a material adverse effect on the company making the representation. The representations in the merger agreement do not survive the effective time of the merger.

Each of Bank of America and FleetBoston has made representations and warranties regarding, among other things:

- · corporate matters, including due organization and qualification;
- capitalization;
- authority relative to execution and delivery of the merger agreement and the absence of conflicts with, or violations of, organizational documents or other obligations as a result of the merger;
- · required governmental filings and consents;
- · the timely filing of reports with governmental entities, and the absence of investigations by regulatory agencies;
- · financial statements;
- · broker's fees payable in connection with the merger;
- the absence of material adverse effects;
- · legal proceedings;
- · tax matters;

- SEC reports;
- compliance with applicable laws;
- · the absence of agreements with regulatory agencies;
- interest rate risk management instruments;
- the absence of undisclosed liabilities;
- environmental liabilities;
- · the inapplicability of state takeover laws and the FleetBoston rights agreement;
- · the absence of knowledge preventing the merger from qualifying as a reorganization;
- the receipt of fairness opinions from financial advisors; and
- the accuracy of information supplied for inclusion in this document and other similar documents.

In addition, FleetBoston has made other representations and warranties about itself to Bank of America regarding, among other things, matters relating to certain contracts and employment matters.

Each of Bank of America and FleetBoston has undertaken customary covenants that place restrictions on it and its subsidiaries until the effective time of the merger. In general, each of Bank of America and FleetBoston agrees to (1) conduct its business in the ordinary course in all material respects, (2) use reasonable best efforts to maintain and preserve intact its business organization, employees and advantageous business relationships (including retaining the services of key officers and employees), and (3) take no action that would adversely affect or materially delay its respective ability to obtain any necessary regulatory approvals, perform its covenants or complete the transaction. FleetBoston further agrees that, except with Bank of America's prior written consent, FleetBoston will not, among other things, undertake the following extraordinary actions:

- incur indebtedness or in any way assume the indebtedness of another person (other than one of its wholly owned subsidiaries), except in the ordinary course of business;
- · adjust, split, combine or reclassify any capital stock of FleetBoston or its subsidiaries;
- make, declare or pay any dividends or other distributions on any shares of its capital stock, other than regular cash dividends;
- grant any stock appreciation rights or stock purchase rights;
- · issue shares or stock options outside the parameters set forth in the merger agreement;
- except as contemplated by the merger agreement, (1) grant any salary increase, (2) pay any pension or retirement allowance, (3) pay any bonus other than customary year-end bonuses for 2003, (4) enter into any employee benefit plan, (5) accelerate the vesting of or lapse of restrictions with respect to FleetBoston stock options, stock appreciation rights or other stock-based awards;
- other than in the ordinary course of business, sell, transfer, mortgage, encumber or otherwise dispose of any assets or properties, or cancel, release or assign any indebtedness, in each case that is material to FleetBoston or its subsidiaries;
- enter into any new line of business or change its lending, investment, underwriting, risk and asset liability management and other banking and operating policies material to FleetBoston or its subsidiaries other than as required by applicable law;
- make any material investment either by purchase of securities, capital contributions, property transfers or purchase of property or assets other than in the ordinary course of business:
- knowingly take any action or fail to take any action reasonably likely to prevent the merger from qualifying as a reorganization;

- amend any charter documents;
- restructure or materially change its investment securities portfolio or its gap position;
- · settle any material claim, except in the ordinary course of business;
- take any action that is reasonably likely to result in any representations or warranties under the merger agreement becoming untrue in any material respect;
- change its tax accounting or financial accounting methods, except as required by applicable law;
- take any action that would materially delay necessary governmental or regulatory approvals; or
- · agree to take or adopt any resolutions by the board of directors in opposition to any of the FleetBoston covenants made in the merger agreement.

Bank of America agrees that, except with FleetBoston's prior written consent, Bank of America will not, among other things, undertake the following extraordinary actions:

- · amend any charter documents;
- · knowingly take any action or fail to take any action reasonably likely to prevent the merger from qualifying as a reorganization;
- take any action that would materially delay necessary governmental or regulatory approvals;
- · take any action that is reasonably likely to result in any representations or warranties under the merger agreement becoming untrue in any material respect; or
- agree to take or adopt any resolutions by the board of directors in opposition to any of the Bank of America covenants made in the merger agreement.

The merger agreement also contains mutual covenants relating to the preparation of this joint proxy statement/prospectus and the holding of special meetings of Bank of America stockholders and FleetBoston stockholders, access to information of the other company and public announcements with respect to the transactions contemplated by the merger agreement.

Declaration and Payment of Dividends

FleetBoston has agreed that, until the merger is completed, it will not pay dividends other than regular quarterly cash dividends not in excess of \$0.35 per share of FleetBoston common stock. FleetBoston and Bank of America also have agreed to coordinate declaration of dividends so that holders of FleetBoston common stock will not receive two dividends, or fail to receive one dividend, for any quarter with respect to their FleetBoston common stock and any Bank of America common stock any holder receives in the merger.

Agreement Not to Solicit Other Offers

FleetBoston also has agreed that it, its subsidiaries and their officers, directors, employees, agents and representatives will not, directly or indirectly:

- · initiate, solicit, encourage or facilitate any inquiries or proposals for any "Acquisition Proposal" (as defined below); or
- · participate in any discussions or negotiations, or enter into any agreement, regarding any "Alternative Transaction" (as defined below).

However, prior to the FleetBoston special meeting, FleetBoston may consider and participate in discussions and negotiations with respect to abona fide Acquisition Proposal if (1) it has first entered into a confidentiality agreement with the party proposing the Acquisition Proposal on terms comparable to the confidentiality agreement with Bank of America and (2) the FleetBoston board of directors determines reasonably in good faith (after consultation with outside legal counsel) that failure to take these actions would cause it to violate its fiduciary duties.

FleetBoston has agreed:

- to notify Bank of America promptly (but in no event later than 24 hours) after it receives any Acquisition Proposal, or any material change to any Acquisition Proposal, or any request for nonpublic information relating to FleetBoston or any of its subsidiaries, and to provide Bank of America with relevant information regarding the Acquisition Proposal or request;
- to keep Bank of America fully informed, on a current basis, of any material changes in the status and any material changes in the terms of any such Acquisition Proposal; and
- to cease any existing discussions or negotiations with any persons with respect to any Acquisition Proposal, and to use reasonable best efforts to cause all persons other than Bank of America who have been furnished with confidential information in connection with an Acquisition Proposal within the 12 months prior to the date of the merger agreement to return or destroy such information.

As used in the merger agreement, an "Acquisition Proposal" means any inquiry or proposal regarding any merger, share exchange, consolidation, sale of shares of capital stock (including, without limitation, by way of a tender offer) or similar transactions involving FleetBoston or any of its subsidiaries that, if completed, would constitute an Alternative Transaction.

As used in the merger agreement, "Alternative Transaction" means any of the following:

- a transaction pursuant to which any person (or group of persons) other than Bank of America or its affiliates, directly or indirectly, acquires or would acquire more than 25% of the outstanding shares of FleetBoston common stock or outstanding voting power or of any new series or new class of FleetBoston preferred stock that would be entitled to a class or series vote with respect to the merger, whether from FleetBoston or pursuant to a tender offer or exchange offer or otherwise;
- · a merger, share exchange, consolidation or other business combination involving FleetBoston (other than the merger being described here);
- any transaction pursuant to which any person (or group of persons) other than Bank of America or its affiliates acquires or would acquire control of assets (including, for this purpose, the outstanding equity securities of subsidiaries of FleetBoston and securities of the entity surviving any merger or business combination including any of FleetBoston's subsidiaries) of FleetBoston, or any of its subsidiaries representing more than 25% of the fair market value of all the assets, net revenues or net income of FleetBoston and its subsidiaries, taken as a whole, immediately prior to such transaction; or
- any other consolidation, business combination, recapitalization or similar transaction involving FleetBoston or any of its subsidiaries, other than the transactions contemplated by the merger agreement, as a result of which the holders of shares of FleetBoston common stock immediately prior to the transaction do not, in the aggregate, own at least 75% of each of the outstanding shares of common stock and the outstanding voting power of the surviving or resulting entity in the transaction immediately after the completion of the transaction in substantially the same proportion as the holders held the shares of FleetBoston common stock immediately prior to the completion of the transaction.

Expenses and Fees

In general, each of Bank of America and FleetBoston will be responsible for all expenses incurred by it in connection with the negotiation and completion of the transactions contemplated by the merger agreement.

However, the costs and expenses of printing and mailing this joint proxy statement/prospectus, and all filing and other fees paid to the SEC in connection with the merger, shall be borne equally by FleetBoston and Bank of America.

Conditions to Complete the Merger

Our respective obligations to complete the merger are subject to the fulfillment or waiver of certain conditions, including:

- the adoption of the merger agreement by the Bank of America stockholders and FleetBoston stockholders;
- the receipt and effectiveness of all governmental and other approvals, registrations and consents, and the expiration of all related waiting periods required to complete
 the merger;
- the absence of any law, statute, regulation, judgment, decree, injunction or other order in effect by any court or other governmental entity that prohibits completion of the transactions contemplated by the merger agreement;
- the registration statement with respect to the Bank of America common stock to be issued in the merger shall have become effective under the Securities Act and no stop order or proceedings for that purpose will have been initiated or threatened by the SEC;
- the approval of the listing of Bank of America common stock and Bank of America preferred stock to be issued in the merger on the NYSE, subject to official notice of issuance:
- the truth and correctness of the representations and warranties of each of us in the merger agreement, subject to the materiality standard provided in the merger agreement, and the performance by each of us in all material respects of our obligations under the merger agreement and the receipt by each of us of certificates from the other to that effect; and
- · the receipt by each of Bank of America and FleetBoston of a legal opinion with respect to certain federal income tax consequences of the merger.

We cannot provide assurance as to when or if all of the conditions to the merger can or will be satisfied or waived by the appropriate party. As of the date of this document, we have no reason to believe that any of these conditions will not be satisfied.

Amendment, Waiver and Termination of the Merger Agreement

Subject to applicable law, the parties may amend the merger agreement by action taken or authorized by their boards of directors or by written agreement. However, after any approval of the transactions contemplated by the merger agreement by the FleetBoston stockholders and Bank of America stockholders, there may not be, without further approval of those stockholders, any amendment of the merger agreement that changes the amount or the form of the consideration to be delivered to the holders of FleetBoston common stock other than as contemplated by the merger agreement. Either party to the merger agreement may waive any inaccuracies in the representations and warranties of the other party, or, subject to applicable law, may waive compliance by the other party with any of the other agreements or conditions contained in the merger agreement. The merger agreement can be terminated at any time prior to closing by mutual consent and by either party in the following circumstances:

- if any of the required regulatory approvals are denied (and the denial is final and nonappealable);
- if the merger has not been completed by October 27, 2004, unless the failure to complete the merger by that date is due to the terminating party's actions;
- if there is a breach by the other party that would cause the failure of the closing conditions described above, unless the breach is capable of being, and is, cured within 45 days of notice of the breach; or

• if either Bank of America stockholders or FleetBoston stockholders vote to reject the transaction, the parties have agreed that neither party can terminate the transaction until October 27, 2004, as long as the other party has not committed a substantial, bad faith breach of its obligation to use reasonable best efforts to negotiate a restructuring of the transaction and to resubmit the transaction to its stockholders for approval.

Effect of Termination. If the merger agreement is terminated, it will become void, and there will be no liability on the part of Bank of America or FleetBoston, except that (1) termination will not relieve a breaching party from liability for any willful breach giving rise to the termination, and (2) the confidentiality agreement between the parties will survive termination. In addition, if the merger agreement is terminated, the stock option agreements will remain in effect in accordance with their terms.

Resales of Bank of America Stock by Affiliates

Affiliates of FleetBoston, as defined under Rule 145 under the Securities Act, generally may not sell their shares of Bank of America common stock acquired in the merger, except pursuant to an effective registration statement under the Securities Act, in compliance with Rule 145 under the Securities Act or pursuant to another applicable exemption from the registration requirements of the Securities Act. Affiliates include directors, executive officers, and beneficial owners of 10% or more of any class of capital stock

Under the merger agreement, FleetBoston agrees to use reasonable best efforts to cause persons that are affiliates of FleetBoston to deliver letter agreements by which each affiliate agrees, among other things, not to offer to sell, transfer or otherwise dispose of any of the shares of Bank of America common stock distributed to him, her or it pursuant to the merger, except in compliance with Rule 145 under the Securities Act, in a transaction that is otherwise exempt from the registration requirements of the Securities Act, or in an offering registered under the Securities Act. Bank of America may place restrictive legends on Bank of America stock certificates that are issued to persons who are deemed to be affiliates of FleetBoston under the Securities Act.

This document does not cover any resales of Bank of America common stock received in the merger by any person who may be deemed an affiliate of FleetBoston.

Employee Benefit Matters

The merger agreement provides that, after completion of the merger and until December 31, 2004 (or a later date if December 31 is not practicable based on the date the merger is completed), Bank of America will provide FleetBoston employees with employee benefits and compensation plans that are substantially similar in the aggregate to those provided to FleetBoston employees prior to the merger, and, thereafter, will provide employee benefits and compensation plans that are equivalent to those provided to similarly-situated Bank of America employees. Eligible FleetBoston employees whose employment is terminated at any time during the first two years following completion of the merger will be entitled to receive severance benefits in accordance with the terms of the applicable FleetBoston severance plan as in effect immediately prior to completion of the merger.

Bank of America will generally provide the FleetBoston employees with service credit for their service with Fleet for purposes of eligibility, participation, vesting and levels of benefits (but generally not for benefit accruals under defined benefit pension plans) under the employee benefit and compensation plans of Bank of America in which such employees are eligible to participate following the merger. Bank of America has agreed to waive specified exclusions and limitations under its welfare benefit plans in which the FleetBoston employees are eligible to participate following the merger to the extent waived under the corresponding FleetBoston plan in which the applicable employee participated prior to the merger and to give FleetBoston employees credit, for the plan year in which they start participating in any such plan, towards applicable deductibles and annual out-of-pocket limits for expenses incurred before such participation.

THE STOCK OPTION AGREEMENTS

The following description, which sets forth the material provisions of the Bank of America stock option agreement and the FleetBoston stock option agreement, is subject to the full text of, and qualified in its entirety by reference to, the stock option agreements, which are attached to this document as Appendices B and C and are incorporated by reference in this document. We urge you to read the stock option agreements carefully and in their entirety.

The Stock Options

At the same time we entered into the merger agreement, we also entered into reciprocal stock option agreements. Under the terms of the stock option granted by FleetBoston to Bank of America, Bank of America may purchase up to 209,496,275 shares of FleetBoston common stock at an exercise price of \$31.80 per share. Under the terms of the stock option granted by Bank of America to FleetBoston, FleetBoston may purchase up to 297,374,945 shares of Bank of America common stock at an exercise price of \$81.86 per share. However, the number of shares issuable upon exercise of the options cannot exceed 19.9% of the Bank of America or FleetBoston common stock outstanding without giving effect to any shares issued under the option. In the event that any additional shares of common stock are either issued or redeemed after the date of the stock option agreements, the number of shares of common stock subject to the option will be adjusted so that such number equals 19.9% of the number of shares of common stock then issued and outstanding without giving effect to any shares of common stock subject or issued under the option. These exercise prices represent our closing stock prices on October 24, 2003, the first business day prior to the execution of the merger agreement and the stock option agreements. The terms of the stock option agreements are similar in most respects and are summarized below.

Purpose of the Stock Option Agreements

The stock option agreements may have the effect of making an acquisition or other business combination of either Bank of America or FleetBoston by a third party more costly because of the need in any transaction to acquire any shares of common stock issued under the stock option agreements or because of any cash payments made under the stock option agreements. The stock option agreements may, therefore, discourage certain third parties from proposing an alternative transaction to the merger, including one that might be more favorable, from a financial point of view, to Bank of America stockholders or FleetBoston stockholders, as the case may be, than the merger.

To our knowledge, no event giving rise to the right to exercise either option has occurred as of the date of this document.

Exercise; Expiration

Each of Bank of America and FleetBoston can exercise its option in whole or in part if both an Initial Triggering Event and a Subsequent Triggering Event occur prior to the occurrence of an Exercise Termination Event, as these terms are described below. The purchase of any shares of Bank of America common stock or FleetBoston common stock pursuant to the options is subject to compliance with applicable law, which may require regulatory approval.

The stock option agreements describe a number of different events as Initial Triggering Events. Generally, an Initial Triggering Event will occur if Bank of America or FleetBoston enter into, propose to enter into, or are the subject of an Acquisition Transaction or a proposed Acquisition Transaction other than the merger agreement.

As used in each stock option agreement, "Initial Triggering Events" generally mean the following:

- Bank of America or FleetBoston, respectively, or any of its subsidiaries, without the other party's prior written consent, enters into an agreement to engage in an "Acquisition Transaction" (as defined below) with a third party or its board of directors recommends that its stockholders approve or accept any Acquisition Transaction with any person other than the other party or its subsidiary;
- Bank of America or FleetBoston, respectively, or any of its subsidiaries, without the other party's prior written consent, authorizes, recommends, proposes or publicly
 announces its intention to authorize, recommend or propose, to engage in an Acquisition Transaction with a third party or its board of directors publicly withdraws or
 modifies, or publicly announces its intention to withdraw or modify, its recommendation that the stockholders approve the transactions contemplated by the merger
 agreement;
- Any third party acquires beneficial ownership or the right to acquire beneficial ownership of 10% or more of the outstanding shares of either Bank of America common stock or FleetBoston common stock;
- Any person, other than Bank of America or FleetBoston, makes abona fide proposal to either party or its stockholders by public announcement or written communication that is or becomes the subject of public disclosure to engage in an Acquisition Transaction;
- After an overture is made by a third party to Bank of America, Fleet Boston or any of their subsidiaries to engage in an Acquisition Transaction, the engaged party breaches any covenant or obligation contained in the merger agreement and the breach entitles the non-breaching party to terminate the merger agreement and the breach has not been cured prior to the date of written notice of the holder's intention to exercise the option; or
- Any third person, without the written consent of Bank of America or FleetBoston, as the case may be, files an application or notice with the Federal Reserve Board or
 other federal or state bank regulatory authority, which application or notice has been accepted for processing, for approval to engage in an Acquisition Transaction.

As used in each stock option agreement, the term "Acquisition Transaction" means:

- a merger, consolidation or share exchange, or any similar transaction, involving Bank of America or FleetBoston or any of their respective significant subsidiaries;
- a purchase, lease or other acquisition or assumption of all or a substantial portion of the assets or deposits of Bank of America or FleetBoston or any of their respective significant subsidiaries;
- a purchase or other acquisition of securities representing 10% or more of the voting power of Bank of America or 10% or more of the voting power of FleetBoston;
- any substantially similar transaction, except that any substantially similar transaction involving only one party and one or more of its subsidiaries or involving only any two or more of these subsidiaries will not be deemed to be an Acquisition Transaction, provided that:
 - any substantially similar transaction is not entered into in violation of the merger agreement, and
 - for Bank of America, any such transaction in which Bank of America stockholders, in the aggregate, would own at least 60% of the voting common shares of
 the surviving or acquiring entity immediately following the completion of the transaction will not be deemed to be an Acquisition Transaction.

Each stock option agreement generally defines the term "Subsequent Triggering Event" to mean any of the following events or transactions:

• the acquisition by a third party of beneficial ownership of 20% or more of the outstanding Bank of America common stock or FleetBoston common stock; or

• Bank of America or FleetBoston enters into an agreement to engage in an Acquisition Transaction with a third party or its board of directors recommends that its stockholders approve or accept any Acquisition Transaction or proposed Acquisition Transaction other than the merger agreement. For this purpose, the percentage referred to in the definition of Acquisition Transaction is 20% instead of 10%.

Each stock option agreement defines the term "Exercise Termination Event" to mean any of the following:

- · completion of the merger;
- termination of the merger agreement in accordance with its terms before an Initial Triggering Event, except a termination of the merger agreement by Bank of America or FleetBoston based on a breach by the other party of a representation, warranty, covenant or other agreement contained in the merger agreement (unless the breach is non-volitional); or
- the passage of 18 months, subject to the extension described below, after termination of the merger agreement, if the termination follows the occurrence of an Initial Triggering Event or is a termination of the merger agreement by Bank of America or FleetBoston based on a breach by the other party of a representation, warranty, covenant or other agreement contained in the merger agreement (unless the breach is non-volitional).

If either option becomes exercisable, it may be exercised, in whole or in part, within 180 days following the Subsequent Triggering Event. Bank of America and FleetBoston's right to exercise its option and certain other rights under the stock option agreements are subject to an extension in order to obtain required regulatory approvals and comply with applicable regulatory waiting periods and to avoid liability under the short-swing trading restrictions contained in Section 16(b) of the Exchange Act.

Rights Under the Stock Option Agreements

Immediately prior to a Repurchase Event, as this term is described below, and prior to an Exercise Termination Event subject to extension as described above, following a request of Bank of America or FleetBoston, the other party may be required to repurchase the option and all or any part of the shares issued under the option. The repurchase of the option will be at a price equal to the number of shares for which the option may be exercised multiplied by the amount by which the market/offer price, as that term is defined in the stock option agreements, exceeds the option price. At the request of the owner of option shares from time to time, Bank of America or FleetBoston may be required to repurchase such number of the option shares from the owner as designated by the owner at a price equal to the market/offer price, as that term is defined in the stock option agreement, multiplied by the number of option shares so designated. The parties agree that the obligation to repurchase the option or option shares under this provision will not terminate upon the occurrence of an Exercise Termination Event unless no Subsequent Triggering Event has occurred prior to the occurrence of an Exercise Termination Event. The term "Repurchase Event" is defined to mean:

- completion of any merger, consolidation, share exchange or similar transaction involving Bank of America or FleetBoston, or any purchase, lease or other acquisition of all or a substantial portion of the assets of Bank of America or FleetBoston, other than any transaction that would not constitute an Acquisition Transaction; or
- acquisition by any person of beneficial ownership of 50% or more of the then-outstanding shares of common stock of the grantor of the option. However, an acquisition will not constitute a Repurchase Event unless a Subsequent Triggering Event has occurred prior to an Exercise Termination Event.

Each stock option agreement also provides that Bank of America and FleetBoston may, at any time during which either party would be required to repurchase the option or any option shares upon proper request or notice, subject to extension as described above, surrender the option and any shares issued under the option held by

Bank of America or FleetBoston to FleetBoston or Bank of America, respectively, for a cash fee equal to \$1.8948 billion, adjusted for the aggregate purchase price previously paid by Bank of America or FleetBoston with respect to any option shares and gains on sales of stock purchased under the option. However, neither Bank of America nor FleetBoston may exercise its surrender right if the other party repurchases the option, or a portion of the option.

If, prior to an Exercise Termination Event, either Bank of America or FleetBoston enters into certain mergers, consolidations or other transactions, certain fundamental changes in its capital stock occur, or it sells all or substantially all of its assets to any person other than the other party or one of its subsidiaries, the option will be converted into, or be exchanged for, a substitute option, at the other party's election, of:

- · the continuing or surviving corporation of a consolidation or merger with Bank of America or FleetBoston,
- · Bank of America or FleetBoston in a merger in which it is the continuing or surviving person,
- the transferee of all or substantially all of the assets of Bank of America or FleetBoston, or
- any person that controls any of these entities, as the case may be.

The substitute option will have the same terms as the original option (including a repurchase right, but based on the closing price of the common stock of the substitute issuer). However, if, because of legal reasons, the terms of the substitute option cannot be the same as those of the original option, the terms of the substitute option will be as similar as possible and at least as advantageous to FleetBoston or Bank of America. Also, the number of shares exercisable under the substitute option is capped at 19.9% of the shares of common stock outstanding prior to exercise. In the event this cap would be exceeded, the issuer of the substitute option will pay Bank of America or FleetBoston, respectively, the difference between the value of a capped and non-capped option.

The stock option agreements provide that the total profit realized by Bank of America or FleetBoston as a result of a stock option agreement will not exceed \$2.3685 billion.

ACCOUNTING TREATMENT

The merger will be accounted for as a "purchase," as that term is used under GAAP, for accounting and financial reporting purposes. Under purchase accounting, the assets (including identifiable intangible assets) and liabilities (including executory contracts and other commitments) of FleetBoston as of the effective time of the merger will be recorded at their respective fair values and added to those of Bank of America. Any excess of purchase price over the fair values is recorded as goodwill. Financial statements of Bank of America issued after the merger would reflect these fair values and would not be restated retroactively to reflect the historical financial position or results of operations of FleetBoston. See "Unaudited Pro Forma Condensed Combined Financial Information" on page 104.

MATERIAL FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER

The following section describes the anticipated material U.S. federal income tax consequences of the merger to holders of FleetBoston stock. This discussion addresses only those FleetBoston stockholders that hold their FleetBoston stock as a capital asset within the meaning of Section 1221 of the Internal Revenue Code, and does not address all the U.S. federal income tax consequences that may be relevant to particular FleetBoston stockholders in light of their individual circumstances or to FleetBoston stockholders that are subject to special rules, such as:

- · financial institutions,
- · insurance companies,

- tax-exempt organizations,
- dealers in securities or currencies,
- traders in securities that elect to use a mark to market method of accounting,
- · persons that hold FleetBoston stock as part of a straddle, hedge, constructive sale or conversion transaction,
- · persons who are not citizens or residents of the United States, and
- · stockholders who acquired their shares of FleetBoston stock through the exercise of an employee stock option or otherwise as compensation.

The following is based upon the Internal Revenue Code, its legislative history, existing and proposed regulations under the Internal Revenue Code and published rulings and decisions, all as currently in effect as of the date of this joint proxy statement/prospectus, and all of which are subject to change, possibly with retroactive effect. Tax considerations under state, local and foreign laws, or federal laws other than those pertaining to income tax, are not addressed in this document. Determining the actual tax consequences of the merger to you may be complex. They will depend on your specific situation and on factors that are not within our control. You should consult with your own tax advisor as to the tax consequences of the merger in your particular circumstances, including the applicability and effect of the alternative minimum tax and any state, local or foreign and other tax laws and of changes in those laws.

Tax Consequences of the Merger Generally. Bank of America and FleetBoston have structured the merger to qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code. It is a condition to FleetBoston's obligation to complete the merger that FleetBoston receive an opinion of its counsel, Wachtell, Lipton, Rosen & Katz, dated the closing date of the merger, that the merger will be treated as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code. It is a condition to Bank of America's obligation to complete the merger that Bank of America receive an opinion of its counsel, Cleary, Gottlieb, Steen & Hamilton, dated the closing date of the merger, that the merger will be treated as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code. In rendering these opinions, counsel may require and rely upon representations contained in letters and certificates to be received from FleetBoston and Bank of America. None of the tax opinions given in connection with the merger or the opinions described below will be binding on the Internal Revenue Service. Neither Bank of America nor FleetBoston intends to request any ruling from the Internal Revenue Service as to the U.S. federal income tax consequences of the merger.

Consequently, no assurance can be given that the Internal Revenue Service will not assert, or that a court would not sustain, a position contrary to any of those set forth below. In addition, if any of the representations or assumptions upon which those opinions are based are inconsistent with the actual facts, the U.S. federal income tax consequences of the merger could be adversely affected. In addition, in connection with the filing of the registration statement, Cleary, Gottlieb, Steen & Hamilton, as counsel to Bank of America, and Wachtell, Lipton, Rosen & Katz, as counsel to FleetBoston, have delivered to Bank of America and FleetBoston, respectively, their opinions, dated January 30, 2004, based on facts, representations and assumptions set forth or referred to in such opinions and factual representations contained in certificates of officers of Bank of America and FleetBoston, all of which must continue to be true and accurate as of the effective time, that the merger will qualify as a "reorganization" within the meaning of Section 368(a) of the Internal Revenue Code and that, accordingly:

- no gain or loss will be recognized by FleetBoston stockholders who receive shares of Bank of America stock in exchange for shares of FleetBoston stock, except with respect to cash received instead of fractional share interests in Bank of America common stock;
- the aggregate basis of the Bank of America stock received in the merger will be the same as the aggregate basis of the FleetBoston stock for which it is exchanged, less any basis attributable to fractional share interests in Bank of America common stock for which cash is received;

- the holding period of Bank of America stock received in exchange for shares of FleetBoston stock will include the holding period of the FleetBoston stock for which
 it is exchanged; and
- · no gain or loss will be recognized by Bank of America or FleetBoston in the merger.

Cash Received Instead of a Fractional Share of Bank of America Common Stock A FleetBoston stockholder who receives cash instead of a fractional share of Bank of America common stock will be treated as having received the fractional share of Bank of America common stock pursuant to the merger and then as having exchanged the fractional share of Bank of America common stock for cash in a redemption by Bank of America. As a result, a FleetBoston stockholder generally will recognize gain or loss equal to the difference between the amount of cash received and the basis in his, her or its fractional share of Bank of America common stock as set forth above. This gain or loss generally will be capital gain or loss, and will be long-term capital gain or loss if, as of the effective date of the merger, the holding period for the shares is greater than one year. The deductibility of capital losses is subject to limitations.

Backup Withholding and Information Reporting. Payments of cash to a holder of FleetBoston common stock instead of a fractional share of Bank of America common stock may, under certain circumstances, be subject to information reporting and backup withholding at a rate of 28% of the cash payable to the holder, unless the holder provides proof of an applicable exemption or furnishes its taxpayer identification number, and otherwise complies with all applicable requirements of the backup withholding rules. Any amounts withheld from payments to a holder under the backup withholding rules are not additional tax and will be allowed as a refund or credit against the holder's U.S. federal income tax liability, provided the required information is furnished to the Internal Revenue Service.

Subject to the limitations and qualifications set forth herein, the foregoing discussion, to the extent it describes matters of law and legal conclusions, constitutes the opinion of Cleary, Gottlieb, Steen & Hamilton and Wachtell, Lipton, Rosen & Katz as to the material U.S. federal income tax consequences of the merger to you under the law in effect as of the date of this document.

DESCRIPTION OF BANK OF AMERICA CAPITAL STOCK

A description of Bank of America's capital stock immediately after the proposed transaction is set forth below. The following statements are brief summaries of, and are subject to the provisions of, Bank of America's amended and restated certificate of incorporation and bylaws, preferred stock certificates of designation and the relevant provisions of the Delaware General Corporation Law.

Description of Common Stock

Bank of America is authorized to issue 5 billion shares of common stock, par value \$.01 per share, of which approximately 1.4 billion shares were outstanding on January 26, 2004. Bank of America common stock trades on the NYSE and on the Pacific Exchange under the symbol "BAC." Bank of America common stock is also listed on the London Stock Exchange, and certain shares of Bank America common stock are listed on the Tokyo Stock Exchange. As of January 26, 2004, approximately 337 million shares were reserved for issuance in connection with various Bank of America employee and director benefit plans and Bank of America's Dividend Reinvestment and Stock Purchase Plan and the conversion of Bank of America's outstanding convertible securities and for other purposes. After taking into account the reserved shares of Bank of America common stock, there were approximately 3.2 billion authorized shares of Bank of America common stock available for issuance as of January 26, 2004.

Voting and Other Rights

Holders of Bank of America common stock are entitled to one vote per share. There are no cumulative voting rights. In general, a majority of votes cast on a matter is sufficient to take action upon routine matters.

Directors are to be elected by a plurality of the votes cast. In general, however, (1) amendments to Bank of America's amended and restated certificate of incorporation must be approved by the affirmative vote of the holders of a majority of the outstanding shares of each class entitled to vote on those amendments as a class, and (2) a merger or dissolution or the sale of all or substantially all of Bank of America's assets must be approved by the affirmative vote of the holders of a majority of the voting power of the then-outstanding voting shares.

In the event of Bank of America's liquidation, holders of Bank of America common stock will be entitled to receiv*pro rata* any assets legally available for distribution to Bank of America stockholders, subject to any prior rights of any Bank of America preferred stock then outstanding.

Bank of America common stock does not have any preemptive rights, redemption privileges, sinking fund privileges, or conversion rights. All the outstanding shares of Bank of America common stock are, and upon proper conversion of any Bank of America preferred stock, all of the shares of Bank of America common stock into which these shares are converted will be, validly issued, fully paid, and nonassessable.

Mellon Investor Services, LLC is the transfer agent and registrar for Bank of America common stock.

Dividends

Subject to the preferential rights of any holders of any outstanding series of Bank of America preferred stock, the holders of Bank of America common stock are entitled to receive dividends or distributions, whether payable in cash or otherwise, as the Bank of America board of directors may declare out of funds legally available for these payments. Stock dividends, if any are declared, may be paid from Bank of America authorized but unissued shares.

Description of Preferred Stock

Bank of America has 100 million shares of preferred stock, par value \$.01 per share, authorized, and may issue Bank of America preferred stock in one or more series, each with such preferences, designations, limitations, conversion rights, and other rights as Bank of America may determine. Bank of America has designated:

- (1) 3 million shares of ESOP Preferred Stock, of which 1,261,824 shares were issued and outstanding at January 26, 2004;
- (2) 35,045 shares of Series B Preferred Stock, of which 7,776 shares were issued and outstanding at January 26, 2004; and
- (3) 20 million shares of \$2.50 Cumulative Convertible Preferred Stock Series BB, which we refer to as the Series BB Preferred Stock, none of which are issued and outstanding.

Any share of Bank of America preferred stock will have the general dividend, voting, and liquidation preference rights stated below unless otherwise fixed by the board of directors of Bank of America, subject to the terms of Bank of America's amended and restated certificate of incorporation.

Dividends. The holders of Bank of America preferred stock will be entitled to receive when, as and if declared by the Bank of America board of directors, cash dividends at those rates as will be fixed by the board of directors of Bank of America, subject to the terms of Bank of America's amended and restated certificate of incorporation. All dividends will be paid out of Bank of America's funds that are legally available for this purpose. Whenever dividends on any non-voting preferred stock are in arrears for six quarterly dividend periods (whether or not consecutive), holders of the non-voting preferred stock will have the right to elect two additional directors to serve on the Bank of America board of directors, and these two additional directors will continue to serve until the dividend arrearage is eliminated.

Voting. The holders of Bank of America preferred stock will have no voting rights except:

- each share of Series B Preferred Stock is entitled to one vote per share;
- each share of ESOP Preferred Stock is entitled to two votes per share;
- as required by applicable law, or
- as specifically approved by us for that particular series.

Liquidation Preference. In the event of Bank of America's voluntary or involuntary liquidation, dissolution, or winding up, the holders of any series of Bank of America preferred stock will be entitled to receive, after distributions to holders of any series or class of Bank of America capital stock ranking superior, an amount equal to the stated or liquidation value of the shares of the series plus an amount equal to accrued and unpaid dividends. If the assets and funds to be distributed among the holders of Bank of America preferred stock will be insufficient to permit full payment to the holders, then the holders of Bank of America preferred stock will share ratably in any distribution of Bank of America's assets in proportion to the amounts that they would otherwise receive on their shares of Bank of America preferred stock if the shares were paid in full.

ESOP Preferred Stock

All shares of ESOP Preferred Stock are held by the trustee under the Bank of America 401(k) Plan. The ESOP Preferred Stock ranks senior to Bank of America common stock, but ranks junior to the Series B Preferred Stock and Series BB Preferred Stock as to dividends and distributions on liquidation. Shares of ESOP Preferred Stock are convertible into Bank of America common stock at a conversion rate of 1.68 shares of Bank of America common stock per share of ESOP Preferred Sock, subject to certain customary anti-dilution adjustments.

Preferential Rights. ESOP Preferred Stock does not have preemptive or preferential rights to purchase or subscribe for shares of Bank of America capital stock, and is not subject to any sinking fund obligations or other obligations to repurchase or retire the series, except as discussed below.

Dividends. ESOP Preferred Stock is entitled to an annual dividend, subject to certain adjustments, of \$3.30 per share, payable semiannually. Unpaid dividends accumulate on the date they first became payable, without interest. While any shares of ESOP Preferred Stock are outstanding, Bank of America may not declare, pay, or set apart for payment any dividend on any other series of Bank of America stock ranking equally with the ESOP Preferred Stock as to dividends unless Bank of America has declared and paid, or set apart for payment, like dividends on ESOP Preferred Stock for all dividend payment periods ending on or before the dividend payment date for the parity Bank of America stock, ratably in proportion to their respective amounts of accumulated and unpaid dividends. Bank of America generally may not declare, pay, or set apart for payment any dividends, except for, among other things, dividends payable solely in shares of Bank of America stock ranking junior to ESOP Preferred Stock as to dividends or upon liquidation, or, make any other distribution on, or make payment on account of the purchase, redemption, or other retirement of, any other class or series of Bank of America capital stock ranking junior to ESOP Preferred Stock as to dividends or upon liquidation, until full cumulative dividends on ESOP Preferred Stock have been declared and paid or set apart for payment when due.

Voting Rights. The holder of ESOP Preferred Stock is entitled to vote on all matters submitted to a vote of the holders of Bank of America common stock and votes together with the holders of Bank of America common stock as one class. Except as otherwise required by applicable law, the holder of ESOP Preferred Stock has no special voting rights. To the extent that the holder of the shares is entitled to vote, each share is entitled to the number of votes equal to the number of shares of Bank of America common stock into which the shares of ESOP Preferred Stock could be converted on the record date for determining Bank of America stockholders entitled to vote, rounded to the nearest whole vote. For purposes of voting at the Bank of America special meeting, each share of ESOP Preferred Stock is entitled to two votes.

Distributions. In the event of Bank of America's voluntary or involuntary dissolution, liquidation, or winding-up, a holder of ESOP Preferred Stock will be entitled to receive, out of Bank of America's assets available for distribution to Bank of America stockholders, \$42.50 per share plus all accrued and unpaid dividends of the shares of ESOP Preferred Stock to the date fixed for distribution. These distributions will be subject to the rights of the holders of any Bank of America Preferred Stock ranking senior to or equally with ESOP Preferred Stock as to distributions upon liquidation, dissolution, or winding-up but before any amount will be paid or distributed among the holders of Bank of America common stock or any other shares of Bank of America stock ranking junior to ESOP Preferred Stock. If, upon Bank of America's voluntary or involuntary dissolution, iquidation, or winding-up, the amounts payable on ESOP Preferred Stock and any other Bank of America stock ranking equally therewith as to any distribution are not paid in full, the holder of ESOP Preferred Stock and the other Bank of America stock will share ratably in any distribution of assets in proportion to the full respective preferential amounts to which they are entitled. After payment of the full amount of the liquidating distribution to which it is entitled, a holder of ESOP Preferred Stock will not be entitled to any further distribution of Bank of America's assets. Any merger, consolidation, or purchase or sale of assets by Bank of America will not be deemed to be a dissolution, liquidation or winding-up of Bank of America's affairs.

Redemption. ESOP Preferred Stock is redeemable, in whole or in part, at Bank of America's option, at any time. The redemption price for the shares of ESOP Preferred Stock, which may be paid in cash or shares of Bank of America common stock, will be \$42.50 per share. The redemption price also must include all accrued and unpaid dividends to the date of redemption. If ESOP Preferred Stock is treated as Tier 1 capital for bank regulatory purposes, the approval of the Federal Reserve Board may be required to redeem ESOP Preferred Stock.

In addition, Bank of America is required to redeem shares of ESOP Preferred Stock at the option of the holder to the extent necessary either to provide for distributions required to be made under the Employee Stock Ownership Plan or to make payments of principal, interest, or premium due and payable on any indebtedness incurred by the holder for the benefit of the Employee Stock Ownership Plan.

Series B Preferred Stock

Preferential Rights. The shares of Series B Preferred Stock rank senior to ESOP Preferred Stock and common stock but rank junior to Series B Preferred Stock as to dividends and upon liquidation. Without the consent of holders of Series B Preferred Stock, Bank of America may issue preferred stock with superior or equal rights or preferences.

Dividends. Holders of shares of Series B Preferred Stock are entitled to receive, when and as declared by the Bank of America board of directors, cumulative cash dividends at an annual dividend rate per share of 7% of the stated value thereof. The stated value per share of the Series B Preferred Stock is \$100. Dividends are payable quarterly. Bank of America cannot declare or pay cash dividends on any shares of Bank of America common stock unless full cumulative dividends on Series B Preferred Stock have been paid or declared and funds sufficient for the payment have been set apart.

Voting Rights. Each share of Series B Preferred Stock has equal voting rights, share for share, with each share of Bank of America common stock.

Distributions. In the event of Bank of America's voluntary or involuntary dissolution, liquidation, or winding up, the holders of Series B Preferred Stock are entitled to receive, after payment of the full liquidation preference on shares of any class of Bank of America preferred stock ranking superior to Series B Preferred Stock, but before any distribution on shares of Bank of America common stock, liquidating dividends of \$100 per share plus accumulated dividends.

Redemption. Shares of Series B Preferred Stock are redeemable, in whole or in part, at the option of the holders, at the redemption price of \$100 per share plus accumulated dividends, provided that (1) full cumulative dividends have been paid, or declared and funds sufficient for payment set apart, upon any class or series of Bank of America preferred stock ranking superior to Series B Preferred Stock; and (2) Bank of America is not then in default or in arrears on any sinking fund or analogous fund or call for tenders obligation or agreement for the purchase or any class or series of Bank of America preferred stock ranking superior to Series B Preferred Stock.

6.75% Perpetual Preferred Stock

Bank of America will issue one share of Bank of America 6.75% Perpetual Preferred Stock for each share of FleetBoston Series VI 6.75% Perpetual Preferred Stock owned by a holder immediately prior to the effective time of the merger. The shares of Bank of America 6.75% Perpetual Preferred Stock will be validly issued, fully paid and nonassessable.

Dividends. The holders of Bank of America 6.75% Perpetual Preferred Stock are entitled to receive dividends at the rate of 6.75% per annum, payable quarterly, before Bank of America may declare or pay any dividend on Bank of America common stock or Bank of America junior preferred stock. The dividends on the Bank of America 6.75% Perpetual Preferred Stock are cumulative. Bank of America will increase the amount of dividends payable in respect of the Bank of America 6.75% Perpetual Preferred Stock if the Internal Revenue Code is amended to reduce the "dividends received deduction," i.e., the percentage of the dividend payable on the Bank of America 6.75% Perpetual Preferred Stock that may be deducted by corporate holders.

Redemption. Bank of America may redeem the Bank of America 6.75% Perpetual Preferred Stock, in whole or in part, at Bank of America's option, on and after April 15, 2006, at \$250 per share, plus accrued and unpaid dividends, if any. Bank of America also may redeem the Bank of America 6.75% Perpetual Preferred Stock prior to April 15, 2006, in whole, at Bank of America's option, if the Internal Revenue Code is amended to reduce the dividends received deduction.

So long as any shares of Bank of America 6.75% Perpetual Preferred Stock are outstanding, Bank of America may not redeem any shares of Bank of America common stock or any other class of Bank of America preferred stock ranking junior to or on a parity with Bank of America 6.75% Perpetual Preferred Stock unless Bank of America has paid full cumulative dividends on all outstanding shares of the Bank of America 6.75% Perpetual Preferred Stock for all past dividend payment periods. Further, if any dividends on the Bank of America 6.75% Perpetual Preferred Stock are in arrears, Bank of America may not redeem any shares of the Bank of America 6.75% Perpetual Preferred Stock unless Bank of America 6.75% Perpetual Preferred Stock, except pursuant to a purchase or exchange offer made on the same terms to holders of all outstanding shares of the Bank of America 6.75% Perpetual Preferred Stock.

Fixed/Adjustable Rate Cumulative Preferred Stock

Bank of America will issue one share of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock for each share of FleetBoston Series VII Fixed/Adjustable Rate Cumulative Preferred Stock owned by a holder as of immediately prior to the effective time of the merger. The shares of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock will be validly issued, fully paid and nonassessable.

Dividends. Through April 1, 2006, the holders of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock are entitled to receive dividends at the rate of 6.60% per annum computed on the basis of the issue price of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock of \$250 per share, payable quarterly, before Bank of America may declare or pay any dividend upon Bank of America common stock or junior Bank of America preferred stock. After that date, the dividend rate on Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock will be a rate per annum equal to 0.50% plus the highest of the Treasury Bill

Rate, the Ten Year Constant Maturity Rate and the Thirty Year Constant Maturity Rate, as each term is defined in the Certificate of Designation establishing the Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock. The applicable rate *per annum* for any dividend period beginning on or after April 1, 2006 will not be less than 7.0% nor greater than 13.0%.

The dividends on Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock are cumulative. Bank of America will increase the amount of dividends that will be payable in respect of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock if the Internal Revenue Code is amended to reduce the dividends received deduction

Redemption. Bank of America may redeem Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock, in whole or in part, at Bank of America's option, on and after April 1, 2006, at \$250 per share, plus accrued and unpaid dividends, if any. Bank of America may also redeem Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock prior to April 1, 2006, in whole, at Bank of America's option, if the Internal Revenue Code is amended to reduce the dividends received deduction.

So long as any shares of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock are outstanding, Bank of America may not redeem any shares of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock, Bank of America common stock or any other class of Bank of America preferred stock ranking junior to or on a parity with Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock, unless Bank of America has paid full cumulative dividends on all outstanding shares of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock for all past dividend payment periods. Further, if any dividends on Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock are in arrears, Bank of America may not redeem any shares of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock, unless Bank of America simultaneously redeems all outstanding shares of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock, except pursuant to a purchase or exchange offer made on the same terms to holders of all outstanding shares of Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock.

COMPARISON OF STOCKHOLDERS' RIGHTS

General

Bank of America and FleetBoston are incorporated in Delaware and Rhode Island, respectively. The rights of FleetBoston stockholders receiving Bank of America common stock in connection with the merger currently are governed by the Rhode Island Business Corporation Act, the FleetBoston restated articles of incorporation and the FleetBoston bylaws. Upon completion of the merger, FleetBoston stockholders automatically will have the right to receive shares of capital stock of Bank of America, and their rights will be governed by the Delaware General Corporation Law, the Bank of America amended and restated certificate of incorporation and the Bank of America bylaws. The following is a summary of the material differences between the rights of holders of Bank of America common stock and the rights of holders of FleetBoston common stock, but does not purport to be a complete description of those differences. These differences may be determined in full by reference to the Delaware General Corporation Law, the Rhode Island Business Corporation Act, the Bank of America amended and restated certificate of incorporation, the FleetBoston restated articles of incorporation, the Bank of America bylaws and the FleetBoston bylaws.

FleetBoston Bank of America

CAPITAL STOCK Authorized Capital

Preferred stock. 16 million shares of FleetBoston preferred stock, par value \$1.00 per share, of which, as of January 26, 2004, (i) 690,000 shares were authorized and 382,450 shares were issued and outstanding as FleetBoston Series VI 6.75% Perpetual Preferred Stock and (ii) 805,000 shares were authorized and 700,000 shares were issued and outstanding as Series VII Fixed/Adjustable Rate Cumulative Preferred Stock. As of January 26, 2004, no shares of FleetBoston preferred stock were reserved for issuance, except for 500,000 shares of cumulative participating junior preferred stock, par value \$1.00 per share, reserved for issuance pursuant to the FleetBoston rights agreement.

Common stock. 5 billion shares of Bank of America common stock, par value \$0.01 per share, of which there were approximately 1,448,613,889 shares issued and outstanding as of January 26, 2004. As of January 26, 2004, no shares of Bank of America common stock were reserved for issuance, except for (1) 333,544,919 shares of Bank of America common stock reserved for issuance under dividend reinvestment plans and upon exercise of stock options issued pursuant to Bank of America employee and director stock plans in effect as of January 26, 2004, and 3,277,040 shares of Bank of America common stock reserved for issuance pursuant to ESOP Preferred Stock, and (2) 297,374,945 shares of Bank of America common stock reserved for issuance pursuant to the Bank of America stock option agreement.

Preferred Stock. 100,000,000 shares of Bank of America preferred stock, par value \$0.01 per share, of which there were 1,269,600 shares issued and outstanding as of January 26, 2004 and no shares reserved for issuance as of January 26, 2004.

FleetBoston Bank of America

VOTING RIGHTS IN AN EXTRAORDINARY TRANSACTION

The Rhode Island Business Corporation Act generally requires approval of a merger, consolidation, dissolution or sale of all or substantially all of a corporation's assets by the affirmative vote of the holders of a majority of the outstanding shares of the corporation entitled to vote on the matter, unless otherwise provided by statute.

The Delaware General Corporation Law generally requires that any merger, consolidation or sale of substantially all the assets of a corporation be approved by a vote of a majority of all outstanding shares entitled to vote thereon. Although a Delaware corporation's certificate of incorporation may provide for a greater vote, the Bank of America amended and restated certificate of incorporation does not require a greater vote.

CHARTER AND BYLAWS AMENDMENT

The Rhode Island Business Corporation Act generally provides that an amendment to a corporation's articles of incorporation requires (1) adoption by the board of directors of a resolution submitting the proposed amendment to the stockholders, and (2) approval by the affirmative vote of the holders of a majority of the outstanding shares of each class entitled to vote as a class on the matter.

The FleetBoston restated articles of incorporation provide that any amendment, alteration, change or repeal of the provisions of the FleetBoston restated articles of incorporation relating to directors and business combinations requires the affirmative vote of 80% of the FleetBoston board of directors, a majority of the continuing directors and the affirmative vote of the holders of 80% or more of the outstanding shares of FleetBoston capital stock entitled to vote generally in the election of directors, voting separately as a class.

The FleetBoston bylaws may be altered, amended or repealed in whole or in part, and new bylaws may be adopted in whole or in part, only by the affirmative vote of 80% of the FleetBoston board of directors and a majority of the continuing directors or by an affirmative vote of the holders of at least 50% of the FleetBoston common stock represented at a meeting and entitled to vote on the matter.

The board of directors' authority to adopt, amend or repeal the corporation's bylaws does not divest or limit the power of stockholders to adopt, amend or repeal the corporation's bylaws. The Rhode Island Business Corporation Act provides that any amendment by the board of directors to the corporation's bylaws may be subsequently changed by the affirmative vote of holders of a majority of the shares entitled to vote on the matter.

Under the Delaware General Corporation Law, a corporation may amend its certificate of incorporation upon the submission of a proposed amendment to stockholders by the board of directors and the subsequent receipt of the affirmative vote of a majority of its outstanding voting shares and the affirmative vote of a majority of the outstanding shares of each class entitled to vote thereon as a class. The Bank of America amended and restated certificate of incorporation provides that Bank of America reserves the right at any time from time to time to amend or repeal any provision of the Bank of America amended and restated certificate of incorporation and that all rights conferred thereby are granted subject to such right of Bank of America, but does not contain any provisions altering the standards for amendment.

Under the Delaware General Corporation Law, bylaws may be adopted, amended or repealed by the stockholders entitled to vote, and by the board of directors if the corporation's certificate of incorporation confers the power to adopt, amend or repeal the corporation's bylaws upon the directors. The Bank of America amended and restated certificate of incorporation confers the power to adopt, amend or repeal the Bank of America bylaws upon both Bank of America stockholders and the Bank of America board of directors.

FleetBoston Bank of America

APPRAISAL RIGHTS

Under the Rhode Island Business Corporation Act, appraisal rights are available only in connection with:

- a merger or consolidation, unless the Rhode Island corporation is to be the surviving corporation and no vote of its stockholders is required to approve the merger,
- · acquisitions that require stockholder approval, and
- sales or exchanges of all or substantially all of the property and assets of a corporation in a transaction requiring stockholder approval.

However, unless otherwise provided in a corporation's articles of incorporation, no appraisal rights are available to holders of shares of any class of stock that, as of the date fixed to determine the stockholders entitled to receive notice of the proposed transaction, are (1) registered on a national securities exchange or included as national market securities in the National Association of Securities Dealers, Inc.'s automated quotation system, or (2) held of record by not less than 2,000 stockholders

There are no provisions in the FleetBoston restated articles of incorporation providing for appraisal rights.

Under the Rhode Island Business Corporation Act, holders of FleetBoston common stock and FleetBoston Series VI Stock have no appraisal or dissenters' rights in connection with the merger because FleetBoston common stock and FleetBoston Series VI Stock are listed on the NYSE.

Under the Rhode Island Business Corporation Act, however, holders of FleetBoston Series VII Stock have appraisal or dissenters' rights in connection with the merger. See the discussion in "The Merger — Appraisal Rights of Dissenting Stockholders" on page 53.

Under the Delaware General Corporation Law, a stockholder of a Delaware corporation such as Bank of America generally has the right to dissent from a merger or consolidation in which the corporation is participating or a sale of all or substantially all of the assets of the corporation, subject to specified procedural requirements. The Delaware General Corporation Law does not confer appraisal rights, however, if the corporation's stock is either (1) listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or (2) held of record by more than 2,000 holders.

Even if a corporation's stock meets the foregoing requirements, however, the Delaware General Corporation Law provides that appraisal rights generally will be permitted if stockholders of the corporation are required to accept for their stock in any merger, consolidation or similar transaction anything other than (1) shares of the corporation surviving or resulting from the transaction, or depository receipts representing shares of the surviving or resulting corporation, or those shares or depository receipts plus cash in lieu of fractional interests, (2) shares of any other corporation, or depository receipts representing shares of the other corporation, or those shares or depository receipts plus cash in lieu of fractional interests, unless those shares or depository receipts are listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or held of record by more than 2,000 holders, or (3) any combination of the foregoing.

Under the Delaware General Corporation Law, holders of Bank of America common stock have no appraisal or dissenters' rights in connection with the merger because Bank of America common stock is listed on the NYSE.

Under the Delaware General Corporation Law, however, holders of Bank of America Series B Preferred Stock and ESOP Preferred Stock have appraisal or dissenters' rights in connection with the merger. See the discussion in "The Merger — Appraisal Rights of Dissenting Stockholders" on page 53.

FleetBoston Bank of America

SPECIAL MEETINGS

The Rhode Island Business Corporation Act provides that a special meeting of stockholders may be called by a corporation's board of directors or by those persons as may be authorized by a corporation's articles of incorporation or bylaws. The FleetBoston bylaws permit special meetings of stockholders to be called by the FleetBoston board of directors pursuant to a resolution adopted by a majority of the FleetBoston board of directors, the Chairman of the FleetBoston board of directors, or the President of FleetBoston.

Under the Delaware General Corporation Law, special stockholder meetings of a corporation may be called by the corporation's board of directors, or by any person or persons authorized to do so by the corporation's certificate of incorporation or bylaws. The Bank of America bylaws provide that a special meeting of stockholders may be called for any purpose by the Bank of America board of directors, or by the Chairman of the Bank of America board of directors, the Chief Executive Officer or President of Bank of America.

BOARD OF DIRECTORS Number of Directors

Under the Rhode Island Business Corporation Act, a corporation must have a board of directors consisting of at least one director. The FleetBoston restated articles of incorporation provide that the FleetBoston board of directors is to consist of 13 members (exclusive of directors to be elected by holders of any one or more series or classes of FleetBoston preferred stock voting separately as a class or classes), unless otherwise determined from time to time by resolution adopted by an affirmative vote of at least 80% of the FleetBoston board of directors and a majority of the continuing directors. Vacancies on the FleetBoston board of directors, regardless of the reason, may only be filled by the FleetBoston board of directors, acting by a vote of 80% of the directors then in office. Currently, the number of FleetBoston directors is 18.

Under Delaware General Corporation Law, the board of directors of a corporation must consist of one or more members, each of whom must be a natural person. The Bank of America bylaws provide that the Bank of America board of directors is to consist of not less than five nor more than 30 members, which number may be changed from time to time within such range by the Bank of America board of directors. Vacancies on the Bank of America board of directors may be filled by the affirmative vote of a majority of the directors then in office. Currently, the number of Bank of America directors is 17.

Stockholder Nominations

Holders of FleetBoston common stock may nominate individuals for election to the FleetBoston board of directors. The procedure for nominations is set forth in the FleetBoston bylaws. The FleetBoston bylaws specify that nominations of individuals for election as directors may be made at a meeting of FleetBoston stockholders by or at the direction of the FleetBoston board of directors, or by any holder of FleetBoston stock entitled to vote on the election of directors that complies with the requisite notice procedure. The notice procedure requires that a stockholder's nomination of an individual for election as a director must be made in writing and received by the Secretary of FleetBoston not less than 30 days prior to the date of

The Delaware General Corporation Law does not contain any specific provisions regarding notice of stockholders' proposals or nominations for directors. The Bank of America bylaws specify that nominations of individuals for election as directors may be made pursuant to Bank of America's notice of meeting, by or at the direction of the Bank of America board of directors or by any holder of Bank of America stock entitled to vote on the election of directors who complies with the requisite notice procedure. The notice procedure requires that a stockholder's nomination of an individual for election as a director must be made in writing and received by the Secretary of Bank of America not later than the

FleetBoston

the meeting of FleetBoston stockholders. If, however, FleetBoston gives FleetBoston stockholders fewer than 40 days' notice or prior public disclosure of the date of the meeting of FleetBoston stockholders, FleetBoston must receive the stockholder's nomination notice not later than the close of business on the seventh day following the first to occur of the publication or mailing of the notice of the meeting date.

Bank of America

close of business on the 75th day nor earlier than the close of business on the 120h day prior to the first anniversary of the date Bank of America commenced mailing its proxy materials for the preceding year's annual meeting. In the event of a special meeting of Bank of America stockholders at which directors are to be elected, any Bank of America stockholder entitled to vote may nominate an individual for election as director if the stockholder's notice is received by the Secretary of Bank of America not later than the close of business on the 15th day following the day on which notice of the meeting is first mailed to Bank of America stockholders.

Classification

The Rhode Island Business Corporation Act permits classification of a Rhode Island corporation's board of directors if the corporation's articles of incorporation so provide. The FleetBoston restated articles of incorporation and the FleetBoston bylaws provide for classification of the FleetBoston board of directors into three classes as nearly equal in number as possible, with each class being elected annually for a three-year term.

The Delaware General Corporation Law permits classification of a Delaware corporation's board of directors if the corporation's certificate of incorporation so provides. The Bank of America amended and restated certificate of incorporation does not provide for a classified board of directors.

Removal

Under the Rhode Island Business Corporation Act, if a corporation's articles of incorporation or bylaws so provide, stockholders may remove a director without cause. However, in the case of a corporation permitting cumulative voting for the election of directors, directors may be removed without cause only if the number of shares voted against removal would not be sufficient to elect the director if voted cumulatively.

Directors of FleetBoston may only be removed for cause and only by a vote of:

- a majority of the continuing directors and a majority of the FleetBoston board of directors as it exists at that time, or
- the holders of 80% or more of the outstanding shares of FleetBoston stock entitled to vote on the election of directors, voting separately as a class at a meeting called for that purpose.

The FleetBoston restated articles of incorporation currently do not provide for cumulative voting

The Delaware General Corporation Law provides that, in the absence of cumulative voting or a classified board, unless a corporation's certificate of incorporation provides that any director or the entire board of directors may be removed only for cause, any director or the entire board of directors may be removed, with or without cause, by the holders of a majority of the shares then entitled to vote in an election of directors. Since the Bank of America amended and restated certificate of incorporation does not include this limitation, Bank of America stockholders may remove one or all of the Bank of America directors with or without cause.

FleetBoston Bank of America

Special Meetings of the Board

Special meetings of the board of directors may be called by any five of the directors then in office, or by the Chairman of the Board or the President.

Special meetings of the board of directors may be called by the Chairman of the Board, the Chief Executive Officer, the President or upon the call of any three directors.

STOCKHOLDER RIGHTS PLAN

On August 16, 2000, FleetBoston adopted a stockholder protection rights agreement, pursuant to which each issued share of FleetBoston common stock has attached to it one right to purchase, under certain conditions, a fraction of a share of FleetBoston junior preferred stock. On October 27, 2003, FleetBoston and the rights agent under the FleetBoston stockholder protection rights agreement amended the rights agreement to prevent the merger agreement, the stock option agreements, and the merger with Bank of America from triggering the provisions of the FleetBoston stockholder protection rights agreement.

Bank of America currently does not have a stockholder rights plan in effect.

STATE ANTI-TAKEOVER STATUTES

Pursuant to the Rhode Island Business Combination Act, a corporation may not engage in any business combination with an "interested shareholder" (generally defined under the Rhode Island Business Combination Act as the beneficial owner of 10% or more of the corporation's outstanding voting stock or an affiliate of the corporation who, within five years prior to the date in question, was the beneficial owner of 10% or more of the corporation's outstanding voting stock) for a period of five years following the date the shareholder became an interested shareholder, unless:

- the corporation's board of directors approved the business combination or transaction prior to the date the shareholder became an interested shareholder:
- no earlier than five years after the interested shareholder's stock
 acquisition date, holders of two-thirds of the outstanding voting stock,
 excluding any stock owned by the interested shareholder or any affiliate or
 associate of the interested shareholder, have approved the business
 combination at a meeting called for that purpose; or
- the business combination meets each of the following conditions: (1) the nature, form and

Section 203 of the Delaware General Corporation Law generally provides that a Delaware corporation such as Bank of America that has not "opted out" of coverage by this section in the prescribed manner may not engage in any "business combination" with an "interested stockholder" for a period of three years following the date that the stockholder became an "interested stockholder" unless:

- prior to that time the corporation's board of directors approved either the business combination or the transaction that resulted in the stockholder becoming an "interested stockholder";
- upon completion of the transaction that resulted in the stockholder becoming an "interested stockholder," the "interested stockholder" owned at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned by individuals who are directors and also officers and shares owned by employee stock

FleetBoston

adequacy of the consideration to be received by the corporation's shareholders in the business combination transaction satisfies certain specific enumerated criteria, (2) the holders of all the outstanding shares of stock of the corporation not beneficially owned by the interested shareholder are entitled to receive the specified consideration in the business combination transaction and (3) the interested shareholder may not acquire additional shares of the corporation's voting stock, except in certain specifically identified transactions.

The restrictions prescribed by the Rhode Island Business Combination Act will not be applicable to any business combination:

- involving a corporation that does not have a class of voting stock registered under the Exchange Act, unless the corporation's articles of incorporation provide otherwise;
- involving a corporation that did not have a class of voting stock registered
 under the Exchange Act at the time the corporation's articles of
 incorporation were amended to provide that the corporation is to be
 subject to these statutory restriction provisions and the interested
 shareholder's stock acquisition date is prior to the effective date of the
 articles of incorporation amendment;
- involving a corporation whose original articles of incorporation contains a provision expressly electing not to be subject to these statutory restrictions or that adopted an amendment expressly electing not to be subject to these statutory restrictions either to the corporation's bylaws prior to March 31, 1991, or to the corporation's articles of incorporation if the articles of incorporation amendment is approved by the affirmative vote of holders, other than the interested shareholders and their affiliates and associates, of two-thirds of the outstanding voting stock, excluding the voting stock of the interested shareholder. However, this amendment to a corporation's articles of incorporation will not be effective until 12 months after the vote of the shareholders and will not apply to any business combination of

Bank of America

ownership plans in which employee participants do not have the right to determine confidentially whether the shares held subject to the stock ownership plan will be tendered in a tender offer or exchange offer; or

 at or subsequent to that time, the business combination is approved by the corporation's board of directors and authorized at an annual or special meeting of stockholders by the affirmative vote of at least 66²/3% of the outstanding voting stock that is not owned by the "interested stockholder."

The three-year prohibition on business combinations with an "interested stockholder" does not apply under certain circumstances, including business combinations with a corporation that does not have a class of voting stock that is:

- · listed on a national security exchange;
- · authorized for quotation on the Nasdaq Stock Market; or
- held of record by more than 2,000 stockholders,

unless, in each case, this result was directly or indirectly caused by the "interested stockholder" or from a transaction in which a person became an "interested stockholder."

An "interested stockholder" generally means any person that:

- is the owner of 15% or more of the outstanding voting stock of the corporation; or
- is an affiliate or associate of the corporation and was the owner of 15% or more of the outstanding voting stock of the corporation at any time within the three-year period immediately prior to the date on which it is sought to be determined whether such person is an "interested stockholder," and the affiliates and associates of such a person.

The term "business combination" is defined to include a wide variety of transactions, including mergers, consolidations, sales or other dispositions of

FleetBoston

- the corporation with an interested shareholder whose stock acquisition date is on or prior to the effective date of the amendment; or
- involving a corporation with an interested shareholder that became an
 interested shareholder inadvertently if the interested shareholder divests
 itself of that number of shares so that it is no longer the beneficial owner
 of 10% of the outstanding voting stock and, but for the inadvertent
 ownership, was not an interested shareholder within the five-year period
 preceding the announcement of the business combination.

The FleetBoston restated articles of incorporation and the FleetBoston bylaws do not contain any provisions opting out of the restrictions prescribed by the statute.

Bank of America

10% or more of a corporation's assets and various other transactions which may benefit an "interested stockholder."

The Bank of America amended and restated certificate of incorporation and the Bank of America bylaws do not contain any provisions opting out of the restrictions prescribed by Section 203 of the Delaware General Corporation Law. The merger does not constitute a prohibited business combination under this statute.

Stockholder Approvals Board Policies

The FleetBoston board of directors has not adopted any specific policies requiring stockholder approval prior to actions by the board.

The Bank of America board of directors has adopted a policy of requiring stockholder approval for severance agreements with senior executives that provide severance benefits in amounts exceeding two times the sum of any senior executive's base salary and bonus.

In addition, the Bank of America board of directors has adopted a policy of requiring stockholder approval prior to the adoption of any stockholder rights plan.

Preferred Stock

Upon completion of the merger, pursuant to the merger agreement, Bank of America will exchange the outstanding shares of FleetBoston Series VI 6.75% Perpetual Preferred Stock and Series VII Fixed/Adjustable Rate Cumulative Preferred Stock for Bank of America 6.75% Perpetual Preferred Stock and Fixed/Adjustable Rate Cumulative Preferred Stock, respectively. The Bank of America preferred stock exchanged for FleetBoston preferred stock will have substantially identical preferences and rights with respect to Bank of America as FleetBoston preferred stock had with respect to FleetBoston.

Selected Provisions in the Restated Articles of Incorporation of FleetBoston

In addition to the above differences, the FleetBoston restated articles of incorporation also contain the following provisions that are not found in the Bank of America amended and restated certificate of incorporation or the Bank of America bylaws.

Business Combinations with Related Persons

The FleetBoston restated articles of incorporation provide that neither FleetBoston nor any of its subsidiaries may engage in business transactions known as "business combinations" with persons known as "related persons" unless the transaction:

- was approved by an 80% vote of the FleetBoston board of directors prior to the time the related person became a related person; and
- is approved by a vote of 80% of the continuing directors and a majority of the entire FleetBoston board of directors and the transaction complies with certain conditions related to price and procedure; or
- if there is not full compliance with the preceding two bullet points, the transaction is approved by a vote of 80% of the outstanding shares of FleetBoston capital stock
 entitled to vote generally in the election of directors, voting as a single class.

A "business combination" includes:

- · any merger or consolidation of FleetBoston or any of its subsidiaries with a related person or any of its affiliates or associates;
- any sale, exchange, lease, transfer or other disposition to or with a related person of all, substantially all or any substantial part (defined as assets having a value of more than 5% of the total consolidated assets of FleetBoston) of the assets of FleetBoston or any of its subsidiaries;
- any purchase, exchange, lease or other acquisition by FleetBoston or any of its subsidiaries of all or any substantial part of the assets or business of a related person or any of its affiliates or associates;
- any reclassification of securities, recapitalization or other transaction that has the effect, directly or indirectly, of increasing the proportionate amount of voting shares
 of FleetBoston or any subsidiary that are beneficially owned by a related person; and
- the acquisition by a related person of beneficial ownership of voting securities, securities convertible into voting securities or any rights, warrants or options to acquire voting securities of a subsidiary of FleetBoston.

A "related person" includes any person that is the beneficial owner of 10% or more of FleetBoston voting stock prior to the completion of a business combination, or any person that is an affiliate of FleetBoston and was the beneficial owner of 10% or more of FleetBoston voting stock at any time within the five years preceding the date on which the FleetBoston board of directors authorizes a binding agreement providing for a business combination.

"Continuing directors" are those individuals who were members of the FleetBoston board of directors prior to the time a related person became the beneficial owner of 10% or more of FleetBoston voting stock or those individuals designated as continuing directors (prior to their initial election as directors) by a majority of the then-continuing directors.

To amend these provisions, the affirmative vote of 80% of the FleetBoston board of directors, a majority vote of the continuing directors and the affirmative vote of 80% of FleetBoston outstanding capital stock entitled to vote on the matter are required. If the amendment is recommended to FleetBoston stockholders by a majority of the FleetBoston board of directors and not less than 80% of the continuing directors, only the vote provided under the Rhode Island Business Corporation Act is required.

ADOPTION OF BANK OF AMERICA'S AMENDED STOCK PLAN

Bank of America currently maintains the Bank of America Corporation 2003 Key Associate Stock Plan, which we refer to as the Stock Plan. Under this plan, Bank of America has reserved a number of shares of Bank of America common stock for issuance to key associates in the form of stock options, stock appreciation rights, restricted stock shares and restricted stock units. The Stock Plan has a five-year term that is scheduled to end December 31, 2007.

Bank of America believes that changes to the Stock Plan are needed primarily for two reasons:

- The merger with FleetBoston will significantly increase the number of key associates eligible for awards under the Stock Plan, and as a result the Stock Plan will have an insufficient number of shares available for future awards. In addition, the FleetBoston management stock plans will be terminated effective as of the date of the merger and no shares will be carried over from the FleetBoston plans to the Stock Plan.
- Design changes are needed to reflect developments since the Stock Plan was originally adopted, including the expensing of stock options (which Bank of America began effective January 1, 2003) and a change in compensation philosophy to place greater weight on non-option awards.

To meet these demands, the Bank of America board has adopted, subject to stockholder approval and the completion of the merger, an amendment and restatement of the Stock Plan proposed to become effective at the completion of the merger. This Amended Stock Plan, which we will refer to formally as the Bank of America Corporation 2003 Key Associate Stock Plan, as amended and restated, is similar in design to the current Stock Plan, but also has some important differences described below. The Amended Stock Plan extends the original Stock Plan's five-year term by one year, from December 31, 2007 to December 31, 2008.

The Amended Stock Plan is intended to serve a critical function in Bank of America's overall compensation program. Bank of America believes that the compensation of its key associates should be significantly linked to Bank of America's business performance in order to enhance Bank of America's long-term success and value. The Amended Stock Plan will serve this compensation philosophy by providing a source of stock-based awards for key associates. Bank of America believes that stock-based awards best align the interests of the key associates with those of Bank of America's stockholders. Awards generally include vesting and other terms and conditions that encourage a long-term stock ownership by key associates and provide incentive for key associates to remain employed with the company. The Amended Stock Plan will also provide Bank of America with a means to better attract, recruit and retain key associates of outstanding ability who will further enhance the long-term success and value of Bank of America through their services.

The following is a summary of the material terms of the Amended Stock Plan, with significant differences from the current Stock Plan identified where applicable. It is qualified in its entirety by reference to the terms of the plan, a copy of which is attached to this joint proxy statement/prospectus as Appendix H. The Amended Stock Plan will become effective at the completion of the merger only if it is first approved by Bank of America stockholders and only if the merger is in fact completed.

Number of Shares

Under the Stock Plan as originally adopted, there were reserved for issuance an aggregate of 100 million shares of Bank of America common stock, plus (a) any remaining shares that were available for awards as of December 31, 2002 under the Bank of America Corporation Key Employee Stock Plan (the predecessor to the Stock Plan) (16,855,936 shares) and (b) any shares covered by awards granted under the Bank of America Corporation Key Employee Stock Plan before January 1, 2003 that again become available because of cancellation or forfeiture of an award. As of December 31, 2003, there is estimated to be remaining under the Stock Plan approximately 86.2 million shares of Bank of America common stock available for awards.

Under the Amended Stock Plan, there would be added an additional 51 million shares of Bank of America common stock available for awards.

As under the Stock Plan, shares covered by awards under the Amended Stock Plan will again be available for awards if and to the extent (a) the award is canceled or forfeited, (b) the exercise price or any tax withholding obligation in connection with the award is satisfied through delivery of shares of Bank of America common stock or (c) the award is settled in cash.

Bank of America anticipates that, subject to applicable law and regulation, it will continue to maintain a general policy of repurchasing a number of shares of Bank of America common stock at least equal to the number of shares of Bank of America common stock issued under the Amended Stock Plan and under Bank of America's other stock option plans.

Types of Awards

The Amended Stock Plan, like the Stock Plan, permits awards of stock options, stock appreciation rights, restricted stock shares and restricted stock units, all of which are described in more detail below. The Amended Stock Plan also includes some important limits on the types of awards that can be made:

- From and after January 1, 2004, no more than 20 million shares of Bank of America common stock may be awarded in the form of restricted stock shares or restricted stock units, except for awards of restricted stock shares or restricted stock units that (a) are awarded in lieu of a portion of annual incentive compensation or (b) become vested and earned solely on the basis of achievement of performance goals. The Stock Plan also contains a similar limit, but the number of shares permitted to be granted in respect of restricted stock shares or restricted stock units is being increased from 5 million to 20 million to reflect the larger size of the company after the merger. This increase is also necessary to allow Bank of America to place a greater focus on non-option awards.
- From and after January 1, 2004, of the total shares reserved, 18 million are set aside to be available exclusively for awards of restricted stock shares or restricted stock units in lieu of a portion of annual incentive compensation under Bank of America's Equity Incentive Plan, Executive Incentive Compensation Plan or similar annual incentive compensation programs.
- Under the Stock Plan, stock appreciation rights may be awarded only to non-U.S.-based key associates who, as a result of local laws, cannot otherwise receive stock options. However, Bank of America's decision to expense stock options has now made stock appreciation rights a viable form of award for U.S.-based associates as well. Accordingly, the Amended Stock Plan allows stock appreciation rights to be awarded to any eligible key associate.

Administration

As under the Stock Plan, the Amended Stock Plan is designed to be administered by the compensation committee of the Bank of America board of directors. (For purposes of awards to executive officers who are "officers" under Section 16 of the Exchange Act or "covered employees" under Section 162(m) of the Internal Revenue Code, the Amended Stock Plan may be administered by any subcommittee of the compensation committee comprised entirely of (i) "non-employee directors" within the meaning of Rule 16b-3 under the Exchange Act and (ii) "outside directors" within the meaning of Section 162(m) of the Internal Revenue Code.) To the extent permitted by law, the compensation committee may designate an individual or committee (which need not consist of directors) to act as the appropriate committee under the Amended Stock Plan for awards to key associates who are not Section 16 "officers" or Section 162(m) "covered employees." Under the Amended Stock Plan, the compensation committee has authority with respect to the following:

• the selection of the key associates to receive awards from time to time,

- the granting of awards in amounts as it determines,
- the imposition of limitations, restrictions and conditions upon awards as it deems appropriate,
- the establishment of performance targets and allocation formulas for awards of restricted stock shares or restricted stock units intended to be "qualified performance-based compensation" under Internal Revenue Code Section 162(m),
- the certification of the attainment of performance goals, if applicable, as required by Internal Revenue Code Section 162(m),
- the interpretation of the Amended Stock Plan and the adoption, amendment and rescission of administrative guidelines and other rules and regulations relating to the Amended Stock Plan,
- · the correction of any defect or omission or reconciliation of any inconsistency in the Amended Stock Plan or any award granted under the Amended Stock Plan, and
- · the making of all other determinations and taking of all other actions necessary or advisable for the implementation and administration of the Amended Stock Plan.

The Committee also has the authority to accelerate the vesting and/or waive any restrictions on any outstanding awards. No awards may be made under the Amended Stock Plan after December 31, 2008.

Eligibility

Only "key associates" of Bank of America and its subsidiaries may participate in the Amended Stock Plan. Key associates are those associates of Bank of America and its subsidiaries who occupy managerial or other important positions and who have made, or are expected to make, important contributions to the business of Bank of America, as determined by the compensation committee, including persons employed outside the United States. Approximately 45,000 associates are expected to be eligible to participate. As mentioned above, the compensation committee in its discretion selects which key associates would in fact receive any awards from time to time.

Awards of Stock Options and Stock Appreciation Rights

The Amended Stock Plan provides for the grant of options to purchase shares of Bank of America common stock at option prices which are not less than the fair market value of a share of Bank of America common stock at the close of business on the date of grant. (The fair market value of a share of Bank of America common stock as of [] was \$[].) The Amended Stock Plan also provides for the grant of stock appreciation rights to key associates. Stock appreciation rights entitle the holder upon exercise to receive either cash or shares of Bank of America common stock or a combination of the two, as the compensation committee in its discretion may determine, with a value equal to the difference between (i) the fair market value on the exercise date of the shares with respect to which a stock appreciation right is exercised and (ii) the fair market value of the shares on the date of grant.

Awards of options under the Amended Stock Plan, which may be either incentive stock options (which qualify for special tax treatment) or nonqualified stock options, are determined by the compensation committee. No more than an aggregate of 20 million shares may be awarded as incentive stock options under the Amended Plan. The terms and conditions of each option and stock appreciation right are to be determined by the compensation committee (or its designees) at the time of grant.

Options and stock appreciation rights granted under the Amended Stock Plan will expire not more than 10 years from the date of grant, and the award agreements entered into with each participant will specify the extent to which options and stock appreciation rights may be exercised during their respective terms, including in the event of the participant's death, disability or termination of employment.

Awards of Restricted Stock Shares and Restricted Stock Units

Under the Amended Stock Plan, the compensation committee may award key associates restricted shares of Bank of America common stock or restricted stock units which represent the right to receive shares of Bank of America common stock (or cash equal to the fair market value of those shares). The award agreement with the participant will contain the terms of the award, including any applicable conditions, which may include the continued service of the participant with Bank of America, the attainment of specified performance goals or any other conditions deemed appropriate by the compensation committee.

Bank of America's practice has been to link the award of restricted stock shares and units to an annual review of key associate performance. Performance is measured against a variety of goals set each year intended to address three areas of achievement: (1) customer satisfaction, (2) associate satisfaction and (3) stockholder satisfaction. A percentage of the annual incentive compensation awarded based on this determination of performance is then delivered in the form of restricted stock shares or restricted stock units under Bank of America's Equity Incentive Program, Executive Incentive Compensation Plan or other annual incentive plans.

Restricted shares will be held in the custody of Bank of America until the applicable restrictions have been satisfied. The participant cannot sell, transfer, pledge, assign or otherwise alienate or hypothecate restricted stock shares until the applicable restrictions are satisfied. Once the restrictions are satisfied, the shares will be delivered to the participant's account, free and clear of any restrictions. During the period of restriction, the participant may exercise full voting rights with respect to the restricted stock shares. The participant will also be credited with dividends with respect to the shares. Dividends may be payable currently or subject to additional restrictions as determined by the compensation committee and reflected in the award agreement.

The award agreement for any restricted stock units will specify whether units that become earned and payable will be settled in shares of Bank of America common stock (with one share of Bank of America common stock to be delivered for each earned and payable restricted stock unit), in cash (equal to the aggregate fair market value of the restricted stock units that are earned and payable), or in a combination of shares and cash. Shares of Bank of America common stock used to pay earned restricted stock units may have additional restrictions, as determined by the compensation committee. Unpaid restricted stock units may have dividend equivalent rights, as determined by the compensation committee and evidenced in the award agreement. Unpaid restricted stock units have no voting rights.

The Amended Stock Plan, like the Stock Plan, places limits on the use of restricted stock shares and restricted stock units:

- From and after January 1, 2004, no more than 20 million shares of Bank of America common stock may be awarded under the Amended Stock Plan as restricted stock shares or restricted stock units, except for awards of restricted stock shares or restricted stock units that (a) are awarded in lieu of a portion of annual incentive compensation or (b) become vested and earned solely on the basis of achievement of performance goals.
- From and after January 1, 2004, of the total shares reserved, 18 million are set aside to be available exclusively for awards of restricted stock shares or restricted stock units in lieu of a portion of annual incentive compensation under Bank of America's Equity Incentive Plan, Executive Incentive Compensation Plan or other annual incentive programs.
- For restricted stock shares or restricted stock units intended to vest solely on the basis of the passage of time, the awards will not vest more quickly than ratably over a three-year period beginning on the first anniversary of the award. Awards may vest more quickly in the event of (a) death, disability or retirement, (b) job loss due to workforce reduction, job elimination or divestiture or (c) under the "change in control" provisions of the plan. Awards made in lieu of annual incentive compensation will not be subject to a time-based vesting requirement (because these awards are already based on an annual review of performance). Also, awards necessary in the recruitment of new key associates or for the retention of key associates acquired in a business combination will not be subject to a time-based vesting requirement.

Stock Plan Benefits Table

Because awards under the Amended Stock Plan are discretionary, awards are generally not determinable at this time. However, if the merger is completed, awards of restricted stock units will be made to some of the executive officers of FleetBoston pursuant to their new employment agreements with Bank of America, and if the Amended Stock Plan is approved, these awards will be granted pursuant to the Amended Stock Plan (as is the case with all commitments to grant equity awards that become effective after the Amended Stock Plan is adopted). See "The Merger — FleetBoston's Directors and Officers Have Financial Interests in the Merger" on page 59. The following table presents information on Bank of America equity compensation plans at December 31, 2003:

	(a) Number of Securities To Be Issued Upon Exercise of Outstanding Options, Warrants and Rights (1)	(b) Weighted-Average Exercise Price of Outstanding Options (1)		(c) Number of Shares Remaining for Future Issuance Under Equity Compensation Plans
Plans approved by stockholders	125,192,323	\$	62.66	86,728,537
Plans not approved by stockholders	39,982,774	\$	57.29	
Total	165,175,097	\$	61.32	86,728,537

Column (a) includes 5,009,407 unvested restricted stock units. The weighted-average exercise price in column (b) does not take into account these restricted stock unit awards.

The following additional information is provided regarding Bank of America and FleetBoston equity awards:

- (a) As of December 31, 2003, there were 3,075,866 unvested shares of Bank of America restricted stock outstanding.
- (b) As of December 31, 2003, (1) the number of shares to be issued upon exercise of outstanding FleetBoston options and upon payment of outstanding FleetBoston stock units was 87,682,158 (equivalent to 48,689,902 shares of Bank of America common stock after application of the exchange ratio), (2) the weighted average exercise price of outstanding FleetBoston options was \$31.88 (equivalent to \$57.41 after application of the exchange ratio), (3) the number of shares available for future issuance under FleetBoston's equity compensation plans, excluding shares reflected in (1) above, was 28,661,550 (equivalent to 15,915,758 shares of Bank of America common stock after application of the exchange ratio) and (4) there were 4,352,818 unvested shares of FleetBoston restricted stock outstanding (equivalent to 2,417,120 shares of Bank of America common stock after application of the exchange ratio).
- (c) Upon completion of the merger, all then outstanding FleetBoston options, stock units and restricted shares will be converted into, and become, rights to acquire or receive shares of Bank of America common stock, but no additional shares from FleetBoston's equity compensation plans will be available for future grant under the Amended Stock Plan or any other equity compensation plans of Bank of America. FleetBoston currently anticipates making equity awards in February 2004 as part of its normal annual award cycle, which awards are currently not expected to exceed approximately 5.25 million shares of FleetBoston common stock (equivalent to about 2.92 million shares of Bank of America common stock after application of the exchange ratio), and may also make awards to newly hired employees in the normal course of business prior to the closing of the merger, which awards are currently not expected to exceed approximately 250,000 shares of FleetBoston common stock (equivalent to 138,825 shares of Bank of America common stock after application of the exchange ratio).

The following is a brief description of the material terms of Bank of America's equity compensation plans that were not approved by stockholders:

2002 Associates Stock Option Plan

The Bank of America Corporation 2002 Associates Stock Option Plan covered all Bank of America employees below a specified executive grade level. Under the plan, eligible employees received a one-time award of a predetermined number of options entitling them to purchase shares of Bank of America common stock. All options are non-qualified and have an exercise price equal to the fair market value on the date of grant. Approximately 54 million options were granted on February 1, 2002 at \$61.36, the closing price for that day. The options vest as follows: 50 percent of the options become exercisable after Bank of America common stock closes at or above \$76.36 per share for ten consecutive trading days (this vesting trigger was achieved during 2003); the remaining 50 percent of the options become exercisable after the common stock closes at or above \$91.36 for ten consecutive trading days. Regardless of the stock price, all options will be fully exercisable beginning February 1, 2006. In addition, the options continue to be exercisable following termination of employment under certain circumstances. At December 31, 2003, approximately 28.0 million options were outstanding under this plan. The options expire on January 31, 2007.

Take Ownership!

The Bank of America Global Associate Stock Option Program (Take Ownership!) covered all Bank of America employees below a specified executive grade level. The options were granted on the first business day of 1999, 2000 and 2001. Under the plan, eligible employees received an award of a predetermined number of stock options entitling them to purchase shares of Bank of America common stock at the fair market value on the grant date. All options are non-qualified. At January 2, 2004, all options issued under this plan were fully vested. These options expire five years after the grant date. In addition, the options continue to be exercisable following termination of employment under certain circumstances. At December 31, 2003, approximately 11.9 million options were outstanding under this plan. No further awards may be granted under this plan.

Barnett Employee Stock Option Plan

Under the Barnett Employee Stock Option Plan, ten-year options to purchase a predetermined number of shares of Bank of America common stock were granted to all employees of Barnett Banks, Inc. (a predecessor to Bank of America) below a specified executive grade level in 1997. All options are non-qualified and have an exercise price equal to the fair market value on the grant date. At December 31, 2003, all options were fully vested. In addition, the options continue to be exercisable following termination of employment under certain circumstances. At December 31, 2003, approximately 100,000 options were outstanding under this plan.

Internal Revenue Code Section 162(m)

Section 162(m) of the Internal Revenue Code generally places a \$1 million annual limit on a company's tax deduction for compensation paid to the chief executive officer or any of the four most highly compensated officers other than the chief executive officer, referred to as the "covered individuals." This limitation does not apply, however, to "qualified performance-based compensation." Because stock options or stock appreciation rights granted under the Amended Stock Plan must have an exercise price equal at least to fair market value at the date of grant, are granted to covered individuals by a compensation committee consisting of at least two outside directors, and the Amended Stock Plan limits the number of shares that may be the subject of awards granted to any key associate during any calendar year, compensation from the exercise of stock options or stock appreciation rights should be treated as "qualified performance-based compensation" for purposes of Section 162(m) of the Internal Revenue Code.

In addition, the Amended Stock Plan authorizes the compensation committee to make awards of restricted stock shares or restricted stock units that are conditioned on the satisfaction of performance criteria. For those

awards intended to meet the requirements of the "qualified performance-based compensation" exception to Section 162(m), the compensation committee must establish the applicable performance conditions prior to or within a specified period after the start of the applicable performance period. The compensation committee may select from the following performance measures for this purpose:

- total revenue (defined as the sum of net interest income on a taxable-equivalent basis and noninterest income),
- net income
- · shareholder value added (which equals the cash basis operating earnings for a year less a charge for the use of capital for the year),
- return on average common shareholders' equity,
- return on average assets,
- earnings per common share (using either diluted earnings or not),
- operating earnings per common share (using either diluted earnings or not),
- · total stockholder return,
- · customer satisfaction (determined based on objective criteria approved by the compensation committee),
- expense management,
- · operating margin,
- operating leverage, or
- · cash flow.

The performance conditions will be stated in the form of an objective, nondiscretionary formula, and the compensation committee will certify in writing the attainment of those performance conditions prior to any payment or distributions with respect to awards. The compensation committee in its discretion may adjust downward any award.

As under the Stock Plan, in no event may an individual receive awards under the Amended Stock Plan for a given calendar year covering in excess of two million shares of Bank of America common stock.

So that options and stock appreciation rights granted under the Amended Stock Plan qualify for the exclusion for performance-based compensation, and to permit the compensation committee to grant other awards under the Amended Stock Plan that are intended to qualify for the exclusion, the Amended Stock Plan is being submitted to Bank of America's stockholders for approval. A vote in favor of approving the Amended Stock Plan will be a vote approving all the material terms and conditions of the plan for purposes of the performance-based exemption under Section 162(m), including the performance measures, eligibility requirements and limits on various stock awards, in each case as described above.

Withholding for Payment of Taxes

As with the Stock Plan, the Amended Stock Plan provides for the withholding and payment by a participant of any payroll or withholding taxes required by applicable law. The Amended Stock Plan permits a participant to satisfy this requirement, with the approval of the compensation committee and subject to the terms of the Amended Stock Plan, by having Bank of America withhold from the participant a number of shares of Bank of America common stock otherwise issuable under the award having a fair market value equal to the amount of applicable payroll and withholding taxes.

Changes in Capitalization and Similar Changes

As with the Stock Plan, in the event of any change in the number of outstanding shares of Bank of America common stock by reason of any stock dividend, split, spin-off, recapitalization, merger, consolidation, combination, exchange of shares or otherwise, the aggregate number of shares of Bank of America common stock with respect to which awards may be made under the Amended Stock Plan, the annual limit on individual awards, the limits on incentive stock options, restricted stock and restricted stock units and the terms, types of shares and number of shares of any outstanding awards under the Amended Stock Plan may be equitably adjusted by the compensation committee in its discretion to preserve the benefit of the award for Bank of America and the participant.

Changes in Control

The Amended Stock Plan, like the Stock Plan, provides that in the event of a change in control of Bank of America, all options and stock appreciation rights will be fully exercisable as of the date of the change in control and will remain exercisable through their full term. However, the Amended Stock Plan has changed the definition of change in control to state that a change in control occurs at completion, rather than at stockholder approval, of a merger, reorganization or other business combination of the company. Outstanding awards of restricted stock shares and restricted stock units will become immediately vested on the date of the change in control, and any applicable performance conditions will be deemed fully satisfied (at the target performance condition, if applicable) as of the date of the change in control.

Amendment and Termination of the Plan

The Bank of America board of directors has the power to amend, modify or terminate the Amended Stock Plan on a prospective basis. Stockholder approval will be obtained for any change to the material terms of the Amended Stock Plan to the extent required by NYSE listing requirements, Internal Revenue Code Section 162(m), or other applicable law.

No Repricings

The Amended Stock Plan, like the Stock Plan, prohibits any amendment to any outstanding option to decrease the exercise price (other than in connection with an adjustment to the capitalization of Bank of America described above) or any cancellation of an outstanding option in consideration for the grant of a new option with a lower exercise price, unless otherwise approved by the stockholders.

Federal Income Tax Treatment

Incentive Stock Options. Options granted under the Amended Stock Plan and intended to qualify as incentive stock options will be subject to the applicable provisions of the Internal Revenue Code, including Section 422. If shares of common stock are issued to an optionee upon the exercise of an incentive stock option, if no "disqualifying disposition" of the shares is made by the optionee within one year after the exercise of the incentive stock option or within two years after the date the incentive stock option was granted, and the options otherwise satisfy all applicable requirements under the Internal Revenue Code to qualify as incentive stock options, then (a) no income will be recognized by the optionee at the time of the grant of the incentive stock option, (b) no income, for regular income tax purposes, will be realized by the optionee at the date of exercise, (c) upon sale of the shares of Bank of America common stock acquired by exercise of the incentive stock option, any amount realized in excess of the option price will be taxed to the optionee, for regular income tax purposes, as a capital gain (at varying rates depending upon the optionee's holding period in the shares and income level) and any loss sustained will be a capital loss, and (d) no deduction will be allowed to Bank of America for federal income tax purposes. If a "disqualifying disposition" of the shares is made, the optionee will realize taxable ordinary income in an amount equal to the excess of the fair market value of the shares purchased at the time of exercise over the option price (the bargain purchase element) and Bank of America will be entitled to a federal income tax deduction equal to that amount. The amount of any gain in excess of the bargain purchase element

realized upon a "disqualifying disposition" will be taxable as capital gain at that time to the holder (at varying rates depending upon the holder's holding period in the shares and income level), for which Bank of America will not be entitled to a federal tax deduction. Upon exercise of an incentive stock option, the optionee may be subject to alternative minimum tax.

Nonqualified Stock Options. With respect to nonqualified stock options granted to optionees under the Amended Stock Plan, in general (a) the optionee will realize no income at the time the nonqualified stock option is granted, (b) at exercise, the optionee will realize ordinary income in an amount equal to the difference between the option price and the fair market value of the shares on the date of exercise, and Bank of America will receive a tax deduction for the same amount, and (c) on disposition, the holder will treat appreciation or depreciation after the date of exercise as capital gain or loss and taxed at varying rates depending upon the holder's holding period in the shares and income level.

Restricted Stock Shares. Upon becoming entitled to receive shares at the end of the applicable restricted period without a forfeiture, the recipient will have ordinary income in an amount equal to the fair market value of the shares at that time. However, a recipient who so elects under Internal Revenue Code Section 83(b) within 30 days after the date of the grant will have ordinary income on the date of the grant equal to the fair market value of the restricted stock shares as if the shares were unrestricted and could be sold immediately. If the shares subject to the election are subsequently forfeited, the recipient will not be entitled to any deduction, refund or loss for tax purposes. Upon the sale of the shares after the forfeiture period has expired, the holding period to determine whether the recipient has long-term or short-term capital gain or loss will begin when the restricted period expires, and the tax basis will be equal to the fair market value of the shares when the restricted period expires. However, if the recipient timely elects under Section 83(b) to be taxed as of the date of grant, the holding period will commence on the date of the grant and the tax basis will be equal to the fair market value of the shares on the date of the grant as if the shares were then unrestricted and could be sold immediately. Bank of America generally will be entitled to a deduction equal to the amount that is taxable as ordinary compensation income to the recipient.

Restricted Stock Units. A participant who is awarded restricted stock units will not recognize income and Bank of America will not be allowed a deduction at the time the award is made. When a participant receives payment for restricted stock units in shares of Bank of America common stock or cash, the fair market value of the shares or the amount of the cash received will be ordinary income to the participant and will be allowed as a deduction for federal income tax purposes to Bank of America. However, if there is a substantial risk that any shares of Bank of America common stock used to pay out earned restricted stock units will be forfeited (for example, because the compensation committee conditions those shares on the performance of future services), the taxable event will be deferred until the risk of forfeiture lapses. In this case, the participant can elect to make an election under Section 83(b) of the Internal Revenue Code as previously described. Bank of America can take the deduction at the time the ordinary income is recognized by the participant.

The Bank of America board recommends a vote "FOR" approval of the 2003 Key Associate Stock Plan, as amended and restated.

AMENDMENT TO THE BANK OF AMERICA AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

The Bank of America board of directors has adopted, subject to stockholder approval, an amendment to Bank of America's amended and restated certificate of incorporation to provide for an increase in the number of shares of Bank of America common stock authorized for issuance from 5 billion to 7.5 billion. In the event this proposal is adopted by Bank of America stockholders, but the merger agreement is terminated (without the merger being completed) prior to the filing of a certificate of amendment with the Secretary of State of the State of Delaware giving effect to the amendment, Bank of America may still file the certificate of amendment, but has not yet decided whether it will do so.

As of January 26, Bank of America had approximately 1,448,613,889 shares of Bank of America common stock issued and outstanding. As of January 26, there were 336,821,959 shares of Bank of America common stock reserved for issuance. Based on the number of FleetBoston common shares outstanding as of January 26, if the merger is completed, Bank of America will issue an additional [] shares of Bank of America common stock to the FleetBoston stockholders. Based on the options and other equity-based awards to purchase FleetBoston common stock to be assumed by Bank of America in the merger, if the merger is completed, Bank of America will reserve for issuance pursuant to the assumed options an additional [] shares of Bank of America common stock.

Although Bank of America's management currently has no definitive plans for the issuance of any additional authorized shares, the authorization of additional shares would permit the issuance of shares for future stock dividends, stock splits, possible acquisitions, stock option plans, and other appropriate corporate purposes. The additional shares of Bank of America common stock will not be entitled to preemptive rights nor will existing stockholders have any preemptive right to acquire any of those shares when issued. Adoption of the amendment requires the affirmative vote of a majority of the votes represented by the aggregate of all of the outstanding shares of Bank of America common stock, Series B Preferred Stock and ESOP Preferred Stock entitled to vote at the Bank of America special meeting. You are entitled to one vote for each share of Bank of America common stock and Series B Preferred Stock you held as of the record date, and two votes for each share of ESOP Preferred Stock you held as of the record date. Abstentions and broker non-votes will have the same effect as a vote against the amendment.

The Bank of America board of directors recommends a vote "FOR" the proposal to amend Bank of America's amended and restated certificate of incorporation to increase the number of authorized shares of Bank of America common stock from 5 billion to 7.5 billion.

COMPARATIVE MARKET PRICES AND DIVIDENDS

Bank of America common stock and FleetBoston common stock are listed on the NYSE, among other stock exchanges. The following table sets forth the high and low closing prices of shares of Bank of America common stock and FleetBoston common stock as reported on the NYSE, and the quarterly cash dividends paid per share in the periods indicated.

		Bank of America Common Stock		FleetBoston Common Stock		
	High	Low	Dividend	High	Low	Dividend
2001						
First Quarter	\$ 55.47	\$ 46.75	\$ 0.56	\$ 43.64	\$ 34.39	\$ 0.33
Second Quarter	61.94	49.59	0.56	42.28	36.06	0.33
Third Quarter	65.00	51.00	0.56	39.73	32.65	0.33
Fourth Quarter	64.99	52.15	0.60	38.64	31.45	0.33
2002						
First Quarter	69.18	58.85	0.60	37.21	31.10	0.35
Second Quarter	76.90	67.45	0.60	36.60	30.70	0.35
Third Quarter	71.94	57.90	0.60	31.75	18.75	0.35
Fourth Quarter	71.42	54.15	0.64	27.49	17.75	0.35
2003						
First Quarter	72.48	65.63	0.64	27.64	21.98	0.35
Second Quarter	79.89	68.00	0.64	31.15	24.55	0.35
Third Quarter	83.53	74.87	0.80	31.54	29.35	0.35
Fourth Quarter	82.50	72.85	0.80	43.65	31.18	0.35

Through [], 2004

Bank of America stockholders and FleetBoston stockholders are advised to obtain current market quotations for Bank of America common stock and FleetBoston common stock. The market price of Bank of America common stock and FleetBoston common stock will fluctuate between the date of this joint proxy statement/prospectus and the completion of the merger. No assurance can be given concerning the market price of Bank of America common stock or FleetBoston common stock before or after the effective date of the merger.

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

The following unaudited pro forma condensed combined financial information and explanatory notes present how the combined financial statements of Bank of America and FleetBoston may have appeared had the businesses actually been combined at the beginning of the period presented. The unaudited pro forma condensed combined financial information shows the impact of the merger of Bank of America and FleetBoston on the companies' respective historical financial positions and results of operations under the purchase method of accounting with Bank of America treated as the acquirer. Under this method of accounting, the assets and liabilities of FleetBoston will be recorded by Bank of America at their estimated fair values as of the date the merger is completed. The unaudited pro forma condensed combined financial information combines the historical financial information of Bank of America and FleetBoston as of and for the nine months ended September 30, 2003 and for the year ended December 31, 2002. The unaudited pro forma condensed combined balance sheet as of September 30, 2003 assumes the merger was completed on that date. The unaudited pro forma condensed combined statements of income give effect to the merger as if the merger had been completed on January 1, 2002.

The merger agreement was announced on October 27, 2003 and provides for each outstanding share of FleetBoston common stock other than shares beneficially owned by FleetBoston and Bank of America to be converted into the right to receive 0.5553 of a share of Bank of America common stock. Shares of FleetBoston preferred stock will be converted on a one-for-one basis into Bank of America preferred stock having the same terms (to the fullest extent possible) as the corresponding FleetBoston preferred stock, except in the case of shares held by preferred stockholders who validly perfect dissenters' appraisal rights. The unaudited pro forma condensed combined financial information has been derived from and should be read in conjunction with the historical consolidated financial statements and the related notes of both Bank of America and FleetBoston which are incorporated in this document by reference. See "Where You Can Find More Information" on page 114.

The unaudited pro forma condensed combined financial information is presented for illustrative purposes only and does not indicate the financial results of the combined companies had the companies actually been combined at the beginning of each period presented and had the impact of possible revenue enhancements, expense efficiencies, asset dispositions and share repurchases, among other factors, been considered. In addition, as explained in more detail in the accompanying notes to the unaudited pro forma condensed combined financial information, the allocation of the purchase price reflected in the pro forma condensed combined financial information is subject to adjustment and may vary from the actual purchase price allocation that will be recorded upon completion of the merger.

Bank of America/FleetBoston Pro Forma Condensed Combined Balance Sheet (unaudited)

The following preliminary unaudited pro forma condensed combined balance sheet combines the historical balance sheets of Bank of America and FleetBoston assuming the companies had been combined on September 30, 2003, on a purchase accounting basis.

September 30, 2003

		September 30, 2003			
	Bank of America	FleetBoston	Pro Forma Adjustments ⁽¹⁾		Bank of America FleetBoston Combined
(Dollars in millions)					
Assets					
Cash and cash equivalents	\$ 22,142	\$ 7,050	\$		\$ 29,192
Time deposits placed and other short-term investments	6,881	3,151			10,032
Federal funds sold and securities purchased under agreements to resell	67,729	3,914			71,643
Trading account assets	65,339	4,151			69,490
Securities	64,885	30,844	107	(A)	95,836
Loans and leases	373,098	126,344	136	(B)	499,578
Allowance for credit losses	(6,716)	(3,128)			(9,844)
Loans and leases, net of allowance for credit losses	366,382	123,216	136		489,734
Premises and equipment, net	5,956	2,415	(413)	(C)	7,958
Goodwill	11,456	4,272	(4,272)	(D)	43,913
	,	-,	32,457	(D)	,.
Core deposit intangibles and other intangibles	966	288	(288)	(E)	5,326
			4,360	(E)	-,
Other assets	125,352	17,097	(1,252)	(F)	141,197
Total assets	\$ 737,088	\$ 196,398	\$ 30,835		\$ 964,321
I Other House					,01,521

Liabilities					
Deposits in domestic offices:	\$ 122,669	e 22.277	\$		\$ 155,046
Noninterest-bearing Interest-bearing	257,586	\$ 32,377 88,610	295	(G)	346,491
	237,380	88,010	293	(G)	340,491
Deposits in foreign offices:	2.650	2,005			4 655
Noninterest-bearing	2,650				4,655
Interest-bearing	25,605	9,523			35,128
Total deposits	408,510	132,515	295		541,320
Federal funds purchased and securities sold under agreements to repurchase	79,775	6,584			86,359
Trading account liabilities	29,744	2,384			32,128
Commercial paper and other short-term borrowings	41,739	7,758			49,497
Accrued expenses and other liabilities	60,413	13,213	249	(H)	74,365
T	66.460	16.250	490	(H)	02.000
Long-term debt	66,462	16,350	1,077	(I)	83,889
Total liabilities	686,643	178,804	2,111		867,558
Shareholders' equity					
Preferred stock	55	271	_	(J)	326
Common stock	15	4,029	(4,029)	(J)	46,062
			46,047	(J)	
Retained earnings	52,320	14,504	(14,504)	(J)	52,320
Accumulated other comprehensive income (loss)	(1,776)	109	(109)	(J)	(1,776)
Other	(169)	(1,319)	1,319	(J)	(169)
Total shareholders' equity	50,445	17,594	28,724		96,763
					- 3,703
Total liabilities and shareholders' equity	\$ 737,088	\$ 196,398	\$ 30,835		\$ 964,321

⁽¹⁾ See Notes to Unaudited Pro Forma Condensed Combined Financial Information.

Bank of America/FleetBoston Pro Forma Condensed Combined Statement of Income (unaudited)

The following preliminary unaudited pro forma condensed combined statement of income combines the historical statements of income of Bank of America and FleetBoston assuming the companies had been combined on January 1, 2002, on a purchase accounting basis.

For the Nine Months Ended September 30, 2003

							,		
		Bank of America orporation	Fle	etBoston		Forma stments ⁽¹⁾		F	k of America leetBoston Combined
(Dollars in millions, except per share information)	_								
Interest income									
Interest and fees on loans and leases	\$	16,088	\$	5,540	\$	(30)	(B)	\$	21,598
Interest and dividends on securities		2,412		976		(3)	(A)		3,385
Trading account assets		3,024		31					3,055
Other interest income		2,051		331					2,382
	_							-	
Total interest income		23,575		6,878		(33)			30,420
Interest expense	_		_					_	
Deposits		3,730		1,056		(53)	(G)		4,733
Short-term borrowings		1,414		292		()	(-)		1,706
Long-term debt		1,584		760		(137)	(I)		2,207
Other interest expense		969		34					1,003
•	_								
Total interest expense		7,697		2,142		(190)			9,649
Total interest expense	<u> </u>	7,077	_	2,142		(170)			7,047
Net interest income		15,878		4,736		157			20,771
Noninterest income									
Service charges		4,182		1,170					5,352
Investment and brokerage services		1,737		1,120					2,857
Investment banking income		1,278		175					1,453
Card income		2,237		461					2,698
Other income		2,945		668					3,613
Total noninterest income	_	12,379		3,594					15,973
Total nonincress meome	<u> </u>	12,575	_	3,374					15,775
Total revenue		28,257		8,330		157			36,744
Provision for credit losses		2,256		830					3,086
Gains on sales of securities		802		104					906
Noninterest expense									
Personnel		7,749		2,507		(20)	(K)		10,236
Occupancy		1,492		388		(19)	(C)		1,861
Equipment		789		339		(25)	(C)		1,103
Other general operating		4,815		1,544		(70) 472	(L) (E)		6,761
	<u> </u>		_			4/2	(L)	_	
Total noninterest expense		14,845		4,778		338			19,961
	_	44.050	_	2.026	_	(404)		_	44.600
Income from continuing operations before income taxes		11,958		2,826		(181)	0.0		14,603
Applicable income tax expense		3,874		1,003		(67)	(M)		4,810
Income from continuing operations	\$	8,084	\$	1,823	\$	(114)		\$	9,793
•	_		_					_	
Income from continuing operations available to common shareholders	\$	8,081	\$	1,809	\$	(114)		\$	9,776
	_		_					_	
Per common share information	\$	5.41	\$	1.73				\$	4.71
Earnings per share-continuing operations	3	3.41	3	1./3				\$	4.71
Diluted earnings per share-continuing operations	\$	5.31	\$	1.72				\$	4.65
0,1			_					_	
Dividends paid	\$	2.08	\$	1.05				\$	2.08
	_		_					_	
Average common shares issued and outstanding (in thousands)		1,494,369	1	1,047,500		(465,823)	(N)		2,076,046
Average diluted common shares issued and outstanding (in thousands)	_	1,523,523		1,050,700	_	(467,246)	(N)		2,106,977
Average unuted common shares issued and outstanding (in thousands)		1,323,323		1,030,700		(407,240)	(IV)		2,100,9//

 $^{(1) \}hspace{0.5cm} \hbox{See Notes to Unaudited Pro Forma Condensed Combined Financial Information}.$

Bank of America/FleetBoston Pro Forma Condensed Combined Statement of Income (unaudited)

The following preliminary unaudited pro forma condensed combined statement of income combines the historical statements of income of Bank of America and FleetBoston assuming the companies had been combined on January 1, 2002, on a purchase accounting basis.

For the Year Ended December 31, 2002

		For th	e Year Ended December 31, 20	002		
	Bank of America Corporation	FleetBoston	Pro Forma Adjustments ⁽¹⁾		F	k of America leetBoston Combined
(Dollars in millions, except per share information)		<u> </u>				
Interest income	e 22.020	e 0.170	6 (20)	(D)	6	20.169
Interest and fees on loans and leases	\$ 22,030 4,035	\$ 8,168 1,422	\$ (30)	(B)	\$	30,168
Interest and dividends on securities Trading account assets	3,811	1,422	(82)	(A)		5,375 3,860
Other interest income	2,285	463				2,748
Other interest income		403				2,740
Total interest income	32,161	10,102	(112)			42,151
Interest expense						
Deposits	5,434	2,072	(87)	(G)		7,419
Short-term borrowings	2,089	417	(87)	(0)		2,506
Long-term debt	2,455	1,148	(230)	(I)		3,373
Other interest expense	1,260	45	(230)	(1)		1,305
Outer interest expense	1,200					1,303
Total interest expense	11,238	3,682	(317)			14,603
					_	
Net interest income	20,923	6,420	205			27,548
Noninterest income		4.500				
Service charges	5,276	1,533				6,809
Investment and brokerage services	2,237	1,559				3,796
Investment banking income	1,545	188				1,733
Card income	2,620	785				3,405
Other income	1,893	965				2,858
Total noninterest income	13,571	5,030				18,601
Total noninterest income	15,571					10,001
Total revenue	34,494	11,450	205			46,149
Provision for credit losses	3,697	2,760				6,457
Gains on sales of securities	630	6				636
Noninterest expense						
Personnel	9,682	3,255	(27)	(K)		12,910
Occupancy	1,780	504	(25)	(C)		2,259
Equipment	1,124	478	(33)	(C)		1,569
Other general operating	5,850	2,167	(93)	(L)		8,609
			685	(E)		
Total noninterest expense	18,436	6,404	507		_	25,347
Total nonimerest expense	18,430	0,404	307			25,347
Income from continuing operations before income taxes	12,991	2,292	(302)			14,981
Applicable income tax expense	3,742	768	(111)	(M)		4,399
Income from continuing operations	\$ 9,249	\$ 1,524	\$ (191)		\$	10,582
Income from continuing operations available to common shareholders	\$ 9,244	\$ 1,506	\$ (191)		\$	10,559
income from continuing operations available to common shareholders	ψ <i>)</i> ,244	Ψ 1,500	\$ (171)		Ψ	10,557
Per common share information						
Earnings per share-continuing operations	\$ 6.08	\$ 1.44			\$	5.03
Diluted earnings per share-continuing operations	\$ 5.91	\$ 1.44			\$	4.93
5.1.						,0
Dividends paid	\$ 2.44	\$ 1.40			\$	2.44
					_	
Average common shares issued and outstanding (in thousands)	1,520,042	1,045,300	(464,845)	(N)		2,100,497
Average diluted common shares issued and outstanding (in thousands)	1,565,467	1,048,700	(466,357)	(N)		2,147,810
	1,505,407	1,040,700	(400,557)	(11)		2,147,310

 $^{(1) \}hspace{0.5cm} \hbox{See Notes to Unaudited Pro Forma Condensed Combined Financial Information}.$

NOTES TO THE UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

Note 1 — Basis of Pro Forma Presentation

The unaudited pro forma condensed combined financial information related to the merger is included for the year ended December 31, 2002 and as of and for the nine months ended September 30, 2003. The pro forma adjustments included herein reflect the conversion of FleetBoston common stock into Bank of America common stock using an exchange ratio of 0.5553 of a share of Bank of America common stock for each of the 1,052,600,000 shares of FleetBoston common stock outstanding at September 30, 2003, \$271 million related to the conversion of 1,082,450 shares of preferred stock and \$1.11 billion for the approximately 96 million shares of FleetBoston common stock issuable under outstanding stock options that will convert into Bank of America stock options. The estimated purchase price of \$46.32 billion, which includes the value of stock options, is based on a per share price for Bank of America common stock of \$76.88, which was the average of the closing prices of Bank of America common stock for the period commencing two trading days before, and ending two trading days after, October 27, 2003, the date of the merger agreement. Bank of America preferred stock to be exchanged was valued using the book value of FleetBoston preferred stock.

The merger will be accounted for using the purchase method of accounting; accordingly, Bank of America's cost to acquire FleetBoston will be allocated to the assets (including identifiable intangible assets) and liabilities (including executory contracts and other commitments) of FleetBoston at their respective fair values on the date the merger is completed.

The unaudited pro forma condensed combined financial information includes estimated adjustments to record the assets and liabilities of FleetBoston at their respective fair values and represents management's estimates based on available information. The pro forma adjustments included herein may be revised as additional information becomes available and as additional analyses are performed. The final allocation of the purchase price will be determined after the merger is completed and after completion of a final analysis to determine the fair values of FleetBoston's tangible, and identifiable intangible, assets and liabilities as of the completion date. Accordingly, the final purchase accounting adjustments and integration charges may be materially different from the pro forma adjustments presented in this document. Increases or decreases in the fair value of the net assets, commitments, executory contracts and other items of FleetBoston as compared to the information shown in this document may change the amount of the purchase price allocated to goodwill and other assets and liabilities and may impact the statement of income due to adjustments in yield and/or amortization of the adjusted assets or liabilities.

The pro forma financial statements do not currently include any amount related to the estimated \$800 million after-tax (\$1.27 billion pre-tax) merger related costs that will be incurred to combine the operations of Bank of America and FleetBoston. The estimated merger related charges will result from action taken with respect to both Bank of America and FleetBoston operations, facilities and associates. The charges will be recorded based on the nature and timing of these integration actions. See Note 3 for a further discussion of these charges.

Certain amounts in the historical consolidated financial statements of FleetBoston have been reclassified to conform with Bank of America's historical financial information presentation. Discontinued operations reported in FleetBoston's historical consolidated statements of income have been excluded. The unaudited pro forma condensed combined financial information presented in this document does not necessarily indicate the results of operations or the combined financial position that would have resulted had the merger been completed at the beginning of the applicable period presented, nor is it indicative of the results of operations in future periods or the future financial position of the combined company.

Note 2 - Pro Forma Adjustments

The unaudited pro forma condensed combined financial information for the merger includes the pro forma balance sheet as of September 30, 2003 assuming the merger was completed on September 30, 2003. The pro forma income statements for the nine months ended September 30, 2003 and the year ended December 31, 2002 were prepared assuming the merger was completed on January 1, 2002.

The unaudited pro forma condensed combined financial information reflects the issuance of 584,509,000 shares of Bank of America common stock with an aggregate value of \$44.94 billion, the issuance of \$271 million of Bank of America preferred stock and the conversion of approximately 96 million FleetBoston stock options with a value of approximately \$1.11 billion at September 30, 2003. Common stock and preferred stock issued in the exchange was valued using the methodology discussed in Note 1 above.

Substantially all of the FleetBoston stock options vest upon completion of the merger and will be converted into Bank of America stock options. The fair value of the Bank of America options that will be issued in exchange for the FleetBoston options was estimated using a Black-Scholes option pricing model. Option pricing models require the use of highly subjective assumptions including expected stock price and volatility that when changed can materially affect fair value estimates. Accordingly, the model does not necessarily provide for a reliable single measure of the fair value of employee stock options. The more significant assumptions used in estimating the fair value of the Bank of America stock options to be issued in the exchange for FleetBoston stock options include a risk-free interest rate of 4.07%, a dividend yield of 4.70%, a weighted average expected life of three years and volatility of 26.57%. The three-year term was based on the weighted average expected term to expiration of these options.

The allocation of the purchase price follows:

	September 3	30, 2003
(Dollars in millions)	-	
Purchase Price		
FleetBoston common stock outstanding (in thousands)	1,052,600	
Exchange ratio	0.5553	
-		
Total Bank of America Common Stock to be issued (in thousands)	584,509	
Purchase price per Bank of America common share	\$ 76.88	
• •		
		\$ 44,937
FleetBoston preferred stock converted to Bank of America preferred stock		271
Fair value of outstanding employee and non-employee stock options		1,110
Total purchase price		\$ 46,318
Net assets acquired		
FleetBoston stockholders' equity	\$ 17,594	
FleetBoston goodwill and other intangible assets	(4,560)	
Estimated adjustments to reflect assets acquired at fair value:		
Securities	107	
Loans and leases	136	
Premises and equipment	(413)	
Identified intangibles	4,360	
Other assets	(1,252)	
Estimated amounts allocated to liabilities assumed at fair value:		
Deposits	(295)	
Personnel related liabilities	(249)	
Deferred income taxes	(490)	
Long-term debt	(1,077)	
	(13,861)	
Goodwill resulting from merger		\$ 32,457
		Ψ 0 2 , . 0 7

The pro forma adjustments included in the unaudited pro forma condensed combined financial information are as follows:

(A) Adjustment to fair-value the securities portfolio. The adjustment will be recognized over the remaining life of the securities portfolio. The impact of the adjustment was to decrease interest income by approximately \$3 million and \$82 million for the nine months ended September 30, 2003 and the twelve months ended December 31, 2002, respectively.

- (B) Adjustment to fair-value of the loan and lease portfolio. The adjustment will be recognized over the estimated remaining life of the loan and lease portfolio. The impact of the adjustment was to decrease interest income by approximately \$30 million for both the nine months ended September 30, 2003 and the twelve months ended December 31, 2002.
- (C) Adjustment to fair-value owned real estate, leased property and related improvements, signage and computer equipment. The effect of these adjustments is to reduce occupancy costs by \$19 million and \$25 million and equipment costs by \$25 million and \$33 million for the nine months ended September 30, 2003 and the twelve months ended December 31, 2002, respectively.
- (D) Adjustment to write off historical FleetBoston goodwill and record goodwill created as a result of the merger.
- (E) Adjustment to write off historical FleetBoston intangible assets (other than goodwill) and to record intangible assets (other than goodwill) resulting from the merger based on estimated fair values. The nature, amount and amortization method of various possible identified intangibles are being studied by management. The adjustments reflected herein are based on current assumptions and valuations, which are subject to change. For purposes of the pro forma adjustments shown here, we have estimated \$4.36 billion of intangibles that consists of a core deposit intangible in the range of \$2.0 to \$3.0 billion, a purchased credit card relationship intangible in the range of \$600 to \$900 million and other customer relationship intangibles in the range of \$200 to \$500 million. We estimate these intangibles will be amortized over a period not to exceed ten years, on an accelerated basis for the core deposit intangible and purchased credit card relationship intangibles and a straight-line basis for the other intangibles. The value of the intangibles represents the estimated future economic benefit resulting from the acquired customer balances and relationships. This value was estimated by considering cash flows from the current balances of accounts, expected growth or attrition in balances, and the estimated life of the relationship. Material changes are possible when our analysis is completed. The impact of these adjustments is to increase other general operating expenses by \$472 million and \$685 million for the nine months ended September 30, 2003 and the twelve months ended December 31, 2002, respectively.
- (F) Adjustments to fair-value other assets including pre-paid pension asset, computer software, deferred costs and other miscellaneous items.
- (G) Adjustment to fair-value fixed-rate deposit liabilities based on current interest rates for similar instruments. The adjustment will be recognized over the estimated remaining term of the related deposit liability. The impact of the adjustment was to decrease interest expense by approximately \$53 million and \$87 million for the nine months ended September 30, 2003 and the twelve months ended December 31, 2002, respectively.
- (H) Adjustment to accrued expenses and other liabilities consists of \$249 million to reflect the fair value of pension liabilities and current interest rate adjustments to deferred compensation plans and \$490 million for deferred tax liabilities, net of deferred tax assets, resulting from the proforma adjustments. Deferred taxes were recorded using Bank of America's statutory rate of 36.9%.
- (I) Adjustment to fair-value outstanding long-term debt instruments. The adjustment will be recognized over the remaining life of the long-term debt instruments. The impact of the adjustment was to decrease interest expense by approximately \$137 million and \$230 million for the nine months ended September 30, 2003 and the twelve months ended December 31, 2002, respectively.
- (J) Adjustment to eliminate FleetBoston's historical shareholders' equity, except for the preferred stock which is expected to be converted on a one-for-one basis into Bank of America preferred stock. Additionally, the adjustment reflects the issuance of Bank of America common stock and the conversion of FleetBoston stock options into Bank of America stock options.
- (K) Adjustment of fixed-rate deferred compensation plans to current interest rates discussed in (H) above.
- (L) Adjustment to reverse amortization of intangible assets recorded in FleetBoston's historical financial statements.

- (M) Adjustment to record the tax effect of the pro forma adjustments using Bank of America's statutory tax rate of 36.9%.
- (N) Weighted average shares were calculated using the historical weighted average shares outstanding of Bank of America and FleetBoston, adjusted using the exchange ratio, to the equivalent shares of Bank of America common stock, for the year ended December 31, 2002 and the nine months ended September 30, 2003. Earnings per share data have been computed based on the combined historical income of Bank of America, income from continuing operations for FleetBoston and the impact of purchase accounting adjustments.

Note 3-Merger Related Charges

In connection with the merger, we have been developing our plan to integrate Bank of America's and FleetBoston's operations. The total integration costs have been estimated to be \$800 million after-tax or \$1.27 billion pre-tax. The specific details of these plans will continue to be refined over the next several months. Currently, our merger integration team is assessing the two companies' operations, including information systems, premises, equipment, benefit plans, supply chain methodologies, service contracts and personnel to determine optimum strategies to realize cost savings.

Our merger integration decisions will impact certain existing FleetBoston facilities (both leased and owned), information systems, supplier contracts and costs associated with the involuntary termination of personnel. Additionally, as part of our formulation of the merger integration plan, certain actions regarding existing Bank of America information systems, premises, equipment, benefit plans, supply chain methodologies, supplier contracts and costs associated with the involuntary termination of personnel may be impacted. To the extent there are costs associated with these actions, the costs will be recorded based on the nature and timing of these integration actions. We expect that such decisions will be completed shortly after the merger. Restructuring charges will be recorded based on the nature and timing of these integration actions.

Included in the costs described above, during the combination of the two companies we will incur additional integration costs consisting of employee retention agreements, conversion costs and incremental communication costs to customers and associates, among other costs. It is expected that these costs will be incurred over a two-year period after completion of the merger. These costs will be expensed as incurred.

Note 4—Estimated Annual Cost Savings

The following estimated annual cost savings represent our estimate only and may not be indicative of the actual amount or nature of the cost savings the combined company actually achieves. The below information does not include the impact of possible revenue opportunities. The following table summarizes Bank of America's estimated annual after-tax cost savings when fully phased in after the merger:

	Annual After- Tax Cost Savings
Overlapping Business Infrastructure/Redundant Processes	\$650 million (A)
Corporate Staff Functions	100 million (B)
Marketing	100 million (C)
Vendor Leverage	175 million (D)
Occupancy	75 million (E)
Total	\$1.1 billion
Occupancy	75 million (E)

(A) Overlapping business infrastructure including technology and operations functions are projected to result in cost savings due to the elimination of redundant systems and software, the elimination of redundant operational support and activities and reduced personnel costs for the combined company.

- (B) Corporate Staff Function cost savings are projected to occur from reduced personnel costs and elimination of duplicative corporate and administrative functions.
- (C) Marketing cost savings are projected from the consolidation of advertising and marketing efforts under a common brand.
- (D) Vendor leverage cost savings are projected to result in an annual cost savings due to purchasing efficiencies from the combined company's negotiating power to achieve reduced rates for services provided.
- (E) Occupancy cost savings are projected to result from consolidation of personnel into a reduced number of office facilities and leased space.

LEGAL MATTERS

The validity of the Bank of America common stock and Bank of America preferred stock being registered in connection with the merger will be passed upon for Bank of America by its Executive Vice President and General Counsel, Paul J. Polking. As of the date of this joint proxy statement/prospectus, Mr. Polking beneficially owned shares of Bank of America common stock.

EXPERTS

The consolidated financial statements of Bank of America and its subsidiaries as of December 31, 2002 and 2001, and for each of the years in the three-year period ended December 31, 2002, incorporated in this joint proxy statement/prospectus by reference to Bank of America's Annual Report on Form 10-K for the year ended December 31, 2002 have been so incorporated in reliance on the report of PricewaterhouseCoopers LLP, independent accountants, given on the authority of said firm as experts in auditing and accounting.

The consolidated financial statements of FleetBoston and its subsidiaries as of December 31, 2002 and 2001, and for each of the years in the three-year period ended December 31, 2002, incorporated in this joint proxy statement/prospectus by reference to FleetBoston's Annual Report on Form 10-K for the year ended December 31, 2002 have been so incorporated in reliance on the report of PricewaterhouseCoopers LLP, independent accountants, given on the authority of said firm as experts in auditing and accounting.

Bank of America and FleetBoston expect representatives of PricewaterhouseCoopers LLP to attend the Bank of America special meeting and the FleetBoston special meeting. These representatives will have an opportunity to make a statement if they desire to do so, and we expect that they will be available to respond to any appropriate questions you may have.

OTHER MATTERS

According to both the Bank of America bylaws and FleetBoston bylaws, business to be conducted at a special meeting of stockholders may only be brought before the meeting pursuant to a notice of meeting. Accordingly, no matters other than the matters described in this joint proxy statement/prospectus will be presented for action at the special meetings or at any adjournment or postponement of the special meetings.

Bank of America 2004 Annual Meeting Stockholder Proposals

For a stockholder proposal to be considered for inclusion in Bank of America's proxy statement and form of proxy relating to the Bank of America annual meeting of stockholders to be held in 2004, the Secretary of Bank of America must have received the proposal, at the address set forth on the first page of this joint proxy statement/prospectus, not later than November 28, 2003. However, if the Bank of America 2004 annual meeting is held on or after May 31, 2004, a stockholder proposal must be received within a reasonable time before Bank of America begins to print and mail its proxy solicitation materials for the Bank of America 2004 annual meeting. Any proposal will be subject to Rule 14a-8 under the Exchange Act.

The Bank of America bylaws provide that in order for a stockholder to make a proposal or a nomination for the election of directors before the Bank of America annual meeting, a stockholder must deliver written notice of such proposal or nomination to the Secretary no later than the close of business on the seventy-fifth (75th) day and no earlier than the close of business on the one hundred and twentieth (120th) day prior to the first anniversary of the date Bank of America commenced mailing its proxy materials for the preceding year's annual meeting. However, if the annual meeting is more than thirty (30) days before or more than seventy (70) days after the anniversary date of the preceding year's annual meeting, the stockholder must deliver the required

notice to Bank of America no earlier than the close of business on the one hundred and twentieth (120°) day prior to the annual meeting and not later than the close of business on the later of the seventy-fifth (75th) day prior to the annual meeting or the tenth (10h) day following the day on which Bank of America first publicly announces the date of the annual meeting.

FleetBoston 2004 Annual Meeting Stockholder Proposals

FleetBoston intends to hold a 2004 annual meeting of stockholders only if the merger agreement is terminated. For a stockholder proposal to be considered for inclusion in FleetBoston's proxy statement and form of proxy relating to the FleetBoston 2004 annual meeting of stockholders (in the event this meeting is held), the Secretary of FleetBoston must have received the proposal, at the address set forth on the first page of this joint proxy statement/prospectus, not later than November 18, 2003. However, if the FleetBoston 2004 annual meeting of stockholders is held on or after May 16, 2004, a stockholder proposal must be received within a reasonable time before FleetBoston begins to print and mail its proxy solicitation materials for the FleetBoston 2004 annual meeting. Any stockholder proposals will be subject to Rule 14a-8 under the Exchange Act.

The FleetBoston bylaws provide that, in order for a stockholder to make nominations for the election of directors before the FleetBoston annual meeting, a stockholder must deliver written notice of such nomination to (and such notice must be received by) the Secretary not less than thirty (30) days before the date of the FleetBoston annual meeting. However, if FleetBoston gives its stockholders less than forty (40) days' notice or prior public disclosure of the date of the annual meeting of stockholders, stockholders must deliver the required notice to FleetBoston not later than the close of business on the seventh (7th) day following the earlier of the day on which FleetBoston mailed notice of the FleetBoston annual meeting to stockholders or the day on which FleetBoston publicly disclosed the meeting date. Further, FleetBoston's bylaws provide that in order for a stockholder to make proposals for business to be brought before the Annual Meeting, a stockholder must deliver notice of the proposal to the Secretary not less than one hundred and twenty (120) days prior to the first anniversary of the date FleetBoston commenced mailing its proxy materials for the preceding year's annual meeting. However, if no annual meeting was held in the preceding year or if the date of the annual meeting has changed by more than thirty (30) calendar days, FleetBoston must receive a proposal within a reasonable time before the company's proxy statement is mailed to stockholders.

ANNUAL MEETINGS

Bank of America currently expects that it will convene and hold its regular 2004 annual meeting of stockholders in May of 2004 following completion of the merger. Bank of America intends to file proxy materials, including a proxy statement and a form of proxy, relating to the 2004 Bank of America annual meeting as soon as practicable. FleetBoston intends to hold a 2004 annual meeting of stockholders only if the merger agreement is terminated.

WHERE YOU CAN FIND MORE INFORMATION

Bank of America has filed with the SEC a registration statement under the Securities Act that registers the distribution to FleetBoston stockholders of the shares of Bank of America common stock to be issued in connection with the merger. The registration statement, including the attached exhibits and schedules, contains additional relevant information about Bank of America and Bank of America stock. The rules and regulations of the SEC allow us to omit certain information included in the registration statement from this document.

You may read and copy this information at the Public Reference Room of the SEC at 450 Fifth Street, N.W., Washington, D.C. 20549. You may obtain information on the operation of the SEC's Public Reference Room by calling the SEC at 1-800-SEC-0330. The SEC also maintains an internet world-wide web site that contains reports, proxy statements and other information about issuers, like Bank of America and FleetBoston, who file electronically with the SEC. The address of the site is http://www.sec.gov. The reports and other information filed by Bank of America with the SEC are also available at Bank of America's internet world-wide web site. The address of the site is http://www.bankofamerica.com. The reports and other information filed by FleetBoston with the SEC are also available at FleetBoston's internet world-wide web site. The address of the site is http://www.fleet.com. We have included the web addresses of the SEC, Bank of America, and FleetBoston as inactive textual references only. Except as specifically incorporated by reference into this joint proxy statement/prospectus, information on those web sites is not part of this joint proxy statement/prospectus.

You also should be able to inspect reports, proxy statements and other information about Bank of America and FleetBoston at the offices of the NYSE at 20 Broad Street. 17th Floor, New York, New York 10005, and, with respect to FleetBoston, the Boston Stock Exchange, 100 Franklin Street, Boston, Massachusetts 02110.

The SEC allows Bank of America and FleetBoston to incorporate by reference information in this document. This means that Bank of America and FleetBoston can disclose important information to you by referring you to another document filed separately with the SEC. The information incorporated by reference is considered to be a part of this document, except for any information that is superseded by information that is included directly in this document.

This document incorporates by reference the documents listed below that Bank of America and FleetBoston previously filed with the SEC. They contain important information about the companies and their financial condition.

Bank of America SEC Filings (SEC file no. 1-06523; CIK no. 0000070858)

Annual Report on Form 10-K

Quarterly Reports on Form 10-Q

Current Reports on Form 8-K

The description of Bank of America common stock set forth in a registration statement filed pursuant to Section 12 of the Exchange Act, as modified on our current report on Form 8-K dated September 28, 1998

Period or Date Filed

Year ended December 31, 2002

Quarters ended March 31, 2003, June 30, 2003 and September 30, 2003

Filed on January 15, 2003; January 23, 2003; March 3, 2003; March 5, 2003; March 31, 2003; April 14, 2003; April 30, 2003; July 14, 2003; July 23, 2003; October 14, 2003; October 21, 2003; October 27, 2003; October 28, 2003; November 5, 2003; November 20, 2003; December 2, 2003; December 5, 2003; December 10, 2003; January 2, 2004; January 15, 2004; and January 29, 2004 (other than those portions furnished under Item 9 or Item 12 of Form 8-K)

FleetBoston SEC Filings (SEC file no. 1-16431; CIK no. 0000050341)

Annual Report on Form 10-K

Quarterly Reports on Form 10-Q

Current Reports on Form 8-K

The description of FleetBoston common stock set forth in the FleetBoston registration statement filed by Industrial National Corporation (predecessor to FleetBoston) on Form 8-B dated May 29, 1970, and the description of the preferred share purchase rights set forth in the FleetBoston registration statement on Form 8-A dated November 7, 2000, and any amendment or report filed for the purpose of updating those descriptions.

Period or Date Filed

Year ended December 31, 2002

Quarters ended March 31, 2003, June 30, 2003 and September 30, 2003

Filed on January 10, 2003; January 16, 2003; February 18, 2003; April 16, 2003; July 16, 2003; July 31, 2003; October 16, 2003; October 28, 2003; December 8, 2003; and January 15, 2004 (other than those portions furnished under Item 9 or Item 12 of Form 8-K)

In addition, Bank of America and FleetBoston also incorporate by reference additional documents that either company files with the SEC between the date of this document and the date of the Bank of America special meeting or the FleetBoston special meeting. These documents include periodic reports, such as Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q and Current Reports on Form 8-K, as well as proxy statements.

Bank of America has supplied all information contained or incorporated by reference in this document relating to Bank of America, as well as all pro forma financial information, and FleetBoston has supplied all information relating to FleetBoston.

Documents incorporated by reference are available from Bank of America and FleetBoston without charge, excluding any exhibits to those documents unless the exhibit is specifically incorporated by reference as an exhibit in this document. You can obtain documents incorporated by reference in this document by requesting them in writing or by telephone from the appropriate company at the following addresses:

Bank of America Corporation

Bank of America Corporate Center Charlotte, North Carolina 28255 Investor Relations Telephone: (704) 386-5681 FleetBoston Financial Corporation

Investor Relations Department FleetBoston Financial Corporation P.O. Box 2016, MA DE 10032F Boston, MA 02106-2106 (617) 434-7858

Bank of America and FleetBoston stockholders requesting documents should do so by March 10, 2004 to receive them before the special meetings. You will not be charged for any of these documents that you request. If you request any incorporated documents from Bank of America or FleetBoston, Bank of America or FleetBoston will mail them to you by first class mail, or another equally prompt means, within one business day after it receives your request.

Neither Bank of America nor FleetBoston has authorized anyone to give any information or make any representation about the merger or our companies that is different from, or in addition to, that contained in this document or in any of the materials that have been incorporated in this document. Therefore, if anyone does give you information of this sort, you should not rely on it. If you are in a jurisdiction where offers to exchange or sell, or solicitations of offers to exchange or purchase, the securities offered by this document or the solicitation of proxies is unlawful, or if you are a person to whom it is unlawful to direct these types of activities, then the offer presented in this document does not extend to you. The information contained in this document speaks only as of the date of this document unless the information specifically indicates that another date applies.

AGREEMENT AND PLAN OF MERGER

by and between

FLEETBOSTON FINANCIAL CORPORATION

and

BANK OF AMERICA CORPORATION

DATED AS OF OCTOBER 27, 2003

3.22

FleetBoston Information

TABLE OF CONTENTS

ARTICLE I

THE MERGER

1.1.	The Merger	A-1
1.2	Effective Time	A-2
1.3	Effects of the Merger	A-2
1.4	Conversion of FleetBoston Capital Stock	A-2
1.5	Bank of America Capital Stock	A-3
1.6	Bank of America Mirror Preferred Stock	A-3
1.7	Options and SARs; Other Stock-Based Awards	A-3
1.8	Certificate of Incorporation of Bank of America	A-4
1.9	Bylaws of Bank of America	A-4
1.10	Tax Consequences	A-4
1.11	Certain Post-Closing Matters	A-5
	ARTICLE II	
	EXCHANGE OF SHARES	
2.1	Bank of America to Make Shares Available	A-5
2.2	Exchange of Shares	A-6
	ARTICLE III	
	REPRESENTATIONS AND WARRANTIES OF FLEETBOSTON	
3.1	Corporate Organization	A-7
3.2	Capitalization	A-8
3.3	Authority; No Violation	A-9
3.4	Consents and Approvals	A-10
3.5	Reports	A-10
3.6	Financial Statements	A-11
3.7	Broker's Fees	A-11
3.8	Absence of Certain Changes or Events	A-11
3.9	Legal Proceedings	A-11
3.10	Taxes and Tax Returns	A-11
3.11	<u>Employees</u>	A-12
3.12	SEC Reports	A-13
3.13	Compliance with Applicable Law	A-13
3.14	Certain Contracts	A-14
3.15	Agreements with Regulatory Agencies	A-14
3.16	Interest Rate Risk Management Instruments	A-14
3.17	Undisclosed Liabilities	A-15
3.18	Environmental Liability	A-15
3.19	State Takeover Laws; FleetBoston Rights Agreement	A-15
3.20	Reorganization	A-15
3.21	Opinions	A-16

A-16

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF BANK OF AMERICA

4.1	Corporate Organization	A-16
4.2	Capitalization	A-17
4.3	Authority; No Violation	A-17
4.4	Consents and Approvals	A-18
4.5	Reports	A-18
4.6	Financial Statements	A-19
4.7	Broker's Fees	A-19
4.8	Absence of Certain Changes or Events	A-19
4.9	Legal Proceedings	A-19
4.10	Taxes and Tax Returns	A-20
4.11	SEC Reports	A-20
4.12	Compliance with Applicable Law	A-20
4.13	Agreements with Regulatory Agencies	A-21
4.14	Interest Rate Risk Management Instruments	A-21
4.15	<u>Undisclosed Liabilities</u>	A-21
4.16	Environmental Liability	A-21
4.17	State Takeover Laws	A-22
4.18	Reorganization	A-22
4.19	<u>Opinions</u>	A-22
4.20	Bank of America Information	A-22
	ARTICLE V	
	COVENANTS RELATING TO CONDUCT OF BUSINESS	
5.1	Conduct of Businesses Prior to the Effective Time	A-22
5.2	FleetBoston Forbearances	A-22
5.3	Bank of America Forbearances	A-24
	ARTICLE VI	
	ARTICLE VI	
	ADDITIONAL AGREEMENTS	
6.1	Regulatory Matters	A-25
6.2	Access to Information	A-26
6.3	Access to information Shareholder Approvals	A-26 A-26
6.4	Legal Conditions to Merger	A-26 A-26
6.5	Affiliates	A-20 A-26
6.6	NYSE Listing	A-26
6.7	Employee Matters	A-27
6.8	Indemnification; Directors' and Officers' Insurance	A-27
6.9	Additional Agreements	A-28
6.10	Advice of Changes	A-28
6.11	Dividends	A-28
6.12	Exemption from Liability Under Section 16(b)	A-28
6.13	No Solicitation	A-29
6.14	Restructuring Efforts	A-30

ARTICLE VII

$\underline{\textbf{CONDITIONS PRECEDENT}}$

7.2 7.3	Conditions to Obligations of Bank of America Conditions to Obligations of FleetBoston	A-31 A-32
	ARTICLE VIII	
	TERMINATION AND AMENDMENT	
8.1	Termination	A-32
8.2	Effect of Termination	A-33
8.3 8.4	Amendment Extension; Waiver	A-33 A-33
0.4	Extension, waiver	A-33
	ARTICLE IX	
	GENERAL PROVISIONS	
9.1	Closing	A-33
9.2	<u>Standard</u>	A-34
9.3	Nonsurvival of Representations, Warranties and Agreements	A-34
9.4	Expenses	A-34
9.5	Notices	A-34
9.6	<u>Interpretation</u>	A-35 A-35
9.7 9.8	Counterparts Entire Agreement	A-35 A-35
9.8	Governing Law	A-35 A-35
9.10	Publicity Publicity	A-35 A-35
9.11	Assignment; Third Party Beneficiaries	A-36
Exhibit A	A – FleetBoston Stock Option Agreement	
Exhibit I	B – Bank of America Stock Option Agreement	
Exhibit (C – Form of Affiliate Letter	A-38

INDEX OF DEFINED TERMS

Agustinion Proposal 6,13(a) Promiser Agreement 6,13(a) Altrealiev Transaction 6,13(a) Articles of Merger 1,7(a) Assumed SAR 1,7(a) Assumed Stock-Dated Award 1,7(a) Assumed Stock-Dated Award 1,7(a) Bank of America Preamble Bank of America (poll Stock) 1,5 Bank of America Ceptificate 4,1(a) Bank of America Comiton Stock 1,4(a) Bank of America Comiton Stock 1,4(a) Bank of America Comiton Stock 1,4(a) Bank of America Common Stock 1,4(a) Bank of America Stock Delice our Schoelide 4,1(a) Bank of America Regulatory Agreement 4,12(a) Bank of America Regulatory Agreement 4,13 Bank of America Stock Plans 4,2(a) Bank of America Stoc		Section
Agreement Alternative Transaction 6.13(a) Afficenative Transaction 6.13(a) Articles of Merger 1.2 Assumed Stock-Dased Award 1.7(b) Assumed Stock Option 1.7(b) Bank of America Option 4.6 Bank of America 10-Q 4.6 Bank of America Dayle 4.1(a) Bank of America Cortificate 1.5 Bank of America Cortificate 4.1(a) Bank of America Convertible Preferred 4.2(a) Bank of America Convertible Preferred 4.7(a) Bank of America Reports 4.7(a) Bank of America Reports 4.7(a) Bank of America Reports 4.7(a) Bank of America Regulatory Agreement 4.1 Bank of America Regulatory Agreement 4.1 Bank of America Subschalary 4.1 Bank of America Subschalary 4.1 Bank of America Subschalary 5.1 Bank of America Subschalary 6.7(a) Bank of America Subschalary 6.7(a) Bank of America Subschalary 6.7(a) Bank of America Subschalary <td>Acquisition Proposal</td> <td>6.13(a)</td>	Acquisition Proposal	6.13(a)
Alternative Transaction 5.136a Articles of Merger 1.2 Assumed SAR 1.76b Assumed Stock Option 1.76b Bank of America Preamble Bank of America (Po) 4.6 Bank of America Capital Stock 1.5 Bank of America Capital Stock 1.5 Bank of America Controllifecte 4.10a Bank of America Common Stock 1.40a Bank of America Common Stock 4.70a Bank of America Disclosure Schedule 4.70a Bank of America Disclosure Schedule 4.70a Bank of America Regulatory Agreement 4.10a Bank of America Regulatory Agreement 4.13 Bank of America Stock Plans 4.20a Bank of America Stock Plans		
Articles of Merger Assumed SNA Assumed SNA Assumed SNA Assumed SNA Bank of America Capital Stock Bank of America Capital Stock Bank of America Certificate Bank of America Corner Stock Bank of America Besideure Schedule Bank of America Besideure Schedule Bank of America Bisoleure Schedule Bank of America Stock Plans Bank of America Bisoleure Schedule Bank of America Bisoleure Schedule Bank of America Subsidiary Bank of America Subsidiary Bank of America Stock Plans Bank of America Beputation Bank of Ameri		
Assumed Stock-Based Award 1.7(b) Assumed Stock Option 1.7(b) Bank of America 4.6 Bank of America Capital Stock 1.5 Bank of America Certificate 4.1(b) Bank of America Common Stock 1.4(b) Bank of America Common Stock 4.2(b) Bank of America Common Stock 4.2(b) Bank of America Disclosure Schedule Art. IV Bank of America Mirror Preferred Stock 4.2(b) Bank of America Reputal Schedule 4.1(b) Bank of America Reputal Schedule 4.1(b) Bank of America Reputal Schedule 4.1(b) Bank of America Schedule 4.1(b) Bank of America Reputal Schedule 4.1(b) Bank of America Schedule 3.1(c) Bank of America Schedule 4.2(a) Bank of America Schedule 4.2(a) Bank of America Schedule 3.1(c) Bank of America Schedule 4.2(a) Bank of		
Assumed Slock Option 1.760 Bank of America Premable Bank of America 10-Q 4.6 Bank of America Capital Slock 1.5 Bank of America Certificate 4.160 Bank of America Certificate 4.160 Bank of America Conventible Preferred 4.260 Bank of America Describile Preferred Slock 1.460 Bank of America Mirror Preferred Slock 4.260 Bank of America Mirror Preferred Slock 4.23 Bank of America Regulatory Agreement 4.13 Bank of America Slock Plans 4.22 Bank of America Slock Plans 4.26 Bank of America Slock Plans 4.28 Bank of America Slock Plans 4.28 Bank of America Slock Plans 4.26 Bank of America Slock Plans 4.26 Bank of America Slock Plans 4.26 Bank	Assumed SAR	1.7(a)
Bank of America Preamble Bank of America (Log) 4.6 Bank of America (Capital Stock) 1.5 Bank of America Certificate 4.1(a) Bank of America Common Stock 1.4(a) Bank of America Converible Preferred 4.2(a) Bank of America Disclosure Schedule Art. IV Bank of America Disclosure Schedule 4.7(a) Bank of America Preferred Stock 4.2(a) Bank of America Reports 4.13 Bank of America Reports 4.11 Bank of America Reports 4.11 Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.6 Bank of America Stock Plans 3.1(b) Certificate of Merica Stock Plans 4.6 Bank of America Stock Plans 9.1 Coloc Certificate o	Assumed Stock-Based Award	1.7(b)
Bank of America 10-Q 4.6 Bank of America Capital Stock 1.5 Bank of America Capital Stock 1.4(a) Bank of America Cortificate 4.1(a) Bank of America Cortificate 4.2(a) Bank of America Cortificate 4.2(a) Bank of America Duclosure Schedule Art. IV Bank of America Duclosure Schedule 4.1(b) Bank of America Wirre Preferred Stock 4.2(a) Bank of America Regulatory Agreement 4.13 Bank of America Sequistory Agreement 4.11 Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.1(a) Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.1(a) Bank of America Stock Plans 4.1(a) Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.2(a)	Assumed Stock Option	1.7(a)
Bank of America Capital Stock 1.5 Bank of America Certificate 1.4(a) Bank of America Certificate 1.4(a) Bank of America Common Stock 4.2(a) Bank of America Disclosure Schedule Art. IV Bank of America Disclosure Schedule 1.4(b) Bank of America Mirror Preferred Stock 4.2(a) Bank of America Regulatory Agreement 4.13 Bank of America Regulatory Agreement 4.13 Bank of America Stock Plans 4.2(a) Bank of America Substidiary 3.1(c) Bank of America Substidiary 6.7(a) Bank of America Substidiary 4.6 Bank of America Substidiary 6.7(a) Bank of America Substidiary 6.7(a) Bank of America Substidiary 9.1 Cortificate of Merger 1.2 Cortificate of Merger 1.2 Closing 9.1 Closing of Plans 9.1 Closing Date 6.7(a) Confectual Evaluation Agreement 5.2(b) Covered Employees 6.7(a) DPC Common Stares 1.4(Bank of America	Preamble
Bank of America Certificate 4.1(a) Bank of America Common Stock 4.2(a) Bank of America Common Stock 4.7(a) Bank of America Disclosure Schedule Art. IV Bank of America Disclosure Schedule 1.4(b) Bank of America Mirror Preferred Stock 4.2(a) Bank of America Regulatory Agreement 4.13 Bank of America Regulatory Agreement 4.11 Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.6 Ben Off America Stock Plans 1.4 Certificate Off America Stock Plans 9.1	Bank of America 10-Q	4.6
Bank of America Common Stock 1.4(a) Bank of America Convertible Prefered 4.2(a) Bank of America Disclosure Schedule Art. IV Bank of America Disclosure Schedule 1.4(b) Bank of America Disclosure Schedule 1.4(b) Bank of America Disclosure Schedule 4.2(a) Bank of America Preferred Stock 4.13 Bank of America Reports 4.11 Bank of America Reports 4.11 Bank of America Stock Plans 4.2(a) Bank of America Subsidiary 3.1(c) Bank of America Subsidiary 4.6 Benefits Transition Date 6.7(a) Bank of America Subsidiary 1.4(c) Certificate 1.4(c) Certificate Offerger 1.2 Conflict Commens 9.1 Closing 9.1 Code Recitate Confidentiality Agreement 3.2 Covered Employees 6.7(a) DFC Common Shares 1.4(a) DFC Preferred Shares 1.4(b) DFC Shares 1.4(b) Effective Time	Bank of America Capital Stock	1.5
Bank of America Convertible Preferred 4.2(a) Bank of America Disclosures Chedule Art. IV Bank of America Disclosures Chedule 1.4(b) Bank of America Mirror Preferred Stock 4.2(a) Bank of America Repulatory Agreement 4.11 Bank of America Regulatory Agreement 4.11 Bank of America Repulatory Agreement 4.11 Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.1(a) Bank of America Stock Plans 4.6 Bank of America Stock Plans 3.1(b) Certificate 6.7(a) BIC Act 1.2 Closing 9.1 Closing 9.1 Closing 9.1 Closing Date 9.1 Co	Bank of America Certificate	4.1(a)
Bank of America Disclosure Schedule 31,40 Bank of America Mirror Preferred Stock 4.2(a) Bank of America Repreferred Stock 4.13 Bank of America Reports 4.11 Bank of America Reports 4.11 Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 3.1(c) Bank of America Stock Plans 4.6 Benefits Transition Date 6.7(a) BHC Act 3.1(b) Certificate 1.2 Certificate of Merger 1.2 Closing 9.1 Closing Plane 9.1 Code Recitals Code Recitals Cofficientiality Agreement 6.2(b) Convertible Note Agreement 3.2 Cowered Employees 6.7(a) DFC Preferred Shares 1.1(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Efficitive Time 1.2 Erschange Act 3.6 Exchange Act 3.6 Exchange Fund 2.1 <	Bank of America Common Stock	1.4(a)
Bank of America Mirror Preferred Stock 1.4(b) Bank of America Referred Stock 4.2(a) Bank of America Regulatory Agreement 4.13 Bank of America Regulatory Agreement 4.11 Bank of America Regulatory Agreement 4.2(a) Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 4.6 Benk of America Stock Plans 4.6 Bank of America Stock Plans 1.1 BC Act 3.2 Cortificator Divers 8.2 Cortificator Divers 9.1 Closing 9.1 Code 8.2 Confidentiality Agreement 6.2(b) Covered Employees 6.7(a) DPC Comerica Stock Plans 1.4(a) DPC Comeric		()
Bank of America Prefered Stock 4.2(a) Bank of America Regulatory Agreement 4.13 Bank of America Reports 4.11 Bank of America Stock Plans 4.2(a) Bank of America Subsidiary 3.1(c) Bank of America Subsidiary 4.6 Benefits Transition Date 6.7(a) BHC Act 3.1(b) Certificate 1.4(c) Certificate of Merger 1.2 Closing 9.1 Code Recitals Code Recitals Confidentiality Agreement 6.2(b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DPC Common Shares 1.4(a) DPC Common Shares 1.4(b) DPC Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.1(c) Exchange Act 3.6 Exchange Act 3.6 Exchange Agent 2.1 Exchange Ratio 3.4 FleetBoston Ho-Q		
Bank of America Reports 4.13 Bank of America Reports 4.11 Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 3.1(c) Bank of America Subsidiary 4.6 Benk of America 2002 10-K 4.1 Certificate of Merger 1.2 Cole Recitals Cole Recitals Confection of Merger 3.2 Conded Employees 6.7(a) Convertible Not Agreement 3.2 Covered Employees 1.4(a) DPC Conarcia Starcia 1.4(a)		
Bank of America Reports 4.11 Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 3.1(c) Bank of America 2002 10-K 4.6 Benefits Transition Date 6.7(a) BHC Act 3.1(b) Certificate 1.4(c) Certificate 1.2 Closing 9.1 Colosing 9.1 Code Recitals Confidentiality Agreement 6.2(b) Convertible Note Agreement 6.7(a) Course Employees 6.7(a) DGLC 1.1(a) DPC Preferred Shares 1.4(a) DPC Preferred Shares 1.4(b) DPC Preferred Shares 1.4(b) DPC Shares 1.2 ERISA 3.1(c) Erick Time 1.2 Exchange Age 3.6 Exchange Age 3.6 Exchange Age 3.1(c) Exchange Fund 2.1 Exchange Fund 3.4 Federal Reserve Board 3.4		
Bank of America Stock Plans 4.2(a) Bank of America Stock Plans 3.1(c) Bank of America 2002 10-K 4.6 Benefits Transition Date 6.7(a) BHC Act 3.1(b) Certificate 1.4(c) Certificate of Merger 1.2 Closing 9.1 Closing Date 9.1 Code Recitals Confidentiality Agreement 6.2(b) Covered Employees 6.7(a) DPC Common Shares 1.1(a) DPC Prefered Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.1(c) Exchange Acen 3.6 Exchange Agent 2.1 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 3.4 FledeIsotson Premble FledeIsotson Articles 3.1(b) FleetBoston Benefit Plans 3.1(b) FleetBoston Benefit Plans 3.1(b) FleetBoston Capital Stock <t< td=""><td></td><td></td></t<>		
Bank of America Subsidiary 3.1(c) Bank of America 2002 10-K 4.6 Benefits Transition Date 6.7(a) BHC Act 3.1(b) Certificate 1.4(c) Certificate 1.2 Closing 9.1 Colosing 9.1 Code Recitals Confidentiality Agreement 6.2(b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1(a) DFC Common Shares 1.4(a) DFC Prefered Shares 1.4(b) DFC Shares 1.4(b) Effective Time 1.2 ERISA 3.1(c) Exchange Agen 2.1 Exchange Agen 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) FleetBoston 3.4 FleetBoston Offices 3.4 FleetBoston Offices 3.1(b) FleetBoston Articles 3.1(b) FleetBoston Dylaws 3.1(b) FleetBo		
Bank of America 2002 10-K 4.6 Benefits Transition Date 6.7(a) BHC Act 3.1(b) Certificate 1.4(c) Certificate of Merger 1.2 Closing 9.1 Closing Date 8e.citals Confidentiality Agreement 6.2(b) Covertible Note Agreement 3.2 Covered Employees 6.7(a) DFL Common Shares 1.4(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 Exchange Act 3.1(c) Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) FleetBoston English 3.6 Exchange Ratio 2.1 FleetBoston Brofit Plans 3.6 FleetBoston In Q 3.6 FleetBoston FleetBoston FleetBoston Fleet Plans 3.1(b) FleetBoston Benefit Plans 3.1(b) FleetBoston Benefit Plans 3.1(b) FleetBoston Capital Stock 1.4(d) <td></td> <td></td>		
Benefits Transition Date 6.7(a) BHC Act 3.1(b) Certificate 1.4(c) Certificate of Merger 1.2 Closing 9.1 Cole 8ecitals Code 8ecitals Confidentiality Agreement 5.2 Convertible Note Agreement 3.2 Covered Employees 6.7(a) DFC Common Shares 1.1(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERNA 3.1(c) Exchange Act 3.6 Exchange Fund 2.1 Exchange Fund 3.4 Fleedral Reserve Board 3.4 Fleedral Reserve Board 3.4 FleetBoston Hicles 3.6 FleetBoston Hicles 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(b) FleetBoston Capital Stock 1.4(b)		
BHC Act 3.1(b) Certificate 1.4(c) Certificate of Merger 1.2 Closing 9.1 Closing Date 9.1 Code Recitals Conflectifility Agreement 6.2(b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1(a) DPC Common Shares 1.4(b) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 Exchange Act 3.1(c) Exchange Agent 2.1 Exchange Ratio 2.1 Fexchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston I0-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(a) FleetBoston Capital Stock 3.1(b)		
Certificate Of Merger 1.2 Closing Date 9.1 Code Recitals Confidentiality Agreement 6.2(b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1(a) DPC Common Shares 1.4(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.1(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 3.4 FleetBoston 3.6 FleetBoston Hereit Plans 3.6 FleetBoston It-Q 3.6 FleetBoston Benefit Plans 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(b) FleetBoston Capital Stock 1.4(b)	=	()
Certificate of Merger 1.2 Closing 9.1 Code Recitals Confidentiality Agreement 5.2 (b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DFLC 1.1(a) DPC Common Shares 1.4(a) DPC Prefered Shares 1.4(b) DPC Shares 1.2 Effective Time 1.2 ERISA 3.1(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) FeetBoston Pundle 3.4 FleetBoston Denoit Plans 3.6 FleetBoston Benefit Plans 3.6 FleetBoston Benefit Plans 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(b) FleetBoston Capital Stock 1.4(b)		()
Closing Date 9.1 Code Recitals Confidentiality Agreement 6.2(b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1(a) DPC Common Shares 1.4(a) DPC Prefered Shares 1.4(b) DPC Shares 1.2 ERISA 3.1(c) Exchange Act 3.1(c) Exchange Agent 2.1 Exchange Ratio 2.1 Exchange Ratio 3.4 Feleabsotn Peteboston 3.6 FleetBoston I0-Q 3.6 FleetBoston Benefit Plans 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(b) FleetBoston Capital Stock 3.1(b)		
Closing Date 9.1 Code Recitals Confidentiality Agreement 6.2(b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1(a) DPC Common Shares 1.4(b) DPC Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.1(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 3.4 Federal Reserve Board 3.4 FleetBoston I0-Q 3.6 FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(a) FleetBoston Benefit Plans 3.1(b) FleetBoston Capital Stock 3.1(b)	g and the same of	
Code Recitals Confidentiality Agreement 6.2 (b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1 (a) DPC Common Shares 1.4 (a) DPC Preferred Shares 1.4 (b) DPC Shares 1.4 (b) Effective Time 1.2 ERISA 3.1 (c) Exchange Act 3.1 (c) Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 2.1 Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston Ho-Q 3.6 FleetBoston Prices 3.1(b) FleetBoston Bylaws 3.1(a) FleetBoston Dylaws 3.1(b) FleetBoston Capital Stock 1.4(b)	- V	
Confidentiality Agreement 6.2(b) Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1(a) DPC Common Shares 1.4(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.1(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston 9.7 FleetBoston I0-Q 3.6 FleetBoston Benefit Plans 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)	Ç	
Convertible Note Agreement 3.2 Covered Employees 6.7(a) DGLC 1.1(a) DPC Common Shares 1.4(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.1(c) Exchange Act 2.1 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 2.1 Federal Reserve Board 3.4 FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.1(b) FleetBoston Bylaw 3.1(b) FleetBoston Capital Stock 1.4(b)		
Covered Employees 6.7(a) DGLC 1.1(a) DPC Common Shares 1.4(b) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.11(c) Exchange Act 3.6 Exchange Fund 2.1 Exchange Fund 2.1 Exchange Ratio 3.4 Federal Reserve Board 3.4 FleetBoston Articles 7 FleetBoston 10-Q 3.6 FleetBoston Benefit Plans 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
DGLC 1.1(a) DPC Common Shares 1.4(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.11(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 3.4 Federal Reserve Board 3.4 FleetBoston 10-Q 3.6 FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
DPC Common Shares 1.4(a) DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.11(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 3.4 Federal Reserve Board 3.4 FleetBoston 10-Q 3.6 FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
DPC Preferred Shares 1.4(b) DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.11(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
DPC Shares 1.4(b) Effective Time 1.2 ERISA 3.11(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
Effective Time 1.2 ERISA 3.11(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
ERISA 3.11(c) Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
Exchange Act 3.6 Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
Exchange Agent 2.1 Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.1(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
Exchange Fund 2.1 Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
Exchange Ratio 1.4(a) Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
Federal Reserve Board 3.4 FleetBoston Preamble FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		()
FleetBoston 10-Q 3.6 FleetBoston Articles 3.1(b) FleetBoston Benefit Plans 3.11(a) FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)	FleetBoston	Preamble
FleetBoston Benefit Plans FleetBoston Bylaws 3.1(a) FleetBoston Capital Stock 3.1(b) FleetBoston Capital Stock 1.4(b)	FleetBoston 10-Q	
FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)	FleetBoston Articles	3.1(b)
FleetBoston Bylaws 3.1(b) FleetBoston Capital Stock 1.4(b)		
FleetBoston Capital Stock 1.4(b)	FleetBoston Bylaws	
FleetBoston Common Stock 1.4(a)	FleetBoston Capital Stock	1.4(b)
	FleetBoston Common Stock	1.4(a)

	Section
FleetBoston Contract	3.14
FleetBoston Disclosure Schedule	Art. III
FleetBoston DRIP	1.7(d)
FleetBoston ERISA Affiliate	3.11(c)
FleetBoston ESPP	1.7(d)
FleetBoston Insiders	6.12
FleetBoston Preferred Stock	3.2
FleetBoston Regulatory Agreement	3.15
FleetBoston Reports	3.12
FleetBoston Rights Agreement	1.4(a)
FleetBoston SAR	1.7(a)
FleetBoston Stock-Based Award	1.7(b)
FleetBoston Stock Plans	1.7(a)
FleetBoston Stock Purchase Plans	1.7(d)
FleetBoston Stock Option	1.7(a)
FleetBoston Shareholder Rights	1.4(a)
FleetBoston Subsidiary	3.1(c)
FleetBoston 2002 10-K Form S-4	3.6 3.4
GAAP	
Governmental Entity	3.1(c) 3.4
HSR Act	3.4
Indemnified Parties	6.8
Injunction	7.1(e)
IRS	3.10(a)
Joint Proxy Statement	3.4
Liens	3.2(b)
Material Adverse Effect	3.1(a)
Merger	Recitals
Merger Consideration	1.1(b)
NYSE	2.2(e)
Other Regulatory Approvals	3.4
Regulatory Agencies	3.5
Requisite Regulatory Approvals	7.1(c)
RIBCA	1.1(a)
SBA	3.4
SEC	3.4
Section 16 Information	6.12
Securities Act	3.12
SRO	3.4
Stock Option Agreements	Recitals
Subsidiary	3.1(c)
Surviving Corporation	Recitals
Takeover Statute	3.19(a)
Tax(es) Trust Account Common Shares	3.10(b)
Trust Account Preferred Shares	1.4(a) 1.4(b)
Trust Account Shares Trust Account Shares	1.4(b) 1.4(b)
Tust Account States	1.4(0)

AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER, dated as of October 27, 2003 (this "Agreement"), by and between FLEETBOSTON FINANCIAL CORPORATION, a Rhode Island corporation ("FleetBoston"), and BANK OF AMERICA CORPORATION, a Delaware corporation ("Bank of America").

WITNESSETH:

WHEREAS, the Boards of Directors of FleetBoston and Bank of America have determined that it is in the best interests of their respective companies and their stockholders to consummate the strategic business combination transaction provided for in this Agreement in which FleetBoston will, on the terms and subject to the conditions set forth in this Agreement, merge with and into Bank of America (the "Merger"), so that Bank of America is the surviving corporation in the Merger (sometimes referred to in such capacity as the "Surviving Corporation"); and

WHEREAS, for Federal income Tax purposes, it is intended that the Merger shall qualify as a reorganization under the provisions of Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Code"), and this Agreement is intended to be and is adopted as a "plan of reorganization" for purposes of Sections 354 and 361 of the Code; and

WHEREAS, as an inducement and condition to Bank of America's entering into this Agreement, FleetBoston is granting to Bank of America an option pursuant to a stock option agreement in the form set forth in *Exhibit A* (the "FleetBoston Stock Option Agreement"), and as an inducement and condition to FleetBoston's entering into this Agreement, Bank of America is granting to FleetBoston an option pursuant to a stock option agreement in the form set forth in *Exhibit B* (the "Bank of America Stock Option Agreement"); and, together with the FleetBoston Stock Option Agreement, the "Stock Option Agreements"); and

WHEREAS, the parties desire to make certain representations, warranties and agreements in connection with the Merger and also to prescribe certain conditions to the Merger.

NOW, THEREFORE, in consideration of the mutual covenants, representations, warranties and agreements contained in this Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties agree as follows:

ARTICLE I

THE MERGER

- 1.1 The Merger. (a) Subject to the terms and conditions of this Agreement, in accordance with the Rhode Island Business Corporation Act (the "RIBCA") and the Delaware General Corporation Law (the "DGCL"), at the Effective Time, FleetBoston shall merge with and into Bank of America. Bank of America shall be the Surviving Corporation in the Merger, and shall continue its corporate existence under the laws of the State of Delaware. As of the Effective Time, the separate corporate existence of FleetBoston shall cease.
 - (b) The parties may at any time change the method of effecting the combination (including by providing for the merger of FleetBoston and a wholly owned subsidiary of Bank of America) if and to the extent requested by either party and consented to by the other party (such consent not to be unreasonably withheld); provided, however, that no such change shall (i) alter or change the amount or kind of consideration to be issued to holders of the capital stock of FleetBoston as provided for in this Agreement (the "Merger Consideration"), (ii) adversely affect the Tax treatment of FleetBoston's shareholders as a result of receiving the Merger Consideration or the Tax treatment of either party pursuant to this Agreement or (iii) materially impede or delay consummation of the transactions contemplated by this Agreement.

- 1.2 Effective Time. The Merger shall become effective as set forth in the articles of merger (the "Articles of Merger") that shall be filed with the Secretary of State of the State of Rhode Island and the certificate of merger (the "Certificate of Merger") that shall be filed with the Secretary of State of the State of Delaware on the Closing Date. The term "Effective Time" shall be the date and time when the Merger becomes effective as set forth in the Articles of Merger and the Certificate of Merger.
 - 1.3 Effects of the Merger. At and after the Effective Time, the Merger shall have the effects set forth in Section 7-1.1-69 of the RIBCA and Section 259 of the DGCL.
- 1.4 Conversion of FleetBoston Capital Stock. At the Effective Time, by virtue of the Merger and without any action on the part of Bank of America, FleetBoston or the holder of any of the following securities:
 - (a) Subject to Section 2.2(e), each share of the common stock, par value \$0.01 per share, of FleetBoston issued and outstanding immediately prior to the Effective Time (together with the preferred share purchase rights ("FleetBoston Shareholder Rights") attached thereto issued pursuant to that certain Rights Agreement dated as of August 16, 2000 (the "FleetBoston Rights Agreement"), between FleetBoston and EquiServe, LP, as Rights Agent, the "FleetBoston Common Stock"), except for shares of FleetBoston Common Stock owned by FleetBoston or Bank of America (other than shares of FleetBoston Common Stock held in trust accounts, managed accounts and the like, or otherwise held in a fiduciary or agency capacity, that are beneficially owned by third parties (any such shares, "Trust Account Common Shares") and other than shares of FleetBoston Common Stock held, directly or indirectly, by FleetBoston or Bank of America in respect of a debt previously contracted (any such shares, "DPC Common Shares")), shall be converted into the right to receive 0.5553 of a share (the "Exchange Ratio") of the common stock, par value \$0.01 per share, of Bank of America (the "Bank of America Common Stock").
 - (b) Each share of FleetBoston Preferred Stock (as defined in Section 3.2) (together with the FleetBoston Common Stock, the "FleetBoston Capital Stock") issued and outstanding immediately prior to the Effective Time, except for shares of FleetBoston Preferred Stock owned by FleetBoston or Bank of America (other than shares of FleetBoston Preferred Stock held by FleetBoston or Bank of America in trust accounts, managed accounts and the like, or otherwise held in a fiduciary or agency capacity, that are beneficially owned by third parties (any such shares, "Trust Account Preferred Shares," and together with the Trust Account Common Shares, the "Trust Account Shares," and other than any shares of FleetBoston Preferred Stock held by FleetBoston or Bank of America in respect of a debt previously contracted (any such shares, "DPC Preferred Shares," and together with the DPC Common Shares, the "DPC Shares")), shall be converted, without any action on the part of the holder thereof, into one share of a newly-designated series of Bank of America Preferred Stock (as defined in Section 4.2) having the same terms (to the fullest extent possible) as the share of FleetBoston Preferred Stock so converted (the "Bank of America Mirror Preferred Stock").
 - (c) All of the shares of FleetBoston Capital Stock converted into the right to receive Bank of America Capital Stock (as defined in Section 1.5) pursuant to this Article I shall no longer be outstanding and shall automatically be cancelled and shall cease to exist as of the Effective Time, and (i) each certificate previously representing any such shares of FleetBoston Common Stock (each a "Certificate") shall thereafter represent only the right to receive (A) a certificate representing the number of whole shares of Bank of America Common Stock and (B) cash in lieu of fractional shares into which the shares of FleetBoston Common Stock represented by such Certificate have been converted pursuant to this Section 1.4 and Section 2.2(e) and (ii) each certificate representing FleetBoston Preferred Stock shall, as of and following the Effective Time, remain outstanding and shall continue to represent the number of shares of Bank of America Mirror Preferred Stock into which such FleetBoston Preferred Stock shall have been converted pursuant to Section 1.4(b) hereof. Certificates previously representing shares of FleetBoston Common Stock shall be exchanged for certificates representing whole shares of Bank of America Common Stock and cash in lieu of fractional shares issued in consideration therefor upon the surrender of such Certificates in accordance with Section 2.2, without any interest thereon. If, prior to the Effective Time, the

outstanding shares of Bank of America Common Stock or FleetBoston Common Stock shall have been increased, decreased, changed into or exchanged for a different number or kind of shares or securities as a result of a reorganization, recapitalization, reclassification, stock dividend, stock split, reverse stock split, or other similar change in capitalization, an appropriate and proportionate adjustment shall be made to the Exchange Ratio.

- (d) Notwithstanding anything in the Agreement to the contrary, at the Effective Time, all shares of FleetBoston Capital Stock that are owned by FleetBoston or Bank of America (other than Trust Account Shares and DPC Shares) shall be cancelled and shall cease to exist and no stock of Bank of America or other consideration shall be delivered in exchange therefor.
- 1.5 Bank of America Capital Stock. At and after the Effective Time, each share of Bank of America Common Stock and each share of Bank of America Preferred Stock (the Bank of America Common Stock and the Bank of America Preferred Stock, the "Bank of America Capital Stock") issued and outstanding immediately prior to the Effective Time shall remain issued and outstanding and shall not be affected by the Merger.
- 1.6 Bank of America Mirror Preferred Stock. Prior to the Effective Time, Bank of America shall take all corporate action necessary to designate the rights and preferences of the Bank of America Mirror Preferred Stock.
- 1.7 Options and SARs; Other Stock-Based Awards. (a) Effective as of the Effective Time, each then outstanding option to purchase shares of FleetBoston Common Stock (each a "FleetBoston Stock Option") and stock appreciation right to purchase shares of FleetBoston Common Stock or receive a cash payment in respect of FleetBoston Common Stock (each a "FleetBoston SAR") granted, pursuant to the equity-based compensation plans identified on Schedule 3.11(d) (the "FleetBoston Stock Plans"), to any current or former employee or director of, or consultant to, FleetBoston or any of its subsidiaries shall be assumed by Bank of America and shall be converted into an option or a stock appreciation right to purchase a number of shares of Bank of America Common Stock (rounded to the nearest whole share) or in the case of an FleetBoston SAR settled in cash to receive a cash payment in respect of a share of Bank of America Common Stock (rounded to the nearest whole share) (an "Assumed Stock Option" or "Assumed SAR", as the case may be) equal to (i) the number of shares of FleetBoston Common Stock subject to such FleetBoston Stock Option or FleetBoston SAR immediately prior to the Effective Time multiplied by (ii) the Exchange Ratio; and the per share exercise price for Bank of America Common Stock issuable (or payable in the case of a FleetBoston SAR settled in cash) upon the exercise of such Assumed Stock Option or Assumed SAR shall be equal to (i) the exercise price per share of FleetBoston Common Stock at which such FleetBoston Stock Option or FleetBoston SAR was exercisable immediately prior to the Effective Time divided by (ii) the Exchange Ratio (rounded to the nearest whole cent); provided, however, that in the case of any FleetBoston Stock Option to which Section 421 of the Code applies by reason of its qualification under Section 422 of the Code, the conversion formula shall be adjusted, if necessary, to comply with Section 424(a) of the Code. Except as otherwise provided herein, the Assumed Stock Options and Assumed SARs shall be subject to the same terms and conditions (including expiration date, vesting and exercise provisions) as were applicable to the corresponding FleetBoston Stock Options and FleetBoston SARs immediately prior to the Effective Time (but taking into account any changes thereto, including the acceleration thereof, provided for in the FleetBoston Stock Plans or other FleetBoston Benefit Plan or in any award agreement thereunder by reason of this Agreement or the transactions contemplated hereby); provided, however, that references to FleetBoston shall be deemed to be references to Bank of America.
 - (b) At the Effective Time, each right of any kind, contingent or accrued, to receive shares of FleetBoston Common Stock or benefits measured by the value of a number of shares of FleetBoston Common Stock, and each award of any kind consisting of shares of FleetBoston Common Stock, granted under the FleetBoston Stock Plans or any other FleetBoston Benefit Plan (including restricted stock, restricted stock units, performance stock units, deferred stock units and dividend equivalents), other than FleetBoston Stock Options and FleetBoston SARs (each, a "FleetBoston Stock-Based Award"), whether vested or unvested, which is outstanding or unsatisfied immediately prior to the Effective Time, shall cease

to represent a right or award with respect to shares of FleetBoston Common Stock and shall be converted, at the Effective Time, into a right or award with respect to shares of Bank of America Common Stock (an "Assumed Stock-Based Award"), on the same terms and conditions (including expiration date, vesting and exercise provisions) as were applicable under the FleetBoston Stock-Based Awards (but taking into account any changes thereto, including the acceleration thereof, provided for in the FleetBoston Stock Plans or other FleetBoston Benefit Plan or in any award agreement thereunder by reason of this Agreement or the transactions contemplated hereby). The number of shares of Bank of America Common Stock subject to each such Assumed Stock-Based Award shall be equal to the number of shares of FleetBoston Common Stock subject to the FleetBoston Stock-Based Award, multiplied by the Exchange Ratio (rounded to the nearest whole share of Bank of America Common Stock). All dividend equivalents credited to the account of each holder of a FleetBoston Stock-Based Award as of the Effective Time shall remain credited to such holder's account immediately following the Effective Time, subject to adjustment in accordance with the foregoing.

- (c) Bank of America has taken all corporate actions necessary to reserve for issuance a sufficient number of shares of Bank of America Common Stock upon the exercise of the Assumed Stock Options and Assumed SARs and the settlement or distribution of the Assumed Stock-Based Awards. On or as soon as practicable following the Closing Date (and in no event more than five business days after the Closing Date), Bank of America shall file a registration statement on an appropriate form or a post-effective amendment to a previously filed registration statement under the Securities Act with respect to the issuance of (or settlement in cash in respect of) the shares of Bank of America Common Stock subject to the Assumed Stock Options, the Assumed SARs and the Assumed Stock-Based Awards and shall use its reasonable efforts to maintain the effectiveness of such registration statement or registration statements (and maintain the current status of the prospectus or prospectuses contained therein) for so long as such equity awards remain outstanding.
- (d) FleetBoston shall take such action as is necessary to (i) cause the exercise (as of a date that is no later than three business days prior to the Effective Date) of each outstanding purchase right under the FleetBoston Employee Stock Purchase Plan (the "FleetBoston ESPP"); and (ii) provide that no further purchase period shall commence under the FleetBoston ESPP following such date; provided, however, that such exercise and cessation of further purchase periods shall be conditioned upon the consummation of the Merger. On such new exercise date, FleetBoston shall apply the funds credited as of such date under the FleetBoston ESPP within each participant's payroll withholding account to the purchase of shares of FleetBoston Common Stock in accordance with the terms of the FleetBoston ESPP. In addition, FleetBoston shall take such action as is necessary to provide that as of no later than three business days prior to the Effective Date no further shares of FleetBoston Common Stock will be purchased under the FleetBoston Automatic Dividend Reinvestment Plan (the "FleetBoston DRIP" and, together with the FleetBoston ESPP, the "FleetBoston Stock Purchase Plans"); provided, however, that such cessation of further purchases shall be conditioned upon the consummation of the Merger. Immediately prior to and effective as of the Effective Time and subject to the consummation of the Merger, FleetBoston shall terminate the FleetBoston Stock Purchase Plans.
- 1.8 Certificate of Incorporation of Bank of America. At the Effective Time, the Bank of America Certificate (as defined in Section 4.1) shall be the certificate of incorporation of the Surviving Corporation until thereafter amended in accordance with applicable law.
- 1.9 Bylaws of Bank of America. At the Effective Time, the Bank of America Bylaws shall be the Bylaws of the Surviving Corporation until thereafter amended in accordance with applicable law.
- 1.10 Tax Consequences. It is intended that the Merger shall constitute a "reorganization" within the meaning of Section 368(a) of the Code, and that this Agreement shall constitute a "plan of reorganization" for purposes of Sections 354 and 361 of the Code.

- 1.11 Certain Post-Closing Matters.
 - (a) Board Composition. As of the Effective Time, and continuing for a period of at least two years following the Effective Time:
 - (i) the ratio of Continuing Bank of America Directors to Continuing FleetBoston Directors serving on the Board of Directors of the Surviving Corporation shall be maintained at 12 to 7:
 - (ii) all vacancies on the Board of Directors of the Surviving Corporation created by the cessation of service of a Continuing Bank of America Director shall be filled by a nominee proposed to the nominating committee of the Board of Directors of the Surviving Corporation by a majority of the remaining Continuing Bank of America Directors; and
 - (iii) all vacancies on the Board of Directors of the Surviving Corporation created by the cessation of service of a Continuing FleetBoston Director shall be filled by a nominee proposed to the nominating committee of the Board of Directors of the Surviving Corporation by a majority of the remaining Continuing FleetBoston Directors.
 - (b) Commitment to the Communities of FleetBoston. From and after the Effective Time:
 - (i) The headquarters of the Surviving Corporation's Wealth Management, Latin American, Asset Based Lending, Small Business and Premier Banking, and Leasing businesses shall each be located in the Boston, Massachusetts metropolitan area;
 - (ii) The Surviving Corporation is committed to maintaining current employment levels of FleetBoston in New England, it being the intention of the Surviving Corporation to increase such employment levels over time at a level equal to the growth in employment of the Surviving Corporation's United States workforce: and
 - (iii) The Surviving Corporation will maintain FleetBoston's annual level of charitable giving in each of New England and the other regions within the FleetBoston footprint, respectively, at levels no less than the levels set in 2003, it being the intention of the Surviving Corporation to increase such giving in those areas over time in a manner equal to the growth in the Surviving Corporation's annual charitable giving in its other principal markets.
- (c) Definition. The terms "Continuing Bank of America Directors" and "Continuing FleetBoston Directors" shall for purposes of this Section 1.11 mean, respectively, the directors of Bank of America or FleetBoston, as the case may be, as of the Effective Time who were selected to be directors of the Surviving Corporation by Bank of America or FleetBoston, as the case may be, prior to the Effective Time, and any additional directors of the Surviving Corporation who take office after the Effective Time who are nominated, or proposed to the nominating committee of the Board of Directors of the Surviving Corporation, by a majority of the Continuing Bank of America Directors or the Continuing FleetBoston Directors, as the case may be.
- (d) Survival/Adoption of Commitments. The commitments set forth in this Section 1.11 shall survive the Effective Time as reflected in a formal resolution of the Board of Directors of the Surviving Corporation to be reflected in the minutes of the Surviving Corporation following the Merger.

ARTICLE II

EXCHANGE OF SHARES

2.1 Bank of America to Make Shares Available. At or prior to the Effective Time, Bank of America shall deposit, or shall cause to be deposited, with a bank or trust company Subsidiary of Bank of America, or another bank or trust company reasonably acceptable to each of FleetBoston and Bank of America (the "Exchange Agent"), for the benefit of the holders of Certificates, for exchange in accordance with this Article II, certificates

representing the shares of Bank of America Common Stock, and cash in lieu of any fractional shares (such cash and certificates for shares of Bank of America Common Stock, together with any dividends or distributions with respect thereto, being referred to as the "Exchange Fund"), to be issued pursuant to Section 1.4 and paid pursuant to Section 2.2(e) in exchange for outstanding shares of FleetBoston Common Stock.

- 2.2 Exchange of Shares. (a) As soon as practicable after the Effective Time, the Exchange Agent shall mail to each holder of record of one or more Certificates a letter of transmittal in customary form as reasonably agreed to by the parties (which shall specify, among other things, that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates to the Exchange Agent) and instructions for use in effecting the surrender of the Certificates in exchange for certificates representing the shares of Bank of America Common Stock and any cash in lieu of fractional shares into which the shares of FleetBoston Common Stock represented by such Certificate or Certificates shall have been converted pursuant to this Agreement. Upon proper surrender of a Certificate or Certificates for exchange and cancellation to the Exchange Agent, together with such properly completed letter of transmittal, duly executed, the holder of such Certificates shall be entitled to receive in exchange therefor, as applicable, (i) a certificate representing the number of whole shares of Bank of America Common Stock to which such holder of FleetBoston Common Stock shall have become entitled pursuant to the provisions of Article I, (ii) a check representing the amount of any cash in lieu of fractional shares which such holder has the right to receive in respect of the Certificate or Certificates surrendered pursuant to the provisions of this Article II, and (iii) a check representing the amount of any dividends or distributions then payable pursuant to Section 2.2(b)(i), and the Certificate or Certificates so surrendered shall forthwith be cancelled. No interest will be paid or accrued on any cash in lieu of fractional shares or on any unpaid dividends and distributions payable to holders of Certificates.
 - (b) No dividends or other distributions declared with respect to Bank of America Common Stock shall be paid to the holder of any unsurrendered Certificate until the holder thereof shall surrender such Certificate in accordance with this *Article II*. After the surrender of a Certificate in accordance with this *Article II*, the record holder thereof shall be entitled to receive (i) the amount of dividends or other distributions with a record date after the Effective Time theretofore paid, without any interest thereon, with respect to the whole shares of Bank of America Common Stock represented by such Certificate and (ii), at the appropriate payment date, the amount of dividends or other distributions with a record date after the Effective Time but prior to surrender and a payment date subsequent to surrender, with respect to shares of Bank of America Common Stock represented by such Certificate.
 - (c) If any certificate representing shares of Bank of America Common Stock is to be issued in a name other than that in which the Certificate or Certificates surrendered in exchange therefor is or are registered, it shall be a condition to the issuance thereof that the Certificate or Certificates so surrendered shall be properly endorsed (or accompanied by an appropriate instrument of transfer) and otherwise in proper form for transfer, and that the person requesting such exchange shall pay to the Exchange Agent in advance any transfer or other Taxes required by reason of the issuance of a certificate representing shares of Bank of America Common Stock in any name other than that of the registered holder of the Certificate or Certificates surrendered, or required for any other reason, or shall establish to the satisfaction of the Exchange Agent that such Tax has been paid or is not payable.
 - (d) After the Effective Time, there shall be no transfers on the stock transfer books of FleetBoston of the shares of FleetBoston Capital Stock that were issued and outstanding immediately prior to the Effective Time other than to settle transfers of FleetBoston Common Stock that occurred prior to the Effective Time. If, after the Effective Time, Certificates representing such shares are presented for transfer to the Exchange Agent, they shall be cancelled and exchanged for certificates representing shares of Bank of America Common Stock as provided in this *Article II*.
 - (e) Notwithstanding anything to the contrary contained in this Agreement, no certificates or scrip representing fractional shares of Bank of America Common Stock shall be issued upon the surrender of Certificates for exchange, no dividend or distribution with respect to Bank of America Common Stock shall be payable on or with respect to any fractional share, and such fractional share interests shall not entitle the owner

thereof to vote or to any other rights of a stockholder of Bank of America. In lieu of the issuance of any such fractional share, Bank of America shall pay to each former stockholder of FleetBoston who otherwise would be entitled to receive such fractional share an amount in cash (rounded to the nearest cent) determined by multiplying (i) the average of the closing-sale prices of Bank of America Common Stock on the New York Stock Exchange, Inc. (the "NYSE") Composite Transactions Tape as reported by The Wall Street Journal for the five full NYSE trading days immediately preceding the date of the Effective Time by (ii) the fraction of a share (rounded to the nearest thousandth when expressed in decimal form) of Bank of America Common Stock to which such holder would otherwise be entitled to receive pursuant to Section 1.4.

(f) Any portion of the Exchange Fund that remains unclaimed by the stockholders of FleetBoston as of the first anniversary of the Effective Time shall be paid to Bank of America. Any former stockholders of FleetBoston who have not theretofore complied with this Article II shall thereafter look only to Bank of America for payment of the shares of Bank of America Common Stock, cash in lieu of any fractional shares and any unpaid dividends and distributions on the Bank of America Common Stock deliverable in respect of each share of FleetBoston Common Stock, as the case may be, such stockholder holds as determined pursuant to this Agreement, in each case, without any interest thereon. Notwithstanding the foregoing, none of Bank of America, FleetBoston, the Exchange Agent or any other person shall be liable to any former holder of shares of FleetBoston Common Stock for any amount delivered in good faith to a public official pursuant to applicable abandoned property, escheat or similar laws.

(g) In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if reasonably required by Bank of America, the posting by such person of a bond in such amount as Bank of America may determine is reasonably necessary as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent will issue in exchange for such lost, stolen or destroyed Certificate the shares of Bank of America Common Stock and any cash in lieu of fractional shares deliverable in respect thereof pursuant to this Agreement.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF FLEETBOSTON

Except as disclosed in (x) the FleetBoston Reports (as defined in Section 3.12) filed prior to the date hereof or (y) the disclosure schedule (the 'FleetBoston Disclosure Schedule'') delivered by FleetBoston to Bank of America prior to the execution of this Agreement (which schedule sets forth, among other things, items the disclosure of which is necessary or appropriate either in response to an express disclosure requirement contained in a provision hereof or as an exception to one or more representations or warranties contained in this Article III, or to one or more of FleetBoston's covenants contained in Article V, provided, however, that, notwithstanding anything in this Agreement to the contrary, (i) no such item is required to be set forth in such schedule as an exception to a representation or warranty if its absence would not result in the related representation or warranty being deemed untrue or incorrect under the standard established by Section 9.2, and (ii) the mere inclusion of an item in such schedule as an exception to a representation or warranty shall not be deemed an admission that such item represents a material exception or material fact, event or circumstance or that such item has had or would be reasonably likely to have a Material Adverse Effect (as defined in Section 3.1) on FleetBoston), FleetBoston hereby represents and warrants to Bank of America as follows:

3.1 Corporate Organization.

(a) FleetBoston is a corporation duly organized, validly existing and in good standing under the laws of the State of Rhode Island. FleetBoston has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the

character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not, either individually or in the aggregate, have a Material Adverse Effect on FleetBoston. As used in this Agreement, the term "Material Adverse Effect" means, with respect to Bank of America, FleetBoston or the Surviving Corporation, as the case may be, a material adverse effect on (i) the business, results of operations or financial condition of such party and its Subsidiaries (as defined below) taken as a whole (provided, however, that, with respect to this clause (i), Material Adverse Effect shall not be deemed to include effects to the extent resulting from (a) changes, after the date hereof, in generally accepted accounting principles or regulatory accounting requirements applicable to banks or savings associations and their holding companies generally, (b) changes, after the date hereof, in laws, rules or regulations of general applicability or interpretations thereof by courts or Governmental Entities (as defined in Section 3.4), (c) actions or omissions of Bank of America or FleetBoston taken with the prior written consent of the other or required hereunder, (d) changes, after the date hereof, in general economic or market conditions affecting banks or their holding companies generally or (e) public disclosure of the transactions contemplated hereby), or (ii) the ability of such party to timely consummate the transactions contemplated by this Agreement.

- (b) FleetBoston is duly registered as a financial holding company under the Bank Holding Company Act of 1956, as amended (the "BHC Act"). True and complete copies of the Restated Articles of Incorporation of FleetBoston (the "FleetBoston Articles") and the Bylaws of FleetBoston (the "FleetBoston Bylaws"), as in effect as of the date of this Agreement, have previously been made available to Bank of America.
- (c) Each of FleetBoston's Subsidiaries (i) is duly organized and validly existing under the laws of its jurisdiction of organization, (ii) is duly qualified to do business and in good standing in all jurisdictions (whether federal, state, local or foreign) where its ownership or leasing of property or the conduct of its business requires it to be so qualified and (iii) has all requisite corporate power and authority to own or lease its properties and assets and to carry on its business as now conducted, except in each of (i) (iii) as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston. As used in this Agreement, the word "Subsidiary" when used with respect to either party, means any bank, corporation, partnership, limited liability company or other organization, whether incorporated or unincorporated, that is consolidated with such party for financial reporting purposes under U.S. generally accepted accounting principles ("GAAP"), and the terms "FleetBoston Subsidiary" and "Bank of America Subsidiary" shall mean any direct or indirect Subsidiary of FleetBoston or Bank of America, respectively.
- 3.2 Capitalization. (a) The authorized capital stock of FleetBoston consists of 2,000,000,000 shares of FleetBoston Common Stock, of which, as of October 24, 2003, 1,052,745,101 shares were issued and outstanding, and 16,000,000 shares of preferred stock, \$1.00 par value ("FleetBoston Preferred Stock"), of which, as of October 24, 2003, (i) 690,000 shares were authorized and 382,450 shares were issued and outstanding as FleetBoston Series VI 6.75% Perpetual Preferred Stock, and (ii) 805,000 shares were authorized and 700,000 shares were issued and outstanding as Series VII 6.60% Cumulative Preferred Stock. As of October 24, 2003, no more than 34,196,003 shares of FleetBoston Common Stock were held in FleetBoston's treasury. As of the date hereof, no shares of FleetBoston Capital Stock were reserved for issuance except for (i) shares of FleetBoston Common Stock reserved for issuance upon the exercise of FleetBoston Stock Options or FleetBoston SARs issued pursuant to the FleetBoston Stock Plans, (ii) 186,336 shares reserved for issuance pursuant to a convertible note agreement (the "Convertible Note Agreement"), (iii) 500,000 shares of Cumulative Participating Junior Preferred Stock, \$1.00 par value, reserved for issuance pursuant to the FleetBoston Stock Option Agreement. All of the issued and outstanding shares of FleetBoston Capital Stock have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights, with no personal liability attaching to the ownership thereof. As of the date of this Agreement, except pursuant to this Agreement, the FleetBoston Stock Option Agreement, the FleetBoston Stock Plans, the FleetBoston Rights Agreements and the Convertible Note Agreement, FleetBoston does not have and is not bound by any

outstanding subscriptions, options, warrants, calls, commitments or agreements of any character calling for the purchase or issuance of any shares of FleetBoston Capital Stock or any other equity securities of FleetBoston or any securities representing the right to purchase or otherwise receive any shares of FleetBoston Capital Stock. FleetBoston has provided Bank of America with a list of (i) the aggregate number of options outstanding under the FleetBoston Stock Plans as of October 21, 2003 and the exercise price for such FleetBoston Stock Options and FleetBoston SARs, (ii) the aggregate number of cash-settled SARs and stock-settled SARs outstanding under the FleetBoston Stock Plans as of October 21, 2003 and (iii) the aggregate number of all FleetBoston Stock-Based Awards outstanding as of October 21, 2003; since October 21, 2003 through the date hereof, FleetBoston has not issued or awarded any options, restricted stock, restricted stock units or SARs under the FleetBoston Stock Plans.

- (b) All of the issued and outstanding shares of capital stock or other equity ownership interests of each "significant subsidiary" (as such term is defined under Regulation S-X of the SEC (as defined in Section 3.4)) of FleetBoston are owned by FleetBoston, directly or indirectly, free and clear of any material liens, pledges, charges and security interests and similar encumbrances ("Liens"), and all of such shares or equity ownership interests are duly authorized and validly issued and are fully paid, nonassessable (subject to 12 U.S.C. §§ 55) and free of preemptive rights. No such significant subsidiary has or is bound by any outstanding subscriptions, options, warrants, calls, commitments or agreements of any character calling for the purchase or issuance of any shares of capital stock or any other equity security of such subsidiary.
- 3.3 Authority; No Violation. (a) FleetBoston has full corporate power and authority to execute and deliver this Agreement and the Stock Option Agreements and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Stock Option Agreements and the consummation of the transactions contemplated hereby have been duly and validly approved by the Board of Directors of FleetBoston. The Board of Directors of FleetBoston has determined that this Agreement and the transactions contemplated hereby are in the best interests of FleetBoston and its shareholders and has directed that this Agreement and the transactions contemplated by this Agreement be submitted to FleetBoston's stockholders for adoption at a duly held meeting of such shareholders and, except for the approval of this Agreement and the transactions contemplated by this Agreement by the affirmative vote of the holders of a majority of the outstanding shares of FleetBoston Common Stock entitled to vote at such meeting, no other corporate proceedings on the part of FleetBoston are necessary to approve this Agreement or the Stock Option Agreements have been duly and validly executed and delivered by FleetBoston and (assuming due authorization, execution and delivery by Bank of America) constitute the valid and binding obligations of FleetBoston, enforceable against FleetBoston in accordance with their terms (except as may be limited by bankruptcy, insolvency, moratorium, reorganization or similar laws affecting the rights of creditors generally and the availability of equitable remedies).
 - (b) Neither the execution and delivery of this Agreement or the Stock Option Agreements by FleetBoston nor the consummation by FleetBoston of the transactions contemplated hereby or thereby, nor compliance by FleetBoston with any of the terms or provisions of this Agreement or the Stock Option Agreements, will (i) violate any provision of the FleetBoston Articles or the FleetBoston Bylaws or (ii) assuming that the consents, approvals and filings referred to in *Section 3.4* are duly obtained and/or made, (A) violate any statute, code, ordinance, rule, regulation, judgment, order, writ, decree or Injunction (as defined in 7.1(e)) applicable to FleetBoston, any of its Subsidiaries or any of their respective properties or assets or (B) violate, conflict with, result in a breach of any provision of or the loss of any benefit under, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation of any Lien upon any of the respective properties or assets of FleetBoston or any of its Subsidiaries under, any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which

FleetBoston or any of its Subsidiaries is a party, or by which they or any of their respective properties or assets may be bound or affected, except for such violations, conflicts, breaches or defaults that are not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston.

- 3.4 Consents and Approvals. Except for (i) the filing of applications and notices, as applicable, with the Board of Governors of the Federal Reserve System (the Federal Reserve Board") under the BHC Act and the Federal Reserve Act, as amended, and approval of such applications and notices, (ii) the filing of any required applications or notices with any foreign or state banking, insurance or other regulatory authorities and approval of such applications and notices (the "Other Regulatory Approvals"), (iii) the filing with the Securities and Exchange Commission (the "SEC") of a Joint Proxy Statement in definitive form relating to the meetings of FleetBoston's and Bank of America's shareholders to be held in connection with this Agreement and the transactions contemplated by this Agreement (the "Joint Proxy Statement") and of a registration statement on Form S-4 (the "Form S-4") in which the Joint Proxy Statement will be included as a prospectus, and declaration of effectiveness of the Form S-4, (iv) the filing of the Articles of Merger with the Rhode Island Secretary pursuant to the RIBCA and the issuance by the Rhode Island Secretary of a Certificate of Merger and the filing of the Certificate of Merger with the Secretary of State of the State of Delaware pursuant to the DGCL, (v) any notices to or filings with the Small Business Administration (the "SBA"), (vi) any notices or filings under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), (vii) any consents, authorizations, approvals, filings or exemptions in connection with compliance with the applicable provisions of federal and state securities laws relating to the regulation of broker-dealers, investment advisers or transfer agents and federal commodities laws relating to the regulation of futures commission merchants and the rules and regulations thereunder and of any applicable industry self-regulatory organization ("SRO"), and the rules of the NYSE, or that are required under consumer finance, mortgage banking and other similar laws, (viii) such filings and approvals as are required to be made or obtained under the securities or "Blue Sky" laws of various states in connection with the issuance of the shares of Bank of America Capital Stock pursuant to this Agreement, (ix) the approval of this Agreement by the requisite vote of stockholders of FleetBoston and (x) filings, if any, required as a result of the particular status of Bank of America, no consents or approvals of or filings or registrations with any court, administrative agency or commission or other governmental authority or instrumentality (each a "Governmental Entity") are necessary in connection with (A) the execution and delivery by FleetBoston of this Agreement or the Stock Option Agreements and (B) the consummation by FleetBoston of the Merger and the other transactions contemplated by this Agreement and the Stock Option Agreements.
- 3.5 Reports. FleetBoston and each of its Subsidiaries have timely filed all reports, registrations and statements, together with any amendments required to be made with respect thereto, that they were required to file since January 1, 2000 with (i) the Federal Reserve Board, (ii) the Federal Deposit Insurance Corporation, (iii) any state regulatory authority, (iv) the SEC, (v) any foreign regulatory authority and (vi) any SRO (collectively, "Regulatory Agencies"), and all other reports and statements required to be filed by them since January 1, 2000, including any report or statement required to be filed pursuant to the laws, rules or regulations of the United States, any state, any foreign entity, or any Regulatory Agency, and have paid all fees and assessments due and payable in connection therewith, except where the failure to file such report, registration or statement or to pay such fees and assessments is not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston. Except for normal examinations conducted by a Regulatory Agency in the ordinary course of the business of FleetBoston and its Subsidiaries, no Regulatory Agency has initiated or has pending any proceedings or, to the knowledge of FleetBoston, investigation into the business or operations of FleetBoston or any of its Subsidiaries since January 1, 2000, except where such proceedings or investigation are not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston. There (i) is no unresolved violation, criticism, or exception by any Regulatory Agency with respect to any report or statement relating to any examinations or inspections of FleetBoston or any of its Subsidiaries and (ii) has been no formal or informal inquiries by, or disagreements or disputes with, any Regulatory Agency with respect to the business, operations, policies or procedures of FleetBoston since January 1, 2000, that are reasonably likely to have, either individually or i

- 3.6 Financial Statements. FleetBoston has previously made available to Bank of America copies of (i) the consolidated balance sheet of FleetBoston and its Subsidiaries as of December 31, 2000, 2001 and 2002, and the related consolidated statements of income, changes in stockholders' equity and cash flows for the years then ended as reported in FleetBoston's Annual Report on Form 10-K for the fiscal year ended December 31, 2002 (the "FleetBoston 2002 10-K") filed with the SEC under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), accompanied by the audit report of PricewaterhouseCoopers LLP, independent public accountants with respect to FleetBoston, and (ii) the unaudited consolidated balance sheet of FleetBoston and its Subsidiaries as of June 30, 2002 and 2003, and the related consolidated statements of income, changes in stockholders' equity and cash flows of the six month periods then ended, as reported in FleetBoston's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 2003 (the "FleetBoston 10-Q"). The December 31, 2002 consolidated balance sheet of FleetBoston (including the related notes, where applicable) fairly presents in all material respects the consolidated financial position of FleetBoston and its Subsidiaries as of the date thereof, and the other financial statements referred to in this Section 3.6 (including the related notes, where applicable) fairly present in all material respects the results of the consolidated operations and changes in stockholders' equity and consolidated financial position of FleetBoston and its Subsidiaries for the respective fiscal periods or as of the respective dates therein set forth, subject to normal year-end audit adjustments in amounts consistent with past experience in the case of unaudited statements; each of such statements (including the related notes, where applicable) complies in all material respects with applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto;
- 3.7 Broker's Fees. Neither FleetBoston nor any FleetBoston Subsidiary nor any of their respective officers or directors has employed any broker or finder or incurred any liability for any broker's fees, commissions or finder's fees in connection with the Merger or related transactions contemplated by this Agreement, other than Morgan Stanley & Co., Inc., the material terms of whose engagement have been disclosed to Bank of America.
- 3.8 Absence of Certain Changes or Events. (a) Except as publicly disclosed in the FleetBoston Reports filed prior to the date of this Agreement, since December 31, 2002, no event or events have occurred that have had or are reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston.
 - (b) Except as publicly disclosed in the FleetBoston Reports filed prior to the date of this Agreement, since December 31, 2002 through and including the date of this Agreement, FleetBoston and its Subsidiaries have carried on their respective businesses in all material respects in the ordinary course.
- 3.9 Legal Proceedings. (a) None of FleetBoston or any of its Subsidiaries is a party to any, and there are no pending or, to the best of FleetBoston's knowledge, threatened, legal, administrative, arbitral or other proceedings, claims, actions or governmental or regulatory investigations of any nature against FleetBoston or any of its Subsidiaries except as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston.
 - (b) There is no Injunction, judgment, or regulatory restriction (other than those of general application that apply to similarly situated financial or bank holding companies or their Subsidiaries) imposed upon FleetBoston, any of its Subsidiaries or the assets of FleetBoston or any of its Subsidiaries that has had or is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston or the Surviving Corporation.
- 3.10 Taxes and Tax Returns. (a) Each of FleetBoston and its Subsidiaries has duly filed all federal, state, foreign and local information returns and Tax returns required to be filed by it on or prior to the date of this Agreement (all such returns being accurate and complete in all material respects) and has duly paid or made

provision for the payment of all Taxes that have been incurred or are due or claimed to be due from it by federal, state, foreign or local taxing authorities other than (i) Taxes that are not yet delinquent or are being contested in good faith, have not been finally determined and have been adequately reserved against or (ii) information returns, Tax returns or Taxes as to which the failure to file, pay or make provision for is not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston. The federal income Tax returns of FleetBoston and its Subsidiaries have been examined by the Internal Revenue Service (the "IRS") for all years to and including 1997 and any liability with respect thereto has been satisfied or any liability with respect to deficiencies asserted as a result of such examination is covered by adequate reserves. There are no material disputes pending, or claims asserted, for Taxes or assessments upon FleetBoston or any of its Subsidiaries for which FleetBoston does not have adequate reserves. Neither FleetBoston nor any of its Subsidiaries is a party to or is bound by any Tax sharing, allocation or indemnification agreement or arrangement (other than such an agreement or arrangement exclusively between or among FleetBoston and its Subsidiaries). Within the past five years, neither FleetBoston nor any of its Subsidiaries has been a "distributing corporation" or a "controlled corporation" in a distribution intended to qualify under Section 355(a) of the Code. No disallowance of a deduction under Section 162(m) of the Code for employee remuneration of any amount paid or payable by FleetBoston or any of its Subsidiaries under any contract, plan, program or arrangement or understanding would be reasonably likely to have, individually or in the aggregate, a Material Adverse Effect on FleetBoston.

- (b) As used in this Agreement, the term "Tax" or "Taxes" means (i) all federal, state, local, and foreign income, excise, gross receipts, gross income, ad valorem, profits, gains, property, capital, sales, transfer, use, payroll, employment, severance, withholding, duties, intangibles, franchise, backup withholding, and other taxes, charges, levies or like assessments together with all penalties and additions to tax and interest thereon and (ii) any liability for Taxes described in clause (i) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign law).
- 3.11 *Employees*. (a) The FleetBoston Disclosure Schedule sets forth a true and complete list of each material benefit or compensation plan, arrangement or agreement, and any material bonus, incentive, deferred compensation, vacation, stock purchase, stock option, severance, employment, change of control or fringe benefit plan, program or agreement that is maintained, or contributed to, for the benefit of current or former directors or employees of FleetBoston and its Subsidiaries or with respect to which FleetBoston or its Subsidiaries may, directly or indirectly, have any liability, as of the date of this Agreement (the "FleetBoston Benefit Plans").
 - (b) FleetBoston has heretofore made available to Bank of America true and complete copies of each of the FleetBoston Benefit Plans and certain related documents, including (i) the actuarial report for such FleetBoston Benefit Plan (if applicable) for each of the last two years and (ii) the most recent determination letter from the IRS (if applicable) for such FleetBoston Benefit Plan.
 - (c) Except as would not reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston, (i) each of the FleetBoston Benefit Plans has been operated and administered in all material respects in compliance with its terms and applicable laws, including the Employee Retirement Income Security Act of 1974, as amended ("ERISA") and the Code, (ii) each of the FleetBoston Benefit Plans intended to be "qualified" within the meaning of Section 401(a) of the Code is so qualified, and there are no existing circumstances or any events that have occurred that will adversely affect the qualified status of any such FleetBoston Benefit Plan, (iii) with respect to each FleetBoston Benefit Plan that is subject to Title IV of ERISA, the present value (as defined under Section 3(26) of ERISA) of accumulated benefit obligations under such FleetBoston Benefit Plan, based upon the actuarial assumptions used for funding purposes in the most recent actuarial report prepared by such FleetBoston Benefit Plan's actuary with respect to such FleetBoston Benefit Plan, did not, as of its latest valuation date, exceed the then current value (as defined under Section 3(26) of ERISA) of the assets of such FleetBoston Benefit Plan allocable to such accrued benefits, (iv) no FleetBoston Benefit Plan provides benefits coverage, including death or medical benefits coverage (whether or not insured), with respect to current or former employees or directors of FleetBoston or its Subsidiaries beyond their retirement or other termination of service, other

than (A) coverage mandated by applicable law, (B) death benefits or retirement benefits under any "employee pension plan" (as such term is defined in Section 3(2) of ERISA), (C) deferred compensation benefits accrued as liabilities on the books of FleetBoston or its Subsidiaries, (D) benefits the full cost of which is borne by the current or former employee or director (or his beneficiary), (E) coverage through the last day of the calendar month in which retirement or other termination of service occurs, or (F) medical expense reimbursement accounts, (v) no liability under Title IV of ERISA has been incurred by FleetBoston, its Subsidiaries or any trade or business, whether or not incorporated, all of which together with FleetBoston would be deemed a "single employer" within the meaning of Section 4001 of ERISA (a "FleetBoston ERISA Affiliate"), that has not been satisfied in full, and no condition exists that presents a material risk to FleetBoston, its Subsidiaries or any FleetBoston ERISA Affiliate of incurring a liability thereunder, (vi) no FleetBoston Benefit Plan is a "multiemployer pension plan" (as such term is defined in Section 3(37) of ERISA), (vii) none of FleetBoston or its Subsidiaries or, to the knowledge of FleetBoston, any other person, including any fiduciary, has engaged in a transaction in connection with which FleetBoston, its Subsidiaries or any FleetBoston Benefit Plan will be subject to either a material civil penalty assessed pursuant to Section 409 or 502(i) of ERISA or a material Tax imposed pursuant to Section 4975 or 4976 of the Code, (viii) to the knowledge of FleetBoston there are no pending, threatened or anticipated claims (other than routine claims for benefits) by, on behalf of or against any of the FleetBoston Benefit Plan or any trusts related thereto and (ix) all contributions or other amounts payable by FleetBoston or its Subsidiaries as of the Effective Time with respect to each FleetBoston Benefit Plan in respect of current or former plan years have been paid or accrued

- (d) Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated by this Agreement will (either alone or in conjunction with any other event) (i) result in any payment (including severance, unemployment compensation, "excess parachute payment" (within the meaning of Section 280G of the Code), forgiveness of indebtedness or otherwise) becoming due to any director or any employee of FleetBoston or any of its Subsidiaries from FleetBoston or any of its Subsidiaries under any FleetBoston Benefit Plan or otherwise, (ii) increase any benefits otherwise payable under any FleetBoston Benefit Plan or (iii) result in any acceleration of the time of payment or vesting of any such benefits.
- 3.12 SEC Reports. FleetBoston has previously made available to Bank of America an accurate and complete copy of each (a) final registration statement, prospectus, report, schedule and definitive proxy statement filed since January 1, 2000 by FleetBoston with the SEC pursuant to the Securities Act of 1933, as amended (the "Securities Act"), or the Exchange Act (the "FleetBoston Reports"), and prior to the date of this Agreement and (b) communication mailed by FleetBoston to its shareholders since January 1, 2000 and prior to the date of this Agreement, and no such FleetBoston Report or communication, as of the date of such FleetBoston Report or communication, contained any untrue statement of a material fact or omitted to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances in which they were made, not misleading, except that information as of a later date (but before the date of this Agreement) shall be deemed to modify information as of an earlier date. Since January 1, 2000, as of their respective dates, all FleetBoston Reports filed under the Securities Act and the Exchange Act complied as to form in all material respects with the published rules and regulations of the SEC with respect thereto.
- 3.13 Compliance with Applicable Law. (a) FleetBoston and each of its Subsidiaries hold all licenses, franchises, permits and authorizations necessary for the lawful conduct of their respective businesses under and pursuant to each, and have complied in all respects with and are not in default in any respect under any, applicable law, statute, order, rule, regulation, policy or guideline of any Governmental Entity relating to FleetBoston or any of its Subsidiaries, except where the failure to hold such license, franchise, permit or authorization or such noncompliance or default is not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston.
 - (b) Except as is not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston, FleetBoston and each FleetBoston Subsidiary have properly administered all

accounts for which it acts as a fiduciary, including accounts for which it serves as a trustee, agent, custodian, personal representative, guardian, conservator or investment advisor, in accordance with the terms of the governing documents, applicable state and federal law and regulation and common law. None of FleetBoston, any FleetBoston Subsidiary, or any director, officer or employee of FleetBoston or of any FleetBoston Subsidiary, has committed any breach of trust or fiduciary duty with respect to any such fiduciary account that is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston, and, except as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston, and the accountings for each such fiduciary account are true and correct and accurately reflect the assets of such fiduciary account.

- 3.14 Certain Contracts. (a) Neither FleetBoston nor any of its Subsidiaries is a party to or bound by any contract, arrangement, commitment or understanding (whether written or oral) (i) that is a "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC) to be performed after the date of this Agreement that has not been filed or incorporated by reference in the FleetBoston Reports filed prior to the date hereof, (ii) that materially restricts the conduct of any material line of business by FleetBoston or upon consummation of the Merger will materially restrict the ability of the Surviving Corporation to engage in any line of business material to Bank of America or FleetBoston in which a financial holding company may lawfully engage, (iii) with or to a labor union or guild (including any collective bargaining agreement) or (iv) including any stock option plan, stock appreciation rights plan, restricted stock plan or stock purchase plan, any of the benefits of which will be increased, or the vesting of the benefits of which will be accelerated, by the execution of this Agreement, the occurrence of any shareholder approval or the consummation of any of the transactions contemplated by this Agreement, or the value of any of the benefits of which will be calculated on the basis of or affected by any of the transactions contemplated by this Agreement, commitment or understanding of the type described in this Section 3.14(a), whether or not set forth in the FleetBoston Disclosure Schedule, is referred to as a "FleetBoston Contract," and neither FleetBoston nor any of its Subsidiaries knows of, or has received notice of, any violation of the above by any of the other parties thereto that is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston.
 - (b) With such exceptions that are not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston, (i) each FleetBoston Contract is valid and binding on FleetBoston or any of its Subsidiaries, as applicable, and is in full force and effect, (ii) FleetBoston and each of its Subsidiaries has in all material respects performed all obligations required to be performed by it to date under each FleetBoston Contract, and (iii) no event or condition exists that constitutes or, after notice or lapse of time or both, will constitute, a material default on the part of FleetBoston or any of its Subsidiaries under any such FleetBoston Contract.
- 3.15 Agreements with Regulatory Agencies. Neither FleetBoston nor any of its Subsidiaries is subject to any cease-and-desist or other order or enforcement action issued by, or is a party to any written agreement, consent agreement or memorandum of understanding with, or is a party to any commitment letter or similar undertaking to, or is subject to any order or directive by, or has been ordered to pay any civil money penalty by, or has been since January 1, 2000, a recipient of any supervisory letter from, or since January 1, 2000, has adopted any policies, procedures or board resolutions at the request or suggestion of any Regulatory Agency or other Governmental Entity that currently restricts in any material respect the conduct of its business or that in any material manner relates to its capital adequacy, its ability to pay dividends, its credit or risk management policies, its management or its business, other than those of general application that apply to similarly situated financial holding companies or their Subsidiaries (each item in this sentence, whether or not set forth in the FleetBoston Disclosure Schedule, a "FleetBoston Regulatory Agreement"), nor has FleetBoston or any of its Subsidiaries been advised since January 1, 2000 by any Regulatory Agency or other Governmental Entity that it is considering issuing, initiating, ordering, or requesting any such FleetBoston Regulatory Agreement.
- 3.16 Interest Rate Risk Management Instruments. Except as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston, (1) all interest rate swaps, caps, floors

and option agreements and other interest rate risk management arrangements, whether entered into for the account of FleetBoston or for the account of a customer of FleetBoston or any of its Subsidiaries, were entered into in the ordinary course of business consistent with past practice and in accordance with prudent banking practice and applicable rules, regulations and policies of any Regulatory Authority and with counterparties believed to be financially responsible at the time and are legal, valid and binding obligations of FleetBoston or one of its Subsidiaries enforceable against it in accordance with their terms (except as may be limited by bankruptcy, insolvency, moratorium, reorganization or similar laws affecting the rights of creditors generally and the availability of equitable remedies), and are in full force and effect, (2) its Subsidiaries have duly performed their obligations thereunder to the extent that such obligations to perform have accrued, and, (3) to FleetBoston's knowledge, there are no breaches, violations or defaults or allegations or assertions of such by any party thereunder.

- 3.17 Undisclosed Liabilities. Except for those liabilities that are reflected or reserved against on the consolidated balance sheet of FleetBoston included in the FleetBoston 10-Q (including any notes thereto) and for liabilities incurred in the ordinary course of business consistent with past practice since June 30, 2003, since such date, neither FleetBoston nor any of its Subsidiaries has incurred any liability of any nature whatsoever (whether absolute, accrued, contingent or otherwise and whether due or to become due) that has had or is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston.
- 3.18 Environmental Liability. There are no legal, administrative, arbitral or other proceedings, claims, actions, causes of action, private environmental investigations or remediation activities or governmental investigations of any nature seeking to impose, or that are reasonably likely to result in the imposition, on FleetBoston of any liability or obligation arising under common law or under any local, state or federal environmental statute, regulation or ordinance including the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, pending or threatened against FleetBoston, which liability or obligation is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston. To the knowledge of FleetBoston, there is no reasonable basis for any such proceeding, claim, action or governmental investigation that would impose any liability or obligation that would be reasonably likely to have, individually or in the aggregate, a Material Adverse Effect on FleetBoston. FleetBoston is not subject to any agreement, order, judgment, decree, letter or memorandum by or with any Governmental Entity or third party imposing any liability or obligation with respect to the foregoing that is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on FleetBoston.
- 3.19 State Takeover Laws; FleetBoston Rights Agreement. (a) The Board of Directors of FleetBoston has approved this Agreement and the FleetBoston Stock Option Agreement and the transactions contemplated hereby and thereby as required to render inapplicable to such agreements and transactions the Rhode Island Business Combination Act of 1990, as amended, Article Ninth of the FleetBoston Articles, the FleetBoston Rights Agreement and, to the knowledge of FleetBoston, any similar "takeover" or "interested stockholder" law (any such laws, "Takeover Statutes").
 - (b) FleetBoston has taken all action, if any, necessary or appropriate so that the entering into of this Agreement and the FleetBoston Stock Option Agreement, and the consummation of the transactions contemplated hereby and thereby do not and will not result in the ability of any person to exercise any Rights (as defined in the FleetBoston Rights Agreement) under the FleetBoston Rights Agreement or enable or require any Rights to separate from the shares of FleetBoston Common Stock to which they are attached or to be triggered or become exercisable. No "Distribution Date" or "Stock Acquisition Date" (as such terms are defined in the FleetBoston Rights Agreement) has occurred.
- 3.20 Reorganization. As of the date of this Agreement, FleetBoston is not aware of any fact or circumstance that could reasonably be expected to prevent the Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.

- 3.21 *Opinions*. Prior to the execution of this Agreement, FleetBoston has received an opinion from Morgan Stanley & Co., Inc. to the effect that as of the date thereof and based upon and subject to the matters set forth therein, the Exchange Ratio is fair to the stockholders of FleetBoston from a financial point of view. Such opinion has not been amended or rescinded as of the date of this Agreement.
- 3.22 FleetBoston Information. The information relating to FleetBoston and its Subsidiaries that is provided by FleetBoston or its representatives for inclusion in the Joint Proxy Statement and the Form S-4, or in any other document filed with any other Regulatory Agency in connection with the transactions contemplated by this Agreement, will not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances in which they are made, not misleading. The Joint Proxy Statement (except for such portions thereof that relate only to Bank of America or any of its Subsidiaries) will comply in all material respects with the provisions of the Exchange Act and the rules and regulations thereunder.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF BANK OF AMERICA

Except as disclosed in (x) the Bank of America Reports (as defined in Section 4.11) filed prior to the date hereof or (y) the disclosure schedule (the Bank of America Disclosure Schedule") delivered by Bank of America to FleetBoston prior to the execution of this Agreement (which schedule sets forth, among other things, items the disclosure of which is necessary or appropriate either in response to an express disclosure requirement contained in a provision hereof or as an exception to one or more representations or warranties contained in this Article IV, or to one or more of Bank of America's covenants contained in Article V, provided, however, that, notwithstanding anything in this Agreement to the contrary, (i) no such item is required to be set forth in such schedule as an exception to a representation or warranty being deemed untrue or incorrect under the standard established by Section 9.2, and (ii) the mere inclusion of an item in such schedule as an exception to a representation or warranty shall not be deemed an admission that such item represents a material exception or material fact, event or circumstance or that such item has had or would be reasonably likely to have a Material Adverse Effect on Bank of America), Bank of America hereby represents and warrants to FleetBoston as follows:

- 4.1 Corporate Organization. (a) Bank of America is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Bank of America has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America. Bank of America is duly registered as a financial holding company under the BHC Act. True and complete copies of the Amended and Restated Certificate of Incorporation (the "Bank of America Certificate") and Bylaws of Bank of America, as in effect as of the date of this Agreement, have previously been made available to FleetBoston.
 - (b) Each Bank of America Subsidiary (i) is duly organized and validly existing under the laws of its jurisdiction of organization, (ii) is duly qualified to do business and in good standing in all jurisdictions (whether Federal, state, local or foreign) where its ownership or leasing of property or the conduct of its business requires it to be so qualified, and (iii) has all requisite corporate power and authority to own or lease its properties and assets and to carry on its business as now conducted, except in each of (i) (iii) as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America.

4.2 Capitalization. (a) The authorized capital stock of Bank of America consists of 5,000,000,000 shares of Bank of America Common Stock, of which, as of October 24, 2003, 1,494,346,458 shares were issued and outstanding, and 100,000,000 shares of preferred stock, \$0.01 par value (the "Bank of America Preferred Stock"), of which, as of October 24, 2003, 1,274,600 shares were issued and outstanding. As of October 24, 2003, no more than 2,000,000 shares of Bank of America Common Stock were held in Bank of America's treasury. As October 24, 2003, no shares of Bank of America Common Stock or Bank of America Stock were reserved for issuance, except for (A) 329,460,493 shares reserved for issuance portions issued pursuant to employee and director stock plans of Bank of America in effect as of the date of this Agreement (the "Bank of America Stock Plans") and shares reserved for issuance pursuant to Bank of America's ESOP Convertible Preferred Stock, Series C (the "Bank of America Convertible Preferred") and (B) 297,374,945 shares reserved for issuance pursuant to the Bank of America Stock Option Agreement. All of the issued and outstanding shares of Bank of America Capital Stock have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights, with no personal liability attaching to the ownership thereof. As of October 24, 2003, except pursuant to this Agreement, the Bank of America Stock Option Agreement, the Bank of America Convertible Preferred, stock repurchase plans entered into by Bank of America from time to time and put options sold by Bank of America on Bank of America Common Stock pursuant to a program designed to partially offset the cost of Bank of America from time to time and put options sold by Bank of America on Bank of America Common Stock pursuant to a program designed to partially offset the cost of Bank of America share repurchase or issuance of any shares of Bank of America Capital Stock or any other equity securities of Bank of America or any securitie

(b) All of the issued and outstanding shares of capital stock or other equity ownership interests of each "significant subsidiary" (as such term is defined under Regulation S-X of the SEC (as defined in Section 3.4)) of Bank of America are owned by Bank of America, directly or indirectly, free and clear of any Liens, and all of such shares or equity ownership interests are duly authorized and validly issued and are fully paid, nonassessable (subject to 12 U.S.C. §§ 55) and free of preemptive rights. No such significant subsidiary has or is bound by any outstanding subscriptions, options, warrants, calls, commitments or agreements of any character calling for the purchase or issuance of any shares of capital stock or any other equity security of such subsidiary or any securities representing the right to purchase or otherwise receive any shares of capital stock or any other equity security of such subsidiary.

4.3 Authority; No Violation. (a) Bank of America has full corporate power and authority to execute and deliver this Agreement and the Stock Option Agreements and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Stock Option Agreements and the consummation of the transactions contemplated hereby have been duly and validly approved by the Board of Directors of Bank of America. The Board of Directors of Bank of America has determined that this Agreement and the transactions contemplated hereby are in the best interests of Bank of America and its stockholders and has directed that this Agreement and the transactions contemplated by this Agreement be submitted to Bank of America's stockholders for adoption at a duly held meeting of such stockholders and, except for the approval of this Agreement and the transactions contemplated by this Agreement by the affirmative vote of the holders of a majority of the outstanding shares of FleetBoston Common Stock entitled to vote at such meeting, no other corporate proceedings on the part of Bank of America are necessary to approve this Agreement or the Stock Option Agreements to consummate the transactions contemplated hereby or thereby. This Agreement and the Stock Option Agreements have been duly and validly executed and delivered by Bank of America and (assuming due authorization, execution and delivery by FleetBoston) constitute the valid and binding obligations of Bank of America, enforceable against Bank of America in accordance with their terms (except as may be limited by bankruptcy, insolvency, moratorium, reorganization or similar laws affecting the rights of creditors generally and the availability of equitable remedies).

(b) Neither the execution and delivery of this Agreement or the Stock Option Agreements by Bank of America, nor the consummation by Bank of America of the transactions contemplated hereby or thereby, nor compliance by Bank of America with any of the terms or provisions of this Agreement or the Stock Option Agreements, will (i) violate any provision of the Bank of America Certificate or the Bank of America Bylaws, or (ii) assuming that the consents, approvals and filings referred to in Section 4.4 are duly obtained and/or made, (A) violate any statute, code, ordinance, rule, regulation, judgment, order, writ, decree or Injunction applicable to Bank of America, any of its Subsidiaries or any of their respective properties or assets or (B) violate, conflict with, result in a breach of any provision of or the loss of any benefit under, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation of any Lien upon any of the respective properties or assets of Bank of America or any of its Subsidiaries under, any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which Bank of America or any of its Subsidiaries is a party, or by which they or any of their respective properties or assets may be bound or affected, except for such violations, conflicts, breaches or defaults that are not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America.

4.4 Consents and Approvals. Except for (i) the filing of applications and notices, as applicable, with the Federal Reserve Board under the BHC Act and the Federal Reserve Act, as amended, and approval of such applications and notices, (ii) the Other Regulatory Approvals, (iii) the filing with the SEC of the Joint Proxy Statement and the filing and declaration of effectiveness of the Form S-4, (iv) the filing of the Articles of Merger with the Rhode Island Secretary pursuant to the RIBCA and the issuance by the Rhode Island Secretary of a Certificate of Merger and the filing of the Certificate of Merger with the Secretary of State of the State of Delaware pursuant to the DGCL, (v) any notices to or filings with the SBA, (vi) any notices or filings under the HSR Act, (vii) any consents, authorizations, approvals, filings or exemptions in connection with compliance with the applicable provisions of federal and state securities laws relating to the regulation of futures commission merchants and the rules and regulations thereunder and of any applicable SRO, and the rules of the NYSE, or that are required under consumer finance, mortgage banking and other similar laws, (viii) such filings and approvals as are required to be made or obtained under the securities or "Blue Sky" laws of various states in connection with the issuance of the shares of Bank of America Capital Stock pursuant to this Agreement, (ix) the approval of this Agreement by the requisite vote of stockholders of Bank of America and (x) filings, if any, required as a result of the particular status of FleetBoston, no consents or approvals of or filings or registrations with any Governmental Entity are necessary in connection with (A) the execution and delivery by Bank of America of this Agreement or the Stock Option Agreements and (B) the consummation by Bank of America of the Merger and the other transactions contemplated by this Agreement and the Stock Option Agreements.

4.5 Reports. Bank of America and each of its Subsidiaries have timely filed all reports, registrations and statements, together with any amendments required to be made with respect thereto, that they were required to file since January 1, 2000 with the Regulatory Agencies, and all other reports and statements required to be filed by them since January 1, 2000, including any report or statement required to be filed pursuant to the laws, rules or regulations of the United States, any state, any foreign entity, or any Regulatory Agency, and have paid all fees and assessments due and payable in connection therewith, except where the failure to file such report, registration or statement or to pay such fees and assessments is not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America. Except for normal examinations conducted by a Regulatory Agency in the ordinary course of the business of Bank of America and its Subsidiaries, no Regulatory Agency has initiated or has pending any proceeding or, to the knowledge of Bank of America, investigation into the business or operations of Bank of America or any of its Subsidiaries since January 1, 2000, except where such proceedings or investigation are not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America. There (i) is no unresolved violation, criticism, or exception by any Regulatory Agency with respect to any report or statement relating to any examinations or

inspections of Bank of America or any of its Subsidiaries, and (ii) has been no formal or informal inquiries by, or disagreements or disputes with, any Regulatory Agency with respect to the business, operations, policies or procedures of Bank of America since January 1, 2000, that are reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America.

- 4.6 Financial Statements. Bank of America has previously made available to FleetBoston copies of (i) the consolidated balance sheet of Bank of America and its Subsidiaries as of December 31, 2000, 2001 and 2002, and the related consolidated statements of income, changes in shareholders' equity and cash flows for the years then ended as reported in Bank of America's Annual Report on Form 10-K for the fiscal year ended December 31, 2002 (the "Bank of America 2002 10-K") filed with the SEC under the Exchange Act, accompanied by the audit report of PricewaterhouseCoopers LLP, independent public accountants with respect to Bank of America, and (ii) the unaudited consolidated balance sheet of Bank of America and its Subsidiaries as of June 30, 2002 and 2003, and the related consolidated statements of income, changes in shareholders' equity and cash flows of the six month periods then ended, as reported in Bank of America's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 2003 (the "Bank of America 10-Q"). The December 31, 2002 consolidated balance sheet of Bank of America (including the related notes, where applicable) fairly presents in all material respects the consolidated financial position of Bank of America and its Subsidiaries as of the date thereof, and the other financial statements referred to in this Section 4.6 (including the related notes, where applicable) fairly present in all material respects the results of the consolidated operations and changes in shareholders' equity and consolidated financial position of Bank of America and its Subsidiaries for the respective fiscal periods or as of the respective dates therein set forth, subject to normal year-end audit adjustments in amounts consistent with past experience in the case of unaudited statements; each of such statements (including the related notes, where applicable) complies in all material respects with applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto; and each of
- 4.7 Broker's Fees. Neither Bank of America nor any Bank of America Subsidiary nor any of their respective officers or directors has employed any broker or finder or incurred any liability for any brokers fees, commissions or finder's fees in connection with the Merger or related transactions contemplated by this Agreement, other than Goldman, Sachs & Co. and Banc of America Securities, the material terms of whose engagement have been disclosed to FleetBoston.
- 4.8 Absence of Certain Changes or Events. (a) Except as publicly disclosed in the Bank of America Reports (as defined in Section 4.11) filed prior to the date of this Agreement, since December 31, 2002, no event or events have occurred that have had or are reasonably likely to have a Material Adverse Effect on Bank of America.
 - (b) Except as publicly disclosed in Bank of America Reports filed prior to the date of this Agreement, from December 31, 2002 through and including the date of this Agreement, Bank of America and the Bank of America Subsidiaries have carried on their respective businesses in all material respects in the ordinary course.
- 4.9 Legal Proceedings. (a) None of Bank of America or any of its Subsidiaries is a party to any, and there are no pending or, to the best of Bank of America's knowledge, threatened, legal, administrative, arbitral or other proceedings, claims, actions or governmental or regulatory investigations of any nature against Bank of America or any of its Subsidiaries except as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America.

- (b) There is no Injunction, judgment, or regulatory restriction (other than those of general application that apply to similarly situated financial or bank holding companies or their Subsidiaries) imposed upon Bank of America, any of its Subsidiaries or the assets of Bank of America or any of its Subsidiaries that has had or is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America or the Surviving Corporation.
- 4.10 Taxes and Tax Returns. Each of Bank of America and its Subsidiaries has duly filed all federal, state, foreign and local information returns and Tax returns required to be filed by it on or prior to the date of this Agreement (all such returns being accurate and complete in all material respects) and has duly paid or made provision for the payment of all Taxes that have been incurred or are due or claimed to be due from it by federal, state, foreign or local taxing authorities other than (i) Taxes or other governmental charges that are not yet delinquent or are being contested in good faith, have not been finally determined and have been adequately reserved against, or (ii) information returns, Tax returns or Taxes as to which the failure to file, pay or make provision for is not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America. The federal income Tax returns of Bank of America and its Subsidiaries have been examined by the IRS for all years to and including 1999 and any liability with respect thereto has been satisfied or any liability with respect to deficiencies asserted as a result of such examination is covered by adequate reserves. There are no material disputes pending, or claims asserted, for Taxes or assessments upon Bank of America or any of its Subsidiaries for which Bank of America does not have adequate reserves. Neither Bank of America nor any of its Subsidiaries is a party to or is bound by any Tax sharing, allocation or indemnification agreement or arrangement (other than such an agreement or arrangement exclusively between or among Bank of America and its Subsidiaries). Within the past five years, neither Bank of America nor any of its Subsidiaries has been a "distributing corporation" or a "controlled corporation" in a distribution intended to qualify under Section 355(a) of the Code. No disallowance of a deduction under Section 162(m) of the Code for employee remuneration of any amount paid or payable by Bank of America or a
- 4.11 SEC Reports. Bank of America has previously made available to FleetBoston an accurate and complete copy of each (a) final registration statement, prospectus, report, schedule and definitive Joint Proxy Statement filed since January 1, 2000 by Bank of America with the SEC pursuant to the Securities Act or the Exchange Act (the "Bank of America Reports") and prior to the date of this Agreement and (b) communication mailed by Bank of America to its stockholders since January 1, 2000 and prior to the date of this Agreement, and no such Bank of America Report or communication, as of the date of such Bank of America Report or communication, contained any untrue statement of a material fact or omitted to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances in which they were made, not misleading, except that information as of a later date (but before the date of this Agreement) shall be deemed to modify information as of an earlier date. Since January 1, 2000, as of their respective dates, all Bank of America Reports filed under the Securities Act and the Exchange Act complied as to form in all material respects with the published rules and regulations of the SEC with respect thereto.
- 4.12 Compliance with Applicable Law. (a) Bank of America and each of its Subsidiaries hold all licenses, franchises, permits and authorizations necessary for the lawful conduct of their respective businesses under and pursuant to each, and have complied in all respects with and are not in default in any respect under any, applicable law, statute, order, rule, regulation, policy or guideline of any Governmental Entity relating to Bank of America or any of its Subsidiaries, except where the failure to hold such license, franchise, permit or authorization or such noncompliance or default is not reasonably likely to, either individually or in the aggregate, have a Material Adverse Effect on Bank of America.
 - (b) Except as is not reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America, Bank of America and each Bank of America Subsidiary have properly administered all accounts for which it acts as a fiduciary, including accounts for which it serves as a trustee,

agent, custodian, personal representative, guardian, conservator or investment advisor, in accordance with the terms of the governing documents, applicable state and federal law and regulation and common law. None of Bank of America, any Bank of America Subsidiary, or any director, officer or employee of Bank of America or of any Bank of America Subsidiary, has committed any breach of trust or fiduciary duty with respect to any such fiduciary account that is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America, and, except as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America and the accountings for each such fiduciary account are true and correct and accurately reflect the assets of such fiduciary account.

- 4.13 Agreements with Regulatory Agencies. Neither Bank of America nor any of its Subsidiaries is subject to any cease-and-desist or other order or enforcement action issued by, or is a party to any written agreement, consent agreement or memorandum of understanding with, or is a party to any commitment letter or similar undertaking to, or is subject to any order or directive by, or has been since January 1, 2000, a recipient of any supervisory letter from, or has been ordered to pay any civil money penalty by, or since January 1, 2000, has adopted any policies, procedures or board resolutions at the request or suggestion of any Regulatory Agency or other Governmental Entity that currently restricts in any material respect the conduct of its business or that in any material manner relates to its capital adequacy, its ability to pay dividends, its credit or risk management policies, its management or its business, other than those of general application that apply to similarly situated financial holding companies or their Subsidiaries (each, whether or not set forth in the Bank of America Disclosure Schedule, a "Bank of America Regulatory Agreement"), nor has Bank of America or any of its Subsidiaries been advised since January 1, 2000, by any Regulatory Agency or other Governmental Entity that it is considering issuing, initiating, ordering or requesting any such Bank of America Regulatory Agreement.
- 4.14 Interest Rate Risk Management Instruments. Except as would not be reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America, (1) all interest rate swaps, caps, floors and option agreements and other interest rate risk management arrangements, whether entered into for the account of Bank of America or for the account of a customer of Bank of America or one of its Subsidiaries, were entered into in the ordinary course of business consistent with past practice and in accordance with prudent banking practice and applicable rules, regulations and policies of any Regulatory Authority and with counterparties believed to be financially responsible at the time and are legal, valid and binding obligations of Bank of America or one of its Subsidiaries enforceable against it in accordance with their terms (except as may be limited by bankruptcy, insolvency, moratorium, reorganization or similar laws affecting the rights of creditors generally and the availability of equitable remedies), and are in full force and effect, (2) each of its Subsidiaries have duly performed their obligations thereunder to the extent that such obligations to perform have accrued, and (3) to Bank of America's knowledge, there are no breaches, violations or defaults or allegations or assertions of such by any party thereunder.
- 4.15 *Undisclosed Liabilities*. Except for those liabilities that are reflected or reserved against on the consolidated balance sheet of Bank of America included in the Bank of America 10-Q (including any notes thereto) and for liabilities incurred in the ordinary course of business consistent with past practice since June 30, 2003, since such date, neither Bank of America nor any of its Subsidiaries has incurred any liability of any nature whatsoever (whether absolute, accrued, contingent or otherwise and whether due or to become due) that, either individually or in the aggregate, has had or is reasonably likely to have, a Material Adverse Effect on Bank of America.
- 4.16 Environmental Liability. There are no legal, administrative, arbitral or other proceedings, claims, actions, causes of action, private environmental investigations or remediation activities or governmental investigations of any nature seeking to impose, or that are reasonably likely to result in the imposition, on Bank of America of any liability or obligation arising under common law or under any local, state or federal environmental statute, regulation or ordinance including the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, pending or threatened against Bank of America, which

liability or obligation is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America. To the knowledge of Bank of America, there is no reasonable basis for any such proceeding, claim, action or governmental investigation that would impose any liability or obligation that would be reasonably likely to have, individually or in the aggregate, a Material Adverse Effect on Bank of America. Bank of America is not subject to any agreement, order, judgment, decree, letter or memorandum by or with any Governmental Entity or third party imposing any liability or obligation with respect to the foregoing that is reasonably likely to have, either individually or in the aggregate, a Material Adverse Effect on Bank of America.

- 4.17 State Takeover Laws. The Board of Directors of Bank of America has approved this Agreement and the Bank of America Stock Option Agreement and the transactions contemplated hereby and thereby as required to render inapplicable to such agreements and transactions Section 203 of the DGCL and, to the knowledge of Bank of America, any Takeover Statutes.
- 4.18 *Reorganization*. As of the date of this Agreement, Bank of America is not aware of any fact or circumstance that could reasonably be expected to prevent the Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.
- 4.19 *Opinions*. Prior to the execution of this Agreement, Bank of America has received an opinion from Goldman, Sachs & Co. to the effect that as of the date thereof and based upon and subject to the matters set forth therein, the Exchange Ratio is fair to Bank of America from a financial point of view. Such opinion has not been amended or rescinded as of the date of this Agreement.
- 4.20 Bank of America Information. The information relating to Bank of America and its Subsidiaries to be contained in the Joint Proxy Statement and the Form S-4, or the information relating to Bank of America and its Subsidiaries that is provided by Bank of America or its representatives for inclusion in any other document filed with any other Regulatory Agency in connection with the transactions contemplated by this Agreement, will not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances in which they are made, not misleading. The Joint Proxy Statement (except for such portions thereof that relate only to FleetBoston or any of its Subsidiaries) will comply with the provisions of the Exchange Act and the rules and regulations thereunder in all material respects. The Form S-4 will comply with the provisions of the Securities Act and the rules and regulations thereunder in all material respects.

ARTICLE V

COVENANTS RELATING TO CONDUCT OF BUSINESS

- 5.1 Conduct of Businesses Prior to the Effective Time During the period from the date of this Agreement to the Effective Time, except as expressly contemplated or permitted by this Agreement (including the FleetBoston Disclosure Schedule and the Bank of America Disclosure Schedule), each of FleetBoston and Bank of America shall, and shall cause each of its respective Subsidiaries to (a) conduct its business in the ordinary course in all material respects, (b) use reasonable best efforts to maintain and preserve intact its business organization, employees and advantageous business relationships and retain the services of its key officers and key employees and (c) take no action that would adversely affect or materially delay the ability of either FleetBoston or Bank of America to obtain any necessary approvals of any Regulatory Agency or other Governmental Entity required for the transactions contemplated hereby or thereby.
- 5.2 FleetBoston Forbearances. During the period from the date of this Agreement to the Effective Time, except as set forth in the FleetBoston Disclosure Schedule and except as expressly contemplated or permitted by

this Agreement, FleetBoston shall not, and shall not permit any of its Subsidiaries to, without the prior written consent of Bank of America (which consent shall not be unreasonably withheld):

- (a) other than in the ordinary course of business consistent with past practice, incur any indebtedness for borrowed money (other than short-term indebtedness incurred to refinance short-term indebtedness and indebtedness of FleetBoston or any of its wholly-owned Subsidiaries to FleetBoston or any of its Subsidiaries), assume, guarantee, endorse or otherwise as an accommodation become responsible for the obligations of any other individual, corporation or other entity, or make any loan or advance (it being understood and agreed that incurrence of indebtedness in the ordinary course of business consistent with past practice shall include the creation of deposit liabilities, purchases of Federal funds, sales of certificates of deposit and entering into repurchase agreements);
 - (b) (i) adjust, split, combine or reclassify any of its capital stock;
 - (ii) make, declare or pay any dividend, or make any other distribution on, or directly or indirectly redeem, purchase or otherwise acquire, any shares of its capital stock or any securities or obligations convertible (whether currently convertible or convertible only after the passage of time or the occurrence of certain events) into or exchangeable for any shares of its capital stock (except (A) for regular quarterly cash dividends at a rate not in excess of \$0.35 per share of FleetBoston Common Stock, subject to Section 6.11, (B) dividends paid by any of the Subsidiaries of FleetBoston to FleetBoston or to any of its wholly-owned Subsidiaries, (C) dividends paid on FleetBoston Preferred Stock outstanding on the date hereof in accordance with the certificate of designation for such FleetBoston Preferred Stock, and (D) the acceptance of shares of FleetBoston Common Stock as payment of the exercise price of stock options or for withholding taxes incurred in connection with the exercise of FleetBoston Stock Options, FleetBoston SARs or the vesting of restricted stock or other FleetBoston Stock-Based Awards, in each case in accordance with past practice and the terms of the applicable award agreements);
 - (iii) grant any stock appreciation rights or grant any individual, corporation or other entity any right to acquire any shares of its capital stock, other than (A) pursuant to the FleetBoston Rights Agreement or any renewal or replacement thereof and (B) grants to newly-hired employees of FleetBoston made in the ordinary course of business consistent with past practice under the FleetBoston Stock Plans subject to the terms set forth on Schedule 5.2(b); or
 - (iv) issue any additional shares of capital stock except (A) pursuant to the exercise of FleetBoston Stock Options or FleetBoston SARs or the satisfaction of any FleetBoston Stock-Based Awards, in each case, outstanding as of the date of this Agreement or issued thereafter in compliance with this Agreement, (B) pursuant to the Convertible Note Agreement, (C) pursuant to the FleetBoston Stock Option Agreement, (D) pursuant to the FleetBoston Rights Agreement or any renewal or replacement thereof or (E) pursuant to the FleetBoston Stock Purchase Plans;
- (c) (i) except for normal increases made in the ordinary course of business consistent with past practice, or as required by applicable law or an existing agreement, increase the wages, salaries, compensation, pension, or other fringe benefits or perquisites payable to any officer, employee, or director, (ii) pay any pension or retirement allowance not required by any existing plan or agreement or by applicable law, (iii) pay any bonus other than customary year-end bonuses for fiscal 2003 determined in the ordinary course consistent with past practice or as required by an existing agreement, (iv) become a party to, amend or commit itself to, any pension, retirement, profit-sharing or welfare benefit plan or agreement or employment agreement with or for the benefit of any employee, other than in the ordinary course of business consistent with past practice or as required by applicable law or any existing agreement, or (v) except as required under any existing plan, grant, or agreement, accelerate the vesting of, or the lapsing of restrictions with respect to, any FleetBoston Stock Options, FleetBoston SARs or other FleetBoston Stock-Based Awards;
- (d) sell, transfer, mortgage, encumber or otherwise dispose of any of its properties or assets that are material to FleetBoston and its Subsidiaries, taken as a whole, to any individual, corporation or other entity

other than a Subsidiary or cancel, release or assign any indebtedness that is material to FleetBoston and its Subsidiaries, taken as a whole, to any such person or any claims held by any such person that are material to FleetBoston and its Subsidiaries, taken as a whole, in each case other than in the ordinary course of business consistent with past practice or pursuant to contracts in force at the date of this Agreement;

- (e) enter into any new line of business that is material to FleetBoston and its Subsidiaries, taken as a whole, or change its lending, investment, underwriting, risk and asset liability management and other banking and operating policies that are material to FleetBoston and its Subsidiaries, taken as a whole, except as required by applicable law, regulation or policies imposed by any Governmental Entity;
- (f) except for transactions in the ordinary course of business consistent with past practice, make any material investment either by purchase of stock or securities, contributions to capital, property transfers, or purchase of any property or assets of any other individual, corporation or other entity, *provided* that nothing contained in this *Section 5.2(f)* shall be deemed to limit FleetBoston's ability to exercise its rights under the Bank of America Stock Option Agreement;
- (g) knowingly take any action, or knowingly fail to take any action, which action or failure to act is reasonably likely to prevent the Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code;
- (h) amend its articles or certificate of incorporation or bylaws, or amend, or redeem the rights issued under, the FleetBoston Rights Agreement (except in connection with entering into the FleetBoston Stock Option Agreement), or otherwise take any action to exempt any person or entity (other than Bank of America or its Subsidiaries) or any action taken by any person or entity from the FleetBoston Rights Agreement or any Takeover Statute or similarly restrictive provisions of its organizational documents or terminate, amend or waive any provisions of any confidentiality or standstill agreements in place with any third parties;
- (i) other than in prior consultation with Bank of America, restructure or materially change its investment securities portfolio or its gap position, through purchases, sales or otherwise, or the manner in which the portfolio is classified or reported;
 - (j) settle any material claim, action or proceeding, except in the ordinary course of business consistent with past practice;
- (k) take any action that is intended or is reasonably likely to result in any of its representations or warranties set forth in this Agreement being or becoming untrue in any material respect at any time prior to the Effective Time, or in any of the conditions to the Merger set forth in *Article VII* not being satisfied or in a violation of any provision of this Agreement, except, in every case, as may be required by applicable law;
- (l) implement or adopt any change in its tax accounting or financial accounting principles, practices or methods, other than as may be required by applicable law, GAAP or regulatory guidelines;
- (m) take any action that would materially impede or delay the ability of the parties to obtain any necessary approvals of any Regulatory Agency or Governmental Entity required for the transactions contemplated by this Agreement; or
 - (n) agree to take, make any commitment to take, or adopt any resolutions of its board of directors in support of, any of the actions prohibited by this Section 5.2.
- 5.3 Bank of America Forbearances. During the period from the date of this Agreement to the Effective Time, except as expressly contemplated or permitted by this Agreement, Bank of America shall not, and shall not permit any of its Subsidiaries to, without the prior written consent of FleetBoston (which consent shall not be unreasonably withheld), (i) amend, repeal or otherwise modify any provision of the Bank of America Articles or the Bank of America Bylaws, (ii) knowingly take any action, or knowingly fail to take any action, which action or failure to act is reasonably likely to prevent the Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code, (iii) take any action that would materially impede or delay the ability of the parties

to obtain any necessary approvals of any Regulatory Agency or other Governmental Entity required for the transactions contemplated hereby, (iv) take any action that is intended or is reasonably likely to result in any of its representations or warranties set forth in this Agreement being or becoming untrue in any material respect at any time prior to the Effective Time, or in any of the conditions to the Merger set forth in *Article VII* not being satisfied or in a violation of any provision of this Agreement, except, in every case, as may be required by applicable law or (v) agree to take, make any commitment to take, or adopt any resolutions of its board of directors in support of, any of the actions prohibited by this *Section 5.3*.

ARTICLE VI

ADDITIONAL AGREEMENTS

- 6.1 Regulatory Matters. (a) Bank of America and FleetBoston shall promptly prepare and file with the SEC the Joint Proxy Statement and Bank of America shall promptly prepare and file with the SEC the Form S-4, in which the Joint Proxy Statement will be included as a prospectus. Each of Bank of America and FleetBoston shall use their reasonable best efforts to have the Form S-4 declared effective under the Securities Act as promptly as practicable after such filing, and FleetBoston and Bank of America shall thereafter mail or deliver the Joint Proxy Statement to its respective shareholders. Bank of America shall also use its reasonable best efforts to obtain all necessary state securities law or "Blue Sky" permits and approvals required to carry out the transactions contemplated by this Agreement, and FleetBoston shall furnish all information concerning FleetBoston and the holders of FleetBoston Capital Stock as may be reasonably requested in connection with any such action.
 - (b) The parties shall cooperate with each other and use their respective reasonable best efforts to promptly prepare and file all necessary documentation, to effect all applications, notices, petitions and filings, to obtain as promptly as practicable all permits, consents, approvals and authorizations of all third parties and Governmental Entities that are necessary or advisable to consummate the transactions contemplated by this Agreement (including the Merger), and to comply with the terms and conditions of all such permits, consents, approvals and authorizations of all such Governmental Entities. FleetBoston and Bank of America shall have the right to review in advance, and, to the extent practicable, each will consult the other on, in each case subject to applicable laws relating to the exchange of information, all the information relating to FleetBoston or Bank of America, as the case may be, and any of their respective Subsidiaries, which appear in any filing made with, or written materials submitted to, any third party or any Governmental Entity in connection with the transactions contemplated by this Agreement. In exercising the foregoing right, each of the parties shall act reasonably and as promptly as practicable. The parties shall consult with each other with respect to the obtaining of all permits, consents, approvals and authorizations of all third parties and Governmental Entities necessary or advisable to consummate the transactions contemplated by this Agreement and each party will keep the other apprised of the status of matters relating to completion of the transactions contemplated by this Agreement.
 - (c) Each of Bank of America and FleetBoston shall, upon request, furnish to the other all information concerning itself, its Subsidiaries, directors, officers and shareholders and such other matters as may be reasonably necessary or advisable in connection with the Joint Proxy Statement, the Form S-4 or any other statement, filing, notice or application made by or on behalf of Bank of America, FleetBoston or any of their respective Subsidiaries to any Governmental Entity in connection with the Merger and the other transactions contemplated by this Agreement.
 - (d) Each of Bank of America and FleetBoston shall promptly advise the other upon receiving any communication from any Governmental Entity consent or approval of which is required for consummation of the transactions contemplated by this Agreement that causes such party to believe that there is a reasonable likelihood that any Requisite Regulatory Approval will not be obtained or that the receipt of any such approval may be materially delayed.

- 6.2 Access to Information. (a) Upon reasonable notice and subject to applicable laws relating to the exchange of information, each of FleetBoston and Bank of America shall, and shall cause each of its Subsidiaries to, afford to the officers, employees, accountants, counsel and other representatives of the other, reasonable access, during normal business hours during the period prior to the Effective Time, to all its properties, books, contracts, commitments and records, and, during such period, the parties shall, and shall cause its Subsidiaries to, make available to the other party (i) a copy of each report, schedule, registration statement and other document filed or received by it during such period pursuant to the requirements of federal securities laws or federal or state banking laws (other than reports or documents that such party is not permitted to disclose under applicable law) and (ii) all other information concerning its business, properties and personnel as the other may reasonably request. Neither FleetBoston nor Bank of America nor any of their Subsidiaries shall be required to provide access to or to disclose information where such access or disclosure would jeopardize the attorney-client privilege of such party or its Subsidiaries or contravene any law, rule, regulation, order, judgment, decree, fiduciary duty or binding agreement entered into prior to the date of this Agreement. The parties shall make appropriate substitute disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply.
 - (b) All information and materials provided pursuant to this Agreement shall be subject to the provisions of the Confidentiality Agreement entered into between the parties as of October 23, 2003 (the "Confidentiality Agreement"). Nothing in this Agreement shall prohibit the disclosure of the tax treatment and tax structure, as those terms are used in Treasury Regulation Section 1.6011-4, of the transactions contemplated by this Agreement (but no other details about the matters covered by this Agreement, including without limitation the identities of the parties) from and after the date of the public announcement by the parties of this Agreement and the Merger.
 - (c) No investigation by either of the parties or their respective representatives shall affect the representations and warranties of the other set forth in this Agreement.
- 6.3 Shareholder Approvals. Each of FleetBoston and Bank of America shall call a meeting of its shareholders to be held as soon as reasonably practicable for the purpose of obtaining the requisite shareholder approvals required in connection with this Agreement and the Merger, and each shall use its reasonable best efforts to cause such meetings to occur as soon as reasonably practicable and on the same date. The Board of Directors of each of FleetBoston and Bank of America shall use its reasonable best efforts to obtain from its respective shareholders the shareholder vote in favor of the approval and adoption of this Agreement required to consummate the transactions contemplated by this Agreement.
- 6.4 Legal Conditions to Merger. Each of Bank of America and FleetBoston shall, and shall cause its Subsidiaries to, use their reasonable best efforts (i) to take, or cause to be taken, all actions necessary, proper or advisable to comply promptly with all legal requirements that may be imposed on such party or its Subsidiaries with respect to the Merger and, subject to the conditions set forth in Article VII, to consummate the transactions contemplated by this Agreement, and (ii) to obtain (and to cooperate with the other party to obtain) any material consent, authorization, order or approval of, or any exemption by, any Governmental Entity and any other third party that is required to be obtained by FleetBoston or Bank of America or any of their respective Subsidiaries in connection with the Merger and the other transactions contemplated by this Agreement.
- 6.5 Affiliates. FleetBoston shall use its reasonable best efforts to cause each director, executive officer and other person who is an "affiliate" (for purposes of Rule 145 under the Securities Act) of FleetBoston to deliver to the other, as soon as practicable after the date of this Agreement, and prior to the date of the meeting of the FleetBoston shareholders to be held pursuant to Section 6.3, a written agreement, in the form of Exhibit C.
- 6.6 NYSE Listing. Bank of America shall cause the shares of Bank of America Common Stock and the Bank of America Mirror Preferred Stock to be issued in the Merger to be approved for listing on the NYSE, subject to official notice of issuance, prior to the Effective Time.

- 6.7 Employee Matters. (a) From the Effective Time through December 31, 2004 or such later date to the extent December 31, 2004 is not practicable based on the occurrence of the Effective Time (such date being referred to herein as the "Benefits Transition Date"), Bank of America shall provide the employees of FleetBoston and its Subsidiaries as of the Effective Time (the "Covered Employees") with employee benefits and compensation plans, programs and arrangements that are substantially similar, in the aggregate, to the employee benefits and compensation plans, programs and arrangements provided by FleetBoston or its Subsidiaries, as the case may be, to such employees immediately prior to the Effective Time. From and after the Benefits Transition Date, Bank of America shall provide the Covered Employees with employee benefits and compensation plans, programs and arrangements that are equivalent to those provided to similarly situated employees of Bank of America and its Subsidiaries. Notwithstanding anything contained herein to the contrary, from and after the Effective Time, a Covered Employee who is terminated during the period commencing at the Effective Time and ending on the second anniversary thereof shall be entitled to receive the severance payments and benefits under the applicable FleetBoston severance plan or policy as in effect immediately prior to the Effective Time (without amendment during such two year period following the Effective Time).
 - (b) From and after the Effective Time, Bank of America shall (i) provide all Covered Employees with service credit for purposes of eligibility, participation, vesting and levels of benefits (but not for benefit accruals under any defined benefit pension plan except as otherwise provided in this Section 6.7(b)), under any employee benefit or compensation plan, program or arrangement adopted, maintained or contributed to by Bank of America or any of its Subsidiaries in which Covered Employees are eligible to participate, for all periods of employment with FleetBoston or any of its Subsidiaries (or their predecessor entities) prior to the Effective Time, (ii) cause any pre-existing conditions or limitations, eligibility waiting periods or required physical examinations under any welfare benefit plans of Bank of America or any of its Subsidiaries to be waived with respect to the Covered Employees and their eligible dependents, to the extent waived under the corresponding plan in which the applicable Covered Employee participated immediately prior to the Effective Time and, with respect to life insurance coverage, up to the Covered Employee's current level of insurability, and (iii) give the Covered Employees and their eligible dependents credit for the plan year in which the Effective Time (or commencement of participation in a plan of Bank of America or any of its Subsidiaries) occurs towards applicable deductibles and annual out-of-pocket limits for expenses incurred prior to the Effective Time (or the date of commencement of participation in a plan of Bank of America or any of its Subsidiaries in which Covered Employees become eligible to participate following the Effective Time (excluding the cash balance pension plans maintained or sponsored by FleetBoston immediately prior to the Effective Time), the Covered Employees' level of benefit accruals under any such plans (for periods of service following the date on which the Covered Employees commence participation in such plans) shall be determined based on the C
 - (c) From and after the Effective Time, Bank of America shall honor all accrued and vested benefit obligations to and contractual rights of current and former employees of FleetBoston and its Subsidiaries under the FleetBoston Benefit Plans. Bank of America agrees to take all action necessary to effectuate and satisfy the agreements and obligations set forth in *Section 6.7(c)* of the FleetBoston Disclosure Schedule.
 - (d) Concurrently with the execution of this Agreement, Bank of America is entering into employment agreements with each of the FleetBoston employees whose names are set forth in *Section 6.7(d)* of the FleetBoston Disclosure Schedule.
- 6.8 Indemnification; Directors' and Officers' Insurance. (a) In the event of any threatened or actual claim, action, suit, proceeding or investigation, whether civil, criminal or administrative, including any such claim, action, suit, proceeding or investigation in which any individual who is now, or has been at any time prior to the date of this Agreement, or who becomes prior to the Effective Time, a director, officer or employee of

FleetBoston or any of its Subsidiaries or who is or was serving at the request of FleetBoston or any of its Subsidiaries as a director, officer, employee or agent of another person (the "Indemnified Parties"), is, or is threatened to be, made a party based in whole or in part on, or arising in whole or in part out of, or pertaining to (i) the fact that he is or was a director, officer or employee of FleetBoston or any of its Subsidiaries or (ii) this Agreement or any of the transactions contemplated by this Agreement, whether asserted or arising before or after the Effective Time, the parties shall cooperate and use their best efforts to defend against and respond thereto. From and after the Effective Time, Bank of America shall indemnify and hold harmless, as and to the fullest extent provided by applicable law, the FleetBoston Certificate, the FleetBoston Bylaws and any agreement set forth in Section 6.8 of the FleetBoston Disclosure Schedule, each such Indemnified Party against any losses, claims, damages, liabilities, costs, expenses (including reimbursement for reasonable fees and expenses incurred in advance of the final disposition of any claim, suit, proceeding or investigation to each Indemnified Party as provided by the FleetBoston Certificate, the FleetBoston Bylaws and any agreement set forth in Section 6.8 of the FleetBoston Disclosure Schedule), judgments, fines and amounts paid in settlement in connection with any such threatened or actual claim, action, suit, proceeding or investigation.

- (b) Bank of America shall use its reasonable best efforts to cause the individuals serving as officers and directors of FleetBoston or any of its Subsidiaries immediately prior to the Effective Time to be covered for a period of six years from the Effective Time by the directors' and officers' liability insurance policy maintained by FleetBoston (provided that Bank of America may substitute therefor policies of at least the same coverage and amounts containing terms and conditions that are not less advantageous than such policy) with respect to acts or omissions occurring prior to the Effective Time that were committed by such officers and directors in their capacity as such.
- (c) The provisions of this Section 6.8 shall survive the Effective Time and are intended to be for the benefit of, and shall be enforceable by, each Indemnified Party and his or her heirs and representatives.
- 6.9 Additional Agreements. In case at any time after the Effective Time any further action is necessary or desirable to carry out the purposes of this Agreement (including any merger between a Subsidiary of Bank of America, on the one hand, and a Subsidiary of FleetBoston, on the other) or to vest the Surviving Corporation with full title to all properties, assets, rights, approvals, immunities and franchises of either party to the Merger, the proper officers and directors of each party and their respective Subsidiaries shall take all such necessary action as may be reasonably requested by, and at the sole expense of, Bank of America.
- 6.10 Advice of Changes. Each of Bank of America and FleetBoston shall promptly advise the other of any change or event (i) having or reasonably likely to have a Material Adverse Effect on it or (ii) that it believes would or would be reasonably likely to cause or constitute a material breach of any of its representations, warranties or covenants contained in this Agreement; provided, however, that no such notification shall affect the representations, warranties, covenants or agreements of the parties (or remedies with respect thereto) or the conditions to the obligations of the parties under this Agreement; provided further that a failure to comply with this Section 6.10 shall not constitute the failure of any condition set forth in Article VII to be satisfied unless the underlying Material Adverse Effect or material breach would independently result in the failure of a condition set forth in Article VI to be satisfied.
- 6.11 Dividends. After the date of this Agreement, each of Bank of America and FleetBoston shall coordinate with the other the declaration of any dividends in respect of Bank of America Common Stock and FleetBoston Common Stock and the record dates and payment dates relating thereto, it being the intention of the parties that holders of FleetBoston Common Stock shall not receive two dividends, or fail to receive one dividend, for any quarter with respect to their shares of FleetBoston Common Stock and any shares of Bank of America Common Stock any such holder receives in exchange therefor in the Merger.
- 6.12 Exemption from Liability Under Section 16(b). Bank of America and FleetBoston agree that, in order to most effectively compensate and retain FleetBoston Insiders (as defined below) in connection with the Merger, both prior to and after the Effective Time, it is desirable that FleetBoston Insiders not be subject to a risk of

liability under Section 16(b) of the Exchange Act to the fullest extent permitted by applicable law in connection with the conversion of shares of FleetBoston Common Stock and FleetBoston Stock Options into shares of Bank of America Common Stock in the Merger, and for that compensatory and retentive purpose agree to the provisions of this Section 6.12. Assuming that FleetBoston delivers to Bank of America the Section 16 Information (as defined below) in a timely fashion, the Board of Directors of Bank of America, or a committee of Non-Employee Directors thereof (as such term is defined for purposes of Rule 16b-3(d) under the Exchange Act), shall adopt a resolution providing that the receipt by FleetBoston Insiders of Bank of America Common Stock in exchange for shares of FleetBoston Common Stock, and of options on Bank of America Common Stock upon conversion of options on FleetBoston Common Stock, in each case pursuant to the transactions contemplated by this Agreement and to the extent such securities are listed in the Section 16 Information, are intended to be exempt from liability pursuant to Section 16(b) under the Exchange Act. "Section 16 Information" shall mean information accurate in all material respects regarding FleetBoston Insiders, the number of shares of FleetBoston Common Stock held by each such FleetBoston Insider and expected to be exchanged for Bank of America Common Stock in the Merger, and the number and description of the options on FleetBoston Common Stock held by each such FleetBoston Insider and expected to be converted into options on Bank of America Common Stock in connection with the Merger; provided that the requirement for a description of any FleetBoston Stock Options shall be deemed to be satisfied if copies of all FleetBoston Stock Plans, and forms of agreements evidencing grants thereunder, under which such FleetBoston Stock Options have been granted, have been made available to Bank of America. "FleetBoston Insiders" shall mean those officers and directors of FleetBoston who ar

6.13 No Solicitation.

(a) None of FleetBoston, its Subsidiaries or any officer, director, employee, agent or representative (including any investment banker, financial advisor, attorney, accountant or other retained representative) of FleetBoston or any of its Subsidiaries shall directly or indirectly (i) solicit, initiate or encourage or facilitate (including by way of furnishing information) or take any other action designed to facilitate any inquiries or proposals regarding any merger, share exchange, consolidation, sale of assets, sale of shares of capital stock (including, without limitation, by way of a tender offer) or similar transactions involving FleetBoston or any of its Subsidiaries that, if consummated, would constitute an Alternative Transaction (any of the foregoing inquiries or proposals being referred to herein as an "Acquisition Proposal"), (ii) participate in any discussions or negotiations regarding an Alternative Transaction or (iii) enter into any agreement regarding any Alternative Transaction. Notwithstanding the foregoing, the Board of Directors of FleetBoston shall be permitted, prior to the meeting of FleetBoston shareholders to be held pursuant to Section 6.13 and subject to compliance with the other terms of this Section 6.13 and to first entering into a confidentiality agreement with the person proposing such Acquisition Proposal on terms substantially similar to, and no less favorable to FleetBoston than, those contained in the Confidentiality Agreement, to consider and participate in discussions and negotiations with respect to a bona fide Acquisition Proposal received by FleetBoston that the Board of Directors of FleetBoston, if and only to the extent that the Board of Directors of FleetBoston reasonably determines in good faith (after consultation with outside legal counsel) that failure to do so would cause it to violate its fiduciary duties

As used in this Agreement, "Alternative Transaction" means any of (i) a transaction pursuant to which any person (or group of persons) other than Bank of America or its affiliates, directly or indirectly, acquires or would acquire more than 25 percent of the outstanding shares of FleetBoston Common Stock or outstanding voting power or of any new series or new class of preferred stock that would be entitled to a class or series vote with respect to the Merger, whether from FleetBoston or pursuant to a tender offer or exchange offer or otherwise, (ii) a merger, share exchange, consolidation or other business combination involving FleetBoston (other than the Merger), (iii) any transaction pursuant to which any person (or group of persons) other than Bank of America or its affiliates acquires or would acquire control of assets (including for this purpose the outstanding equity securities of subsidiaries of FleetBoston and securities of the entity surviving any merger or business

combination including any of FleetBoston's Subsidiaries) of FleetBoston, or any of its subsidiaries representing more than 25 percent of the fair market value of all the assets, net revenues or net income of FleetBoston and its subsidiaries, taken as a whole, immediately prior to such transaction, or (iv) any other consolidation, business combination, recapitalization or similar transaction involving FleetBoston or any of its subsidiaries, other than the transactions contemplated by this Agreement, as a result of which the holders of shares of FleetBoston Common Stock immediately prior to such transaction do not, in the aggregate, own at least 75 percent of each of the outstanding shares of common stock and the outstanding voting power of the surviving or resulting entity in such transaction immediately after the consummation thereof in substantially the same proportion as such holders held the shares of FleetBoston Common Stock immediately prior to the consummation thereof.

- (b) FleetBoston shall notify Bank of America promptly (but in no event later than 24 hours) after receipt of any Acquisition Proposal, or any material modification of or material amendment to any Acquisition Proposal, or any request for nonpublic information relating to FleetBoston or any of its Subsidiaries or for access to the properties, books or records of FleetBoston or any Subsidiary by any Person or entity that informs the Board of Directors of FleetBoston or any Subsidiary that it is considering making, or has made, an Acquisition Proposal. Such notice to Bank of America shall be made orally and in writing, and shall indicate the identity of the Person making the Acquisition Proposal or intending to make or considering making an Acquisition Proposal or requesting non-public information or access to the books and records of FleetBoston or any Subsidiary, and the material terms of any such Acquisition Proposal or modification or amendment to an Acquisition Proposal. FleetBoston shall keep Bank of America fully informed, on a current basis, of any material changes in the status and any material changes or modifications in the terms of any such Acquisition Proposal, indication or request. FleetBoston shall also promptly, and in any event within 24 hours, notify Bank of America, orally and in writing, if it enters into discussions or negotiations concerning any Acquisition Proposal in accordance with Section 6.13(a).
- (c) Nothing contained in this Section 6.13 shall prohibit FleetBoston or its Subsidiaries from taking and disclosing to its shareholders a position required by Rule 14e-2(a) or Rule 14d-9 promulgated under the Exchange Act.
- (d) FleetBoston and its Subsidiaries shall immediately cease and cause to be terminated any existing discussions or negotiations with any Persons (other than Bank of America) conducted heretofore with respect to any of the foregoing, and shall use reasonable best efforts to cause all Persons other than Bank of America who have been furnished confidential information regarding FleetBoston in connection with the solicitation of or discussions regarding an Acquisition Proposal within the 12 months prior to the date hereof promptly to return or destroy such information. FleetBoston agrees not to, and to cause its Subsidiaries not to, release any third party from the confidentiality and standstill provisions of any agreement to which FleetBoston or its Subsidiaries is or may become a party, and shall immediately take all steps necessary to terminate any approval that may have been heretofore given under any such provisions authorizing any person to make an Acquisition Proposal.
- (e) FleetBoston shall ensure that the officers, directors and all employees, agents and representatives (including any investment bankers, financial advisors, attorneys, accountants or other retained representatives) of FleetBoston or its Subsidiaries are aware of the restrictions described in this *Section 6.13* as reasonably necessary to avoid violations thereof. It is understood that any violation of the restrictions set forth in this *Section 6.13* by any officer, director, employee, agent or representative (including any investment banker, financial advisor, attorney, accountant or other retained representative) of FleetBoston or its Subsidiaries, at the direction or with the consent of FleetBoston or its Subsidiaries, shall be deemed to be a breach of this *Section 6.13* by FleetBoston.
- 6.14 Restructuring Efforts. If either Bank of America or FleetBoston shall have failed to obtain the requisite vote or votes of its shareholders for the consummation of the transactions contemplated by this Agreement at a duly held meeting of its shareholders or at any adjournment or postponement thereof, each of the parties shall in good faith use its reasonable best efforts to negotiate a restructuring of the transaction provided for herein (it

being understood that neither party shall have any obligation to alter or change the amount or kind of the Merger Consideration in a manner adverse to such party or its shareholders) and to resubmit the transaction to their respective shareholders for approval.

ARTICLE VII

CONDITIONS PRECEDENT

- 7.1 Conditions to Each Party's Obligation To Effect the Merger. The respective obligations of the parties to effect the Merger shall be subject to the satisfaction at or prior to the Effective Time of the following conditions:
 - (a) Stockholder Approval. This Agreement shall have been approved and adopted by the requisite affirmative vote of the holders of FleetBoston Common Stock entitled to vote thereon and by the requisite affirmative vote of the holders of Bank of America Common Stock entitled to vote thereon.
 - (b) NYSE Listing. The shares of Bank of America Common Stock and Bank of America Mirror Preferred Stock to be issued to the holders of FleetBoston Capital Stock upon consummation of the Merger shall have been authorized for listing on the NYSE, subject to official notice of issuance.
 - (c) Regulatory Approvals. All regulatory approvals set forth in Sections 3.4 and 4.4 required to consummate the transactions contemplated by this Agreement, including the Merger, shall have been obtained and shall remain in full force and effect and all statutory waiting periods in respect thereof shall have expired (all such approvals and the expiration of all such waiting periods being referred as the "Requisite Regulatory Approvals").
 - (d) Form S-4. The Form S-4 shall have become effective under the Securities Act and no stop order suspending the effectiveness of the Form S-4 shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the SEC.
 - (e) No Injunctions or Restraints; Illegality. No order, injunction or decree issued by any court or agency of competent jurisdiction or other legal restraint or prohibition (an "Injunction") preventing the consummation of the Merger or any of the other transactions contemplated by this Agreement shall be in effect. No statute, rule, regulation, order, Injunction or decree shall have been enacted, entered, promulgated or enforced by any Governmental Entity that prohibits or makes illegal consummation of the Merger.
- 7.2 Conditions to Obligations of Bank of America. The obligation of Bank of America to effect the Merger is also subject to the satisfaction, or waiver by Bank of America, at or prior to the Effective Time, of the following conditions:
 - (a) Representations and Warranties. Subject to the standard set forth in Section 9.2, the representations and warranties of FleetBoston set forth in this Agreement shall be true and correct as of the date of this Agreement and as of the Effective Time as though made on and as of the Effective Time (except that representations and warranties that by their terms speak specifically as of the date of this Agreement or another date shall be true and correct as of such date); and Bank of America shall have received a certificate signed on behalf of FleetBoston by the Chief Executive Officer or the Chief Financial Officer of FleetBoston to the foregoing effect.
 - (b) Performance of Obligations of FleetBoston. FleetBoston shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date; and Bank of America shall have received a certificate signed on behalf of FleetBoston by the Chief Executive Officer or the Chief Financial Officer of FleetBoston to such effect.
 - (c) Federal Tax Opinion. Bank of America shall have received the opinion of its counsel, Cleary, Gottlieb, Steen & Hamilton, in form and substance reasonably satisfactory to Bank of America, dated the Closing Date, substantially to the effect that, on the basis of facts, representations and assumptions set forth

in such opinion that are consistent with the state of facts existing at the Effective Time, the Merger will be treated as a reorganization within the meaning of Section 368(a) of the Code. In rendering such opinion, counsel may require and rely upon representations contained in certificates of officers of FleetBoston and Bank of America, reasonably satisfactory in form and substance to it.

- 7.3 Conditions to Obligations of FleetBoston. The obligation of FleetBoston to effect the Merger is also subject to the satisfaction or waiver by FleetBoston at or prior to the Effective Time of the following conditions:
 - (a) Representations and Warranties. Subject to the standard set forth in Section 9.2, the representations and warranties of Bank of America set forth in this Agreement shall be true and correct as of the date of this Agreement and as of the Effective Time as though made on and as of the Effective Time (except that representations and warranties that by their terms speak specifically as of the date of this Agreement or another date shall be true and correct as of such date); and FleetBoston shall have received a certificate signed on behalf of Bank of America by the Chief Executive Officer or the Chief Financial Officer of Bank of America to the foregoing effect.
 - (b) Performance of Obligations of Bank of America. Bank of America shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and FleetBoston shall have received a certificate signed on behalf of Bank of America by the Chief Executive Officer or the Chief Financial Officer of Bank of America to such effect.
 - (c) Federal Tax Opinion. FleetBoston shall have received the opinion of its counsel, Wachtell, Lipton, Rosen & Katz, in form and substance reasonably satisfactory to FleetBoston, dated the Closing Date, substantially to the effect that, on the basis of facts, representations and assumptions set forth in such opinion that are consistent with the state of facts existing at the Effective Time, the Merger will be treated as a reorganization within the meaning of Section 368(a) of the Code. In rendering such opinion, counsel may require and rely upon representations contained in certificates of officers of FleetBoston and Bank of America, reasonably satisfactory in form and substance to it.

ARTICLE VIII

TERMINATION AND AMENDMENT

- 8.1 *Termination*. This Agreement may be terminated at any time prior to the Effective Time, whether before or after approval of the matters presented in connection with the Merger by the shareholders of FleetBoston or Bank of America:
 - (a) by mutual consent of FleetBoston and Bank of America in a written instrument, if the Board of Directors of each so determines by a vote of a majority of the members of its respective entire Board of Directors;
 - (b) by either the Board of Directors of FleetBoston or the Board of Directors of Bank of America if any Governmental Entity that must grant a Requisite Regulatory Approval has denied approval of the Merger and such denial has become final and nonappealable or any Governmental Entity of competent jurisdiction shall have issued a final and nonappealable order permanently enjoining or otherwise prohibiting the consummation of the transactions contemplated by this Agreement;
 - (c) by either the Board of Directors of FleetBoston or the Board of Directors of Bank of America if the Merger shall not have been consummated on or before the first anniversary of the date of this Agreement unless the failure of the Closing to occur by such date shall be due to the failure of the party seeking to terminate this Agreement to perform or observe the covenants and agreements of such party set forth in this Agreement;
 - (d) by either the Board of Directors of Bank of America or the Board of Directors of FleetBoston if there shall have been a breach of any of the covenants or agreements or any of the representations or

warranties set forth in this Agreement on the part of FleetBoston, in the case of a termination by Bank of America, or Bank of America, in the case of a termination by FleetBoston, which breach, either individually or in the aggregate, would result in, if occurring or continuing on the Closing Date, the failure of the conditions set forth in Section 7.2 or 7.3, as the case may be, and which is not cured within 45 days following written notice to the party committing such breach or by its nature or timing cannot be cured within such time period; or

- (e) by either Bank of America or FleetBoston, if its Board of Directors determines in good faith by a majority vote that the other party has substantially engaged in bad faith in breach of its obligations under Section 6.14 of this Agreement.
- 8.2 Effect of Termination. In the event of termination of this Agreement by either FleetBoston or Bank of America as provided in Section 8.1, this Agreement shall forthwith become void and have no effect, and none of FleetBoston, Bank of America, any of their respective Subsidiaries or any of the officers or directors of any of them shall have any liability of any nature whatsoever under this Agreement, or in connection with the transactions contemplated by this Agreement, except that (i) Sections 6.2(b), 8.2, 9.3, 9.4, 9.9 and 9.10 shall survive any termination of this Agreement, and (ii) notwithstanding anything to the contrary contained in this Agreement, neither FleetBoston nor Bank of America shall be relieved or released from any liabilities or damages arising out of its willful breach of any provision of this Agreement. Notwithstanding the foregoing, in the event of any termination of this Agreement, the Stock Option Agreements shall remain in full force and effect in accordance with their respective terms.
- 8.3 Amendment. Subject to compliance with applicable law and Section 1.1(b), this Agreement may be amended by the parties, by action taken or authorized by their respective Boards of Directors, at any time before or after approval of the matters presented in connection with Merger by the shareholders of FleetBoston or Bank of America; provided, however, that after any approval of the transactions contemplated by this Agreement by the shareholders of FleetBoston and Bank of America, there may not be, without further approval of such shareholders, any amendment of this Agreement that changes the amount or the form of the consideration to be delivered under this Agreement to the holders of FleetBoston Common Stock, other than as contemplated by this Agreement. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties.
- 8.4 Extension; Waiver. At any time prior to the Effective Time, the parties, by action taken or authorized by their respective Board of Directors, may, to the extent legally allowed, (a) extend the time for the performance of any of the obligations or other acts of the other party, (b) waive any inaccuracies in the representations and warranties contained in this Agreement and (c) waive compliance with any of the agreements or conditions contained in this Agreement; provided, however, that after any approval of the transactions contemplated by this Agreement by the shareholders of FleetBoston and Bank of America, there may not be, without further approval of such shareholders, any extension or waiver of this Agreement or any portion hereof that reduces the amount or changes the form of the consideration to be delivered to the holders of FleetBoston Common Stock under this Agreement, other than as contemplated by this Agreement. Any agreement on the part of a party to any such extension or waiver shall be valid only if set forth in a written instrument signed on behalf of such party, but such extension or waiver or failure to insist on strict compliance with an obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

ARTICLE IX

GENERAL PROVISIONS

9.1 Closing. On the terms and subject to conditions set forth in this Agreement, the closing of the Merger (the "Closing") shall take place at 10:00 a.m. on a date and at a place to be specified by the parties, which date shall be no later than five business days after the satisfaction or waiver (subject to applicable law) of the latest to

occur of the conditions set forth in Article VII (other than those conditions that by their nature are to be satisfied or waived at the Closing), unless extended by mutual agreement of the parties (the "Closing Date").

- 9.2 Standard. No representation or warranty of FleetBoston contained in Article III or of Bank of America contained in Article IV shall be deemed untrue or incorrect for any purpose under this Agreement, and no party hereto shall be deemed to have breached a representation or warranty for any purpose under this Agreement, in any case as a consequence of the existence or absence of any fact, circumstance or event unless such fact, circumstance or event, individually or when taken together with all other facts, circumstances or events inconsistent with any representations or warranties contained in Article III, in the case of FleetBoston, or Article IV, in the case of Bank of America, has had or would be reasonably likely to have a Material Adverse Effect with respect to FleetBoston or Bank of America, respectively (disregarding for purposes of this Section 9.2 any materiality or Material Adverse Effect qualification contained in any representations or warranties). Notwithstanding the immediately preceding sentence, the representations and warranties contained in Section 3.2(a), in the case of FleetBoston, and Section 4.2(a), in the case of Bank of America, shall be deemed untrue and incorrect if not true and correct in all material respects.
- 9.3 Nonsurvival of Representations, Warranties and Agreements. None of the representations, warranties, covenants and agreements set forth in this Agreement or in any instrument delivered pursuant to this Agreement shall survive the Effective Time, except for Section 6.8 and for those other covenants and agreements contained in this Agreement that by their terms apply or are to be performed in whole or in part after the Effective Time. After the Effective Time, the provisions of Section 1.11(a) may be amended by a vote of 75% of the Board of Directors of the Surviving Corporation. The provisions of Section 1.11(b) shall be subject to the continuing interpretation of the Board of Directors of the Surviving Corporation, which shall have the sole authority for interpreting such provisions consistent with the purposes reflected therein.
- 9.4 Expenses. All costs and expenses incurred in connection with this Agreement and the transactions contemplated by this Agreement shall be paid by the party incurring such expense; provided, however, that the costs and expenses of printing and mailing the Joint Proxy Statement, and all filing and other fees paid to the SEC in connection with the Merger, shall be borne equally by FleetBoston and Bank of America.
- 9.5 Notices. All notices and other communications in connection with this Agreement shall be in writing and shall be deemed given if delivered personally, sent via facsimile (with confirmation), mailed by registered or certified mail (return receipt requested) or delivered by an express courier (with confirmation) to the parties at the following addresses (or at such other address for a party as shall be specified by like notice):

(a) if to FleetBoston, to:

FleetBoston Financial Corporation 100 Federal Street, 26th Floor MA DE 10026A Boston, Massachusetts 02110 Attention: General Counsel Facsimile: (617) 434-6525

with a copy to:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, New York 10019 Attention: Edward D. Herlihy, Esq.

Facsimile: (212) 403-2000

and

(b) if to Bank of America, to:

100 South Tryon Street NC1-002-29-01 Charlotte, North Carolina 28255 Attention: General Counsel Facsimile: (704) 370-3515

Bank of America Corporation

with a copy to:

Cleary, Gottlieb, Steen & Hamilton 2000 Pennsylvania Avenue, NW Washington, D.C. 20006-1801 Attention: John C. Murphy, Jr., Esq. Facsimile: (202) 974-1999

- 9.6 Interpretation. When a reference is made in this Agreement to Articles, Sections, Exhibits or Schedules, such reference shall be to a Article or Section of or Exhibit or Schedule to this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The FleetBoston Disclosure Schedule and the Bank of America Disclosure Schedule, as well as all other schedules and all exhibits hereto, shall be deemed part of this Agreement and included in any reference to this Agreement. This Agreement shall not be interpreted or construed to require any person to take any action, or fail to take any action, if to do so would violate any applicable law.
- 9.7 Counterparts. This Agreement may be executed in two or more counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each of the parties and delivered to the other party, it being understood that each party need not sign the same counterpart.
- 9.8 Entire Agreement. This Agreement (including the documents and the instruments referred to in this Agreement), together with the Confidentiality Agreement and the Stock Option Agreements, constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter of this Agreement, other than the Confidentiality Agreement and the Stock Option Agreements.
- 9.9 Governing Law. This Agreement shall be governed and construed in accordance with the internal laws of the State of New York applicable to contracts made and wholly-performed within such state, without regard to any applicable conflicts of law principles.
- 9.10 *Publicity*. Neither FleetBoston nor Bank of America shall, and neither FleetBoston nor Bank of America shall permit any of its Subsidiaries to, issue or cause the publication of any press release or other public announcement with respect to, or otherwise make any public statement concerning, the transactions contemplated by this Agreement without the prior consent (which consent shall not be unreasonably withheld) of Bank of America, in the case of a proposed announcement or statement by FleetBoston, or FleetBoston, in the case of a proposed announcement or statement by Bank of America; *provided*, *however*, that either party may, without the prior consent of the other party (but after prior consultation with the other party to the extent practicable under the circumstances) issue or cause the publication of any press release or other public announcement to the extent required by law or by the rules and regulations of the NYSE.

9.11 Assignment; Third Party Beneficiaries. Neither this Agreement nor any of the rights, interests or obligations under this Agreement shall be assigned by either of the parties (whether by operation of law or otherwise) without the prior written consent of the other party. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by each of the parties and their respective successors and assigns. Except as otherwise specifically provided in Section 6.8, this Agreement (including the documents and instruments referred to in this Agreement) is not intended to and does not confer upon any person other than the parties hereto any rights or remedies under this Agreement.

 $Remainder\ of\ Page\ Intentionally\ Left\ Blank$

IN WITNESS WHEREOF, FleetBoston and Bank of America have caused this Agreement to be executed by their respective officers thereunto duly authorized as of the date first above written.

FLEETBOSTON FINANCIAL CORPORATION

By: /s/ Charles K. Gifford

Name: Charles K. Gifford

Title: Chairman and Chief Executive Officer

BANK OF AMERICA CORPORATION

By: /s/ Kenneth D. Lewis

Name: Kenneth D. Lewis

Title: Chairman, President and Chief Executive Officer

 $Signature\ Page\ to\ Agreement\ and\ Plan\ of\ Merger$

A-37

Exhibit C

Form of Affiliate Letter

Bank of America Corporation

Ladies and Gentlemen:

I have been advised that as of the date hereof I may be deemed to be an "affiliate" of FleetBoston Financial Corporation, a Rhode Island corporation (*FleetBoston*"), as the term "affiliate" is defined for purposes of paragraphs (c) and (d) of Rule 145 of the Rules and Regulations (the "Rules and Regulations") of the Securities and Exchange Commission (the "Commission") under the Securities Act of 1933, as amended (the "Act"). I have been further advised that pursuant to the terms of the Agreement and Plan of Merger dated as of October 27, 2003 (the "Merger Agreement"), by and between Bank of America Corporation, a Delaware corporation ("Bank of America"), and FleetBoston, FleetBoston shall be merged with and into Bank of America (the "Merger") and each share of the common stock, par value \$0.01 per share, of FleetBoston (*'FleetBoston Common Stock*") shall be converted into the right to receive 0.5553 of a share of common stock, par value \$0.01 per share, of Bank of America (*'Bank of America Common Stock*"). All terms used in this letter but not defined herein shall have the meanings ascribed thereto in the Merger Agreement.

I represent, warrant and covenant to Bank of America that in the event I receive any Bank of America Common Stock as a result of the Merger:

- (a) I shall not make any sale, transfer or other disposition of Bank of America Common Stock in violation of the Act or the Rules and Regulations,
- (b) I have carefully read this letter and the Merger Agreement and discussed its requirements and other applicable limitations upon my ability to sell, transfer or otherwise dispose of Bank of America Common Stock to the extent I believed necessary with my counsel or counsel for FleetBoston.
- (c) I have been advised that the issuance of Bank of America Common Stock to me pursuant to the Merger will be registered with the Commission under the Act on a Registration Statement on Form S-4. However, I have also been advised that, since at the time the Merger will be submitted for a vote of the stockholders of FleetBoston I may be deemed to have been an affiliate of FleetBoston and the distribution by me of Bank of America Common Stock has not been registered under the Act, I may not sell, transfer or otherwise dispose of Bank of America Common Stock issued to me in the Merger unless (i) such sale, transfer or other disposition has been registered under the Act, (ii) such sale, transfer or other disposition is made in conformity with the volume and other limitations of Rule 145 promulgated by the Commission under the Act, or (iii) in the opinion of counsel reasonably acceptable to Bank of America, such sale, transfer or other disposition is otherwise exempt from registration under the Act.
- (d) I understand that Bank of America is under no obligation to register the sale, transfer or other disposition of Bank of America Common Stock by me or on my behalf under the Act or to take any other action necessary in order to make compliance with an exemption from such registration available.
- (e) I also understand that stop transfer instructions will be given to Bank of America's transfer agents with respect to Bank of America Common Stock and that there will be placed on the certificates for Bank of America Common Stock issued to me, or any substitutions therefor, a legend stating in substance:

"The securities represented by this certificate have been issued in a transaction to which Rule 145 promulgated under the Securities Act of 1933 applies and may only be sold or otherwise transferred in compliance with the requirements of Rule 145 or pursuant to a registration statement under said act or an exemption from such registration."

(f) I also understand that unless the transfer by me of my Bank of America Common Stock has been registered under the Act or is a sale made in conformity with the provisions of Rule 145, Bank of America reserves the right to put the following legend on the certificates issued to my transferee:

"The shares represented by this certificate have not been registered under the Securities Act of 1933 and were acquired from a person who received such shares in a transaction to which Rule 145 promulgated under the Securities Act of 1933 applies. The shares have been acquired by the holder not with a view to, or for resale in connection with, any distribution thereof within the meaning of the Securities Act of 1933 and may not be sold, pledged or otherwise transferred except in accordance with an exemption from the registration requirements of the Securities Act of 1933."

It is understood and agreed that the legends set forth above shall be removed by delivery of substitute certificates without such legend, and/or the issuance of a letter to Bank of America's transfer agent removing such stop transfer instructions, and the above restrictions on sale will cease to apply, if (A) one year (or such other period as may be required by Rule 145(d)(2) under the Securities Act or any successor thereto) shall have elapsed from the Closing Date and the provisions of such Rule are then available to me; or (B) if two years (or such other period as may be required by Rule 145(d)(3) under the Securities Act or any successor thereto) shall have elapsed from the Effective Date and the provisions of such Rule are then available to me; or (C) I shall have delivered to Bank of America (i) a copy of a letter from the staff of the Commission, or an opinion of counsel in form and substance reasonably satisfactory to Bank of America, or other evidence reasonably satisfactory to Bank of America, to the effect that such legend and/or stop transfer instructions are not required for purposes of the Securities Act or (ii) reasonably satisfactory evidence or representations that the securities represented by such certificates are being or have been transferred in a transaction made in conformity with the provisions of Rule 145 under the Securities Act or pursuant to an effective registration under the Securities Act.

I recognize and agree that the foregoing provisions also apply to (i) my spouse, (ii) any relative of mine or my spouse occupying my home, (iii) any trust or estate in which I, my spouse or any such relative owns at least 10% beneficial interest or of which any of us serves as trustee, executor or in any similar capacity and (iv) any corporate or other organization in which I, my spouse or any such relative owns at least 10% of any class of equity securities or of the equity interest.

It is understood and agreed that this Letter Agreement shall terminate and be of no further force and effect if the Merger Agreement is terminated in accordance with its terms.

Execution of this letter should not be construed as an admission on my part that I am an "affiliate" of FleetBoston as described in the first paragraph of this letter or as a waiver of any rights I may have to object to any claim that I am such an affiliate on or after the date of this letter.

	Very truly yours,	
	Ву:	
	Nar	me:
Accepted this [] day of [], 2003		
Bank of America Corporation		
By:		
Name: Title:		

Appendix B

THE TRANSFER OF THIS AGREEMENT IS SUBJECT TO CERTAIN PROVISIONS CONTAINED HEREIN AND TO RESALE RESTRICTIONS UNDER THE SECURITIES ACT OF 1933, AS AMENDED

STOCK OPTION AGREEMENT, dated October 27, 2003, between FleetBoston Financial Corporation, a Rhode Island corporation ("Issuer"), and Bank of America Corporation, a Delaware corporation ("Grantee").

<u>W I T N E S S E T H</u>:

WHEREAS, Grantee and Issuer have entered into an Agreement and Plan of Merger of even date herewith (the "Merger Agreement"), which agreement has been executed by the parties hereto in connection with this Stock Option Agreement (the "Agreement"); and

WHEREAS, as a condition to Grantee's entering into the Merger Agreement and in consideration therefor, Issuer has agreed to grant Grantee the Option (as hereinafter defined);

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants and agreements set forth herein and in the Merger Agreement, the parties hereto agree as follows:

- 1. (a) Issuer hereby grants to Grantee an unconditional, irrevocable option (the "Option") to purchase, subject to the terms hereof, up to 209,496,275 fully paid and nonassessable shares of Issuer's Common Stock, par value \$0.01 per share ("Common Stock"), at a price of \$31.80 per share (the "Option Price"); provided, however, that in no event shall the number of shares of Common Stock for which this Option is exercisable exceed 19.9% of the Issuer's issued and outstanding shares of Common Stock without giving effect to any shares subject to or issued pursuant to the Option. The number of shares of Common Stock that may be received upon the exercise of the Option and the Option Price are subject to adjustment as herein set forth.
 - (b) In the event that any additional shares of Common Stock are either (i) issued or otherwise become outstanding after the date of this Agreement (other than pursuant to this Agreement) or (ii) redeemed, repurchased, retired or otherwise cease to be outstanding after the date of this Agreement, the number of shares of Common Stock subject to the Option shall be increased or decreased, as appropriate, so that, after such issuance, such number equals 19.9% of the number of shares of Common Stock then issued and outstanding without giving effect to any shares subject or issued pursuant to the Option. Nothing contained in this Section 1(b) or elsewhere in this Agreement shall be deemed to authorize Issuer or Grantee to breach any provision of the Merger Agreement.
- 2. (a) The Holder (as hereinafter defined) may exercise the Option, in whole or part, and from time to time, if, but only if, both an Initial Triggering Event (as hereinafter defined) and a Subsequent Triggering Event (as hereinafter defined) shall have occurred prior to the occurrence of an Exercise Termination Event (as hereinafter defined), provided that the Holder shall have sent the written notice of such exercise (as provided in subsection (e) of this Section 2) within 180 days following such Subsequent Triggering Event. Each of the following shall be an "Exercise Termination Event": (i) the Effective Time (as defined in the Merger Agreement) of the Merger; (ii) termination of the Merger Agreement in accordance with the provisions thereof if such termination occurs prior to the occurrence of an Initial Triggering Event except a termination by Grantee pursuant to Section 8.1(d) of the Merger Agreement (unless the breach by Issuer giving rise to such right of termination is non-volitional); or (iii) the passage of 18 months after termination of the Merger Agreement if such termination follows the occurrence of an Initial Triggering Event or is a termination by Grantee pursuant to Section 8.1(d) of the Merger Agreement (unless the breach by Issuer giving rise to such right of termination is non-volitional). The term "Holder" shall mean the holder or holders of the Option.
 - (b) The term "Initial Triggering Event" shall mean any of the following events or transactions occurring after the date hereof:
 - (i) Issuer or any of its Subsidiaries (each an "Issuer Subsidiary"), without having received Grantee's prior written consent, shall have entered into an agreement to engage in an Acquisition

Transaction (as hereinafter defined) with any person (the term "person" for purposes of this Agreement having the meaning assigned thereto in Sections 3(a)(9) and 13(d)(3) of the Securities Exchange Act of 1934, as amended (the "1934 Act"), and the rules and regulations thereunder) other than Grantee or any of its Subsidiaries (each a "Grantee Subsidiary") or the Board of Directors of Issuer shall have recommended that the stockholders of Issuer approve or accept any Acquisition Transaction with any person other than Grantee or a Subsidiary of Grantee. For purposes of this Agreement, "Acquisition Transaction" shall mean (w) a merger, consolidation or share exchange, or any similar transaction, involving Issuer or any Significant Subsidiary (as defined in Rule 1-02 of Regulation S-X promulgated by the Securities and Exchange Commission (the "SEC")) of Issuer, (x) a purchase, lease or other acquisition or assumption of all or a substantial portion of the assets or deposits of Issuer or any Significant Subsidiary of Issuer, (y) a purchase or other acquisition (including by way of merger, consolidation, share exchange or otherwise) of securities representing 10% or more of the voting power of Issuer, or (z) any substantially similar transaction; provided, however, that in no event shall any merger, consolidation, purchase or similar transaction involving only the Issuer and one or more of its Subsidiaries or involving only any two or more of such Subsidiaries, be deemed to be an Acquisition Transaction, provided that any such transaction is not entered into in violation of the terms of the Merger Agreement;

- (ii) Issuer or any Issuer Subsidiary, without having received Grantee's prior written consent, shall have authorized, recommended, proposed or publicly announced its intention to authorize, recommend or propose, to engage in an Acquisition Transaction with any person other than Grantee or a Grantee Subsidiary, or the Board of Directors of Issuer shall have publicly withdrawn or modified, or publicly announced its intention to withdraw or modify, in any manner adverse to Grantee, its recommendation that the stockholders of Issuer approve the transactions contemplated by the Merger Agreement;
- (iii) Any person other than Grantee, any Grantee Subsidiary or any Issuer Subsidiary acting in a fiduciary capacity in the ordinary course of its business shall have acquired beneficial ownership or the right to acquire beneficial ownership of 10% or more of the outstanding shares of Common Stock (the term "beneficial ownership" for purposes of this Agreement having the meaning assigned thereto in Section 13(d) of the 1934 Act, and the rules and regulations thereunder);
- (iv) Any person other than Grantee or any Grantee Subsidiary shall have made a bona fide proposal to Issuer or its stockholders by public announcement or written communication that is or becomes the subject of public disclosure to engage in an Acquisition Transaction;
- (v) After an overture is made by a third party to Issuer or its stockholders to engage in an Acquisition Transaction, Issuer shall have breached any covenant or obligation contained in the Merger Agreement and such breach (x) would entitle Grantee to terminate the Merger Agreement and (y) shall not have been cured prior to the Notice Date (as defined below); or
- (vi) Any person other than Grantee or any Grantee Subsidiary, other than in connection with a transaction to which Grantee has given its prior written consent, shall have filed an application or notice with the Federal Reserve Board, or other federal or state bank regulatory authority, which application or notice has been accepted for processing, for approval to engage in an Acquisition Transaction.
- (c) The term "Subsequent Triggering Event" shall mean either of the following events or transactions occurring after the date hereof:
 - (i) The acquisition by any person of beneficial ownership of 20% or more of the then outstanding Common Stock; or
- (ii) The occurrence of the Initial Triggering Event described in paragraph (i) of subsection (b) of this Section 2, except that the percentage referred to in clause (y) shall be 20%.
- (d) Issuer shall notify Grantee promptly in writing of the occurrence of any Initial Triggering Event or Subsequent Triggering Event of which it has knowledge (together, a "Triggering Event"), it being

understood that the giving of such notice by Issuer shall not be a condition to the right of the Holder to exercise the Option.

- (e) In the event the Holder is entitled to and wishes to exercise the Option, it shall send to Issuer a written notice (the date of which being herein referred to as the "Notice Date") specifying (i) the total number of shares it will purchase pursuant to such exercise and (ii) a place and date not earlier than three business days nor later than 60 business days from the Notice Date for the closing of such purchase (the "Closing Date"); provided that if prior notification to or approval of the Federal Reserve Board or any other regulatory agency is required in connection with such purchase, the Holder shall as soon as reasonably practicable file the required notice or application for approval and shall expeditiously process the same and the period of time that otherwise would run pursuant to this sentence shall run instead from the date on which any required notification periods have expired or been terminated or such approvals have been obtained and any requisite waiting period or periods shall have passed. Any exercise of the Option shall be deemed to occur on the Notice Date relating thereto.
- (f) At the closing referred to in subsection (e) of this Section 2, the Holder shall pay to Issuer the aggregate purchase price for the shares of Common Stock purchased pursuant to the exercise of the Option in immediately available funds by wire transfer to a bank account designated by Issuer, *provided* that failure or refusal of Issuer to designate such a bank account shall not preclude the Holder from exercising the Option.
- (g) At such closing, simultaneously with the delivery of immediately available funds as provided in subsection (f) of this Section 2, Issuer shall deliver to the Holder a certificate or certificates representing the number of shares of Common Stock purchased by the Holder and, if the Option should be exercised in part only, a new Option evidencing the rights of the Holder thereof to purchase the balance of the shares purchasable hereunder, and the Holder shall deliver to Issuer this Agreement and a letter agreeing that the Holder will not offer to sell or otherwise dispose of such shares in violation of applicable law or the provisions of this Agreement.
 - (h) Certificates for Common Stock delivered at a closing hereunder may be endorsed with a restrictive legend that shall read substantially as follows:
 - "The transfer of the shares represented by this certificate is subject to certain provisions of an agreement between the registered holder hereof and Issuer and to resale restrictions arising under the Securities Act of 1933, as amended. A copy of such agreement is on file at the principal office of Issuer and will be provided to the holder hereof without charge upon receipt by Issuer of a written request therefor."

It is understood and agreed that: (i) the reference to the resale restrictions of the Securities Act of 1933, as amended (the "1933 Act"), in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the Holder shall have delivered to Issuer a copy of a letter from the staff of the SEC, or an opinion of counsel, in form and substance reasonably satisfactory to Issuer, to the effect that such legend is not required for purposes of the 1933 Act; (ii) the reference to the provisions to this Agreement in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the shares have been sold or transferred in compliance with the provisions of this Agreement and under circumstances that do not require the retention of such reference; and (iii) the legend shall be removed in its entirety if the conditions in the preceding clauses (i) and (ii) are both satisfied. In addition, such certificates shall bear any other legend as may be required by law.

(i) Upon the giving by the Holder to Issuer of the written notice of exercise of the Option provided for under subsection (e) of this Section 2 and the tender of the applicable purchase price in immediately available funds, the Holder shall be deemed to be the holder of record of the shares of Common Stock issuable upon such exercise, notwithstanding that the stock transfer books of Issuer shall then be closed or that certificates representing such shares of Common Stock shall not then be actually delivered to the

Holder. Issuer shall pay all expenses, and any and all United States federal, state and local taxes and other charges that may be payable in connection with the preparation, issue and delivery of stock certificates under this Section 2 in the name of the Holder or its assignee, transferee or designee.

- 3. Issuer agrees: (i) that it shall at all times maintain, free from preemptive rights, sufficient authorized but unissued or treasury shares of Common Stock so that the Option may be exercised without additional authorization of Common Stock after giving effect to all other options, warrants, convertible securities and other rights to purchase Common Stock; (ii) that it will not, by charter amendment or through reorganization, consolidation, merger, dissolution or sale of assets, or by any other voluntary act, avoid or seek to avoid the observance or performance of any of the covenants, stipulations or conditions to be observed or performed hereunder by Issuer; (iii) promptly to take all action as may from time to time be required (including (x) complying with all premerger notification, reporting and waiting period requirements specified in 15 U.S.C. 18a and regulations promulgated thereunder and (y) in the event, under the Bank Holding Company Act of 1956, as amended (the "BHCA"), or the Change in Bank Control Act of 1978, as amended, or any state banking law, prior approval of or notice to the Federal Reserve Board or to any state or other regulatory authority is necessary before the Option may be exercised, cooperating fully with the Holder in preparing such applications or notices and providing such information to the Federal Reserve Board or such other regulatory authority as they may require) in order to permit the Holder to exercise the Option and Issuer duly and effectively to issue shares of Common Stock pursuant hereto; and (iv) promptly to take all action provided herein to protect the rights of the Holder against dilution.
- 4. This Agreement (and the Option granted hereby) are exchangeable, without expense, at the option of the Holder, upon presentation and surrender of this Agreement at the principal office of Issuer, for other Agreements providing for Options of different denominations entitling the holder thereof to purchase, on the same terms and subject to the same conditions as are set forth herein, in the aggregate the same number of shares of Common Stock purchasable hereunder. The terms "Agreement" and "Option" as used herein include any Stock Option Agreements and related Options for which this Agreement (and the Option granted hereby) may be exchanged. Upon receipt by Issuer of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Agreement, and (in the case of loss, theft or destruction) of reasonably satisfactory indemnification, and upon surrender and cancellation of this Agreement, if mutilated, Issuer will execute and deliver a new Agreement of like tenor and date. Any such new Agreement executed and delivered shall constitute an additional contractual obligation on the part of Issuer, whether or not the Agreement so lost, stolen, destroyed or mutilated shall at any time be enforceable by anyone.
- 5. In addition to the adjustment in the number of shares of Common Stock that are purchasable upon exercise of the Option pursuant to Section 1 of this Agreement, the number of shares of Common Stock purchasable upon the exercise of the Option and the Option Price shall be subject to adjustment from time to time as provided in this Section 5. In the event of any change in, or distributions in respect of, the Common Stock by reason of stock dividends, split-ups, mergers, recapitalizations, combinations, subdivisions, conversions, exchanges of shares, distributions on or in respect of the Common Stock that would be prohibited under the terms of the Merger Agreement, or the like, the type and number of shares of Common Stock purchasable upon exercise hereof and the Option Price shall be appropriately adjusted in such manner as shall fully preserve the economic benefits provided hereunder and proper provision shall be made in any agreement governing any such transaction to provide for such proper adjustment and the full satisfaction of the Issuer's obligations hereunder.
- 6. Upon the occurrence of a Subsequent Triggering Event that occurs prior to an Exercise Termination Event, Issuer shall, at the request of Grantee delivered within 180 days of such Subsequent Triggering Event (whether on its own behalf of any subsequent holder of this Option (or part thereof) or any of the shares of Common Stock issued pursuant hereto), promptly prepare, file and keep current a shelf registration statement under the 1933 Act covering this Option and any shares issued and issuable pursuant to this Option and shall use its reasonable best efforts to cause such registration statement to become effective and remain current in order to permit the sale or other disposition of this Option and any shares of Common Stock issued upon total or

partial exercise of this Option ("Option Shares") in accordance with any plan of disposition requested by Grantee. Issuer will use its reasonable best efforts to cause such registration statement first to become effective and then to remain effective for such period not in excess of 180 days from the day such registration statement first becomes effective or such shorter time as may be reasonably necessary to effect such sales or other dispositions. Grantee shall have the right to demand two such registrations. The foregoing notwithstanding, if, at the time of any request by Grantee for registration of the Option or Option Shares as provided above, Issuer is in registration with respect to an underwritten public offering of shares of Common Stock, and if in the good faith judgment of the managing underwriter or managing underwriters, or, if none, the sole underwriter or underwriters, of such offering the inclusion of the Holder's Option or Option Shares would interfere with the successful marketing of the shares of Common Stock offered by Issuer, the number of Option Shares otherwise to be covered in the registration statement contemplated hereby may be reduced; provided, however, that after any such required reduction the number of Option Shares to be included in such offering for the account of the Holder shall constitute at least 25% of the total number of shares to be sold by the Holder and Issuer in the aggregate; and provided further, however, that if such reduction occurs, then the Issuer shall file a registration statement for the balance as promptly as practicable and no reduction shall thereafter occur. Each such Holder shall provide all information reasonably requested by Issuer for inclusion in any registration statement to be filed hereunder. If requested by any such Holder in connection with such registration. Issuer shall become a party to any underwriting agreement relating to the sale of such shares, but only to the extent of obligating itself in respect of representations, warranties, indemnities and other agreements customarily included in secondary offering underwriting agreements for the Issuer. Upon receiving any request under this Section 6 from any Holder, Issuer agrees to send a copy thereof to any other person known to Issuer to be entitled to registration rights under this Section 6, in each case by promptly mailing the same, postage prepaid, to the address of record of the persons entitled to receive such copies. Notwithstanding anything to the contrary contained herein, in no event shall Issuer be obligated to effect more than two registrations pursuant to this Section 6 by reason of the fact that there shall be more than one Grantee as a result of any assignment or division of this Agreement.

7. (a) Immediately prior to the occurrence of a Repurchase Event (as defined below), (i) following a request of the Holder, delivered prior to an Exercise Termination Event, Issuer (or any successor thereto) shall repurchase the Option from the Holder at a price (the "Option Repurchase Price") equal to the number of shares for which this Option may then be exercised multiplied by the amount by which (A) the Market/Offer Price (as defined below) exceeds (B) the Option Price, and (ii) at the request of the owner of Option Shares from time to time (the "Owner"), delivered within 90 days of such occurrence (or such later period as provided in Section 10), Issuer shall repurchase such number of the Option Shares from the Owner as the Owner shall designate at a price (the "Option Share Repurchase Price") equal to the Market/Offer Price multiplied by the number of Option Shares so designated. The term "Market/Offer Price" shall mean the highest of (i) the price per share of Common Stock at which a tender offer or exchange offer therefor has been made, (ii) the price per share of Common Stock to be paid by any third party pursuant to an agreement with Issuer, (iii) the highest closing price for shares of Common Stock within the six-month period immediately preceding the date the Holder gives notice of the required repurchase of this Option or the Owner gives notice of the required repurchase of Option Shares, as the case may be, or (iv) in the event of a sale of all or a substantial portion of Issuer's assets, the sum of the price paid in such sale for such assets and the current market value of the remaining assets of Issuer as determined by a nationally recognized investment banking firm selected by the Holder or Owner, as the case may be, and reasonably acceptable to the Issuer. divided by the number of shares of Common Stock of Issuer outstanding at the time of such sale. In determining the Market/Offer Price, the value of consideration other than cash shall be determined by a nationally recognized investment banki

(b) The Holder and the Owner, as the case may be, may exercise its right to require Issuer to repurchase the Option and any Option Shares pursuant to this Section 7 by surrendering for such purpose to Issuer, at its principal office, this Agreement or certificates for Option Shares, as applicable, accompanied by a written notice or notices stating that the Holder or the Owner, as the case may be, elects to require Issuer to repurchase this Option and/or the Option Shares in accordance with the provisions of this Section 7. Within

the latter to occur of (x) five business days after the surrender of the Option and/or certificates representing Option Shares and the receipt of such notice or notices relating thereto and (y) the time that is immediately prior to the occurrence of a Repurchase Event, Issuer shall deliver or cause to be delivered to the Holder the Option Repurchase Price and/or to the Owner the Option Share Repurchase Price therefor or the portion thereof, if any, that Issuer is not then prohibited under applicable law and regulation from so delivering.

- (c) To the extent that Issuer is prohibited under applicable law or regulation from repurchasing the Option and/or the Option Shares in full, Issuer shall immediately so notify the Holder and/or the Owner and thereafter deliver or cause to be delivered, from time to time, to the Holder and/or the Owner, as appropriate, the portion of the Option Repurchase Price and the Option Share Repurchase Price, respectively, that it is no longer prohibited from delivering, within five business days after the date on which Issuer is no longer so prohibited; *provided*, *however*, that if Issuer at any time after delivery of a notice of repurchase pursuant to paragraph (b) of this Section 7 is prohibited under applicable law or regulation from delivering to the Holder and/or the Owner, as appropriate, the Option Repurchase Price and the Option Share Repurchase Price, respectively, in full (and Issuer hereby undertakes to use its best efforts to obtain all required regulatory and legal approvals and to file any required notices, in each case as promptly as practicable in order to accomplish such repurchase), the Holder or Owner may revoke its notice of repurchase of the Option or the Option Shares either in whole or to the extent of the prohibition, whereupon, in the latter case, Issuer shall promptly (i) deliver to the Holder and/or the Owner, as appropriate, that portion of the Option Repurchase Price or the Option Share Repurchase Price that Issuer is not prohibited from delivering; and (ii) deliver, as appropriate, either (A) to the Holder, a new Stock Option Agreement evidencing the right of the Holder to purchase that number of shares of Common Stock obtained by multiplying the number of shares of Common Stock for which the surrendered Stock Option Agreement was exercisable at the time of delivery of the notice of repurchase by a fraction, the numerator of which is the Option Repurchase Price less the portion thereof theretofore delivered to the Holder and the denominator of which is the Option Repurchase Price for the
- (d) For purposes of this Section 7, a Repurchase Event shall be deemed to have occurred (i) upon the consummation of any merger, consolidation, share exchange or similar transaction involving Issuer or any purchase, lease or other acquisition of all or a substantial portion of the assets of Issuer, other than any such transaction which would not constitute an Acquisition Transaction pursuant to the provisos to Section 2(b)(i) hereof or (ii) upon the acquisition by any person of beneficial ownership of 50% or more of the then outstanding shares of Common Stock, provided that no such event shall constitute a Repurchase Event unless a Subsequent Triggering Event shall have occurred prior to an Exercise Termination Event. The parties hereto agree that Issuer's obligations to repurchase the Option or Option Shares under this Section 7 shall not terminate upon the occurrence of an Exercise Termination Event unless no Subsequent Triggering Event shall have occurred prior to the occurrence of an Exercise Termination Event.
- 8. (a) In the event that prior to an Exercise Termination Event, Issuer shall enter into an agreement (i) to consolidate with or merge into any person, other than Grantee or one of its Subsidiaries, and shall not be the continuing or surviving corporation of such consolidation or merger, (ii) to permit any person, other than Grantee or one of its Subsidiaries, to merge into Issuer and Issuer shall be the continuing or surviving corporation, but, in connection with such merger, the then outstanding shares of Common Stock shall be changed into or exchanged for stock or other securities of any other person or cash or any other property or the then outstanding shares of Common Stock shall after such merger represent less than 50% of the outstanding voting shares and voting shares quivalents of the merged company, or (iii) to sell or otherwise transfer all or substantially all of its assets to any person, other than Grantee or one of its Subsidiaries, then, and in each such case, the agreement governing such transaction shall make proper provision so that the Option shall, upon the consummation of any such transaction and upon the terms and conditions set forth herein, be converted into, or exchanged for, an option (the "Substitute Option"), at the election of the Holder, of either (x) the Acquiring Corporation (as hereinafter defined) or (y) any person that controls the Acquiring Corporation.

- (b) The following terms have the meanings indicated:
- (A) "Acquiring Corporation" shall mean (i) the continuing or surviving corporation of a consolidation or merger with Issuer (if other than Issuer), (ii) Issuer in a merger in which Issuer is the continuing or surviving person, and (iii) the transferee of all or substantially all of Issuer's assets.
 - (B) "Substitute Common Stock" shall mean the common stock issued by the issuer of the Substitute Option upon exercise of the Substitute Option.
 - (C) "Assigned Value" shall mean the Market/Offer Price, as defined in Section 7.
- (D) "Average Price" shall mean the average closing price of a share of the Substitute Common Stock for the one year immediately preceding the consolidation, merger or sale in question, but in no event higher than the closing price of the shares of Substitute Common Stock on the day preceding such consolidation, merger or sale; *provided* that if Issuer is the issuer of the Substitute Option, the Average Price shall be computed with respect to a share of common stock issued by the person merging into Issuer or by any company which controls or is controlled by such person, as the Holder may elect.
- (c) The Substitute Option shall have the same terms as the Option, *provided*, that if the terms of the Substitute Option cannot, for legal reasons, be the same as the Option, such terms shall be as similar as possible and in no event less advantageous to the Holder. The issuer of the Substitute Option shall also enter into an agreement with the then Holder or Holders of the Substitute Option in substantially the same form as this Agreement, which shall be applicable to the Substitute Option.
- (d) The Substitute Option shall be exercisable for such number of shares of Substitute Common Stock as is equal to the Assigned Value multiplied by the number of shares of Common Stock for which the Option is then exercisable, divided by the Average Price. The exercise price of the Substitute Option per share of Substitute Common Stock shall then be equal to the Option Price multiplied by a fraction, the numerator of which shall be the number of shares of Common Stock for which the Option is then exercisable and the denominator of which shall be the number of Substitute Common Stock for which the Substitute Option is exercisable.
- (e) In no event, pursuant to any of the foregoing paragraphs, shall the Substitute Option be exercisable for more than 19.9% of the shares of Substitute Common Stock outstanding prior to exercise of the Substitute Option. In the event that the Substitute Option would be exercisable for more than 19.9% of the shares of Substitute Common Stock outstanding prior to exercise but for this clause (e), the issuer of the Substitute Option (the "Substitute Option Issuer") shall make a cash payment to Holder equal to the excess of (i) the value of the Substitute Option without giving effect to the limitation in this clause (e) over (ii) the value of the Substitute Option after giving effect to the limitation in this clause (e). This difference in value shall be determined by a nationally recognized investment banking firm selected by the Holder or the Owner, as the case may be, and reasonably acceptable to the Acquiring Corporation.
- (f) Issuer shall not enter into any transaction described in subsection (a) of this Section 8 unless the Acquiring Corporation and any person that controls the Acquiring Corporation assume in writing all the obligations of Issuer hereunder.
- 9. (a) At the request of the holder of the Substitute Option (the "Substitute Option Holder"), the Substitute Option Issuer shall repurchase the Substitute Option from the Substitute Option Holder at a price (the "Substitute Option Repurchase Price") equal to the amount by which (i) the Highest Closing Price (as hereinafter defined) exceeds (ii) the exercise price of the Substitute Option, multiplied by the number of shares of Substitute Common Stock for which the Substitute Option may then be exercised, and at the request of the owner (the "Substitute Share Owner") of shares of Substitute Common Stock (the "Substitute Shares"), the Substitute Option Issuer shall repurchase the Substitute Shares at a price (the "Substitute Share Repurchase Price") equal to the Highest

Closing Price multiplied by the number of Substitute Shares so designated. The term "Highest Closing Price" shall mean the highest closing price for shares of Substitute Common Stock within the six-month period immediately preceding the date the Substitute Option Holder gives notice of the required repurchase of the Substitute Option or the Substitute Share Owner gives notice of the required repurchase of the Substitute Shares, as applicable.

- (b) The Substitute Option Holder and the Substitute Share Owner, as the case may be, may exercise its respective right to require the Substitute Option Issuer to repurchase the Substitute Option and the Substitute Shares pursuant to this Section 9 by surrendering for such purpose to the Substitute Option Issuer, at its principal office, the agreement for such Substitute Option (or, in the absence of such an agreement, a copy of this Agreement) and certificates for Substitute Shares accompanied by a written notice or notices stating that the Substitute Option Holder or the Substitute Share Owner, as the case may be, elects to require the Substitute Option Issuer to repurchase the Substitute Option and/or the Substitute Shares in accordance with the provisions of this Section 9. As promptly as practicable, and in any event within five business days after the surrender of the Substitute Option and/or certificates representing Substitute Shares and the receipt of such notice or notices relating thereto, the Substitute Option Issuer shall deliver or cause to be delivered to the Substitute Option Holder the Substitute Option Repurchase Price and/or to the Substitute Share Owner the Substitute Share Repurchase Price therefor or, in either case, the portion thereof which the Substitute Option Issuer is not then prohibited under applicable law and regulation from so delivering.
- (c) To the extent that the Substitute Option Issuer is prohibited under applicable law or regulation from repurchasing the Substitute Option and/or the Substitute Shares in part or in full, the Substitute Option Issuer following a request for repurchase pursuant to this Section 9 shall immediately so notify the Substitute Option Holder and/or the Substitute Share Owner and thereafter deliver or cause to be delivered, from time to time, to the Substitute Option Holder and/or the Substitute Share Owner, as appropriate, the portion of the Substitute Share Repurchase Price, respectively, which it is no longer prohibited from delivering, within five business days after the date on which the Substitute Option Issuer is no longer so prohibited; provided, however, that if the Substitute Option Issuer is at any time after delivery of a notice of repurchase pursuant to subsection (b) of this Section 9 prohibited under applicable law or regulation from delivering to the Substitute Option Holder and/or the Substitute Share Owner, as appropriate, the Substitute Option Repurchase Price and the Substitute Share Repurchase Price, respectively, in full (and the Substitute Option Issuer shall use its best efforts to obtain all required regulatory and legal approvals, in each case as promptly as practicable, in order to accomplish such repurchase), the Substitute Option Holder or Substitute Share Owner may revoke its notice of repurchase of the Substitute Option or the Substitute Shares either in whole or to the extent of the prohibition, whereupon, in the latter case, the Substitute Option Issuer shall promptly (i) deliver to the Substitute Option Holder or Substitute Share Owner, as appropriate, that portion of the Substitute Option Repurchase Price or the Substitute Share Repurchase Price that the Substitute Option Issuer is not prohibited from delivering; and (ii) deliver, as appropriate, either (A) to the Substitute Option Holder, a new Substitute Option evidencing the right of the Substitute Option Holder to purchase that number of shares of the Substitute Common Stock obtained by multiplying the number of shares of the Substitute Common Stock for which the surrendered Substitute Option was exercisable at the time of delivery of the notice of repurchase by a fraction, the numerator of which is the Substitute Option Repurchase Price less the portion thereof theretofore delivered to the Substitute Option Holder and the denominator of which is the Substitute Option Repurchase Price, or (B) to the Substitute Share Owner, a certificate for the Substitute Common Shares it is then so prohibited from repurchasing.
- 10. The 90-day or 180-day periods for exercise of certain rights under Sections 2, 6, 7 and 13 shall be extended: (i) to the extent necessary to obtain all regulatory approvals for the exercise of such rights and for the expiration of all statutory waiting periods; and (ii) to the extent necessary to avoid liability under Section 16(b) of the 1934 Act by reason of such exercise.

- 11. Issuer hereby represents and warrants to Grantee as follows:
- (a) Issuer has full corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized by the Board of Directors of Issuer and no other corporate proceedings on the part of Issuer are necessary to authorize this Agreement or to consummate the transactions so contemplated. This Agreement has been duly and validly executed and delivered by Issuer.
- (b) Issuer has taken all necessary corporate action to authorize and reserve and to permit it to issue, and at all times from the date hereof through the termination of this Agreement in accordance with its terms will have reserved for issuance upon the exercise of the Option, that number of shares of Common Stock equal to the maximum number of shares of Common Stock at any time and from time to time issuable hereunder, and all such shares, upon issuance pursuant hereto, will be duly authorized, validly issued, fully paid, nonassessable, and will be delivered free and clear of all claims, liens, encumbrance and security interests and not subject to any preemptive rights.
- (c) Issuer has taken all action (including if required redeeming all of the Rights or amending or terminating the Rights Agreement) so that the entering into of this Option Agreement, the acquisition of shares of Common Stock hereunder and the other transactions contemplated hereby do not and will not result in the grant of any rights to any person under the Rights Agreement or enable or require the Rights to be exercised, distributed or triggered.
- 12. Grantee hereby represents and warrants to Issuer that:
- (a) Grantee has all requisite corporate power and authority to enter into this Agreement and, subject to any approvals or consents referred to herein, to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Grantee. This Agreement has been duly executed and delivered by Grantee.
- (b) The Option is not being, and any shares of Common Stock or other securities acquired by Grantee upon exercise of the Option will not be, acquired with a view to the public distribution thereof and will not be transferred or otherwise disposed of except in a transaction registered or exempt from registration under the 1933 Act.
- 13. Neither of the parties hereto may assign any of its rights or obligations under this Agreement or the Option created hereunder to any other person, without the express written consent of the other party, except that in the event a Subsequent Triggering Event shall have occurred prior to an Exercise Termination Event, Grantee, subject to the express provisions hereof, may assign in whole or in part its rights and obligations hereunder within 180 days following such Subsequent Triggering Event (or such later period as provided in Section 10); provided, however, that until the date 15 days following the date on which the Federal Reserve Board approves an application by Grantee under the BHCA to acquire the shares of Common Stock subject to the Option, Grantee may not assign its rights under the Option except in (i) a widely dispersed public distribution, (ii) a private placement in which no one party acquires the right to purchase in excess of 2% of the voting shares of Issuer, (iii) an assignment to a single party (e.g., a broker or investment banker) for the purpose of conducting a widely dispersed public distribution on Grantee's behalf, or (iv) any other manner approved by the Federal Reserve Board.
- 14. Each of Grantee and Issuer will use its best efforts to make all filings with, and to obtain consents of, all third parties and governmental authorities necessary to the consummation of the transactions contemplated by this Agreement, including without limitation making application to list the shares of Common Stock issuable hereunder on the New York Stock Exchange upon official notice of issuance and applying to the Federal Reserve

Board under the BHCA for approval to acquire the shares issuable hereunder, but Grantee shall not be obligated to apply to state banking authorities for approval to acquire the shares of Common Stock issuable hereunder until such time, if ever, as it deems appropriate to do so.

- 15. (a) Grantee may, at any time during which Issuer would be required to repurchase the Option or any Option Shares pursuant to Section 7 upon proper request or notice, surrender the Option (together with any Option Shares issued to and then owned by Grantee) to Issuer in exchange for a cash fee equal to the Surrender Price (as defined below); provided, however, that Grantee may not exercise its rights pursuant to this Section 15 if Issuer has repurchased the Option (or any portion thereof) or any Option Shares pursuant to Section 7. The "Surrender Price" shall be equal to (i) \$1.8948 billion, plus (ii) if applicable, the aggregate purchase price previously paid pursuant hereto by Grantee with respect to any Option Shares, minus (iii) if applicable, the sum of (A) the excess of (1) the net cash amounts, if any, received by Grantee pursuant to the arms' length sale of Option Shares (or any other securities into which such Option Shares were converted or exchanged) to any party not affiliated with Grantee, over (2) the aggregate purchase price previously paid pursuant hereto by Grantee with respect to such Option Shares and (B) the net cash amounts, if any, received by Grantee pursuant to an arms' length sale of a portion of the Option to any party not affiliated with Grantee.
 - (b) Grantee may exercise its right to surrender the Option and any Option Shares pursuant to this Section 15 by surrendering to Issuer, at its principal office, this Agreement together with certificates for Option Shares, if any, accompanied by a written notice stating (i) that Grantee elects to surrender the Option and Option Shares, if any, in accordance with the provisions of this Section 15 and (ii) the Surrender Price. The Surrender Price shall be payable in immediately available funds on or before the second business day following receipt of such notice by Issuer.
 - (c) To the extent that Issuer is prohibited under applicable law or regulation from paying the Surrender Price to Grantee in full, Issuer shall immediately so notify Grantee and thereafter deliver or cause to be delivered, from time to time, to Grantee, the portion of the Surrender Price that Issuer is no longer prohibited from paying, within five business days after the date on which Issuer is no longer so prohibited, *provided*, *however*, that if Issuer at any time after delivery of a notice of surrender pursuant to paragraph (b) of this Section 15 is prohibited under applicable law or regulation from paying to Grantee the Surrender Price in full (i) Issuer shall (A) use its reasonable best efforts to obtain all required regulatory and legal approvals and to file any required notices as promptly as practicable in order to make such payments, (B) within five days of the submission or receipt of any documents relating to any such regulatory and legal approvals, provide Grantee with copies of the same, and (C) keep Grantee advised of both the status of any such request for regulatory and legal approvals, as well as any discussions with any relevant regulatory or other third party reasonably related to the same and (ii) Grantee may revoke such notice of surrender by delivery of a notice of revocation to Issuer and, upon delivery of such notice of revocation, the Exercise Termination Date shall be extended to a date six months from the date on which the Exercise Termination Date would have occurred if not for the provisions of this Section 15(c) (during which period Grantee may exercise any of its rights hereunder, including any and all rights pursuant to this Section 15).
 - (d) Grantee shall have rights substantially identical to those set forth in paragraphs (a),(b) and (c) of this Section 15 with respect to the Substitute Option and the Substitute Option Issuer during any period in which the Substitute Option Issuer would be required to repurchase the Substitute Option pursuant to Section 9.
- 16. (a) Notwithstanding any other provision of this Agreement, in no event shall the Grantee's Total Profit (as hereinafter defined) exceed \$2.3685 billion and, if it otherwise would exceed such amount, the Grantee, at its sole election, shall either (i) reduce the number of shares of Common Stock subject to this Option, (ii) deliver to Issuer for cancellation Option Shares previously purchased by Grantee, (iii) pay cash to Issuer, or (iv) any combination thereof, so that Grantee's actually realized Total Profit shall not exceed \$2.3685 billion after taking into account the foregoing actions.

- (b) Notwithstanding any other provision of this Agreement, this Option may not be exercised for a number of shares as would, as of the date of exercise, result in a Notional Total Profit (as defined below) of more than \$2.3685 billion; *provided* that nothing in this sentence shall restrict any exercise of the Option permitted hereby on any subsequent date.
- (c) As used herein, the term "Total Profit" shall mean the aggregate amount (before taxes) of the following: (i) the amount received by Grantee pursuant to Issuer's repurchase of the Option (or any portion thereof) pursuant to Section 7, (ii) (x) the amount received by Grantee pursuant to Issuer's repurchase of Option Shares pursuant to Section 7, less (y) the Grantee's purchase price for such Option Shares, (iii) (x) the net cash amounts received by Grantee pursuant to the sale of Option Shares (or any other securities into which such Option Shares are converted or exchanged) to any unaffiliated party, less (y) the Grantee's purchase price of such Option Shares, (iv) any amounts received by Grantee on the transfer of the Option (or any portion thereof) to any unaffiliated party, and (v) any amount equivalent to the foregoing with respect to the Substitute Option.
- (d) As used herein, the term "Notional Total Profit" with respect to any number of shares as to which Grantee may propose to exercise this Option shall be the Total Profit determined as of the date of such proposed exercise assuming that this Option were exercised on such date for such number of shares and assuming that such shares, together with all other Option Shares held by Grantee and its affiliates as of such date, were sold for cash at the closing market price for the Common Stock as of the close of business on the preceding trading day (less customary brokerage commissions).
- 17. The parties hereto acknowledge that damages would be an inadequate remedy for a breach of this Agreement by either party hereto and that the obligations of the parties hereto shall be enforceable by either party hereto through injunctive or other equitable relief.
- 18. If any term, provision, covenant or restriction contained in this Agreement is held by a court or a federal or state regulatory agency of competent jurisdiction to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants and restrictions contained in this Agreement shall remain in full force and effect, and shall in no way be affected, impaired or invalidated. If for any reason such court or regulatory agency determines that the Holder is not permitted to acquire, or Issuer is not permitted to repurchase pursuant to Section 7, the full number of shares of Common Stock provided in Section 1(a) hereof (as adjusted pursuant to Section 1(b) or 5 hereof), it is the express intention of Issuer to allow the Holder to acquire or to require Issuer to repurchase such lesser number of shares as may be permissible, without any amendment or modification hereof.
- 19. All notices, requests, claims, demands and other communications hereunder shall be deemed to have been duly given when delivered in person, by cable, telegram, telecopy or telex, or by registered or certified mail (postage prepaid, return receipt requested) at the respective addresses of the parties set forth in the Merger Agreement.
- 20. This Agreement shall be governed by and construed in accordance with the laws of the State of New York, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws thereof (except to the extent that mandatory provisions of federal or state law apply).
 - 21. This Agreement may be executed in two counterparts, each of which shall be deemed to be an original, but all of which shall constitute one and the same agreement.
- 22. Except as otherwise expressly provided herein, each of the parties hereto shall bear and pay all costs and expenses incurred by it or on its behalf in connection with the transactions contemplated hereunder, including fees and expenses of its own financial consultants, investment bankers, accountants and counsel.
- 23. Except as otherwise expressly provided herein or in the Merger Agreement, this Agreement contains the entire agreement between the parties with respect to the transactions contemplated hereunder and supersedes all

prior arrangements or understandings with respect thereof, written or oral. The terms and conditions of this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns. Nothing in this Agreement, expressed or implied, is intended to confer upon any party, other than the parties hereto, and their respective successors and permitted assigns, any rights, remedies, obligations or liabilities under or by reason of this Agreement, except as expressly provided herein.

24. Capitalized terms used in this Agreement and not defined herein shall have the meanings assigned thereto in the Merger Agreement.

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be executed on its behalf by its officers thereunto duly authorized, all as of the date first above written.

BANK OF AMERICA CORPORATION (Grantee)

By: /s/ Kenneth D. Lewis

Name: Kenneth D. Lewis

Title: Chairman, President and Chief Executive Officer

FLEETBOSTON FINANCIAL CORPORATION (Issuer)

By: /s/ Charles K. Gifford

Name: Charles K. Gifford

Title: Chairman and Chief Executive Officer

Signature Page to FleetBoston Stock Option Agreement

Appendix C

THE TRANSFER OF THIS AGREEMENT IS SUBJECT TO CERTAIN PROVISIONS CONTAINED HEREIN AND TO RESALE RESTRICTIONS UNDER THE SECURITIES ACT OF 1933, AS AMENDED

STOCK OPTION AGREEMENT, dated October 27, 2003, between Bank of America Corporation, a Delaware corporation ("Issuer"), and FleetBoston Financial Corporation, a Rhode Island corporation ("Grantee").

<u>W I T N E S S E T H</u>:

WHEREAS, Grantee and Issuer have entered into an Agreement and Plan of Merger of even date herewith (the "Merger Agreement"), which agreement has been executed by the parties hereto in connection with this Stock Option Agreement (the "Agreement"); and

WHEREAS, as a condition to Grantee's entering into the Merger Agreement and in consideration therefor, Issuer has agreed to grant Grantee the Option (as hereinafter defined);

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants and agreements set forth herein and in the Merger Agreement, the parties hereto agree as follows:

- 1. (a) Issuer hereby grants to Grantee an unconditional, irrevocable option (the "Option") to purchase, subject to the terms hereof, up to 297,374,945 fully paid and nonassessable shares of Issuer's Common Stock, par value \$0.01 per share ("Common Stock"), at a price of \$81.86 per share (the "Option Price"); provided, however, that in no event shall the number of shares of Common Stock for which this Option is exercisable exceed 19.9% of the Issuer's issued and outstanding shares of Common Stock without giving effect to any shares subject to or issued pursuant to the Option. The number of shares of Common Stock that may be received upon the exercise of the Option and the Option Price are subject to adjustment as herein set forth.
 - (b) In the event that any additional shares of Common Stock are either (i) issued or otherwise become outstanding after the date of this Agreement (other than pursuant to this Agreement) or (ii) redeemed, repurchased, retired or otherwise cease to be outstanding after the date of this Agreement, the number of shares of Common Stock subject to the Option shall be increased or decreased, as appropriate, so that, after such issuance, such number equals 19.9% of the number of shares of Common Stock then issued and outstanding without giving effect to any shares subject or issued pursuant to the Option. Nothing contained in this Section 1(b) or elsewhere in this Agreement shall be deemed to authorize Issuer or Grantee to breach any provision of the Merger Agreement.
- 2. (a) The Holder (as hereinafter defined) may exercise the Option, in whole or part, and from time to time, if, but only if, both an Initial Triggering Event (as hereinafter defined) and a Subsequent Triggering Event (as hereinafter defined) shall have occurred prior to the occurrence of an Exercise Termination Event (as hereinafter defined), provided that the Holder shall have sent the written notice of such exercise (as provided in subsection (e) of this Section 2) within 180 days following such Subsequent Triggering Event. Each of the following shall be an "Exercise Termination Event": (i) the Effective Time (as defined in the Merger Agreement) of the Merger; (ii) termination of the Merger Agreement in accordance with the provisions thereof if such termination occurs prior to the occurrence of an Initial Triggering Event except a termination by Grantee pursuant to Section 8.1(d) of the Merger Agreement (unless the breach by Issuer giving rise to such right of termination is non-volitional); or (iii) the passage of 18 months after termination of the Merger Agreement if such termination follows the occurrence of an Initial Triggering Event or is a termination by Grantee pursuant to Section 8.1(d) of the Merger Agreement (unless the breach by Issuer giving rise to such right of termination is non-volitional). The term "Holder" shall mean the holder or holders of the Option.
 - (b) The term "Initial Triggering Event" shall mean any of the following events or transactions occurring after the date hereof:
 - (i) Issuer or any of its Subsidiaries (each an "Issuer Subsidiary"), without having received Grantee's prior written consent, shall have entered into an agreement to engage in an Acquisition

Transaction (as hereinafter defined) with any person (the term "person" for purposes of this Agreement having the meaning assigned thereto in Sections 3(a)(9) and 13(d)(3) of the Securities Exchange Act of 1934, as amended (the "1934 Act"), and the rules and regulations thereunder) other than Grantee or any of its Subsidiaries (each a "Grantee Subsidiary") or the Board of Directors of Issuer shall have recommended that the stockholders of Issuer approve or accept any Acquisition Transaction with any person other than Grantee or a Subsidiary of Grantee. For purposes of this Agreement, "Acquisition Transaction" shall mean (w) a merger, consolidation or share exchange, or any similar transaction, involving Issuer or any Significant Subsidiary (as defined in Rule 1-02 of Regulation S-X promulgated by the Securities and Exchange Commission (the "SEC")) of Issuer, (x) a purchase, lease or other acquisition or assumption of all or a substantial portion of the assets or deposits of Issuer or any Significant Subsidiary of Issuer, (y) a purchase or other acquisition (including by way of merger, consolidation, share exchange or otherwise) of securities representing 10% or more of the voting power of Issuer, or (z) any substantially similar transaction; provided, however, that in no event shall any merger, consolidation, purchase or similar transaction involving only any two or more of such Subsidiaries, be deemed to be an Acquisition Transaction, provided that any such transaction is not entered into in violation of the terms of the Merger Agreement; and, *provided further*, that in no event shall any merger, consolidation, share exchange or similar transaction in which the shareholders of B, as such, shall in the aggregate own at least 60% of the voting common shares (or equivalents) of the surviving or acquiring entity immediately following the consummation of such transaction be deemed to be an Acquisition Transaction;

- (ii) Issuer or any Issuer Subsidiary, without having received Grantee's prior written consent, shall have authorized, recommended, proposed or publicly announced its intention to authorize, recommend or propose, to engage in an Acquisition Transaction with any person other than Grantee or a Grantee Subsidiary, or the Board of Directors of Issuer shall have publicly withdrawn or modified, or publicly announced its intention to withdraw or modify, in any manner adverse to Grantee, its recommendation that the stockholders of Issuer approve the transactions contemplated by the Merger Agreement;
- (iii) Any person other than Grantee, any Grantee Subsidiary or any Issuer Subsidiary acting in a fiduciary capacity in the ordinary course of its business shall have acquired beneficial ownership or the right to acquire beneficial ownership of 10% or more of the outstanding shares of Common Stock (the term "beneficial ownership" for purposes of this Agreement having the meaning assigned thereto in Section 13(d) of the 1934 Act, and the rules and regulations thereunder);
- (iv) Any person other than Grantee or any Grantee Subsidiary shall have made a bona fide proposal to Issuer or its stockholders by public announcement or written communication that is or becomes the subject of public disclosure to engage in an Acquisition Transaction;
- (v) After an overture is made by a third party to Issuer or its stockholders to engage in an Acquisition Transaction, Issuer shall have breached any covenant or obligation contained in the Merger Agreement and such breach (x) would entitle Grantee to terminate the Merger Agreement and (y) shall not have been cured prior to the Notice Date (as defined below); or
- (vi) Any person other than Grantee or any Grantee Subsidiary, other than in connection with a transaction to which Grantee has given its prior written consent, shall have filed an application or notice with the Federal Reserve Board, or other federal or state bank regulatory authority, which application or notice has been accepted for processing, for approval to engage in an Acquisition Transaction.
- (c) The term "Subsequent Triggering Event" shall mean either of the following events or transactions occurring after the date hereof:
 - (i) The acquisition by any person of beneficial ownership of 20% or more of the then outstanding Common Stock; or

- (ii) The occurrence of the Initial Triggering Event described in paragraph (i) of subsection (b) of this Section 2, except that the percentage referred to in clause (v) shall be 20%.
- (d) Issuer shall notify Grantee promptly in writing of the occurrence of any Initial Triggering Event or Subsequent Triggering Event of which it has knowledge (together, a "Triggering Event"), it being understood that the giving of such notice by Issuer shall not be a condition to the right of the Holder to exercise the Option.
- (e) In the event the Holder is entitled to and wishes to exercise the Option, it shall send to Issuer a written notice (the date of which being herein referred to as the "Notice Date") specifying (i) the total number of shares it will purchase pursuant to such exercise and (ii) a place and date not earlier than three business days nor later than 60 business days from the Notice Date for the closing of such purchase (the "Closing Date"); provided that if prior notification to or approval of the Federal Reserve Board or any other regulatory agency is required in connection with such purchase, the Holder shall as soon as reasonably practicable file the required notice or application for approval and shall expeditiously process the same and the period of time that otherwise would run pursuant to this sentence shall run instead from the date on which any required notification periods have expired or been terminated or such approvals have been obtained and any requisite waiting period or periods shall have passed. Any exercise of the Option shall be deemed to occur on the Notice Date relating thereto.
- (f) At the closing referred to in subsection (e) of this Section 2, the Holder shall pay to Issuer the aggregate purchase price for the shares of Common Stock purchased pursuant to the exercise of the Option in immediately available funds by wire transfer to a bank account designated by Issuer, *provided* that failure or refusal of Issuer to designate such a bank account shall not preclude the Holder from exercising the Option.
- (g) At such closing, simultaneously with the delivery of immediately available funds as provided in subsection (f) of this Section 2, Issuer shall deliver to the Holder a certificate or certificates representing the number of shares of Common Stock purchased by the Holder and, if the Option should be exercised in part only, a new Option evidencing the rights of the Holder thereof to purchase the balance of the shares purchasable hereunder, and the Holder shall deliver to Issuer this Agreement and a letter agreeing that the Holder will not offer to sell or otherwise dispose of such shares in violation of applicable law or the provisions of this Agreement.
 - (h) Certificates for Common Stock delivered at a closing hereunder may be endorsed with a restrictive legend that shall read substantially as follows:
 - "The transfer of the shares represented by this certificate is subject to certain provisions of an agreement between the registered holder hereof and Issuer and to resale restrictions arising under the Securities Act of 1933, as amended. A copy of such agreement is on file at the principal office of Issuer and will be provided to the holder hereof without charge upon receipt by Issuer of a written request therefor."

It is understood and agreed that: (i) the reference to the resale restrictions of the Securities Act of 1933, as amended (the "1933 Act"), in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the Holder shall have delivered to Issuer a copy of a letter from the staff of the SEC, or an opinion of counsel, in form and substance reasonably satisfactory to Issuer, to the effect that such legend is not required for purposes of the 1933 Act; (ii) the reference to the provisions to this Agreement in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the shares have been sold or transferred in compliance with the provisions of this Agreement and under circumstances that do not require the retention of such reference; and (iii) the legend shall be removed in its entirety if the conditions in the preceding clauses (i) and (ii) are both satisfied. In addition, such certificates shall bear any other legend as may be required by law.

(i) Upon the giving by the Holder to Issuer of the written notice of exercise of the Option provided for under subsection (e) of this Section 2 and the tender of the applicable purchase price in immediately available funds, the Holder shall be deemed to be the holder of record of the shares of Common Stock issuable upon such exercise, notwithstanding that the stock transfer books of Issuer shall then be closed or

that certificates representing such shares of Common Stock shall not then be actually delivered to the Holder. Issuer shall pay all expenses, and any and all United States federal, state and local taxes and other charges that may be payable in connection with the preparation, issue and delivery of stock certificates under this Section 2 in the name of the Holder or its assignee, transferee or designee.

- 3. Issuer agrees: (i) that it shall at all times maintain, free from preemptive rights, sufficient authorized but unissued or treasury shares of Common Stock so that the Option may be exercised without additional authorization of Common Stock after giving effect to all other options, warrants, convertible securities and other rights to purchase Common Stock; (ii) that it will not, by charter amendment or through reorganization, consolidation, merger, dissolution or sale of assets, or by any other voluntary act, avoid or seek to avoid the observance or performance of any of the covenants, stipulations or conditions to be observed or performed hereunder by Issuer; (iii) promptly to take all action as may from time to time be required (including (x) complying with all premerger notification, reporting and waiting period requirements specified in 15 U.S.C. 18a and regulations promulgated thereunder and (y) in the event, under the Bank Holding Company Act of 1956, as amended (the "BHCA"), or the Change in Bank Control Act of 1978, as amended, or any state banking law, prior approval of or notice to the Federal Reserve Board or to any state or other regulatory authority is necessary before the Option may be exercised, cooperating fully with the Holder in preparing such applications or notices and providing such information to the Federal Reserve Board or such other regulatory authority as they may require) in order to permit the Holder to exercise the Option and Issuer duly and effectively to issue shares of Common Stock pursuant hereto; and (iv) promptly to take all action provided herein to protect the rights of the Holder against dilution.
- 4. This Agreement (and the Option granted hereby) are exchangeable, without expense, at the option of the Holder, upon presentation and surrender of this Agreement at the principal office of Issuer, for other Agreements providing for Options of different denominations entitling the holder thereof to purchase, on the same terms and subject to the same conditions as are set forth herein, in the aggregate the same number of shares of Common Stock purchasable hereunder. The terms "Agreement" and "Option" as used herein include any Stock Option Agreements and related Options for which this Agreement (and the Option granted hereby) may be exchanged. Upon receipt by Issuer of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Agreement, and (in the case of loss, theft or destruction) of reasonably satisfactory indemnification, and upon surrender and cancellation of this Agreement, if mutilated, Issuer will execute and deliver a new Agreement of like tenor and date. Any such new Agreement executed and delivered shall constitute an additional contractual obligation on the part of Issuer, whether or not the Agreement so lost, stolen, destroyed or mutilated shall at any time be enforceable by anyone.
- 5. In addition to the adjustment in the number of shares of Common Stock that are purchasable upon exercise of the Option pursuant to Section 1 of this Agreement, the number of shares of Common Stock purchasable upon the exercise of the Option and the Option Price shall be subject to adjustment from time to time as provided in this Section 5. In the event of any change in, or distributions in respect of, the Common Stock by reason of stock dividends, split-ups, mergers, recapitalizations, combinations, subdivisions, conversions, exchanges of shares, distributions on or in respect of the Common Stock that would be prohibited under the terms of the Merger Agreement, or the like, the type and number of shares of Common Stock purchasable upon exercise hereof and the Option Price shall be appropriately adjusted in such manner as shall fully preserve the economic benefits provided hereunder and proper provision shall be made in any agreement governing any such transaction to provide for such proper adjustment and the full satisfaction of the Issuer's obligations hereunder.
- 6. Upon the occurrence of a Subsequent Triggering Event that occurs prior to an Exercise Termination Event, Issuer shall, at the request of Grantee delivered within 180 days of such Subsequent Triggering Event (whether on its own behalf or on behalf of any subsequent holder of this Option (or part thereof) or any of the shares of Common Stock issued pursuant hereto), promptly prepare, file and keep current a shelf registration statement under the 1933 Act covering this Option and any shares issued and issuable pursuant to this Option and shall use its reasonable best efforts to cause such registration statement to become effective and remain current in order to permit the sale or other disposition of this Option and any shares of Common Stock issued upon total or

partial exercise of this Option ("Option Shares") in accordance with any plan of disposition requested by Grantee. Issuer will use its reasonable best efforts to cause such registration statement first to become effective and then to remain effective for such period not in excess of 180 days from the day such registration statement first becomes effective or such shorter time as may be reasonably necessary to effect such sales or other dispositions. Grantee shall have the right to demand two such registrations. The foregoing notwithstanding, if, at the time of any request by Grantee for registration of the Option or Option Shares as provided above, Issuer is in registration with respect to an underwritten public offering of shares of Common Stock, and if in the good faith judgment of the managing underwriter or managing underwriters, or, if none, the sole underwriter or underwriters, of such offering the inclusion of the Holder's Option or Option Shares would interfere with the successful marketing of the shares of Common Stock offered by Issuer, the number of Option Shares otherwise to be covered in the registration statement contemplated hereby may be reduced; provided, however, that after any such required reduction the number of Option Shares to be included in such offering for the account of the Holder shall constitute at least 25% of the total number of shares to be sold by the Holder and Issuer in the aggregate; and provided further, however, that if such reduction occurs, then the Issuer shall file a registration statement for the balance as promptly as practicable and no reduction shall thereafter occur. Each such Holder shall provide all information reasonably requested by Issuer for inclusion in any registration statement to be filed hereunder. If requested by any such Holder in connection with such registration. Issuer shall become a party to any underwriting agreement relating to the sale of such shares, but only to the extent of obligating itself in respect of representations, warranties, indemnities and other agreements customarily included in secondary offering underwriting agreements for the Issuer. Upon receiving any request under this Section 6 from any Holder, Issuer agrees to send a copy thereof to any other person known to Issuer to be entitled to registration rights under this Section 6, in each case by promptly mailing the same, postage prepaid, to the address of record of the persons entitled to receive such copies. Notwithstanding anything to the contrary contained herein, in no event shall Issuer be obligated to effect more than two registrations pursuant to this Section 6 by reason of the fact that there shall be more than one Grantee as a result of any assignment or division of this Agreement.

7. (a) Immediately prior to the occurrence of a Repurchase Event (as defined below), (i) following a request of the Holder, delivered prior to an Exercise Termination Event, Issuer (or any successor thereto) shall repurchase the Option from the Holder at a price (the "Option Repurchase Price") equal to the number of shares for which this Option may then be exercised multiplied by the amount by which (A) the Market/Offer Price (as defined below) exceeds (B) the Option Price, and (ii) at the request of the owner of Option Shares from time to time (the "Owner"), delivered within 90 days of such occurrence (or such later period as provided in Section 10), Issuer shall repurchase such number of the Option Shares from the Owner as the Owner shall designate at a price (the "Option Share Repurchase Price") equal to the Market/Offer Price multiplied by the number of Option Shares so designated. The term "Market/Offer Price" shall mean the highest of (i) the price per share of Common Stock at which a tender offer or exchange offer therefor has been made, (ii) the price per share of Common Stock to be paid by any third party pursuant to an agreement with Issuer, (iii) the highest closing price for shares of Common Stock within the six-month period immediately preceding the date the Holder gives notice of the required repurchase of this Option or the Owner gives notice of the required repurchase of Option Shares, as the case may be, or (iv) in the event of a sale of all or a substantial portion of Issuer's assets, the sum of the price paid in such sale for such assets and the current market value of the remaining assets of Issuer as determined by a nationally recognized investment banking firm selected by the Holder or Owner, as the case may be, and reasonably acceptable to the Issuer. divided by the number of shares of Common Stock of Issuer outstanding at the time of such sale. In determining the Market/Offer Price, the value of consideration other than cash shall be determined by a nationally recognized investment banki

(b) The Holder and the Owner, as the case may be, may exercise its right to require Issuer to repurchase the Option and any Option Shares pursuant to this Section 7 by surrendering for such purpose to Issuer, at its principal office, this Agreement or certificates for Option Shares, as applicable, accompanied by a written notice or notices stating that the Holder or the Owner, as the case may be, elects to require Issuer to repurchase this Option and/or the Option Shares in accordance with the provisions of this Section 7. Within

the latter to occur of (x) five business days after the surrender of the Option and/or certificates representing Option Shares and the receipt of such notice or notices relating thereto and (y) the time that is immediately prior to the occurrence of a Repurchase Event, Issuer shall deliver or cause to be delivered to the Holder the Option Repurchase Price and/or to the Owner the Option Share Repurchase Price therefor or the portion thereof, if any, that Issuer is not then prohibited under applicable law and regulation from so delivering.

- (c) To the extent that Issuer is prohibited under applicable law or regulation from repurchasing the Option and/or the Option Shares in full, Issuer shall immediately so notify the Holder and/or the Owner and thereafter deliver or cause to be delivered, from time to time, to the Holder and/or the Owner, as appropriate, the portion of the Option Repurchase Price and the Option Share Repurchase Price, respectively, that it is no longer prohibited from delivering, within five business days after the date on which Issuer is no longer so prohibited; *provided*, *however*, that if Issuer at any time after delivery of a notice of repurchase pursuant to paragraph (b) of this Section 7 is prohibited under applicable law or regulation from delivering to the Holder and/or the Owner, as appropriate, the Option Repurchase Price and the Option Share Repurchase Price, respectively, in full (and Issuer hereby undertakes to use its best efforts to obtain all required regulatory and legal approvals and to file any required notices, in each case as promptly as practicable in order to accomplish such repurchase), the Holder or Owner may revoke its notice of repurchase of the Option or the Option Shares either in whole or to the extent of the prohibition, whereupon, in the latter case, Issuer shall promptly (i) deliver to the Holder and/or the Owner, as appropriate, that portion of the Option Repurchase Price or the Option Share Repurchase Price that Issuer is not prohibited from delivering; and (ii) deliver, as appropriate, either (A) to the Holder, a new Stock Option Agreement evidencing the right of the Holder to purchase that number of shares of Common Stock obtained by multiplying the number of shares of Common Stock obtained by multiplying the number of shares of Common Stock for which the surrendered Stock Option Agreement was exercisable at the time of delivery of the notice of repurchase by a fraction, the numerator of which is the Option Repurchase Price less the portion thereof theretofore delivered to the Holder and the
- (d) For purposes of this Section 7, a Repurchase Event shall be deemed to have occurred (i) upon the consummation of any merger, consolidation, share exchange or similar transaction involving Issuer or any purchase, lease or other acquisition of all or a substantial portion of the assets of Issuer, other than any such transaction which would not constitute an Acquisition Transaction pursuant to the provisos to Section 2(b)(i) hereof or (ii) upon the acquisition by any person of beneficial ownership of 50% or more of the then outstanding shares of Common Stock, provided that no such event shall constitute a Repurchase Event unless a Subsequent Triggering Event shall have occurred prior to an Exercise Termination Event. The parties hereto agree that Issuer's obligations to repurchase the Option or Option Shares under this Section 7 shall not terminate upon the occurrence of an Exercise Termination Event unless no Subsequent Triggering Event shall have occurred prior to the occurrence of an Exercise Termination Event.
- 8. (a) In the event that prior to an Exercise Termination Event, Issuer shall enter into an agreement (i) to consolidate with or merge into any person, other than Grantee or one of its Subsidiaries, and shall not be the continuing or surviving corporation of such consolidation or merger, (ii) to permit any person, other than Grantee or one of its Subsidiaries, to merge into Issuer and Issuer shall be the continuing or surviving corporation, but, in connection with such merger, the then outstanding shares of Common Stock shall be changed into or exchanged for stock or other securities of any other person or cash or any other property or the then outstanding shares of Common Stock shall after such merger represent less than 50% of the outstanding voting shares and voting shares quivalents of the merged company, or (iii) to sell or otherwise transfer all or substantially all of its assets to any person, other than Grantee or one of its Subsidiaries, then, and in each such case, the agreement governing such transaction shall make proper provision so that the Option shall, upon the consummation of any such transaction and upon the terms and conditions set forth herein, be converted into, or exchanged for, an option (the "Substitute Option"), at the election of the Holder, of either (x) the Acquiring Corporation (as hereinafter defined) or (y) any person that controls the Acquiring Corporation.

- (b) The following terms have the meanings indicated:
- (A) "Acquiring Corporation" shall mean (i) the continuing or surviving corporation of a consolidation or merger with Issuer (if other than Issuer), (ii) Issuer in a merger in which Issuer is the continuing or surviving person, and (iii) the transferee of all or substantially all of Issuer's assets.
 - (B) "Substitute Common Stock" shall mean the common stock issued by the issuer of the Substitute Option upon exercise of the Substitute Option.
 - (C) "Assigned Value" shall mean the Market/Offer Price, as defined in Section 7.
- (D) "Average Price" shall mean the average closing price of a share of the Substitute Common Stock for the one year immediately preceding the consolidation, merger or sale in question, but in no event higher than the closing price of the shares of Substitute Common Stock on the day preceding such consolidation, merger or sale; *provided* that if Issuer is the issuer of the Substitute Option, the Average Price shall be computed with respect to a share of common stock issued by the person merging into Issuer or by any company which controls or is controlled by such person, as the Holder may elect.
- (c) The Substitute Option shall have the same terms as the Option, *provided*, that if the terms of the Substitute Option cannot, for legal reasons, be the same as the Option, such terms shall be as similar as possible and in no event less advantageous to the Holder. The issuer of the Substitute Option shall also enter into an agreement with the then Holder or Holders of the Substitute Option in substantially the same form as this Agreement, which shall be applicable to the Substitute Option.
- (d) The Substitute Option shall be exercisable for such number of shares of Substitute Common Stock as is equal to the Assigned Value multiplied by the number of shares of Common Stock for which the Option is then exercisable, divided by the Average Price. The exercise price of the Substitute Option per share of Substitute Common Stock shall then be equal to the Option Price multiplied by a fraction, the numerator of which shall be the number of shares of Common Stock for which the Option is then exercisable and the denominator of which shall be the number of Substitute Common Stock for which the Substitute Option is exercisable.
- (e) In no event, pursuant to any of the foregoing paragraphs, shall the Substitute Option be exercisable for more than 19.9% of the shares of Substitute Common Stock outstanding prior to exercise of the Substitute Option. In the event that the Substitute Option would be exercisable for more than 19.9% of the shares of Substitute Common Stock outstanding prior to exercise but for this clause (e), the issuer of the Substitute Option (the "Substitute Option Issuer") shall make a cash payment to Holder equal to the excess of (i) the value of the Substitute Option without giving effect to the limitation in this clause (e) over (ii) the value of the Substitute Option after giving effect to the limitation in this clause (e). This difference in value shall be determined by a nationally recognized investment banking firm selected by the Holder or the Owner, as the case may be, and reasonably acceptable to the Acquiring Corporation.
- (f) Issuer shall not enter into any transaction described in subsection (a) of this Section 8 unless the Acquiring Corporation and any person that controls the Acquiring Corporation assume in writing all the obligations of Issuer hereunder.
- 9. (a) At the request of the holder of the Substitute Option (the "Substitute Option Holder"), the Substitute Option Issuer shall repurchase the Substitute Option from the Substitute Option Holder at a price (the "Substitute Option Repurchase Price") equal to the amount by which (i) the Highest Closing Price (as hereinafter defined) exceeds (ii) the exercise price of the Substitute Option, multiplied by the number of shares of Substitute Common Stock for which the Substitute Option may then be exercised, and at the request of the owner (the "Substitute Share Owner") of shares of Substitute Common Stock (the "Substitute Shares"), the Substitute Option Issuer shall repurchase the Substitute Shares at a price (the "Substitute Share Repurchase Price") equal to the Highest Closing Price multiplied by the number of Substitute Shares so designated. The term "Highest Closing Price" shall mean the highest closing price for shares of Substitute Common Stock within the six-month period immediately preceding the

date the Substitute Option Holder gives notice of the required repurchase of the Substitute Option or the Substitute Share Owner gives notice of the required repurchase of the Substitute Shares, as applicable.

- (b) The Substitute Option Holder and the Substitute Share Owner, as the case may be, may exercise its respective right to require the Substitute Option Issuer to repurchase the Substitute Option and the Substitute Shares pursuant to this Section 9 by surrendering for such purpose to the Substitute Option Issuer, at its principal office, the agreement for such Substitute Option (or, in the absence of such an agreement, a copy of this Agreement) and certificates for Substitute Shares accompanied by a written notice or notices stating that the Substitute Option Holder or the Substitute Share Owner, as the case may be, elects to require the Substitute Option Issuer to repurchase the Substitute Option and/or the Substitute Shares in accordance with the provisions of this Section 9. As promptly as practicable, and in any event within five business days after the surrender of the Substitute Option and/or certificates representing Substitute Shares and the receipt of such notice or notices relating thereto, the Substitute Option Issuer shall deliver or cause to be delivered to the Substitute Option Holder the Substitute Option Repurchase Price and/or to the Substitute Share Owner the Substitute Share Repurchase Price therefor or, in either case, the portion thereof which the Substitute Option Issuer is not then prohibited under applicable law and regulation from so delivering.
- (c) To the extent that the Substitute Option Issuer is prohibited under applicable law or regulation from repurchasing the Substitute Option and/or the Substitute Shares in part or in full, the Substitute Option Issuer following a request for repurchase pursuant to this Section 9 shall immediately so notify the Substitute Option Holder and/or the Substitute Share Owner and thereafter deliver or cause to be delivered, from time to time, to the Substitute Option Holder and/or the Substitute Share Owner, as appropriate, the portion of the Substitute Share Repurchase Price, respectively, which it is no longer prohibited from delivering, within five business days after the date on which the Substitute Option Issuer is no longer so prohibited; provided, however, that if the Substitute Option Issuer is at any time after delivery of a notice of repurchase pursuant to subsection (b) of this Section 9 prohibited under applicable law or regulation from delivering to the Substitute Option Holder and/or the Substitute Share Owner, as appropriate, the Substitute Option Repurchase Price and the Substitute Share Repurchase Price, respectively, in full (and the Substitute Option Issuer shall use its best efforts to obtain all required regulatory and legal approvals, in each case as promptly as practicable, in order to accomplish such repurchase), the Substitute Option Holder or Substitute Share Owner may revoke its notice of repurchase of the Substitute Option or the Substitute Shares either in whole or to the extent of the prohibition, whereupon, in the latter case, the Substitute Option Issuer shall promptly (i) deliver to the Substitute Option Holder or Substitute Share Owner, as appropriate, that portion of the Substitute Option Repurchase Price or the Substitute Share Repurchase Price that the Substitute Option Issuer is not prohibited from delivering; and (ii) deliver, as appropriate, either (A) to the Substitute Option Holder, a new Substitute Option evidencing the right of the Substitute Option Holder to purchase that number of shares of the Substitute Common Stock obtained by multiplying the number of shares of the Substitute Common Stock for which the surrendered Substitute Option was exercisable at the time of delivery of the notice of repurchase by a fraction, the numerator of which is the Substitute Option Repurchase Price less the portion thereof theretofore delivered to the Substitute Option Holder and the denominator of which is the Substitute Option Repurchase Price, or (B) to the Substitute Share Owner, a certificate for the Substitute Common Shares it is then so prohibited from repurchasing.
- 10. The 90-day or 180-day periods for exercise of certain rights under Sections 2, 6, 7 and 13 shall be extended: (i) to the extent necessary to obtain all regulatory approvals for the exercise of such rights and for the expiration of all statutory waiting periods; and (ii) to the extent necessary to avoid liability under Section 16(b) of the 1934 Act by reason of such exercise.
 - 11. Issuer hereby represents and warrants to Grantee as follows:
 - (a) Issuer has full corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized by the Board of Directors of Issuer and no other corporate proceedings on the part of Issuer are necessary to authorize this

Agreement or to consummate the transactions so contemplated. This Agreement has been duly and validly executed and delivered by Issuer.

- (b) Issuer has taken all necessary corporate action to authorize and reserve and to permit it to issue, and at all times from the date hereof through the termination of this Agreement in accordance with its terms will have reserved for issuance upon the exercise of the Option, that number of shares of Common Stock equal to the maximum number of shares of Common Stock at any time and from time to time issuable hereunder, and all such shares, upon issuance pursuant hereto, will be duly authorized, validly issued, fully paid, nonassessable, and will be delivered free and clear of all claims, liens, encumbrance and security interests and not subject to any preemptive rights.
- (c) Issuer has taken all action (including if required redeeming all of the Rights or amending or terminating the Rights Agreement) so that the entering into of this Option Agreement, the acquisition of shares of Common Stock hereunder and the other transactions contemplated hereby do not and will not result in the grant of any rights to any person under the Rights Agreement or enable or require the Rights to be exercised, distributed or triggered.
- 12. Grantee hereby represents and warrants to Issuer that:
- (a) Grantee has all requisite corporate power and authority to enter into this Agreement and, subject to any approvals or consents referred to herein, to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Grantee. This Agreement has been duly executed and delivered by Grantee.
- (b) The Option is not being, and any shares of Common Stock or other securities acquired by Grantee upon exercise of the Option will not be, acquired with a view to the public distribution thereof and will not be transferred or otherwise disposed of except in a transaction registered or exempt from registration under the 1933 Act.
- 13. Neither of the parties hereto may assign any of its rights or obligations under this Agreement or the Option created hereunder to any other person, without the express written consent of the other party, except that in the event a Subsequent Triggering Event shall have occurred prior to an Exercise Termination Event, Grantee, subject to the express provisions hereof, may assign in whole or in part its rights and obligations hereunder within 180 days following such Subsequent Triggering Event (or such later period as provided in Section 10); provided, however, that until the date 15 days following the date on which the Federal Reserve Board approves an application by Grantee under the BHCA to acquire the shares of Common Stock subject to the Option, Grantee may not assign its rights under the Option except in (i) a widely dispersed public distribution, (ii) a private placement in which no one party acquires the right to purchase in excess of 2% of the voting shares of Issuer, (iii) an assignment to a single party (e.g., a broker or investment banker) for the purpose of conducting a widely dispersed public distribution on Grantee's behalf, or (iv) any other manner approved by the Federal Reserve Board.
- 14. Each of Grantee and Issuer will use its best efforts to make all filings with, and to obtain consents of, all third parties and governmental authorities necessary to the consummation of the transactions contemplated by this Agreement, including without limitation making application to list the shares of Common Stock issuable hereunder on the New York Stock Exchange upon official notice of issuance and applying to the Federal Reserve Board under the BHCA for approval to acquire the shares issuable hereunder, but Grantee shall not be obligated to apply to state banking authorities for approval to acquire the shares of Common Stock issuable hereunder until such time, if ever, as it deems appropriate to do so.
- 15. (a) Grantee may, at any time during which Issuer would be required to repurchase the Option or any Option Shares pursuant to Section 7 upon proper request or notice, surrender the Option (together with any Option Shares issued to and then owned by Grantee) to Issuer in exchange for a cash fee equal to the Surrender

Price (as defined below); provided, however, that Grantee may not exercise its rights pursuant to this Section 15 if Issuer has repurchased the Option (or any portion thereof) or any Option Shares pursuant to Section 7. The "Surrender Price" shall be equal to (i) \$1.8948 billion, plus (ii) if applicable, the aggregate purchase price previously paid pursuant hereto by Grantee with respect to any Option Shares, minus (iii) if applicable, the sum of (A) the excess of (1) the net cash amounts, if any, received by Grantee pursuant to the arms' length sale of Option Shares (or any other securities into which such Option Shares were converted or exchanged) to any party not affiliated with Grantee, over (2) the aggregate purchase price previously paid pursuant hereto by Grantee with respect to such Option Shares and (B) the net cash amounts, if any, received by Grantee pursuant to an arms' length sale of a portion of the Option to any party not affiliated with Grantee.

- (b) Grantee may exercise its right to surrender the Option and any Option Shares pursuant to this Section 15 by surrendering to Issuer, at its principal office, this Agreement together with certificates for Option Shares, if any, accompanied by a written notice stating (i) that Grantee elects to surrender the Option and Option Shares, if any, in accordance with the provisions of this Section 15 and (ii) the Surrender Price. The Surrender Price shall be payable in immediately available funds on or before the second business day following receipt of such notice by Issuer.
- (c) To the extent that Issuer is prohibited under applicable law or regulation from paying the Surrender Price to Grantee in full, Issuer shall immediately so notify Grantee and thereafter deliver or cause to be delivered, from time to time, to Grantee, the portion of the Surrender Price that Issuer is no longer prohibited from paying, within five business days after the date on which Issuer is no longer so prohibited, *provided*, *however*, that if Issuer at any time after delivery of a notice of surrender pursuant to paragraph (b) of this Section 15 is prohibited under applicable law or regulation from paying to Grantee the Surrender Price in full (i) Issuer shall (A) use its reasonable best efforts to obtain all required regulatory and legal approvals and to file any required notices as promptly as practicable in order to make such payments, (B) within five days of the submission or receipt of any documents relating to any such regulatory and legal approvals, provide Grantee with copies of the same, and (C) keep Grantee advised of both the status of any such request for regulatory and legal approvals, as well as any discussions with any relevant regulatory or other third party reasonably related to the same and (ii) Grantee may revoke such notice of surrender by delivery of a notice of revocation to Issuer and, upon delivery of such notice of revocation, the Exercise Termination Date shall be extended to a date six months from the date on which the Exercise Termination Date would have occurred if not for the provisions of this Section 15(c) (during which period Grantee may exercise any of its rights hereunder, including any and all rights pursuant to this Section 15).
- (d) Grantee shall have rights substantially identical to those set forth in paragraphs (a),(b) and (c) of this Section 15 with respect to the Substitute Option and the Substitute Option Issuer during any period in which the Substitute Option Issuer would be required to repurchase the Substitute Option pursuant to Section 9.
- 16. (a) Notwithstanding any other provision of this Agreement, in no event shall the Grantee's Total Profit (as hereinafter defined) exceed \$2.3685 billion and, if it otherwise would exceed such amount, the Grantee, at its sole election, shall either (i) reduce the number of shares of Common Stock subject to this Option, (ii) deliver to Issuer for cancellation Option Shares previously purchased by Grantee, (iii) pay cash to Issuer, or (iv) any combination thereof, so that Grantee's actually realized Total Profit shall not exceed \$2.3685 billion after taking into account the foregoing actions.
 - (b) Notwithstanding any other provision of this Agreement, this Option may not be exercised for a number of shares as would, as of the date of exercise, result in a Notional Total Profit (as defined below) of more than \$2.3685 billion; *provided* that nothing in this sentence shall restrict any exercise of the Option permitted hereby on any subsequent date.
 - (c) As used herein, the term "Total Profit" shall mean the aggregate amount (before taxes) of the following: (i) the amount received by Grantee pursuant to Issuer's repurchase of the Option (or any portion thereof) pursuant to Section 7, (ii) (x) the amount received by Grantee pursuant to Issuer's repurchase of

Option Shares pursuant to Section 7, less (y) the Grantee's purchase price for such Option Shares, (iii) (x) the net cash amounts received by Grantee pursuant to the sale of Option Shares (or any other securities into which such Option Shares are converted or exchanged) to any unaffiliated party, less (y) the Grantee's purchase price of such Option Shares, (iv) any amounts received by Grantee on the transfer of the Option (or any portion thereof) to any unaffiliated party, and (v) any amount equivalent to the foregoing with respect to the Substitute Option.

- (d) As used herein, the term "Notional Total Profit" with respect to any number of shares as to which Grantee may propose to exercise this Option shall be the Total Profit determined as of the date of such proposed exercise assuming that this Option were exercised on such date for such number of shares and assuming that such shares, together with all other Option Shares held by Grantee and its affiliates as of such date, were sold for cash at the closing market price for the Common Stock as of the close of business on the preceding trading day (less customary brokerage commissions).
- 17. The parties hereto acknowledge that damages would be an inadequate remedy for a breach of this Agreement by either party hereto and that the obligations of the parties hereto shall be enforceable by either party hereto through injunctive or other equitable relief.
- 18. If any term, provision, covenant or restriction contained in this Agreement is held by a court or a federal or state regulatory agency of competent jurisdiction to be invalid, void or unenforceable, the remainder of the terms, provisions and covenants and restrictions contained in this Agreement shall remain in full force and effect, and shall in no way be affected, impaired or invalidated. If for any reason such court or regulatory agency determines that the Holder is not permitted to acquire, or Issuer is not permitted to repurchase pursuant to Section 7, the full number of shares of Common Stock provided in Section 1(a) hereof (as adjusted pursuant to Section 1(b) or 5 hereof), it is the express intention of Issuer to allow the Holder to acquire or to require Issuer to repurchase such lesser number of shares as may be permissible, without any amendment or modification hereof.
- 19. All notices, requests, claims, demands and other communications hereunder shall be deemed to have been duly given when delivered in person, by cable, telegram, telecopy or telex, or by registered or certified mail (postage prepaid, return receipt requested) at the respective addresses of the parties set forth in the Merger Agreement.
- 20. This Agreement shall be governed by and construed in accordance with the laws of the State of New York, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws thereof (except to the extent that mandatory provisions of federal or state law apply).
 - 21. This Agreement may be executed in two counterparts, each of which shall be deemed to be an original, but all of which shall constitute one and the same agreement.
- 22. Except as otherwise expressly provided herein, each of the parties hereto shall bear and pay all costs and expenses incurred by it or on its behalf in connection with the transactions contemplated hereunder, including fees and expenses of its own financial consultants, investment bankers, accountants and counsel.
- 23. Except as otherwise expressly provided herein or in the Merger Agreement, this Agreement contains the entire agreement between the parties with respect to the transactions contemplated hereunder and supersedes all prior arrangements or understandings with respect thereof, written or oral. The terms and conditions of this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns. Nothing in this Agreement, expressed or implied, is intended to confer upon any party, other than the parties hereto, and their respective successors and permitted assigns, any rights, remedies, obligations or liabilities under or by reason of this Agreement, except as expressly provided herein.
 - 24. Capitalized terms used in this Agreement and not defined herein shall have the meanings assigned thereto in the Merger Agreement.

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be executed on its behalf by its officers thereunto duly authorized, all as of the date first above written.

BANK OF AMERICA CORPORATION (Issuer)

By: /s/ Kenneth D. Lewis

Name: Kenneth D. Lewis

Title: Chairman, President and Chief Executive Officer

FLEETBOSTON FINANCIAL CORPORATION (Grantee)

By: /s/ Charles K. Gifford

Name: Charles K. Gifford

Title: Chairman and Chief Executive Officer

Signature Page to Bank of America Stock Option Agreement

Appendix D

Opinion of Goldman, Sachs & Co.

PERSONAL AND CONFIDENTIAL

October 27, 2003 Board of Directors Bank of America Corporation

Bank of America Corporate Center 100 North Tryon Street Charlotte, North Carolina 28255

Ladies and Gentlemen:

You have requested our opinion as to the fairness from a financial point of view to Bank of America Corporation (the "Company") of the exchange ratio (the "Exchange Ratio") of 0.5553 shares of common stock, par value \$0.01 per share (the "Company Common Stock"), of the Company to be issued in respect of each share of common stock, par value \$0.01 per share (the "FleetBoston Common Stock"), of FleetBoston Financial Corporation ("FleetBoston") pursuant to the Agreement and Plan of Merger, dated as of October 27, 2003 (the "Agreement"), between the Company and FleetBoston.

Goldman, Sachs & Co. and its affiliates, as part of their investment banking business, are continually engaged in performing financial analyses with respect to businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, competitive biddings, secondary distributions of listed and unlisted securities, private placements and other transactions as well as for estate, corporate and other purposes. We have acted as financial advisor to the Company in connection with the transaction contemplated by the Agreement (the "Transaction"). We expect to receive fees for our services in connection with the Transaction, the principal portion of which fees are contingent upon consummation of the Transaction, and the Company has agreed to reimburse our expenses and indemnify us against certain liabilities arising out of our engagement. We also have provided certain investment banking services to the Company from time to time, including having acted as co-manager on several note and capital securities offerings, having acted as manager of a public offering of Company Common Stock in November 2001, and having acted as financial advisor in connection with the restructuring of the Company's Equicredit home equity business in April 2002, and we also participate in certain joint ventures from time to time with respect to certain investment arrangements. We also have provided certain investment banking services to FleetBoston from time to time, including having acted as co-manager or lead-manager on several note and capital securities offerings, and having acted as financial advisor to FleetBoston in connection with the acquisitions of Summit Bancorp in March 2001 and Liberty Financial Companies in November 2001, and we participate in a joint venture with FleetBoston with respect to certain loan arrangements. We also may provide investment banking services to the Company or FleetBoston in the future. In connection with the above-described investment banking services, we have received, and may receive, compensation. In addition, Goldman, Sachs & Co. is a full service securities firm engaged, either directly or through its affiliates, in securities trading, investment management, financial planning and benefits counseling, financing and brokerage activities for both companies and individuals. In the ordinary course of their trading, investment management, financing and brokerage activities, Goldman, Sachs & Co. and its affiliates may actively trade the debt and equity securities of the Company and FleetBoston (or related derivative securities) for their own account and for the accounts of their customers and may at any time hold long and short positions of such securities.

In connection with this opinion, we have reviewed, among other things, the Agreement; annual reports to stockholders and Annual Reports on Form 10-K of the Company and FleetBoston for the five fiscal years ended December 31, 2002; certain interim reports to stockholders and Quarterly Reports on Form 10-Q of the Company and FleetBoston; certain other communications from the Company and FleetBoston to their respective

stockholders; certain First Call research analyst estimates for each of the Company and FleetBoston approved for use in connection with this opinion by the management of the Company; certain internal financial analyses and forecasts for the Company prepared by its management; certain internal financial analyses and forecasts for the combined company on a pro forma basis prepared by management of the Company, including certain cost savings and operating synergies projected by the management of the Company to result from the Transaction. We also have held discussions with members of the senior managements of the Company and FleetBoston regarding their assessment of the strategic rationale for, and the potential benefits of, the Transaction and the past and current business operations, financial condition and future prospects of the Company and FleetBoston. In addition, we have reviewed the reported price and trading activity for the shares of FleetBoston Common Stock and the shares of the Company Common Stock, compared certain financial and stock market information for the Company and FleetBoston with similar information for certain other companies the securities of which are publicly traded, reviewed the financial terms of certain recent business combinations in the banking industry specifically and in other industries generally and performed such other studies and analyses as we considered appropriate.

We have relied upon the accuracy and completeness of all of the financial, accounting, legal, tax and other information discussed with or reviewed by us and have assumed such accuracy and completeness for purposes of rendering this opinion. In that regard, we have assumed with your consent that the forecasts reviewed by us have been reasonably prepared on a basis reflecting the best currently available estimates and judgments of the Company and FleetBoston, as the case may be. We also have assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the Transaction will be obtained without any adverse effect on the Company or FleetBoston or on the expected benefits of the Transaction in any way material to our analysis. We are not experts in the evaluation of loan and lease portfolios for purposes of assessing the adequacy of the allowances for losses with respect thereto and, accordingly, we have assumed that such allowances for losses are in the aggregate adequate to cover such losses. In addition, we have not reviewed individual credit files nor have we made an independent evaluation or appraisal of the assets and liabilities (including any derivative or off-balance-sheet assets and liabilities) of the Company or FleetBoston or any of their respective subsidiaries, and we have not been furnished with any such evaluation or appraisal. Our opinion does not address the underlying business decision of the Company to engage in the Transaction. In addition, we are not expressing any opinion herein as to the prices at which shares of Company Common Stock or FleetBoston Common Stock will trade at any time. Our advisory services and the opinion expressed herein are provided for the information and assistance of the Board of Directors of the Company in connection with its consideration of the Transaction and such opinion does not constitute a recommendation as to how any holder of Company Common Stock should vote with respect to such transaction.

Based upon and subject to the foregoing and based upon such other matters as we consider relevant, it is our opinion that, as of the date hereof, the Exchange Ratio pursuant to the Agreement is fair from a financial point of view to the Company.

Very truly yours,
/s/ Goldman, Sachs & Co.
(GOLDMAN, SACHS & CO.)

Appendix E

Opinion of Morgan Stanley & Co. Incorporated

Board of Directors FleetBoston Financial 100 Federal Street Boston, MA 02110 October 26, 2003

Members of the Board:

We understand that FleetBoston Financial ("FleetBoston") and Bank of America Corporation ("Bank of America") propose to enter into an Agreement and Plan of Merger, substantially in the form of the draft dated October 26, 2003 (the "Merger Agreement"), which provides, among other things, for the merger of FleetBoston with and into Bank of America (the "Merger"). Pursuant to the Merger, the separate corporate existence of FleetBoston shall cease and each outstanding share of common stock, par value \$.01 per share, of FleetBoston (the "FleetBoston Common Stock"), other than shares held in treasury or held by Bank of America (except for shares held in a fiduciary or agency capacity owned by third parties or held in respect of previously contracted debt), will be converted into 0.5553 shares (the "Exchange Ratio") of common stock, par value \$.01 per share, of Bank of America (the "Bank of America Common Stock"). The terms and conditions of the Merger are more fully set forth in the Merger Agreement.

You have asked for our opinion as to whether the Exchange Ratio pursuant to the Merger Agreement is fair from a financial point of view to the holders of shares of FleetBoston Common Stock.

For purposes of the opinion set forth herein, we have:

- i. reviewed certain publicly available financial statements and other information of FleetBoston and Bank of America, respectively;
- ii. reviewed certain internal financial statements and other financial and operating data concerning FleetBoston and Bank of America prepared by the managements of FleetBoston and Bank of America, respectively, including among other things, limited financial forecasts and profit plans for each company;
- iii. discussed the past and current operations and financial condition and the prospects of FleetBoston and Bank of America, including information relating to certain strategic, financial and operational benefits anticipated from the Merger, with senior executives of FleetBoston and Bank of America, respectively;
 - iv. reviewed the pro forma impact of the Merger on Bank of America's earnings per share, consolidated capitalization and financial ratios;
 - v. reviewed the reported prices and trading activity of FleetBoston Common Stock and Bank of America Common Stock;
- vi. compared the financial performance of FleetBoston and Bank of America and the prices and trading activity of the FleetBoston Common Stock and the Bank of America Common Stock with that of certain other comparable publicly-traded companies and their securities;
 - vii. reviewed the financial terms, to the extent publicly available, of certain precedent transactions;
 - viii. participated in discussions among representatives of FleetBoston and Bank of America and their financial and legal advisors;
 - ix. reviewed the draft Merger Agreement and certain related documents; and
 - x. considered such other factors and performed such other analyses as we have deemed appropriate.

We have assumed and relied upon without independent verification the accuracy and completeness of the information reviewed by us for the purposes of this opinion. With respect to the financial forecasts, profit plans, information regarding certain strategic, financial and operational benefits anticipated from the Merger, we have assumed that they have been reasonably prepared on bases reflecting the best currently available estimates and

judgments of the future financial performance of FleetBoston and Bank of America. We have assumed that the Merger will be consummated in accordance with the terms of the Merger Agreement, including, among other things, that the Merger will be treated as a tax-free reorganization pursuant to the Internal Revenue Code of 1986, as amended. We have not made any independent valuation or appraisal of the assets or liabilities (including any hedge or derivative position) of FleetBoston and Bank of America, nor have we been furnished with any such appraisals, and we have not made any independent examination of the loan loss reserves or examined any individual loan credit files of FleetBoston or Bank of America. In addition, we have assumed that in connection with the receipt of all necessary government, regulatory or other consents and approvals for the Merger, no restrictions will be imposed that would have any adverse effect on FleetBoston or Bank of America or on the benefits expected to be derived from the Merger. Our opinion is necessarily based on financial, economic, market and other conditions as in effect on, and the information made available to us as of, the date hereof.

We have acted as financial advisor to the Board of Directors of FleetBoston in connection with this transaction and will receive a fee for our services which is contingent upon the completion of the Merger. In the past, Morgan Stanley & Co. Incorporated and its affiliates have provided financial advisory and financing services for FleetBoston and Bank of America and have received fees for the rendering of these services. In the ordinary course of our business, Morgan Stanley and its affiliates may from time to time trade in the securities or indebtedness of FleetBoston and Bank of America for its own account, the accounts of investment funds and other clients under the management of Morgan Stanley and for the accounts of its customers and, accordingly, may at any time hold a long or short position in such securities or indebtedness for any such account. In addition, Morgan Stanley and its affiliates may from time to time act as a counterparty to either FleetBoston or Bank of America and have received compensation for such activities

It is understood that this letter is for the information of the Board of Directors of FleetBoston and may not be used for any other purpose without our prior written consent, except that this opinion may be included in its entirety in any filing made by FleetBoston in respect of the Merger with the Securities and Exchange Commission. In addition, this opinion does not in any matter address the prices at which the Bank of America Common Stock will trade following the consummation of the Merger, and Morgan Stanley expresses no opinion or recommendation as how the shareholders of the FleetBoston should vote at the shareholders' meeting held in connection with the Merger.

Based on and subject to the foregoing, we are of the opinion on the date hereof that the Exchange Ratio pursuant to the Merger Agreement is fair from a financial point of view to the holders of shares of FleetBoston Common Stock.

Very truly yours,

MORGAN STANLEY & CO. INCORPORATED

By: /s/ Jonathan M. Pruzan

Jonathan M. Pruzan Managing Director

Appendix F

BUSINESS CORPORATION ACT OF RHODE ISLAND TITLE 7. CORPORATIONS, ASSOCIATIONS AND PARTNERSHIPS CHAPTER 1.1. RHODE ISLAND BUSINESS CORPORATION ACT

7-1.1-74 RIGHTS OF DISSENTING SHAREHOLDERS.

- (a) Any shareholder electing to exercise the right of dissent shall file with the corporation, prior to or at the meeting of shareholders at which the proposed corporate action is submitted to a vote, a written objection to the proposed corporate action. If the proposed corporate action is approved by the required vote and the shareholder has not voted in favor of it, the shareholder may, within ten (10) days after the date on which the vote was taken, or if a corporation is to be merged without a vote of its shareholders into another corporation, any of its shareholders may, within fifteen (15) days after the plan of the merger has been mailed to the shareholders, make written demand on the corporation, or, in the case of a merger or consolidation, on the surviving or new corporation, domestic or foreign, for payment of the fair value of the shareholder's shares. If the proposed corporate action is effected, the corporation shall pay to the shareholder, upon surrender of the certificate or certificates representing the shares, the fair value of the shares as of the day prior to the date on which the vote was taken approving the proposed corporate action, excluding any appreciation or depreciation in anticipation of the corporate action. Any shareholder failing to make demand within the ten (10) day period or the fifteen (15) day period, as the case may be, is bound by the terms of the proposed corporate action. Any shareholder making the demand is thereafter only entitled to payment as provided in this section and is not entitled to vote or to exercise any other rights of a shareholder.
- (b) No demand may be withdrawn unless the corporation consents to it. If, however, the demand is withdrawn upon consent, or if the proposed corporate action is abandoned or rescinded or the shareholders revoke the authority to effect the action, or if, in the case of a merger, on the date of the filing of the articles of merger the surviving corporation is the owner of all the outstanding shares of the other corporations, domestic and foreign, that are parties to the merger, or if no demand or petition for the determination of fair value by a court has been made or filed within the time provided in this section, or if a court of competent jurisdiction determines that the shareholder is not entitled to the relief provided by this section, then the right of the shareholder to be paid the fair value of his or her shares ceases and his or her status as a shareholder is restored, without prejudice to any corporate proceedings taken during the interim.
- (c) Within ten (10) days after the corporate action is effected, the corporation, or, in the case of a merger or consolidation, the surviving or new corporation, domestic or foreign, shall give written notice of the action to each dissenting shareholder who has made demand as provided in these provisions, and shall make a written offer to each shareholder to pay for the shares at a specified price deemed by the corporation to be the fair value of the shares. The notice and offer shall be accompanied by a balance sheet of the corporation the shares of which the dissenting shareholder holds, as of the latest available date and not more than twelve (12) months prior to the making of the offer, and a profit and loss statement of the corporation for the twelve (12) month period ended on the date of the balance sheet.
- (d) If within thirty (30) days after the date on which the corporate action was effected the fair value of the shares is agreed upon between any dissenting shareholder and the corporation, payment for the shares shall be made within ninety (90) days after the date on which the corporate action was effected, upon surrender of the certificate or certificates representing the shares. Upon payment of the agreed value, the dissenting shareholder ceases to have any interest in the shares.
- (e) If within the period of thirty (30) days a dissenting shareholder and the corporation do not agree on the matter, then the corporation, within thirty (30) days after receipt of written request for the filing from any dissenting shareholder given within sixty (60) days after the date on which the corporate action was effected, shall, or at its election at any time within the period of sixty (60) days may, file a petition in any court of competent jurisdiction in the county in this state where the registered office of the corporation is located praying

that the fair value of the shares be found and determined. If, in the case of a merger or consolidation, the surviving or new corporation is a foreign corporation without a registered office in this state, the petition shall be filed in the county where the registered office of the domestic corporation was last located. If the corporation fails to institute the proceeding as provided, any dissenting shareholder may do so in the name of the corporation. All dissenting shareholders, wherever they reside, shall be made parties to the proceeding as an action against their shares quasi in rem. A copy of the petition shall be served on each dissenting shareholder who is a resident of this state and shall be served by registered or certified mail on each dissenting shareholder who is a nonresident. Service on nonresidents shall also be made by publication as provided by law. The jurisdiction of the court shall be plenary and exclusive. All shareholders who are parties to the proceeding are entitled to judgment against the corporation for the amount of the fair value of their shares. The court may, if it so elects, appoint one or more persons as appraisers to receive evidence and recommend a decision on the question of fair value. The appraisers have the power and authority that is specified in the order of their appointment or an amendment of the order. The judgment is payable only upon and concurrently with the surrender to the corporation of the certificates representing the shares. Upon payment of the judgment, the dissenting shareholder ceases to have any interest in the shares.

- (f) The judgment shall include an allowance for interest at the rate of interest on judgments in civil actions from the date on which the vote was taken on the proposed corporate action to the date of payment.
- (g) The costs and expenses of any proceeding shall be determined by the court and assessed against the corporation, but all or any part of the costs and expenses may be apportioned and assessed as the court deems equitable against any or all of the dissenting shareholders who are parties to the proceeding to whom the corporation has made an offer to pay for the shares if the court finds that the action of the shareholders in failing to accept the offer was arbitrary or vexatious or not in good faith. The expenses include reasonable compensation for and reasonable expenses of the appraisers, but exclude the fees and expenses of counsel for and experts employed by any party; but if the fair value of the shares as determined materially exceeds the amount which the corporation offered to pay for the shares, or if no offer was made, the court in its discretion may award to any shareholder who is a party to the proceeding a sum that the court determines to be reasonable compensation to any expert or experts employed by the shareholder in the proceeding.
- (h) Within twenty (20) days after demanding payment for his or her shares, each shareholder demanding payment shall submit the certificate or certificates representing his or her shares to the corporation for notation on the certificate that the demand has been made. His or her failure to do so shall, at the option of the corporation, terminate his or her rights under this section unless a court of competent jurisdiction, for good and sufficient cause shown, directs otherwise. If shares represented by a certificate on which notation has been made are transferred, each new certificate issued for the shares shall bear similar notation, together with the name of the original dissenting holder of the shares, and a transferee of the shares acquires by the transfer no rights in the corporation other than those which the original dissenting shareholder had after making demand for payment of the fair value of the shares.
- (i) Shares acquired by a corporation pursuant to payment of the agreed value for the shares or to payment of the judgment entered for the shares, as provided in this section, may be held and disposed of by the corporation as in the case of other treasury shares. However, in the case of a merger or consolidation, they may be held and disposed of as the plan of merger or consolidation otherwise provides.

Appendix G

DELAWARE GENERAL CORPORATION LAW TITLE 8 CORPORATIONS CHAPTER 1. GENERAL CORPORATION LAW Subchapter IX. Merger, Consolidation or Conversion

§ 262 Appraisal rights.

(a) Any stockholder of a corporation of this State who holds shares of stock on the date of the making of a demand pursuant to subsection (d) of this section with respect to such shares, who continuously holds such shares through the effective date of the merger or consolidation, who has otherwise complied with subsection (d) of this section and who has neither voted in favor of the merger or consolidation nor consented thereto in writing pursuant to §228 of this title shall be entitled to an appraisal by the Court of Chancery of the fair value of the stockholder's shares of stock under the circumstances described in subsections (b) and (c) of this section. As used in this section, the word "stockholder" means a holder of record of stock in a stock corporation and also a member of record of a nonstock corporation; the words "stock" and "share" mean and include what is ordinarily meant by those words and also membership interest of a member of a nonstock corporation; and the words "depository receipt" mean a receipt or other instrument issued by a depository representing an interest in one or more shares, or fractions thereof, solely of stock of a corporation, which stock is deposited with the depository.

- (b) Appraisal rights shall be available for the shares of any class or series of stock of a constituent corporation in a merger or consolidation to be effected pursuant to §251 (other than a merger effected pursuant to §251(g) of this title), §252, §254, §257, §258, §263 or §264 of this title:
 - (1) Provided, however, that no appraisal rights under this section shall be available for the shares of any class or series of stock, which stock, or depository receipts in respect thereof, at the record date fixed to determine the stockholders entitled to receive notice of and to vote at the meeting of stockholders to act upon the agreement of merger or consolidation, were either (i) listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or (ii) held of record by more than 2,000 holders; and further provided that no appraisal rights shall be available for any shares of stock of the constituent corporation surviving a merger if the merger did not require for its approval the vote of the stockholders of the surviving corporation as provided in subsection (f) of §251 of this title.
 - (2) Notwithstanding paragraph (1) of this subsection, appraisal rights under this section shall be available for the shares of any class or series of stock of a constituent corporation if the holders thereof are required by the terms of an agreement of merger or consolidation pursuant to §§251, 252, 254, 257, 258, 263 and 264 of this title to accept for such stock anything except:
 - a. Shares of stock of the corporation surviving or resulting from such merger or consolidation, or depository receipts in respect thereof;
 - b. Shares of stock of any other corporation, or depository receipts in respect thereof, which shares of stock (or depository receipts in respect thereof) or depository receipts at the effective date of the merger or consolidation will be either listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or held of record by more than 2,000 holders;
 - c. Cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a. and b. of this paragraph; or
 - d. Any combination of the shares of stock, depository receipts and cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a., b. and c. of this paragraph.

- (3) In the event all of the stock of a subsidiary Delaware corporation party to a merger effected under §253 of this title is not owned by the parent corporation immediately prior to the merger, appraisal rights shall be available for the shares of the subsidiary Delaware corporation.
- (c) Any corporation may provide in its certificate of incorporation that appraisal rights under this section shall be available for the shares of any class or series of its stock as a result of an amendment to its certificate of incorporation, any merger or consolidation in which the corporation is a constituent corporation or the sale of all or substantially all of the assets of the corporation. If the certificate of incorporation contains such a provision, the procedures of this section, including those set forth in subsections (d) and (e) of this section, shall apply as nearly as is practicable.
 - (d) Appraisal rights shall be perfected as follows:
 - (1) If a proposed merger or consolidation for which appraisal rights are provided under this section is to be submitted for approval at a meeting of stockholders, the corporation, not less than 20 days prior to the meeting, shall notify each of its stockholders who was such on the record date for such meeting with respect to shares for which appraisal rights are available pursuant to subsections (b) or (c) hereof that appraisal rights are available for any or all of the shares of the constituent corporations, and shall include in such notice a copy of this section. Each stockholder electing to demand the appraisal of such stockholder's shares shall deliver to the corporation, before the taking of the vote on the merger or consolidation, a written demand for appraisal of such stockholder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such stockholder's shares. A proxy or vote against the merger or consolidation shall not constitute such a demand. A stockholder electing to take such action must do so by a separate written demand as herein provided. Within 10 days after the effective date of such merger or consolidation, the surviving or resulting corporation shall notify each stockholder of each constituent corporation who has complied with this subsection and has not voted in favor of or consented to the merger or consolidation of the date that the merger or consolidation has become effective; or
 - (2) If the merger or consolidation was approved pursuant to §228 or §253 of this title, then either a constituent corporation before the effective date of the merger or consolidation or the surviving or resulting corporation within 10 days thereafter shall notify each of the holders of any class or series of stock of such constituent corporation who are entitled to appraisal rights of the approval of the merger or consolidation and that appraisal rights are available for any or all shares of such class or series of stock of such constituent corporation, and shall include in such notice a copy of this section. Such notice may, and, if given on or after the effective date of the merger or consolidation, shall, also notify such stockholders of the effective date of the merger or consolidation. Any stockholder entitled to appraisal rights may, within 20 days after the date of mailing of such notice, demand in writing from the surviving or resulting corporation the appraisal of such holder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such holder's shares. If such notice did not notify stockholders of the effective date of the merger or consolidation, either (i) each such constituent corporation shall send a second notice before the effective date of the merger or consolidation notifying each of the holders of any class or series of stock of such constituent corporation that are entitled to appraisal rights of the effective date of the merger or consolidation or (ii) the surviving or resulting corporation shall send such a second notice to all such holders on or within 10 days after such effective date; provided, however, that if such second notice is sent more than 20 days following the sending of the first notice, such second notice need only be sent to each stockholder who is entitled to appraisal rights and who has demanded appraisal of such holder's shares in accordance with this subsection. An affidavit of the secretary or assistant secretary or of the transfer agent of the corporation that is required to give either notice that such notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein. For purposes of determining the stockholders entitled to receive either notice, each constituent corporation may fix, in advance, a record date that shall be not more than 10 days prior to the date the notice is given, provided, that if the notice is given on or after the effective date of the merger or consolidation, the record date shall be such effective date. If no record date is

fixed and the notice is given prior to the effective date, the record date shall be the close of business on the day next preceding the day on which the notice is given.

- (e) Within 120 days after the effective date of the merger or consolidation, the surviving or resulting corporation or any stockholder who has complied with subsections (a) and (d) hereof and who is otherwise entitled to appraisal rights, may file a petition in the Court of Chancery demanding a determination of the value of the stock of all such stockholders. Notwithstanding the foregoing, at any time within 60 days after the effective date of the merger or consolidation, any stockholder shall have the right to withdraw such stockholder's demand for appraisal and to accept the terms offered upon the merger or consolidation. Within 120 days after the effective date of the merger or consolidation, any stockholder who has complied with the requirements of subsections (a) and (d) hereof, upon written request, shall be entitled to receive from the corporation surviving the merger or resulting from the consolidation a statement setting forth the aggregate number of shares not voted in favor of the merger or consolidation and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. Such written statement shall be mailed to the stockholder within 10 days after such stockholder's written request for such a statement is received by the surviving or resulting corporation or within 10 days after expiration of the period for delivery of demands for appraisal under subsection (d) hereof, whichever is later.
- (f) Upon the filing of any such petition by a stockholder, service of a copy thereof shall be made upon the surviving or resulting corporation, which shall within 20 days after such service file in the office of the Register in Chancery in which the petition was filed a duly verified list containing the names and addresses of all stockholders who have demanded payment for their shares and with whom agreements as to the value of their shares have not been reached by the surviving or resulting corporation. If the petition shall be filed by the surviving or resulting corporation, the petition shall be accompanied by such a duly verified list. The Register in Chancery, if so ordered by the Court, shall give notice of the time and place fixed for the hearing of such petition by registered or certified mail to the surviving or resulting corporation and to the stockholders shown on the list at the addresses therein stated. Such notice shall also be given by 1 or more publications at least 1 week before the day of the hearing, in a newspaper of general circulation published in the City of Wilmington, Delaware or such publication as the Court deems advisable. The forms of the notices by mail and by publication shall be approved by the Court, and the costs thereof shall be borne by the surviving or resulting corporation.
- (g) At the hearing on such petition, the Court shall determine the stockholders who have complied with this section and who have become entitled to appraisal rights. The Court may require the stockholders who have demanded an appraisal for their shares and who hold stock represented by certificates to submit their certificates of stock to the Register in Chancery for notation thereon of the pendency of the appraisal proceedings; and if any stockholder fails to comply with such direction, the Court may dismiss the proceedings as to such stockholder.
- (h) After determining the stockholders entitled to an appraisal, the Court shall appraise the shares, determining their fair value exclusive of any element of value arising from the accomplishment or expectation of the merger or consolidation, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value. In determining such fair value, the Court shall take into account all relevant factors. In determining the fair rate of interest, the Court may consider all relevant factors, including the rate of interest which the surviving or resulting corporation would have had to pay to borrow money during the pendency of the proceeding. Upon application by the surviving or resulting corporation or by any stockholder entitled to participate in the appraisal proceedings, the Court may, in its discretion, permit discovery or other pretrial proceedings and may proceed to trial upon the appraisal prior to the final determination of the stockholder entitled to an appraisal. Any stockholder whose name appears on the list filed by the surviving or resulting corporation pursuant to subsection (f) of this section and who has submitted such stockholder's certificates of stock to the Register in Chancery, if such is required, may participate fully in all proceedings until it is finally determined that such stockholder is not entitled to appraisal rights under this section.
- (i) The Court shall direct the payment of the fair value of the shares, together with interest, if any, by the surviving or resulting corporation to the stockholders entitled thereto. Interest may be simple or compound, as the Court may direct. Payment shall be so made to each such stockholder, in the case of holders of uncertificated

stock forthwith, and the case of holders of shares represented by certificates upon the surrender to the corporation of the certificates representing such stock. The Court's decree may be enforced as other decrees in the Court of Chancery may be enforced, whether such surviving or resulting corporation be a corporation of this State or of any state.

- (j) The costs of the proceeding may be determined by the Court and taxed upon the parties as the Court deems equitable in the circumstances. Upon application of a stockholder, the Court may order all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorney's fees and the fees and expenses of experts, to be charged pro rata against the value of all the shares entitled to an appraisal.
- (k) From and after the effective date of the merger or consolidation, no stockholder who has demanded appraisal rights as provided in subsection (d) of this section shall be entitled to vote such stock for any purpose or to receive payment of dividends or other distributions on the stock (except dividends or other distributions payable to stockholders of record at a date which is prior to the effective date of the merger or consolidation); provided, however, that if no petition for an appraisal shall be filed within the time provided in subsection (e) of this section, or if such stockholder shall deliver to the surviving or resulting corporation a written withdrawal of such stockholder's demand for an appraisal and an acceptance of the merger or consolidation, either within 60 days after the effective date of the merger or consolidation as provided in subsection (e) of this section or thereafter with the written approval of the corporation, then the right of such stockholder to an appraisal shall cease. Notwithstanding the foregoing, no appraisal proceeding in the Court of Chancery shall be dismissed as to any stockholder without the approval of the Court, and such approval may be conditioned upon such terms as the Court deems just.
- (1) The shares of the surviving or resulting corporation to which the shares of such objecting stockholders would have been converted had they assented to the merger or consolidation shall have the status of authorized and unissued shares of the surviving or resulting corporation.

Bank of America Corporation 2003 Key Associate Stock Plan, As Amended and Restated

Original Effective Date: January 1, 2003 Amended and Restated Effective Date: [, 2004]

Contents

	Page
Article 1. Establishment, Duration and Purpose	1
Article 2. Definitions	1
Article 3. Administration	5
Article 4. Shares Subject to the Plan	6
Article 5. Eligibility and Participation	7
Article 6. Stock Options	7
Article 7. Stock Appreciation Rights	9
Article 8. Restricted Stock and Restricted Stock Units	10
Article 9. Performance Measures	11
Article 10. Beneficiary Designation	12
Article 11. Deferrals	12
Article 12. Rights of Key Associates	12
Article 13. Change in Control	13
Article 14. Amendment, Modification, and Termination	15
Article 15. Withholding	15
Article 16. Indemnification	15
Article 17. Successors	15
Article 18. Legal Construction	16

Bank of America Corporation 2003 Key Associate Stock Plan, As Amended and Restated

Article 1. Establishment, Duration and Purpose

- 1.1 Establishment and Duration of the Plan The Company originally established this Plan effective as of January 1, 2003, and the Plan as originally established was approved by the Company's stockholders. The Plan has been amended and restated effective , 2004 in connection with the merger with FleetBoston Financial Corporation in order to (i) authorize the issuance of additional Shares under the Plan, (ii) modify the permitted mix of Option and non-Option awards under the Plan, (iii) extend the term of the Plan by one year and (iv) make certain other design changes consistent with changes in the economic and business environment since the Plan was originally adopted. The amendment and restatement of the Plan is subject to stockholder approval. The Plan as amended and restated shall remain in effect until the earlier (i) the date that no additional Shares are available for issuance under the Plan, (ii) the date that the Plan has been terminated in accordance with Article 14 or (iii) the close of business on December 31, 2008.
- 1.2 Purpose of the Plan. The Company believes that the compensation of its Key Associates should be significantly linked to the Company's business performance in order to enhance the long-term success and value of the Company. The Plan serves this compensation philosophy by providing a source of stock-based Awards for Key Associates that are intended to further motivate Key Associates to increase the value of the Company's common stock, thereby linking the personal interests of the Key Associates with those of the Company's stockholders. Terms and conditions placed on Awards further encourage the long-term retention of Key Associates. The Plan also provides the Company with a means to better attract and recruit Key Associates of outstanding ability who will further enhance the long-term success and value of the Company through their services.

Article 2. Definitions

Whenever used in the Plan, the following terms shall have the meanings set forth below and, when the meaning is intended, the initial letter of the word is capitalized:

- "Award" means, individually or collectively, a grant under this Plan of Nonqualified Stock Options, Incentive Stock Options, Stock Appreciation Rights, Restricted Stock or Restricted Stock Units.
- "Award Agreement" means an agreement between the Company and each Participant setting forth the terms and provisions applicable to Awards granted under this Plan.
- "Beneficial Owner" or "Beneficial Ownership" shall have the meaning ascribed to such term in Rule 13d-3 of the General Rules and Regulations under the Exchange Act.
 - "Board" or "Board of Directors" means the Board of Directors of the Company.
 - "Change in Control" of the Company means, and shall be deemed to have occurred upon, any of the following events:
 - (a) The acquisition by any Person of Beneficial Ownership of twenty-five percent (25%) or more of either:
 - (i) The then-outstanding Shares (the "Outstanding Shares"); or
 - (ii) The combined voting power of the then-outstanding voting securities of the Company entitled to vote generally in the election of Directors (the "Outstanding Voting Securities");

provided, however, that the following acquisitions shall not constitute a Change in Control for purposes of this subparagraph (a): (A) any acquisition directly from the Company, (B) any acquisition by the Company or any of its Subsidiaries, (C) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or any of its Subsidiaries, or (D) any acquisition by any corporation pursuant to a transaction which complies with clauses (i), (ii) and (iii) of subparagraph (c) below: or

- (b) Individuals who, as of the Effective Date, constitute the Board of Directors (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board of Directors; provided, however, that any individual who becomes a Director subsequent to the Effective Date and whose election, or whose nomination for election by the Company's stockholders, to the Board of Directors was either (i) approved by a vote of at least a majority of the Directors then comprising the Incumbent Board or (ii) recommended by a corporate governance committee comprised entirely of Directors who are then Incumbent Board members shall be considered as though such individual were a member of the Incumbent Board, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest, other actual or threatened solicitation of proxies or consents or an actual or threatened tender offer; or
- (c) Consummation of a reorganization, merger, or consolidation or sale or other disposition of all or substantially all of the assets of the Company (a "Business Combination"), in each case, unless following such Business Combination, (i) all or substantially all of the Persons who were the Beneficial Owners, respectively, of the Outstanding Shares and Outstanding Voting Securities immediately prior to such Business Combination own, directly or indirectly, more than fifty percent (50%) of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from the Business Combination (including, without limitation, a corporation which as a result of such transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions as their ownership, immediately prior to such Business Combination, of the Outstanding Shares and Outstanding Voting Securities, as the case may be (provided, however, that for purposes of this clause (i), any shares of common stock or voting securities of such resulting corporation received by such Beneficial Owners in such Business Combination other than as the result of such Beneficial Owners' ownership of Outstanding Shares or Outstanding Voting Securities immediately prior to such Business Combination shall not be considered to be owned by such Beneficial Owners for the purposes of calculating their percentage of ownership of the outstanding common stock and voting power of the resulting corporation), (ii) no Person (excluding any corporation resulting from such Business Combination or any employee benefit plan (or related trust) of the Company or such corporation resulting from the Business Combination) beneficially owns, directly or indirectly, twenty-five percent (25%) or more of, respectively, the then outstanding shares of common stock of the corporation resulting from the Business Combination or the combined voting power of the then outstanding voting securities of such corporation unless such Person owned twenty-five percent (25%) or more of the Outstanding Shares or Outstanding Voting Securities immediately prior to the Business Combination and (iii) at least a majority of the members of the board of directors of the corporation resulting from such Business Combination were members of the Incumbent Board at the time of the execution of the initial agreement, or the action of the Board, providing for such Business Combination; or
 - (d) Approval by the Company's stockholders of a complete liquidation or dissolution of the Company.

"Code" means the Internal Revenue Code of 1986, as amended from time to time. References to the Code shall include the valid and binding governmental regulations, court decisions and other regulatory and judicial authority issued or rendered thereunder.

"Committee" means the Compensation Committee of the Board of Directors; provided, however, that (i) with respect to Awards to any Key Associates who are Insiders, Committee means all of the members of the Compensation Committee who are "non-employee directors" within the meaning of Rule 16b-3 adopted under

the Exchange Act, and (ii) with respect to Awards to any Key Associates who are Named Executive Officers intended to comply with the Performance-Based Exception, Committee means all of the members of the Compensation Committee who are "outside directors" within the meaning of Section 162(m) of the Code. Committee may also mean any individual or committee of individuals (who need not be Directors) that the Compensation Committee may appoint from time to time to administer the Plan with respect to Awards to Key Associates who are not Insiders or Named Executive Officers, in accordance with and subject to the requirements of Section 3.2.

- "Company" means Bank of America Corporation, a Delaware corporation, and any successor as provided in Article 17 herein.
- "Director" means any individual who is a member of the Board of Directors of the Company.
- "Disability" with respect to a Participant, means "disability" as defined from time to time under any long-term disability plan of the Company or Subsidiary with which the Participant is employed.
- "Earnings Per Share" means "earnings per common share" of the Company based on all earnings (either diluted or without regard to dilution, as selected by the Committee) determined in accordance with generally accepted accounting principles that would be reported in the Company's Annual Report to Stockholders or Annual Report on Form 10-K.
 - "Effective Date" means January 1, 2003.
 - "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, or any successor act thereto.
- "Fair Market Value" of a Share on any date means the closing price of a Share as reflected in the report of composite trading of New York Stock Exchange listed securities for that day (or, if no Shares were publicly traded on that day, the immediately preceding day that Shares were so traded) published in The Wall Street Journal (Eastern Edition) or in any other publication selected by the Committee:provided, however, that if the Shares are misquoted or omitted by the selected publication(s), the Committee shall directly solicit the information from officials of the stock exchanges or from other informed independent market sources.
- "Incentive Stock Option" or "ISO" means an option to purchase Shares granted to a Key Associate under Article 6 herein, and designated as an Incentive Stock Option which is intended to meet the requirements of Section 422 of the Code.
- "Insider" shall mean an individual who is, on the relevant date, an officer, director or ten percent (10%) beneficial owner of any class of the Company's equity securities that is registered pursuant to Section 12 of the Exchange Act, all as defined under Section 16 of the Exchange Act and the rules thereunder.
- "Key Associate" means an employee of the Company or any Subsidiary, including an officer of the Company or a Subsidiary, in a managerial or other important position who, by virtue of such employee's ability, qualifications and performance, has made, or is expected to make, important contributions to the Company or its Subsidiaries, all as determined by the Committee in its discretion.
- "Named Executive Officer" means, for a calendar year, a Participant who is one of the group of "covered employees" for such calendar year within the meaning of Code Section 162(m) or any successor statute.
- "Net Income" means "net income" of the Company determined in accordance with generally accepted accounting principles that would be reported in the Company's Annual Report to Stockholders or Annual Report on Form 10-K.

- "Nonqualified Stock Option" or "NQSO" means an option to purchase Shares granted to a Key Associate under Article 6 herein, and which is not intended to meet the requirements of Code Section 422.
- "Operating Earnings Per Share" means "earnings per common share" of the Company based only on operating earnings (either diluted or without regard to dilution, as selected by the Committee) determined in accordance with generally accepted accounting principles that would be reported in the Company's Annual Report to Stockholders or Annual Report on Form 10-K.
 - "Option" means an Incentive Stock Option or a Nonqualified Stock Option.
 - "Option Price" means the price at which a Share may be purchased by a Participant pursuant to an Option.
- "Participant" means a Key Associate, a former Key Associate or any permitted transferee under the Plan of a Key Associate or former Key Associate who has outstanding an Award granted under the Plan.
- "Performance-Based Exception" means the performance-based exception set forth in Code Section 162(m)(4)(C) from the deductibility limitations of Code Section 162(m).
- "Period of Restriction" means the period during which the transfer of Shares of Restricted Stock or an Award of Restricted Stock Units is limited in some way (based on the passage of time, the achievement of performance goals, or upon the occurrence of other events as determined by the Committee, at its discretion), and the Shares of Restricted Stock or the Restricted Stock Units are subject to a substantial risk of forfeiture, as provided in Article 8 herein.
- "Person" shall have the meaning ascribed to such term in Section 3(a)(9) of the Exchange Act and used in Sections 13(d) and 14(d) thereof, including a "group" within the meaning of Section 13(d).
- "Plan" means the incentive compensation plan set forth herein known as the "Bank of America Corporation 2003 Key Associate Stock Plan," as the same may be amended from time to time.
 - "Prior Plan" means the Bank of America Corporation Key Employee Stock Plan, as amended and restated effective September 24, 1998.
 - "Restricted Stock" means an Award of Shares, subject to a Period of Restriction, that is granted to a Key Associate under Article 8 herein.
- "Restricted Stock Unit" means an Award, subject to a Period of Restriction, that is granted to a Key Associate under Article 8 herein and is settled either (i) by the delivery of one (1) Share for each Restricted Stock Unit or (ii) in cash in an amount equal to the Fair Market Value of one (1) Share for each Restricted Stock Unit, all as specified in the applicable Award Agreement. The Award of a Restricted Stock Unit represents the mere promise of the Company to deliver a Share or the appropriate amount of cash, as applicable, at the end of the Period of Restriction (or such later date as provided by the Award Agreement) in accordance with and subject to the terms and conditions of the applicable Award Agreement, and is not intended to constitute a transfer of "property" within the meaning of Section 83 of the Code.
- "Return on Assets" means "return on average assets" of the Company determined in accordance with generally accepted accounting principles that would be reported in the Company's Annual Report to Stockholders or Annual Report on Form 10-K.
- "Return on Equity" means "return on average common stockholders' equity" of the Company determined in accordance with generally accepted accounting principles that would be reported in the Company's Annual Report to Stockholders or Annual Report on Form 10-K.

"Shareholder Value Added" means the "shareholder value added" performance measure of the Company for a year determined in accordance with generally accepted accounting principles that would be reported in the Company's Annual Report to Stockholders or Annual Report on Form 10-K for the year. In that regard, Shareholder Value Added for a year equals the cash basis operating earnings for the year less a charge for the use of capital for the year. For purposes of any Award intended to satisfy the Performance-Based Exception incorporating Shareholder Value Added as a performance criteria, the Committee shall approve the charge for the use of capital for use in determining Shareholder Value Added for the year within the time required under Code Section 162(m).

"Shares" means the shares of common stock of the Company.

"Stock Appreciation Right" or "SAR" means an Award designated as an SAR that is granted to a Key Associate under Article 7 herein.

"Subsidiary" means any corporation, partnership, joint venture, affiliate, or other entity in which the Company owns more than fifty percent (50%) of the voting stock or voting ownership interest, as applicable, or any other business entity designated by the Committee as a Subsidiary for purposes of the Plan.

"Total Revenue" means the sum of (i) net interest income on a taxable equivalent basis of the Company and (ii) noninterest income of the Company, such amounts determined in accordance with generally accepted accounting principles that would be reported in the Company's Annual Report to Stockholders or Annual Report on Form 10-K

"Total Stockholder Return" means the percentage change of an initial investment in Shares over a specified period assuming reinvestment of all dividends during the period.

Article 3. Administration

- 3.1 Authority of the Committee. The Plan shall be administered by the Committee. Except as limited by law, or by the Certificate of Incorporation or Bylaws of the Company, and subject to the provisions herein, the Committee shall have full power to select Key Associates who shall participate in the Plan; determine the sizes and types of Awards; determine the terms and conditions of Awards in a manner consistent with the Plan; construe and interpret the Plan and any agreement or instrument entered into under the Plan; establish, amend, or waive rules and regulations for the Plan's administration; and (subject to the provisions of Article 14 herein), amend the terms and conditions of any outstanding Award to the extent such terms and conditions are within the discretion of the Committee as provided in the Plan. Further, the Committee shall make all other determinations which may be necessary or advisable for the administration of the Plan.
- 3.2 Delegation. To the extent permitted by applicable law, the Committee may delegate its authority as identified herein to any individual or committee of individuals (who need not be Directors), including without limitation the authority to make Awards to Key Associates who are not Insiders or Named Executive Officers. To the extent that the Committee delegates its authority to make Awards as provided by this Section 3.2, all references in the Plan to the Committee's authority to make Awards and determinations with respect thereto shall be deemed to include the Committee's delegate. Any such delegate shall serve at the pleasure of, and may be removed at any time by, the Committee.
- 3.3 Decisions Binding. All determinations and decisions made by the Committee pursuant to the provisions of the Plan and all related orders and resolutions of the Board shall be final, conclusive and binding on all persons, including the Company, its stockholders, employees, Participants, and their estates and beneficiaries.

Article 4. Shares Subject to the Plan

- **4.1** Number of Shares Available for Grants. Subject to the provisions of this Article 4, the aggregate number of Shares that may be issued to Participants pursuant to Awards granted under the Plan shall not exceed the sum of (A) one hundred million (100,000,000) Shares plus (B) the number of Shares available for awards under the Prior Plan as of December 31, 2002 plus (C) any Shares that were subject to an award under the Prior Plan which award is canceled, terminates, expires or lapses for any reason from and after the Effective Date plus (D) effective as of the consummation of the merger with FleetBoston Financial Corporation and subject to stockholder approval of this amended and restated Plan, fifty-one million (51,000,000) Shares.
- **4.2 Lapsed Awards** If any Award is canceled, terminates, expires, or lapses for any reason, any Shares subject to such Award shall not count against the aggregate number of Shares that may be issued under the Plan set forth in Section 4.1 above.
- 4.3 Shares Used to Pay Option Price and Withholding Taxes If, in accordance with the terms of the Plan, a Participant pays the Option Price for an Option or satisfies any tax withholding requirement with respect to any taxable event arising as a result of this Plan by either tendering previously owned Shares or having the Company withhold Shares, then such Shares surrendered to pay the Option Price or used to satisfy such tax withholding requirements shall not count against the aggregate number of Shares that may be issued under the Plan set forth in Section 4.1 above.
- 4.4 Other Items Not Included. The following items shall not count against the aggregate number of Shares that may be issued under the Plan set forth in Section 4.1 above: (i) the payment in cash of dividends or dividend equivalents under any outstanding Award; (ii) any Award that is settled in cash rather than by issuance of Shares; or (iii) Awards granted through the assumption of, or in substitution for, outstanding awards previously granted to individuals who become Key Associates as a result of a merger, consolidation, acquisition or other corporate transaction involving the Company or any Subsidiary.
- **4.5 Award Limits.** Notwithstanding any provision herein to the contrary, the following provisions shall apply (subject to adjustment in accordance with Section 4.6 below):
 - (i) in no event shall a Participant receive an Award or Awards during any one (1) calendar year covering in the aggregate more than two million (2,000,000) Shares (whether such Award or Awards may be settled in Shares, cash or any combination of Shares and cash);
 - (ii) in no event shall there be granted during the term of the Plan Incentive Stock Options covering more than an aggregate of twenty million (20,000,000) Shares;
 - (iii) in no event shall there be granted from and after January 1, 2004 Shares of Restricted Stock or Restricted Stock Units covering more than an aggregate of:
 - (A) twenty million (20,000,000) Shares, plus
 - (B) the number of Shares covering any Award made under this subparagraph (iii) from and after January 1, 2004 that again become available for issuance (x) under Section 4.2 above because the Award is canceled, terminates, expires, or lapses for any reason or (y) under Section 4.3 above because the Shares are used to pay withholding taxes;

provided, however, that the limitation of this subparagraph (iii) shall not apply to any Award of Restricted Stock or Restricted Stock Units either (X) described in subparagraph (iv) below or (Y) which is earned solely on the basis of the achievement of performance goals; and

- (iv) of the total Shares available for issuance under the Plan, the sum of:
 - (A) eighteen million (18,000,000) Shares, plus

(B) the number of Shares covering any Award made under this subparagraph (iv) from and after January 1, 2004 that again become available for issuance (x) under Section 4.2 above because the Award is canceled, terminates, expires, or lapses for any reason or (y) under Section 4.3 above because the Shares are used to pay withholding taxes,

are set aside to be available exclusively for awards of Restricted Stock or Restricted Stock Units from and after January 1, 2004 in lieu of a portion of annual incentive compensation under the Bank of America Corporation Equity Incentive Plan, the Bank of America Corporation Executive Incentive Compensation Plan or any similar annual incentive compensation program as determined by the Committee.

- 4.6 Adjustments in Authorized Shares. In the event of any change in corporate capitalization, such as a stock split, or a corporate transaction, such as any merger, consolidation, separation, including a spin-off, or other distribution of stock or property of the Company, any reorganization (whether or not such reorganization comes within the definition of such term in Code Section 368) or any partial or complete liquidation of the Company, such adjustment shall be made in the number and class of Shares which may be issued under the Plan and in the number and class of and/or price of Shares subject to outstanding Awards granted under the Plan, as may be determined to be appropriate and equitable by the Committee, in its sole discretion, to prevent dilution or enlargement of rights; provided, however, that the number of Shares subject to any Award shall always be a whole number.
- **4.7 Source of Shares.** Shares issued under the Plan may be original issue shares, treasury stock or shares purchased in the open market or otherwise, all as determined by the Chief Financial Officer of the Company (or the Chief Financial Officer's designee) from time to time, unless otherwise determined by the Committee.

Article 5. Eligibility and Participation

- **5.1** Eligibility. Persons eligible to participate in this Plan are all Key Associates of the Company, as determined by the Committee, including Key Associates who are Directors, but excluding Directors who are not Key Associates.
- 5.2 Actual Participation. Subject to the provisions of the Plan, the Committee may, from time to time, select from all eligible Key Associates those to whom Awards shall be granted and shall determine the nature and amount of each Award.
- 5.3 Non-U.S. Associates. Notwithstanding any provision of the Plan to the contrary, in order to foster and promote achievement of the purposes of the Plan or to comply with provisions of laws in other countries in which the Company operates or has employees, the Committee, in its sole discretion, shall have the power and authority to (i) determine which Key Associates (if any) employed outside the United States are eligible to participate in the Plan, (ii) modify the terms and conditions of any Awards made to such Key Associates and (iii) establish subplans and modified Option exercise and other terms and procedures to the extent such actions may be necessary or advisable.

Article 6. Stock Options

- **6.1 Grant of Options.** Subject to the terms and provisions of the Plan, Options may be granted to Key Associates in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee.
- **6.2 Award Agreement.** Each Option grant shall be evidenced by an Award Agreement that shall specify the Option Price, the duration of the Option, the number of Shares to which the Option pertains, and such other provisions as the Committee shall determine. The Award Agreement also shall specify whether the Option is

intended to be an ISO within the meaning of Section 422 of the Code, or an NQSO whose grant is intended not to fall under Code Section 422.

- **6.3 Option Price**. The Option Price for each grant of an Option under this Plan shall be at least equal to one hundred percent (100%) of the Fair Market Value of a Share on the date the Option is granted.
- **6.4 Duration of Options**. Each Option shall expire at such time as the Committee shall determine at the time of grant: provided, however, that no Option shall be exercisable later than the tenth (10th) anniversary date of its grant.
- **6.5** Exercise of Options. Options granted under this Article 6 shall be exercisable at such times and be subject to such restrictions and conditions as the Committee shall in each instance approve and which shall be set forth in the applicable Award Agreement, which need not be the same for each grant or for each Participant.
- **6.6 Payment.** Options shall be exercised by the delivery of a notice of exercise to the Company, setting forth the number of Shares with respect to which the Option is to be exercised, accompanied by full payment for the Shares. To be effective, notice of exercise must be made in accordance with procedures established by the Company from time to time.

The Option Price due upon exercise of any Option shall be payable to the Company in full either: (a) in cash or its equivalent, or (b) by tendering previously acquired Shares having an aggregate Fair Market Value at the time of exercise equal to the total Option Price (provided that the Shares which are tendered must have been held by the Participant for at least six (6) months prior to their tender to satisfy the Option Price unless such Shares had been acquired by the Participant on the open market), or (c) by a combination of (a) and (b).

As soon as practicable after notification of exercise and full payment, the Company shall deliver the Shares to the Participant in an appropriate amount based upon the number of Shares purchased under the Option(s).

Notwithstanding the foregoing, the Committee also may allow (i) cashless exercises as permitted under Federal Reserve Board's Regulation T, subject to applicable securities law restrictions, or (ii) exercises by any other means which the Committee determines to be consistent with the Plan's purpose and applicable law.

- 6.7 Restrictions on Share Transferability. The Committee may impose such restrictions on any Shares acquired pursuant to the exercise of an Option granted under this Article 6 as it may deem advisable, including, without limitation, restrictions under applicable Federal securities laws, under the requirements of any stock exchange or market upon which such Shares are then listed and/or traded, and under any blue sky or state securities laws applicable to such Shares.
- 6.8 Termination of Employment. Each Participant's Option Award Agreement shall set forth the extent to which the Participant shall have the right to exercise the Option following termination of the Participant's employment with the Company and its Subsidiaries. Such provisions shall be determined in the sole discretion of the Committee, shall be included in the Award Agreement entered into with Participants, need not be uniform among all Options issued pursuant to this Article 6, and may reflect distinctions based on the reasons for termination of employment. In that regard, if an Award Agreement permits exercise of an Option following the death of the Participant, the Award Agreement shall provide that such Option shall be exercisable to the extent provided therein by any person that may be empowered to do so under the Participant's will, or if the Participant shall fail to make a testamentary disposition of the Option or shall have died intestate, by the Participant's executor or other legal representative.

6.9 Nontransferability of Options.

(a) Incentive Stock Options. No ISO granted under this Article 6 may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and

distribution. Further, all ISOs granted to a Participant under the Plan shall be exercisable during his or her lifetime only by such Participant except to the extent otherwise permitted by applicable law.

- (b) **Nonqualified Stock Options**. Except as otherwise provided in a Participant's Award Agreement, no NQSO granted under this Article 6 may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution. Further, except as otherwise provided in a Participant's Award Agreement, all NQSOs granted to a Participant under this Article 6 shall be exercisable during his or her lifetime only by such Participant.
- **6.10** No Rights. A Participant granted an Option shall have no rights as a stockholder of the Company with respect to the Shares covered by such Option except to the extent that Shares are issued to the Participant upon the due exercise of the Option.
- **6.11** No Repricing. Except for adjustments made pursuant to Section 4.6, the Option Price for any outstanding Option granted under the Plan may not be decreased after the date of grant nor may any outstanding Option granted under the Plan be surrendered to the Company as consideration for the grant of a new Option with a lower exercise price without approval of the Company's stockholders.

Article 7. Stock Appreciation Rights

- 7.1 Grant of SARs. Subject to the terms and conditions of the Plan, SARs may be granted to Key Associates at any time and from time to time as shall be determined by the Committee. The Committee shall have complete discretion in determining the number of SARs granted to each Participant (subject to Article 4 herein) and, consistent with the provisions of the Plan, in determining the terms and conditions pertaining to such SARs. The grant price of an SAR shall equal the Fair Market Value of a Share on the date of grant of the SAR.
 - 7.2 Exercise of SARs. SARs may be exercised upon whatever terms and conditions the Committee, in its sole discretion, imposes upon them.
- 7.3 SAR Agreement. Each SAR grant shall be evidenced by an Award Agreement that shall specify the grant price, the term of the SAR, and such other provisions as the Committee shall determine.
- 7.4 Term of SARs. The term of an SAR granted under the Plan shall be determined by the Committee, in its sole discretion provided, however, that such term shall not exceed ten (10) years.
- 7.5 Payment of SAR Amount. Upon exercise of an SAR, a Participant shall be entitled to receive payment from the Company in an amount determined by multiplying:
 - (a) The difference between the Fair Market Value of a Share on the date of exercise over the grant price; by
 - (b) The number of Shares with respect to which the SAR is exercised.

At the discretion of the Committee or as otherwise provided in the applicable Award Agreement, the payment upon SAR exercise shall be in cash, in Shares of equivalent value, or in some combination thereof.

- 7.6 Other Restrictions. Notwithstanding any other provision of the Plan, the Committee may impose such conditions on exercise of an SAR (including, without limitation, the right of the Committee to limit the time of exercise to specified periods) as may be required to satisfy the requirements of Section 16 (or any successor rule) of the Exchange Act or for any other purpose deemed appropriate by the Committee.
- 7.7 Termination of Employment. Each SAR Award Agreement shall set forth the extent to which the Participant shall have the right to exercise the SAR following termination of the Participant's employment with

the Company and its Subsidiaries. Such provisions shall be determined in the sole discretion of the Committee, shall be included in the Award Agreement entered into with Participants, need not be uniform among all SARs issued pursuant to the Plan, and may reflect distinctions based on the reasons for termination of employment. In that regard, if an Award Agreement permits exercise of an SAR following the death of the Participant, the Award Agreement shall provide that such SAR shall be exercisable to the extent provided therein by any person that may be empowered to do so under the Participant's will, or if the Participant shall fail to make a testamentary disposition of the SAR or shall have died intestate, by the Participant's executor or other legal representative.

- 7.8 Nontransferability of SARs. Except as otherwise provided in a Participant's Award Agreement, no SAR granted under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution. Further, except as otherwise provided in a Participant's Award Agreement, all SARs granted to a Participant under the Plan shall be exercisable during his or her lifetime only by such Participant.
- 7.9 No Rights. A Participant granted an SAR shall have no rights as a stockholder of the Company with respect to the Shares covered by such SAR except to the extent that Shares are issued to the Participant upon the due exercise of the SAR.

Article 8. Restricted Stock and Restricted Stock Units

- **8.1 Grant of Restricted Stock.** Subject to the terms and provisions of the Plan, the Committee, at any time and from time to time, may grant Shares of Restricted Stock or Restricted Stock Units to eligible Key Associates in such amounts as the Committee shall determine.
- **8.2 Restricted Stock Agreement** Each grant of Restricted Stock or Restricted Stock Units shall be evidenced by an Award Agreement that shall specify the Period or Periods of Restriction, the number of Shares of Restricted Stock or the number of Restricted Stock Units granted, and such other provisions as the Committee shall determine.
- 8.3 Transferability. Except as provided in this Article 8, the Shares of Restricted Stock or Restricted Stock Units granted herein may not be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated until the end of the applicable Period of Restriction established by the Committee and specified in the Award Agreement, or upon earlier satisfaction of any other conditions, as specified by the Committee in its sole discretion and set forth in the Award Agreement. All rights with respect to the Restricted Stock or Restricted Stock Units granted to a Participant under the Plan shall be available during his or her lifetime only to such Participant.
- 8.4 Other Restrictions. The Committee shall impose such other conditions and/or restrictions on any Shares of Restricted Stock or Restricted Stock Units granted pursuant to the Plan as it may deem advisable including, without limitation, a requirement that Participants pay a stipulated purchase price for each Share of Restricted Stock or each Restricted Stock Unit, restrictions based upon the achievement of specific performance goals (Company-wide, divisional, and/or individual), time-based restrictions on vesting following the attainment of the performance goals, and/or restrictions under applicable Federal or state securities laws.

The Company shall retain the Shares of Restricted Stock in the Company's possession until such time as all conditions and/or restrictions applicable to such Shares have been satisfied.

Except as otherwise provided in this Article 8, Shares of Restricted Stock covered by each Restricted Stock grant made under the Plan shall become freely transferable by the Participant after the last day of the Period of Restriction.

- **8.5 Settlement of Restricted Stock Units.** Any Restricted Stock Units that become payable in accordance with the terms and conditions of the applicable Award Agreement shall be settled in cash, Shares, or a combination of cash and Shares as determined by the Committee in its discretion or as otherwise provided for under the Award Agreement.
- **8.6** Voting Rights. During the Period of Restriction, Participants holding Shares of Restricted Stock granted hereunder may exercise full voting rights with respect to those Shares. There shall be no voting rights with respect to Restricted Stock Units.
- 8.7 Dividends and Other Distributions. During the Period of Restriction, Participants holding Shares of Restricted Stock granted hereunder may receive regular cash dividends paid with respect to the underlying Shares while the Restricted Stock is held by the Company. The Committee may apply any restrictions to the dividends that the Committee deems appropriate. The Committee, in its discretion, may also grant dividend equivalents rights with respect to earned but unpaid Restricted Stock Units as evidenced by the applicable Award Agreement.
- 8.8 Termination of Employment. Each Restricted Stock or Restricted Stock Unit Award Agreement shall set forth the extent to which the Participant shall have the right to receive unvested Restricted Shares or Restricted Stock Units following termination of the Participant's employment with the Company and its Subsidiaries. Such provisions shall be determined in the sole discretion of the Committee, shall be included in the Award Agreement entered into with Participants, need not be uniform among all Shares of Restricted Stock or Restricted Stock Units issued pursuant to the Plan, and may reflect distinctions based on the reasons for termination of employment. In amplification but not limitation of the foregoing, in the case of an award of Restricted Stock or Restricted Stock Units to a Named Executive Officer which is intended to qualify for the Performance-Based Exception, the Award Agreement may provide that such Restricted Stock or Restricted Stock Units may become payable in the event of a termination of employment by reason of death, Disability or Change in Control, regardless of whether the related performance goal has been previously attained.
- 8.9 Limitation on Vesting for Certain Awards. Notwithstanding any provision of the Plan to the contrary, an Award of Restricted Stock or Restricted Stock Units that vests solely on the basis of the passage of time (e.g., not on the basis of any performance standards) shall not vest more quickly than ratably over the three (3) year period beginning on the first anniversary of the Award, except that the Award may vest sooner under any of the following circumstances as more specifically set forth in the applicable Award Agreement: (i) the Participant's death, (ii) the Participant's Disability, (iii) the Participant's "retirement" as defined in the Award Agreement consistent with the Company's retirement policies and programs, (iv) a Participant's termination of employment with the Company and its Subsidiaries due to workforce reduction, job elimination or divestiture as determined by the Committee, (v) a Change in Control consistent with the provisions of Article 13 hereof or (vi) in connection with establishing the terms and conditions of employment of a Key Associate necessary for the recruitment of the Key Associate or as the result of a business combination or acquisition by the Company or any of its Subsidiaries. The provisions of this Section 8.9 shall not apply to any Award of Restricted Stock or Restricted Stock Units that is made to a Key Associate as a portion of the Key Associate, including any such program applicable to an Insider or Named Executive Officer).

Article 9. Performance Measures

The performance measure(s) to be used for purposes of Awards to Named Executive Officers which are designed to qualify for the Performance-Based Exception shall be chosen from among the following alternatives:

- · Earnings Per Share;
- Net Income;

- Operating Earnings Per Share;
- Return On Assets:
- Return On Equity;
- · Shareholder Value Added;
- Total Revenue;
- · Total Stockholder Return;
- · customer satisfaction (determined based on objective criteria approved by the Committee);
- expense management;
- operating margin;
- operating leverage; or
- · cash flow.

The Committee shall have the discretion to adjust the determinations of the degree of attainment of the preestablished performance goals; provided, however, that Awards which are designed to qualify for the Performance-Based Exception, and which are held by Named Executive Officers, may not be adjusted upward (the Committee shall retain the discretion to adjust such Awards downward).

In the event that applicable tax and/or securities laws change to permit Committee discretion to alter the governing performance measures without obtaining stockholder approval of such changes, the Committee shall have sole discretion to make such changes without obtaining stockholder approval.

Article 10. Beneficiary Designation

Except as otherwise provided in an Award Agreement, each Participant under the Plan may, from time to time, name any beneficiary or beneficiaries (who may be named contingently or successively) to whom any benefit under the Plan is to be paid in case of his or her death before he or she receives any or all of such benefit. Each such designation shall revoke all prior designations by the same Participant, shall be in a form prescribed by the Company, and will be effective only when filed by the Participant in writing with the Company during the Participant's lifetime. In the absence of any such designation, benefits remaining unpaid at the Participant's death shall be paid to the Participant's estate.

Article 11. Deferrals

The Committee may permit a Participant to defer such Participant's receipt of the payment of cash or the delivery of Shares that would otherwise be due to such Participant by virtue of the exercise of an Option or SAR or the lapse or waiver of restrictions with respect to Restricted Stock or Restricted Stock Units. If any such deferral election is required or permitted, the Committee shall, in its sole discretion, establish rules and procedures for such payment deferrals.

Article 12. Rights of Key Associates

12.1 Employment. Nothing in the Plan shall interfere with or limit in any way the right of the Company to terminate any Participant's employment at any time, nor confer upon any Participant any right to continue in the employ of the Company.

For purposes of this Plan, a transfer of a Participant's employment between the Company and a Subsidiary, or between Subsidiaries, shall not be deemed to be a termination of employment. Upon such a transfer, the Committee may make such adjustments to outstanding Awards as it deems appropriate to reflect the changed reporting relationships.

12.2 Participation. No Key Associate shall have the right to be selected to receive an Award under this Plan, or, having been so selected, to be selected to receive a future Award.

Article 13. Change in Control

- 13.1 Treatment of Outstanding Awards. Upon the occurrence of a Change in Control, unless otherwise specifically prohibited under applicable laws, or by the rules and regulations of any governing governmental agencies or national securities exchanges:
 - (a) Any and all outstanding Options and SARs held by persons employed with the Company or any Subsidiary on the date of the Change in Control shall become immediately exercisable, and shall remain exercisable throughout their entire term;
 - (b) Any restriction periods and restrictions imposed on outstanding Shares of Restricted Stock or Restricted Stock Units held by persons employed with the Company or any Subsidiary on the date of the Change in Control shall lapse;
 - (c) If applicable, the target payout opportunities attainable under all outstanding Awards of Restricted Stock or Restricted Stock Units held by persons employed with the Company or any Subsidiary on the date of the Change in Control shall be deemed to have been fully earned for the entire performance period(s) as of the effective date of the Change in Control, and the vesting of all such Awards shall be accelerated as of the effective date of the Change in Control; and
 - (d) Subject to Article 14 herein, the Committee shall have the authority to make any modifications to the Awards as determined by the Committee to be appropriate before the effective date of the Change in Control.
- 13.2 Limitation on Change-in-Control Benefits. It is the intention of the Company and the Participants to reduce the amounts payable or distributable to a Participant hereunder if the aggregate Net After Tax Receipts (as defined below) to the Participant would thereby be increased, as a result of the application of the excise tax provisions of Section 4999 of the Code. Accordingly, anything in this Plan to the contrary notwithstanding, in the event that the certified public accountants regularly employed by the Company immediately prior to any "change" described below (the "Accounting Firm") shall determine that receipt of all Payments (as defined below) would subject the Participant to tax under Section 4999 of the Code, it shall determine whether some amount of Payments would meet the definition of a "Reduced Amount" (as defined below). If the Accounting Firm determines that there is a Reduced Amount, the aggregate Payments shall be reduced to such Reduced Amount in accordance with the provisions of Section 13.2(b) below.
 - (a) For purposes of this Section 13.2(a):
 - (i) A "Payment" shall mean any payment or distribution in the nature of compensation to or for the benefit of a Participant who is a "disqualified individual" within the meaning of Section 280G(c) of the Code and which is contingent on a "change" described in Section 280G(b)(2)(A)(i) of the Code with respect to the Company, whether paid or payable pursuant to this Plan or otherwise;
 - (ii) "Plan Payment" shall mean a Payment paid or payable pursuant to this Plan (disregarding this Section 13.2);
 - (iii) "Net After Tax Receipt" shall mean the Present Value of a Payment, net of all taxes imposed on the Participant with respect thereto under Sections 1 and 4999 of the Code, determined by applying the highest marginal rate under Section 1 of the Code which applied to the Participant's Federal taxable income for the immediately preceding taxable year;

- (iv) "Present Value" shall mean such value determined in accordance with Section 280G(d)(4) of the Code; and
- (v) "Reduced Amount" shall mean the smallest aggregate amount of Payments which (A) is less than the sum of all Payments and (B) results in aggregate Net After Tax Receipts which are equal to or greater than the Net After Tax Receipts which would result if all Payments were paid to or for the benefit of the Participant.
- (b) If the Accounting Firm determines that aggregate Payments should be reduced to the Reduced Amount, the Committee shall promptly give the Participant notice to that effect and a copy of the detailed calculation thereof, and the Participant may then elect, in the Participant's sole discretion, which and how much of the Payments, including without limitation Plan Payments, shall be eliminated or reduced (as long as after such election the Present Value of the aggregate Payments is equal to the Reduced Amount), and shall advise the Committee in writing of such election within ten (10) days of the Participant's receipt of notice. If no such election is made by the Participant within such ten (10) day period, the Committee may elect which of the Payments, including without limitation Plan Payments, shall be eliminated or reduced (as long as after such election the Present Value of the aggregate Payments is equal to the Reduced Amount) and shall notify the Participant promptly of such election. All determinations made by the Accounting Firm under this Section 13.2 shall be binding upon the Company and the Participant and shall be made within sixty (60) days immediately following the event constituting the "change" referred to above. As promptly as practicable following such determination, the Company shall pay to or distribute for the benefit of the Participant such Payments as are then due to the Participant under this Plan.
- (c) At the time of the initial determination by the Accounting Firm hereunder, it is possible that amounts will have been paid or distributed by the Company to or for the benefit of the Participant pursuant to this Plan which should not have been so paid or distributed ("Overpayment") or that additional amounts which will have not been paid or distributed by the Company to or for the benefit of the Participant pursuant to this Plan could have been so paid or distributed ("Underpayment"), in each case, consistent with the calculation of the Reduced Amount hereunder. In the event that the Accounting Firm, based either upon the assertion of a deficiency by the Internal Revenue Service against the Company or the Participant which the Accounting Firm believes has a high probability of success or controlling precedent or other substantial authority, determines that an Overpayment has been made, any such Overpayment paid or distributed by the Company to or for the benefit of the Participant shall be treated for all purposes as a loan ab initio to the Participant which the Participant shall repay to the Company together with interest at the applicable Federal rate provided for in Section 7872(f)(2) of the Code; provided, however, that no such loan shall be deemed to have been made and no amount shall be payable by the Participant to the Company if and to the extent (i) such deemed loan and payment would not either reduce the amount on which the Participant is subject to tax under Section 1 and Section 4999 of the Code or generate a refund of such taxes or (ii) the Participant is subject to the prohibition on personal loans under Section 402 of the Sarbanes-Oxley Act of 2002.

In the event that the Accounting Firm, based upon controlling precedent or other substantial authority, determines that an Underpayment has occurred, any such Underpayment shall be promptly paid by the Company to or for the benefit of the Participant together with interest at the applicable Federal rate provided for in Section 7872(f)(2) of the Code.

13.3 Termination, Amendment, and Modifications of Change-in-Control Provisions. Notwithstanding any other provision of this Plan or any Award Agreement provision, the provisions of this Article 13 may not be terminated, amended, or modified on or after the date of a Change in Control to affect adversely any Award theretofore granted under the Plan without the prior written consent of the Participant with respect to said Participant's outstanding Awards; provided, however, the Board of Directors, upon recommendation of the Committee, may terminate, amend, or modify this Article 13 at any time and from time to time prior to the date of a Change in Control.

Article 14. Amendment, Modification, and Termination

- 14.1 Amendment, Modification, and Termination. The Board may at any time and from time to time, alter, amend, suspend or terminate the Plan in whole or in part; provided, however, that an amendment to the Plan may be conditioned on the approval of the stockholders of the Company if and to the extent the Board determines that stockholder approval is necessary or appropriate.
- 14.2 Awards Previously Granted. No termination, amendment, or modification of the Plan shall adversely affect in any material way any Award previously granted under the Plan, without the written consent of the Participant holding such Award.
- 14.3 Acceleration of Award Vesting; Waiver of Restrictions. Notwithstanding any provision of this Plan or any Award Agreement provision to the contrary, the Committee, in its sole and exclusive discretion, shall have the power at any time to (i) accelerate the vesting of any Award granted under the Plan, including, without limitation, acceleration to such a date that would result in said Awards becoming immediately vested, or (ii) waive any restrictions of any Award granted under the Plan.
- 14.4 No Repricing. Nothing in the provisions of this Article 14 shall be construed to permit the repricing of any outstanding Option as otherwise prohibited by the provisions of Section 6.12.

Article 15. Withholding

- 15.1 Tax Withholding. The Company shall have the power and the right to deduct or withhold, or require a Participant to remit to the Company, an amount sufficient to satisfy Federal, state, and local taxes (including the Participant's FICA obligation) required by law to be withheld with respect to any taxable event arising as a result of this Plan.
- 15.2 Share Withholding. The Company may cause any tax withholding obligation described in Section 15.1 to be satisfied by the Company withholding Shares having a Fair Market Value on the date the tax is to be determined equal to the minimum statutory total tax which could be imposed on the transaction. In the alternative, the Company may permit Participants to elect to satisfy the tax withholding obligation, in whole or in part, by either (i) having the Company withhold Shares having a Fair Market Value on the date the tax is to be determined equal to the minimum statutory total tax which could be imposed on the transaction or (ii) tendering previously acquired Shares having an aggregate Fair Market Value equal to the minimum statutory total tax which could be imposed on the transaction (provided that the Shares which are tendered must have been held by the Participant for at least six (6) months prior to their tender unless such Shares had been acquired by the Participant on the open market). All such elections shall be irrevocable, made in writing, signed by the Participant, and shall be subject to any restrictions or limitations that the Committee, in its sole discretion, deems appropriate.

Article 16. Indemnification

Provisions for the indemnification of officers and directors of the Company in connection with the administration of the Plan shall be as set forth in the Company's Certificate of Incorporation and Bylaws as in effect from time to time.

Article 17. Successors

All obligations of the Company under the Plan with respect to Awards granted hereunder shall be binding on any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation, or otherwise, of all or substantially all of the business and/or assets of the Company.

Article 18. Legal Construction

- 18.1 Gender and Number. Except where otherwise indicated by the context, any masculine term used herein also shall include the feminine; the plural shall include the singular and the singular shall include the plural.
- 18.2 Severability. In the event any provision of the Plan shall be held illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included.
- 18.3 Requirements of Law. The granting of Awards and the issuance of Shares under the Plan shall be subject to all applicable laws, rules, and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.
- 18.4 Securities Law Compliance. With respect to Insiders, transactions under this Plan are intended to comply with all applicable conditions of Rule 16b-3 or its successors under the Exchange Act. To the extent any provision of the plan or action by the Committee fails to so comply, it shall be deemed null and void, to the extent permitted by law and deemed advisable by the Committee.
- 18.5 No Conflict. Unless otherwise provided for by an Award Agreement, in the event of any conflict between the terms of the Plan and the terms of an Award Agreement, the terms of the Plan shall control.
- 18.6 Governing Law. To the extent not preempted by Federal law, the Plan, and all agreements hereunder, shall be construed in accordance with and governed by the laws of the State of Delaware.

PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

ITEM 20. Indemnification of Directors and Officers.

Section 145(a) of the General Corporation Law of the State of Delaware ("Delaware Corporation Law") provides, in general, that a corporation shall have the power to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation), because the person is or was a director or officer of the corporation. Such indemnity may be against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with such action, suit or proceeding, if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation and if, with respect to any criminal action or proceeding, the person did not have reasonable cause to believe the person's conduct was unlawful.

Section 145(b) of the Delaware Corporation Law provides, in general, that a corporation shall have the power to indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the corporation to procure a judgment in its favor because the person is or was a director or officer of the corporation, against any expenses (including attorneys' fees) actually and reasonably incurred by the person in connection with the defense or settlement of such action or suit if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation, except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

Section 145(g) of the Delaware Corporation Law provides, in general, that a corporation shall have the power to purchase and maintain insurance on behalf of any person who is or was a director or officer of the corporation against any liability asserted against the person in any such capacity, or arising out of the person's status as such, whether or not the corporation would have the power to indemnify the person against such liability under the provisions of the Delaware Corporation Law.

Article VIII of Bank of America's bylaws provides for indemnification to the fullest extent authorized by the Delaware Corporation Law for any person who is or was a director or officer of Bank of America who is or was involved or threatened to be made so involved in any proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such person is or was serving as a director or officer of Bank of America or was serving at the request of Bank of America as a director or officer of any other enterprise. Such indemnification is provided only if the director or officer acted in good faith and in a manner that the director or officer reasonably believed to be in, or not opposed to, the best interests of Bank of America, and with respect to any criminal proceeding, had no reasonable cause to believe that the conduct was unlawful.

The foregoing is only a general summary of certain aspects of Delaware law and Bank of America's bylaws dealing with indemnification of directors and officers, and does not purport to be complete. It is qualified in its entirety by reference to the detailed provisions of Section 145 of the Delaware Corporation Law and Article VIII of the bylaws of Bank of America.

ITEM 21. Exhibits and Financial Statement Schedules.

(a) Exhibits. The following is a list of Exhibits to this Registration Statement:

Exhibit No.	Description
2(a)	Agreement and Plan of Merger, by and between FleetBoston and Bank of America, dated as of October 27, 2003 (included in Part I as Appendix A to the accompanying joint proxy statement/prospectus)
3(a)	Amended and Restated Certificate of Incorporation of Bank of America, as in effect on the date hereof, incorporated by reference to Exhibit 99.1 of Bank of America's Current Report on Form 8-K filed May 7, 1999
3(b)	Amended and Restated Bylaws of Bank of America, as in effect on the date hereof, incorporated by reference to Exhibit 99.2 of Bank of America's Current Report on Form 8-K filed May 7, 1999
4(a)	Certificate of Designation for Bank of America 6.75% Perpetual Preferred Stock
4(b)	Certificate of Designation for Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock
5(a)	Opinion of Paul J. Polking, Esq., Executive Vice President and General Counsel of Bank of America, as to the validity of the shares of Bank of America common stock and Bank of America preferred stock being registered
8(a)	Opinion of Cleary, Gottlieb, Steen & Hamilton, counsel to Bank of America, as to material U.S. federal tax matters
8(b)	Opinion of Wachtell, Lipton, Rosen & Katz, counsel to FleetBoston, as to material U.S. federal tax matters
10(a)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and Eugene M. McQuade
10(b)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and H. Jay Sarles
10(c)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and Bradford H. Warner
10(d)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and Brian T. Moynihan
10(e)	Form of the Bank of America Corporation 2003 Key Associate Stock Plan, incorporated by reference to Exhibit 10(j) of Bank of America's Annual Report on Form 10-K filed March 3, 2003
10(f)	Form of the Bank of America Corporation 2003 Key Associate Stock Plan, as amended and restated (included in Part I as Appendix H to the accompanying joint proxy statement/prospectus)
23(a)	Consent of Paul J. Polking, Esq., Executive Vice President and General Counsel of Bank of America (included in Exhibit 5(a) to this Registration Statement)
23(b)	Consent of PricewaterhouseCoopers LLP as to Bank of America
23(c)	Consent of PricewaterhouseCoopers LLP as to FleetBoston
23(d)	Consent of Cleary, Gottlieb, Steen & Hamilton (included in Exhibit 8(a) to this Registration Statement)
23(e)	Consent of Wachtell, Lipton, Rosen & Katz (included in Exhibit 8(b) to this Registration Statement)
24(a)*	Power of Attorney and Certified Resolutions
99(a)	Consent of Goldman, Sachs & Co.

Exhibit No.	Description
99(b)	Consent of Morgan Stanley & Co. Incorporated
99(c)	Stock Option Agreement, dated as of October 27, 2003, by and between FleetBoston (as issuer) and Bank of America (as grantee) (included as Appendix B to the accompanying joint proxy statement/prospectus)
99(d)	Stock Option Agreement, dated as of October 27, 2003, by and between Bank of America (as issuer) and FleetBoston (as grantee) (included as Appendix C to the accompanying joint proxy statement/prospectus)
99(e)	Notice of Special Meeting of Stockholders of Bank of America (included in the accompanying joint proxy statement/prospectus)
99(f)	Notice of Special Meeting of Stockholders of FleetBoston (included in the accompanying joint proxy statement/prospectus)
99(g)	Form of Proxy for Special Meeting of Stockholders of Bank of America
99(h)	Form of Proxy for Special Meeting of Stockholders of FleetBoston

^{*} Filed with the initial Registration Statement on Form S-4, dated December 4, 2003.

(b) Financial Statement Schedules.

Not Applicable.

(c) Fairness Opinions.

The opinion of each of Goldman Sachs and Morgan Stanley is included in Part I as Appendix D and E, respectively, to the accompanying joint proxy statement/prospectus.

ITEM 22. Undertakings.

The undersigned Registrant hereby undertakes:

- (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement: (i) to include any prospectus required by Section 10(a)(3) of the Securities Act of 1933; (ii) to reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement (notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20% change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement); and (iii) to include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement.
- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (4) That, for purposes of determining any liability under the Securities Act of 1933, each filing of the Registrant's annual report pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934 that is

incorporated by reference in this registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

- (5) That prior to any public reoffering of the securities registered hereunder through use of a prospectus which is a part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the Registrant undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.
- (6) That every prospectus (i) that is filed pursuant to paragraph (5) above, or (ii) that purports to meet the requirements of Section 10(a)(3) of the Securities Act of 1933 and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to this registration statement and will not be used until such amendment has become effective, and that for the purpose of determining liabilities under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (7) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of Bank of America pursuant to the foregoing provisions, or otherwise, Bank of America has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by Bank of America of expenses incurred or paid by a director, officer or controlling person of Bank of America in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, Bank of America will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.
- (8) To respond to requests for information that is incorporated by reference into the prospectus pursuant to Items 4, 10(b), 11 or 13 of this form, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the registration statement through the date of responding to the request.
- (9) To supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in this registration statement when it became effective.

[Remainder of Page Intentionally Left Blank]

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the Registrant has duly caused this Amendment No. 2 to Form S-4 Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the city of Charlotte, state of North Carolina, on January 30, 2004.

BANK OF AMERICA CORPORATION

By: /s/ Kenneth D. Lewis*

Kenneth D. Lewis Chairman, President and Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1933, this Amendment No. 2 to Form S-4 Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

Signature ——	Title	Date
/s/ Kenneth D. Lewis*	Chairman, President, Chief Executive Officer and Director	January 30, 2004
Kenneth D. Lewis	(Principal Executive Officer)	
/s/ James H. Hance, Jr.*	Vice Chairman, Chief Financial Officer and Director	January 30, 2004
James H. Hance, Jr.	(Principal Financial Officer)	
/s/ Marc D. Oken*	Executive Vice President and Principal Financial Executive	January 30, 2004
Marc D. Oken	(Principal Accounting Officer)	
/s/ John R. Belk*	Director	January 30, 2004
John R. Belk		
/s/ Charles W. Coker*	Director	January 30, 2004
Charles W. Coker		
/s/ Frank Dowd, IV*	Director	January 30, 2004
Frank Dowd, IV		
/s/ Kathleen F. Feldstein*	Director	January 30, 2004
Kathleen F. Feldstein		
/s/ Paul Fulton*	Director	January 30, 2004
Paul Fulton		
/s/ Donald E. Guinn*	Director	January 30, 2004
Donald E. Guinn		

Signature	Title	Date
/s/ Walter E. Massey*	Director	January 30, 2004
Walter E. Massey		
/s/ C. Steven McMillan*	Director	January 30, 2004
C. Steven McMillan		
/s/ Patricia E. Mitchell*	Director	January 30, 2004
Patricia E. Mitchell		
/s/ Edward L. Romero*	Director	January 30, 2004
Edward L. Romero		
/s/ O. Temple Sloan, Jr.*	Director	January 30, 2004
O. Temple Sloan, Jr.		
/s/ Meredith R. Spangler*	Director	January 30, 2004
Meredith R. Spangler		
/s/ Ronald Townsend*	Director	January 30, 2004
Ronald Townsend		
/s/ Jackie M. Ward*	Director	January 30, 2004
Jackie M. Ward		
/s/ Virgil R. Williams*	Director	January 30, 2004
Virgil R. Williams		
*By: /s/ Rachel R. Cummings		
Rachel R. Cummings Attorney-in-Fact		

EXHIBIT INDEX

Exhibit No.	Description
2(a)	Agreement and Plan of Merger, by and between FleetBoston and Bank of America, dated as of October 27, 2003 (included in Part I as Appendix A to the accompanying joint proxy statement/prospectus)
3(a)	Amended and Restated Certificate of Incorporation of Bank of America, as in effect on the date hereof, incorporated by reference to Exhibit 99.1 of Bank of America's Current Report on Form 8-K filed May 7, 1999
3(b)	Amended and Restated Bylaws of Bank of America, as in effect on the date hereof, incorporated by reference to Exhibit 99.2 of Bank of America's Current Report on Form 8-K filed May 7, 1999
4(a)	Certificate of Designation for Bank of America 6.75% Perpetual Preferred Stock
4(b)	Certificate of Designation for Bank of America Fixed/Adjustable Rate Cumulative Preferred Stock
5(a)	Opinion of Paul J. Polking, Esq., Executive Vice President and General Counsel of Bank of America, as to the validity of the shares of Bank of America common stock and Bank of America preferred stock being registered
8(a)	Opinion of Cleary, Gottlieb, Steen & Hamilton, counsel to Bank of America, as to material U.S. federal tax matters
8(b)	Opinion of Wachtell, Lipton, Rosen & Katz, counsel to FleetBoston, as to material U.S. federal tax matters
10(a)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and Eugene M. McQuade
10(b)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and H. Jay Sarles
10(c)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and Bradford H. Warner
10(d)*	Employment Agreement, dated as of October 27, 2003, by and between Bank of America and Brian T. Moynihan
10(e)	Form of the Bank of America Corporation 2003 Key Associate Stock Plan, incorporated by reference to Exhibit 10(j) of Bank of America's Annual Report on Form 10-K filed March 3, 2003
10(f)	Form of the Bank of America Corporation 2003 Key Associate Stock Plan, as amended and restated (included in Part I as Appendix H to the accompanying joint proxy statement/prospectus)
23(a)	Consent of Paul J. Polking, Esq., Executive Vice President and General Counsel of Bank of America (included in Exhibit 5(a) to this Registration Statement)
23(b)	Consent of PricewaterhouseCoopers LLP as to Bank of America
23(c)	Consent of PricewaterhouseCoopers LLP as to FleetBoston
23(d)	Consent of Cleary, Gottlieb, Steen & Hamilton (included in Exhibit 8(a) to this Registration Statement)
23(e)	Consent of Wachtell, Lipton, Rosen & Katz (included in Exhibit 8(b) to this Registration Statement)
24(a)*	Power of Attorney and Certified Resolutions
99(a)	Consent of Goldman, Sachs & Co.
99(b)	Consent of Morgan Stanley & Co. Incorporated
99(c)	Stock Option Agreement, dated as of October 27, 2003, by and between FleetBoston (as issuer) and Bank of America (as grantee) (included as Appendix B to the accompanying joint proxy statement/prospectus)

Exhibit No.	Description
99(d)	Stock Option Agreement, dated as of October 27, 2003, by and between Bank of America (as issuer) and FleetBoston (as grantee) (included as Appendix C to the accompanying joint proxy statement/prospectus)
99(e)	Notice of Special Meeting of Stockholders of Bank of America (included in the accompanying joint proxy statement/prospectus)
99(f)	Notice of Special Meeting of Stockholders of FleetBoston (included in the accompanying joint proxy statement/prospectus)
99(g)	Form of Proxy for Special Meeting of Stockholders of Bank of America
99(h)	Form of Proxy for Special Meeting of Stockholders of FleetBoston

^{*} Filed with the initial Registration Statement on Form S-4, dated December 4, 2003.

CERTIFICATE OF DESIGNATION

OF

6.75% PERPETUAL PREFERRED STOCK

OF

BANK OF AMERICA CORPORATION

(Pursuant to Section 151 of the Delaware Corporation Law)

Bank of America Corporation, a corporation organized and existing under the General Corporation Law of the State of Delaware (the "Corporation"), hereby certifies that the following resolutions were adopted by the Board of Directors of the Corporation (the "Board of Directors") pursuant to the authority of the Board of Directors as required by Section 151 of the General Corporation Law of the State of Delaware, at a meeting duly convened and held on January 28, 2004:

RESOLVED, that pursuant to the authority granted to and vested in the Board of Directors in accordance with the provisions of the Amended and Restated Certificate of Incorporation of the Corporation, the Board of Directors hereby creates a series of the Corporation's previously authorized preferred stock, without par value (the "Preferred Stock") and hereby states the designation and number thereof and fixes the voting powers, preferences and relative, participating, optional and other special rights, and the qualifications, limitations and restrictions thereof, as follows:

(a) <u>Designation</u>. The designation of the series of Preferred Stock shall be "6.75% Perpetual Preferred Stock" (hereinafter called this "Series") and the number of shares constituting this Series is Six Hundred Ninety Thousand (690,000).

(b) Dividend Rate.

(1) The holders of shares of this Series shall be entitled to receive dividends thereon at a rate of 6.75% per annum computed on the basis of an issue price thereof of \$250 per share, and no more, payable quarterly out of the funds of the Corporation legally available for the payment of dividends. Such dividends shall be cumulative from the date of original issue of such shares and shall be payable, when, as and if declared by the Board, on January 15, April 15, July 15 and October 15 of each year, commencing April 15, 2004 (a "Dividend Payment Date"). Each such dividend shall be paid to the holders of record of shares of this Series as they appear on the stock register of the Corporation on such record date, not exceeding 30 days preceding the payment date thereof, as shall be fixed by the Board. Dividends on account of arrears for any past quarters may be declared and paid at any time, without reference to any regular dividend payment date, to holders of record on such date, not exceeding 45 days preceding the payment date thereof, as may be fixed by the Board.

(2) If one or more amendments to the Internal Revenue Code of 1986, as amended (the "Code"), are enacted that change the percentage of the dividends received deduction (currently 70%) as specified in Section 243(a)(1) of the Code or any successor provision (the "Dividends Received Percentage"), the amount of each dividend payable per share of this Series for dividend payments made on or after the date of enactment of such change shall be adjusted by multiplying the amount of the dividend payable determined as described above (before adjustment) by a factor which shall be the number determined in accordance with the following formula (the "DRD Formula"), and rounding the result to the nearest cent:

For the purposes of the DRD Formula, "DRP" means the Dividends Received Percentage applicable to the dividend in question. No amendment to the Code, other than a change in the percentage of the dividends received deduction set forth in Section 243(a)(1) of the Code or any successor provision, will give rise to an adjustment. Notwithstanding the foregoing provisions, in the event that, with respect to any such amendment, the Corporation shall receive either an unqualified opinion of independent recognized tax counsel or a private letter ruling or similar form of authorization from the Internal Revenue Service to the effect that such an amendment would not apply to dividends payable on shares of this Series, then any such amendment shall not result in the adjustment provided for pursuant to the DRD Formula. The Corporation's calculation of the dividends payable as so adjusted and as certified accurate as to calculation and reasonable as to method by the independent certified public accountants then regularly engaged by the Corporation shall be final and not subject to review.

If any amendment to the Code which reduces the Dividends Received Percentage is enacted after a dividend payable on a Dividend Payment Date has been declared, the amount of dividend payable on such Dividend Payment Date will not be increased; but instead, an amount, equal to the excess of (x) the product of the dividends paid by the Corporation on such Dividend Payment Date and the DRD Formula (where the DRP used in the DRD Formula would be equal to the reduced Dividends Received Percentage) and (y) the dividends paid by the Corporation on such Dividend Payment Date, will be payable to holders of record on the next succeeding Dividend Payment Date in addition to any other amounts payable on such date.

(3) No full dividends shall be declared or paid or set apart for payment on the Preferred Stock of any series ranking, as to dividends, on a parity with or junior to this Series for any period unless full cumulative dividends have been or contemporaneously are declared and paid or declared and a sum sufficient for the payment thereof set apart for such payment on this Series for all dividend payment periods terminating on or prior to the date of payment of such full cumulative dividends. When dividends are not paid in full, as aforesaid, upon the shares of this Series and any other preferred stock ranking on a parity as to dividends with this Series, all dividends declared upon shares of this Series

and any other class or series of preferred stock of the Corporation ranking on a parity as to dividends with this Series shall be declared pro rata so that the amount of dividends declared per share on this Series and such other preferred stock shall in all cases bear to each other the same ratio that accrued dividends per share on the shares of this Series and such other preferred stock bear to each other. Holders of shares of this Series shall not be entitled to any dividend, whether payable in cash, property or stocks, in excess of full cumulative dividends, as herein provided, on this Series. No interest, or sum of money in lieu of interest, shall be payable in respect of any dividend payment or payments on this Series which may be in arrears.

- (4) So long as any shares of this Series are outstanding, no dividend (other than a dividend in Common Stock or in any other stock ranking junior to this Series as to dividends and upon liquidation and other than as provided in subsection (3) of this Section (b)) shall be declared or paid or set aside for payment or other distribution declared or made upon the Common Stock or upon any other stock ranking junior to or on a parity with this Series as to dividends or upon liquidation, nor shall any Common Stock nor any other stock of the Corporation ranking junior to or on a parity with this Series as to dividends or upon liquidation be redeemed, purchased or otherwise acquired for any consideration (or any moneys be paid to or made available for a sinking fund for the redemption of any shares of any such stock) by the Corporation (except by conversion into or exchange for stock of the Corporation ranking junior to this Series as to dividends and upon liquidation) unless, in each case, the full cumulative dividends on all outstanding shares of this Series shall have been paid for all past dividend payment periods.
 - (5) Dividends payable on this Series for any period shall be computed on the basis of a 360-day year consisting of twelve 30-day months.

(c) Redemption.

- (1) (A) The shares of this Series shall not be redeemable prior to April 15, 2006. On and after April 15, 2006, the Corporation, at its option, may redeem shares of this Series, in whole or in part, at any time or from time to time, at a redemption price of \$250 per share, plus accrued and unpaid dividends thereon to the date fixed for redemption.
- (B) In the event that fewer than all the outstanding shares of this Series are to be redeemed pursuant to subsection (1)(A), the number of shares to be redeemed shall be determined by the Board and the shares to be redeemed shall be determined by lot or pro rata as may be determined by the Board or by any other method as may be determined by the Board in its sole discretion to be equitable.
- (2) (A) Notwithstanding subsection (1) above, if the Dividends Received Percentage is equal to or less than 40% and, as a result, the amount of dividends on the shares of this Series payable on any Dividend Payment Date will be or is adjusted upwards as described in Section (b)(2) above, the Corporation, at its option, may redeem all, but not less than all, of the outstanding shares of this Series; provided, that within sixty days of the date on which an amendment to the Code is enacted which reduces the

Dividends Received Percentage to 40% or less, the Corporation sends notice to holders of shares of this Series of such redemption in accordance with subsection (3) below.

(B) Any redemption of the Perpetual Preferred Stock in accordance with this subsection (2) shall be at the applicable redemption price set forth in the following table, in each case plus accrued and unpaid dividends (whether or not declared) thereon to the date fixed for redemption, including any changes in dividends payable due to changes in the Dividends Received Percentage.

Redemption Period	Redemption Price Per Share	Redemption Price Per Depositary Share
		
April 2, 2004 to April 14, 2004	253.75	50.75
April 15, 2004 to April 14, 2005	252.50	50.50
April 15, 2005 to April 14, 2006	251.25	50.25
On or after April 15, 2006	250.00	50.00

- (3) In the event the Corporation shall redeem shares of this Series pursuant to subsections (1) or (2) above, notice of such redemption shall be given by first class mail, postage prepaid, mailed not less than 30 nor more than 60 days prior to the redemption date, to each holder of record of the shares to be redeemed, at such holder's address as the same appears on the stock register of the Corporation. Each such notice shall state: (i) the redemption date; (ii) the number of shares of this Series to be redeemed and, if fewer than all the shares held by such holder are to be redeemed, the number of such shares to be redeemed from such holder; (iii) the redemption price; (iv) the place or places where certificates for such shares are to be surrendered for payment of the redemption price; and (v) that dividends on the shares to be redeemed will cease to accrue on such redemption date.
- (4) Notice having been mailed as aforesaid, from and after the redemption date (unless default shall be made by the Corporation in providing money for the payment of the redemption price) dividends on the shares of this Series so called for redemption under either subsection (1) or (2) above shall cease to accrue, and said shares shall no longer be deemed to be outstanding, and all rights of the holders thereof as stockholders of the Corporation (except the right to receive from the Corporation the redemption price) shall cease. Upon surrender in accordance with said notice of the certificates for any shares so redeemed (properly endorsed or assigned for transfer, if the Board shall so require and the notice shall so state), such shares shall be redeemed by the Corporation at the applicable redemption price. In case fewer than all the shares represented by any such certificate are redeemed, a new certificate shall be issued representing the unredeemed shares without cost to the holder thereof.
- (5) Notwithstanding the foregoing provisions of this Section (c), if any dividends on this Series are in arrears, no shares of this Series shall be redeemed unless all outstanding shares of this Series are simultaneously redeemed, and the Corporation shall not purchase or otherwise acquire any shares of this Series; provided, however, that the foregoing shall not prevent the purchase or acquisition of shares of this Series pursuant to

a purchase or exchange offer made on the same terms to holders of all outstanding shares of this Series.

(d) Liquidation Rights.

- (1) Upon the dissolution, liquidation or winding up of the Corporation, the holders of the shares of this Series shall be entitled to receive and be paid out of the assets of the Corporation available for distribution to its stockholders, before any payment or distribution shall be made on the Common Stock or on any other class of stock ranking junior to the shares of this Series upon liquidation, the amount of \$250 per share, plus a sum equal to all dividends (whether or not earned or declared) on such shares accrued and unpaid thereon to the date of final distribution.
- (2) Neither the sale of all or substantially all the property or business of the Corporation nor the merger or consolidation of the Corporation into or with any other corporation or the merger or consolidation of any other corporation into or with the Corporation, shall be deemed to be a dissolution, liquidation or winding up, voluntary or involuntary, for the purposes of this Section (d).
- (3) After the payment to the holders of the shares of this Series of the full preferential amounts provided for in this Section (d), the holders of this Series as such shall have no right or claim to any of the remaining assets of the Corporation.
- (4) In the event the assets of the Corporation available for distribution to the holders of shares of this Series upon any dissolution, liquidation or winding up of the Corporation, whether voluntary or involuntary, shall be insufficient to pay in full all amounts to which such holders are entitled pursuant to paragraph (1) of this Section (d), no such distribution shall be made on account of any shares of any other class or series of Preferred Stock ranking on a parity with the shares of this Series upon such dissolution, liquidation or winding up unless proportionate distributive amounts shall be paid on account of the shares of this Series, ratably, in proportion to the full distributable amounts for which holders of all such parity shares are respectively entitled upon such dissolution, liquidation or winding up.
- (e) <u>Conversion or Exchange</u>. The holders of shares of this Series shall not have any rights herein to convert such shares into or exchange such shares for shares of any other class or classes or of any other series of any class or classes of capital stock of the Corporation.
 - (f) Voting. The shares of this Series shall not have any voting powers, either general or special, except that:
 - (1) Unless the vote or consent of the holders of a greater number of shares shall then be required by law, the consent of the holders of at least 66 2/3% of all of the shares of this Series at the time outstanding, given in person or by proxy, either in writing or by a vote at a meeting called for the purpose at which the holders of shares of this Series shall vote together as a separate class, shall be necessary for authorizing, effecting or validating the amendment, alteration or repeal of any of the provisions of the

Corporation's Amended and Restated Certificate of Incorporation or of any certificate amendatory thereof or supplemental thereto (including any Certificate of the Voting Powers, Designations, Preferences and Relative, Participating, Optional or Other Special Rights, and the Qualifications, Limitations or Restrictions thereof, or any similar document relating to any series of Preferred Stock) which would adversely affect the preferences, rights, powers or privileges of this Series;

- (2) Unless the vote or consent of the holders of a greater number of shares shall then be required by law, the consent of the holders of at least 66 2/3% of all of the shares of this Series and all other series of Preferred Stock ranking on a parity with shares of this Series, either as to dividends or upon liquidation, at the time outstanding, given in person or by proxy, either in writing or by a vote at a meeting called for the purpose at which the holders of shares of this Series and such other series of Preferred Stock shall vote together as a single class without regard to series, shall be necessary for authorizing, effecting, increasing or validating the creation, authorization or issue of any shares of any class of stock of the Corporation ranking prior to the shares of this Series as to dividends or upon liquidation, or the reclassification of any authorized stock of the Corporation into any such prior shares, or the creation, authorization or issue of any obligation or security convertible, into or evidencing the right to purchase any such prior shares.
- (3) If, at the time of any annual meeting of stockholders for the election of directors, a default in preference dividends on any series of the Preferred Stock or any other class or series of preferred stock of the Corporation (other than any other class or series of the Corporation's preferred stock expressly entitled to elect additional directors to the Board by a vote separate and distinct from the vote provided for in this paragraph (3) ("Voting Preferred")) shall exist, the number of directors constituting the Board shall be increased by two (without duplication of any increase made pursuant to the terms of any other class or series of the Corporation's preferred stock other than any Voting Preferred) and the holders of the Corporation's preferred stock of all classes and series (other than any such Voting Preferred) shall have the right at such meeting, voting together as a single class without regard to class or series, to the exclusion of the holders of Common Stock and the Voting Preferred, to elect two directors of the Corporation to fill such newly created directorships. Such right shall continue until there are no dividends in arrears upon shares of any class or series of the Corporation's preferred stock ranking prior to or on a parity with shares of this Series as to dividends (other than any Voting Preferred). Each director elected by the holders of shares of any series of the Preferred Stock or any other class or series of the Corporation's preferred stock in an election provided for by this paragraph (3) (herein called a "Preferred Director") shall continue to serve as such director for the full term for which he shall have been elected, notwithstanding that prior to the end of such term a default in preference dividends shall cease to exist. Any Preferred Director may be removed by, and shall not be removed except by, the vote of the holders of record of the outstanding shares of the Corporation's preferred stock entitled to have originally voted for such director's election, voting together a

exist (other than any Voting Preferred) (A) any vacancy in the office of a Preferred Director may be filled (except as provided in the following clause (B)) by an instrument in writing signed by the remaining Preferred Director and filed with the Corporation and (B) in the case of the removal of any Preferred Director, the vacancy may be filled by the vote of the holders of the outstanding shares of the Corporation's preferred stock entitled to have originally voted for the removed director's election, voting together as a single class without regard to class or series, at the same meeting at which such removal shall be voted. Each director appointed as aforesaid shall be deemed for all purposes hereto to be a Preferred Director.

Whenever the term of office of the Preferred Directors shall end and a default in preference dividends shall no longer exist, the number of directors constituting the Board shall be reduced by two. For purposes hereof, a "default in preference dividends" on any series of the Preferred Stock or any other class or series of preferred stock of the Corporation shall be deemed to have occurred whenever the amount of accrued dividends upon such class or series of the Corporation's preferred stock shall be equivalent to six full quarterly dividends or more, and, having so occurred, such default shall be deemed to exist thereafter until, but only until, all accrued dividends on all such shares of the Corporation's preferred stock of each and every series then outstanding (other than any Voting Preferred or shares of any class or series ranking junior to shares of this Series as to dividends) shall have been paid to the end of the last preceding quarterly dividend period.

- (4) Without limiting the foregoing, under any circumstances in which the Series would have additional rights under Rhode Island law if the Corporation were incorporated under the Rhode Island Business Corporation Act (rather than the Delaware General Corporation Law), holders of shares of the Series shall be entitled to such rights, including, without limitation, voting rights under Chapter 7-1.1-55, voting and notice rights under Chapter 7-1.1-67 and dissenters' rights under Chapters 7-1.1-73 and 7-1.1-74 of the Rhode Island Business Corporation Act (as such Chapters may be amended from time to time).
- (g) <u>Reacquired Shares</u>. Shares of this Series which have been issued and reacquired through redemption or purchase shall, upon compliance with an applicable provision of the Delaware General Corporation Law, have the status of authorized and unissued shares of Preferred Stock and may be reissued but only as part of a new series of Preferred Stock to be created by resolution or resolutions of the Board.
- (h) Relation to Existing Preferred Classes of Stock. Shares of this Series are equal in rank and preference with all other series of the Preferred Stock outstanding on the date of original issue of the shares of this Series and are senior in rank and preference to the Common Stock.
 - (i) Relation to Other Preferred Classes of Stock. For purposes of this resolution, any stock of any class or classes of the Corporation shall be deemed to rank:

- (1) prior to the shares of this Series, either as to dividends or upon liquidation, if the holders of such class or classes shall be entitled to the receipt of dividends or of amounts distributable upon dissolution, liquidation or winding up of the Corporation, as the case may be, in preference or priority to the holders of shares of this Series;
- (2) on a parity with shares of this Series, either as to dividends or upon liquidation, whether or not the dividend rates, dividend payment dates or redemption or liquidation prices per share or sinking fund provisions, if any, be different from those of this Series, if the holders of such stock shall be entitled to the receipt of dividends or of amounts distributable upon dissolution, liquidation or winding up of the Corporation, as the case may be, in proportion to their respective dividend rates or liquidation prices, without preference or priority, one over the other, as between the holders of such stock and the holders of shares of this Series; and
- (3) junior to the shares of this Series, either as to dividends or upon liquidation, if such class shall be Common Stock or if the holders of shares of this Series shall be entitled to receipt of dividends or of amounts distributable upon dissolution, liquidation or winding up of the Corporation, as the case may be, in preference or priority to the holders of shares of such class or classes.

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Designation to be duly executed by Kenneth D. Lewis, its Chairman, President and Chief Executive Officer, and attested to by Rachel R. Cummings, its Corporate Secretary, and has caused the corporate seal to be affixed hereto, this 28th day of January, 2004.

BANK OF AMERICA CORPORATION

By: /s/ Kenneth D. Lewis

Chairman, President and Chief Executive Officer

ATTEST:

/S/ RACHEL R. CUMMINGS

Corporate Secretary (Corporate Seal)

CERTIFICATE OF DESIGNATION

OF

FIXED/ADJUSTABLE RATE CUMULATIVE

PREFERRED STOCK

OF

BANK OF AMERICA CORPORATION

(Pursuant to Section 151 of the Delaware Corporation Law)

Bank of America Corporation, a corporation organized and existing under the General Corporation Law of the State of Delaware (the "Corporation"), hereby certifies that the following resolutions were adopted by the Board of Directors of the Corporation (the "Board of Directors") pursuant to the authority of the Board of Directors as required by Section 151 of the General Corporation Law of the State of Delaware, at a meeting duly convened and held on January 28, 2004:

RESOLVED, that pursuant to the authority granted to and vested in the Board of Directors in accordance with the provisions of the Amended and Restated Certificate of Incorporation of the Corporation, the Board of Directors hereby creates a series of the Corporation's previously authorized preferred stock, without par value (the "Preferred Stock") and hereby states the designation and number thereof and fixes the voting powers, preferences and relative, participating, optional and other special rights, and the qualifications, limitations and restrictions thereof, as follows:

(a) <u>Designation</u>. The designation of the series of Preferred Stock shall be "Fixed/Adjustable Rate Cumulative Preferred Stock" (hereinafter called this "Series") and the number of shares constituting this Series is Eight Hundred Five Thousand (805,000).

(b) Dividend Rate

(1) The holders of shares of this Series shall be entitled to receive dividends thereon at a rate of 6.60% per annum computed on the basis of an issue price thereof of \$250 per share, and no more, payable quarterly out of the funds of the Corporation legally available for the payment of dividends. Such dividends shall be cumulative from the date of original issue of such shares and shall be payable, when, as and if declared by the Board, on January 1, April 1, July 1 and October 1 of each year, commencing July 1, 2004 (a "Dividend Payment Date") through April 1, 2006. Each such dividend shall be paid to the holders of record of shares of this Series as they appear on the stock register of the Corporation on such record date, not exceeding 30 days preceding the payment date thereof, as shall be fixed by the Board. Dividends on account of arrears for any past quarters may be declared and paid at any time, without reference to any regular dividend

payment date, to holders of record on such date, not exceeding 45 days preceding the payment date thereof, as may be fixed by the Board.

After April 1, 2006, dividends on this Series will be payable quarterly, as, if and when declared by the Board of Directors or a duly authorized committee thereof on each Dividend Payment Date at the Applicable Rate from time to time in effect. The Applicable Rate per annum for any dividend period beginning on or after April 1, 2006 will be equal to .50% plus the highest of the Treasury Bill Rate, the Ten Year Constant Maturity Rate and the Thirty Year Constant Maturity Rate (each as defined below under "Adjustable Rate Dividends"), as determined in advance of such dividend period. The Applicable Rate per annum for any dividend period beginning on or after April 1, 2006 will not be less than 7.0% nor greater than 13.0% (without taking into account any adjustments as described below in subsection (3) of this Section (b)).

- (2) Except as provided below in this paragraph, the "Applicable Rate" per annum for any dividend period beginning on or after April 1, 2006 will be equal to .50% plus the Effective Rate (as defined below), but not less than 7.0% nor greater than 13.0% (without taking into account any adjustments as described below in subsection (3) of this Section (b)). The "Effective Rate" for any dividend period beginning on or after April 1, 2006 will be equal to the highest of the Treasury Bill Rate, the Ten Year Constant Maturity Rate and the Thirty Year Constant Maturity Rate (each as defined below) for such dividend period. In the event that the Corporation determines in good faith that for any reason:
 - (i) any one of the Treasury Bill Rate, the Ten Year Constant Maturity Rate or the Thirty Year Constant Maturity Rate cannot be determined for any dividend period, then the Effective Rate for such dividend period will be equal to the higher of whichever two of such rates can be so determined;
 - (ii) only one of the Treasury Bill Rate, the Ten Year Constant Maturity Rate or the Thirty Year Constant Maturity Rate can be determined for any dividend period, then the Effective Rate for such dividend period will be equal to whichever such rate can be so determined; or
 - (iii) none of the Treasury Bill Rate, the Ten Year Constant Maturity Rate or the Thirty Year Constant Maturity Rate can be determined for any dividend period, then the Effective Rate for the preceding dividend period will be continued for such dividend period.

Except as described below in this paragraph, the "Treasury Bill Rate" for each dividend period will be the arithmetic average of the two most recent weekly per annum market discount rates (or the one weekly per annum market discount rate, if only one such rate is published during the relevant Calendar Period (as defined below)) for three-month U.S. Treasury bills, as published weekly by the Federal Reserve Board (as defined below) during the Calendar Period immediately preceding the last ten calendar days preceding the dividend period for which the dividend rate on this Series is being determined. In the event that the Federal Reserve Board does not publish such a weekly

per annum market discount rate during any such Calendar Period, then the Treasury Bill Rate for such dividend period will be the arithmetic average of the two most recent weekly per annum market discount rates (or the one weekly per annum market discount rate, if only one such rate is published during the relevant Calendar Period) for three-month U.S. Treasury bills, as published weekly during such Calendar Period by any Federal Reserve Bank or by any U.S. Government department or agency selected by the Corporation. In the event that a per annum market discount rate for three-month U.S. Treasury bills is not published by the Federal Reserve Board or by any Federal Reserve Bank or by any U.S. Government department or agency during such Calendar Period, then the Treasury Bill Rate for such dividend period will be the arithmetic average of the two most recent weekly per annum market discount rates (or the one weekly per annum market discount rate, if only one such rate is published during the relevant Calendar Period) for all of the U.S. Treasury bills then having remaining maturities of not less than 80 nor more than 100 days, as published during such Calendar Period by the Federal Reserve Board or, if the Federal Reserve Board does not publish such rates, by any Federal Reserve Bank or by any U.S. Government Department or agency selected by the Corporation. In the event that the Corporation determines in good faith that for any reason no such U.S. Treasury Bill Rates are published as provided above during such Calendar Period, then the Treasury Bill Rate for such dividend period will be the arithmetic average of the per annum market discount rates based upon the closing bids during such Calendar Period for each of the issues of marketable non-interest-bearing U.S. Treasury securities with a remaining maturity of not less than 80 nor more than 100 days from the date of each such quotation, as chosen and quoted daily for each business day in New York City (or less frequently if daily quotations are not generally available) to the Corporation by at least three recognized dealers in U.S. Government securities selected by the Corporation. In the event that the Corporation determines in good faith that for any reason the Corporation cannot determine the Treasury Bill Rate for any dividend period as provided above in this paragraph, the Treasury Bill Rate for such dividend period will be the arithmetic average of the per annum market discount rates based upon the closing bids during such Calendar Period for each of the issues of marketable interest-bearing U.S. Treasury securities with a remaining maturity of not less than 80 nor more than 100 days, as chosen and quoted daily for each business day in New York City (or less frequently if daily quotations are not generally available) to the Corporation by at least three recognized dealers in U.S. Government securities selected by the Corporation.

Except as described below in this paragraph, the "Ten Year Constant Maturity Rate" for each dividend period will be the arithmetic average of the two most recent weekly per annum Ten Year Average Yields (as defined below) (or the one weekly per annum Ten Year Average Yield, if only one such yield is published during the relevant Calendar Period), as published weekly by the Federal Reserve Board during the Calendar Period immediately preceding the last ten calendar days preceding the dividend period for which the dividend rate on this Series is being determined. In the event that the Federal Reserve Board does not publish such a weekly per annum Ten Year Average Yield during such Calendar Period, then the Ten Year Constant Maturity Rate for such dividend period will be the arithmetic average of the two most recent weekly per annum Ten Year

Average Yields (or the one weekly per annum Ten Year Average Yield, if only such yield is published during the relevant Calendar Period), as published weekly during such Calendar Period by any Federal Reserve Bank or by any U.S. Government department or agency selected by the Corporation. In the event that a per annum Ten Year Average Yield is not published by the Federal Reserve Board or by any Federal Reserve Bank or by any U.S. Government department or agency during such Calendar Period, then the Ten Year Constant Maturity Rate for such dividend period will be the arithmetic average of the two most recent weekly per annum average yields to maturity (or the one weekly per annum average yield to maturity, if only one such yield is published during the relevant Calendar Period) for all of the actively traded marketable U.S. Treasury fixed interest rate securities (other than Special Securities (as defined below)) then having remaining maturities of not less than eight nor more than twelve years, as published during such Calendar Period by the Federal Reserve Board or, if the Federal Reserve Board does not publish such yields, by any Federal Reserve Bank or by any U.S. Government department or agency selected by the Corporation. In the event that the Corporation determines in good faith that for any reason the Corporation cannot determine the Ten Year Constant Maturity Rate for any dividend period as provided above in this paragraph, then the Ten Year Constant Maturity Rate for such dividend period will be the arithmetic average of the per annum average yields to maturity based upon the closing bids during such Calendar Period for each of the issues of actively traded marketable U.S. Treasury fixed interest rate securities (other than Special Securities) with a final maturity date not less than eight nor more than twelve years from the date of each such quotation, as chosen and quoted daily for each business day in New York City (or less frequently if daily quotations are not generally available) to the Corporat

Except as described below in this paragraph, the "Thirty Year Constant Maturity Rate" for each dividend period will be the arithmetic average of the two most recent weekly per annum Thirty Year Average Yields (as defined below) (or the one weekly per annum Thirty Year Yield, if only one such yield is published during the relevant Calendar Period), as published weekly by the Federal Reserve Board during the Calendar Period immediately preceding the last ten calendar days preceding the dividend period for which the dividend rate on this Series is being determined. In the event that the Federal Reserve Board does not publish such a weekly per annum Thirty Year Average Yield during such Calendar Period, then the Thirty Year Constant Maturity Rate for such dividend period will be the arithmetic average of the two most recent weekly per annum Thirty Year Average Yields (or the one weekly per annum Thirty Year Average Yield, if only one such yield is published during the relevant Calendar Period), as published weekly during such Calendar Period by any Federal Reserve Bank or by any U.S. Government department or agency during such Calendar Period, then the Thirty Year Constant Maturity Rate for such dividend period will be the arithmetic average of the two most recent weekly per annum average yields to maturity (or the one weekly per annum average yield to maturity, if only one such yield is

published during the relevant Calendar Period) for all of the actively traded marketable U.S. Treasury fixed interest rate securities (other than Special Securities) then having remaining maturities of not less than twenty-eight nor more than thirty years, as published during such Calendar Period by the Federal Reserve Board or, if the Federal Reserve Board does not publish such yields, by any Federal Reserve Bank or by any U.S. Government department or agency selected by the Corporation. In the event that the Corporation determines in good faith that for any reason the Corporation cannot determine the Thirty Year Constant Maturity Rate for any dividend period as provided above in this paragraph, then the Thirty Year Constant Maturity Rate for such dividend period will be the arithmetic average of the per annum average yields to maturity based upon the closing bids during such Calendar Period for each of the issues of actively traded marketable U.S. Treasury fixed interest rate securities (other than Special Securities) with a final maturity date not less than twenty-eight nor more than thirty years from the date of such quotation, as chosen and quoted daily for each business day in New York City (or less frequently if daily quotations are not generally available) to the Corporation by at least three recognized dealers in U.S. Government securities selected by the Corporation.

The Treasury Bill Rate, the Ten Year Constant Maturity Rate and the Thirty Year Constant Maturity Rate will each be rounded to the nearest five hundredths of a percent.

The Applicable Rate with respect to each dividend period beginning on or after April 1, 2006 will be calculated as promptly as practicable by the Corporation according to the appropriate method described above. The Corporation will cause notice of each Applicable Rate to be enclosed with the dividend payment checks next mailed to the holders of this Series.

As used above, the term "Calendar Period" means a period of fourteen calendar days; the term "Federal Reserve Board" means the Board of Governors of the Federal Reserve System; the term "Special Securities" means securities which can, at the option of the holder, be surrendered at face value in payment of any Federal estate tax or which provide tax benefits to the holder and are priced to reflect such tax benefits or which were originally issued at a deep or substantial discount; the term "Ten Year Average Yield" means the average yield to maturity for actively traded marketable U.S. Treasury fixed interest rate securities (adjusted to constant maturities of ten years); and the term "Thirty Year Average Yield" means the average yield to maturity for actively traded marketable U.S. Treasury fixed interest rate securities (adjusted to constant maturities of thirty years.)

(3) If one or more amendments to the Internal Revenue Code of 1986, as amended (the "Code"), are enacted that change the percentage of the dividends received deduction (currently 70%) as specified in Section 243(a)(1) of the Code or any successor provision (the "Dividends Received Percentage"), the amount of each dividend payable per share of this Series for dividend payments made on or after the date of enactment of such change shall be adjusted by multiplying the amount of the dividend payable determined as described above (before adjustment) by a factor which shall be the number

determined in accordance with the following formula (the "DRD Formula"), and rounding the result to the nearest cent:

For the purposes of the DRD Formula, "DRP" means the Dividends Received Percentage applicable to the dividend in question. No amendment to the Code, other than a change in the percentage of the dividends received deduction set forth in Section 243(a)(1) of the Code or any successor provision, will give rise to an adjustment. Notwithstanding the foregoing provisions, in the event that, with respect to any such amendment, the Corporation shall receive either an unqualified opinion of independent recognized tax counsel or a private letter ruling or similar form of authorization from the Internal Revenue Service to the effect that such an amendment would not apply to dividends payable on shares of this Series, then any such amendment shall not result in the adjustment provided for pursuant to the DRD Formula. The Corporation's calculation of the dividends payable as so adjusted and as certified accurate as to calculation and reasonable as to method by the independent certified public accountants then regularly engaged by the Corporation shall be final and not subject to review.

If any amendment to the Code which reduces the Dividends Received Percentage is enacted after a dividend payable on a Dividend Payment Date has been declared, the amount of dividend payable on such Dividend Payment Date will not be increased; but instead, an amount, equal to the excess of (x) the product of the dividends paid by the Corporation on such Dividend Payment Date and the DRD Formula (where the DRP used in the DRD Formula would be equal to the reduced Dividends Received Percentage) and (y) the dividends paid by the Corporation on such Dividend Payment Date, will be payable to holders of record on the next succeeding Dividend Payment Date in addition to any other amounts payable on such date.

(4) No full dividends shall be declared or paid or set apart for payment on the Preferred Stock of any series ranking, as to dividends, on a parity with or junior to this Series for any period unless full cumulative dividends have been or contemporaneously are declared and paid or declared and a sum sufficient for the payment thereof set apart for such payment on this Series for all dividend payment periods terminating on or prior to the date of payment of such full cumulative dividends. When dividends are not paid in full, as aforesaid, upon the shares of this Series and any other preferred stock ranking on a parity as to dividends with this Series, all dividends declared upon shares of this Series and any other class or series of preferred stock of the Corporation ranking on a parity as to dividends with this Series shall be declared pro rata so that the amount of dividends declared per share on this Series and such other preferred stock shall in all cases bear to each other the same ratio that accrued dividends per share on the shares of this Series and such other preferred stock bear to each other. Holders of shares of this Series shall not be entitled to any dividend, whether payable in cash, property or stocks, in excess of full cumulative dividends, as herein provided, on this Series. No interest, or sum of money in

lieu of interest, shall be payable in respect of any dividend payment or payments on this Series which may be in arrears.

- (5) So long as any shares of this Series are outstanding, no dividend (other than a dividend in Common Stock or in any other stock ranking junior to this Series as to dividends and upon liquidation and other than as provided in subsection (4) of this Section (b)) shall be declared or paid or set aside for payment or other distribution declared or made upon the Common Stock or upon any other stock ranking junior to or on a parity with this Series as to dividends or upon liquidation, nor shall any Common Stock nor any other stock of the Corporation ranking junior to or on a parity with this Series as to dividends or upon liquidation be redeemed, purchased or otherwise acquired for any consideration (or any moneys be paid to or made available for a sinking fund for the redemption of any shares of any such stock) by the Corporation (except by conversion into or exchange for stock of the Corporation ranking junior to this Series as to dividends and upon liquidation) unless, in each case, the full cumulative dividends on all outstanding shares of this Series shall have been paid for all past dividend payment periods.
 - (6) Dividends payable on this Series for any period shall be computed on the basis of a 360-day year consisting of twelve 30-day months.

(c) <u>Redemption</u>.

- (1) (A) The shares of this Series shall not be redeemable prior to April 1, 2006. On and after April 1, 2006, the Corporation, at its option, may redeem shares of this Series, in whole or in part, at any time or from time to time, at a redemption price of \$250 per share, plus accrued and unpaid dividends thereon to the date fixed for redemption.
- (B) In the event that fewer than all the outstanding shares of this Series are to be redeemed pursuant to subsection (1)(A), the number of shares to be redeemed shall be determined by the Board and the shares to be redeemed shall be determined by lot or pro rata as may be determined by the Board or by any other method as may be determined by the Board in its sole discretion to be equitable.
- (2) (A) Notwithstanding subsection (1) above, if the Dividends Received Percentage is equal to or less than 40% and, as a result, the amount of dividends on the shares of this Series payable on any Dividend Payment Date will be or is adjusted upwards as described in Section (b)(2) above, the Corporation, at its Option, may redeem all, but not less than all, of the outstanding shares of this Series; provided, that within sixty days of the date on which an amendment to the Code is enacted which reduces the Dividends Received Percentage to 40% or less, the Corporation sends notice to holders of shares of this Series of such redemption in accordance with subsection (3) below.
- (B) Any redemption of this Series in accordance with this subsection (2) shall be at the applicable redemption price set forth in the following table, in each case plus accrued and unpaid dividends (whether or not declared) thereon to the date fixed for

redemption, including any changes in dividends payable due to changes in the Dividends Received Percentage.

Redemption Period	Redemption Price Per Share	Redemption Price Per Depositary Share
April 2, 2004 to March 31, 2005	252.50	50.50
April 1, 2005 to March 31, 2006	251.25	50.25
On or after April 1, 2006	250.00	50.00

- (3) In the event the Corporation shall redeem shares of this Series pursuant to subsections (1) or (2) above, notice of such redemption shall be given by first class mail, postage prepaid, mailed not less than 30 nor more than 60 days prior to the redemption date, to each holder of record of the shares to be redeemed, at such holder's address as the same appears on the stock register of the Corporation. Each such notice shall state: (i) the redemption date; (ii) the number of shares of this Series to be redeemed and, if fewer than all the shares held by such holder are to be redeemed, the number of such shares to be redeemed from such holder; (iii) the redemption price; (iv) the place or places where certificates for such shares are to be surrendered for payment of the redemption price; and (v) that dividends on the shares to be redeemed will cease to accrue on such redemption date.
- (4) Notice having been mailed as aforesaid, from and after the redemption date (unless default shall be made by the Corporation in providing money for the payment of the redemption price) dividends on the shares of this Series so called for redemption under either subsection (1) or (2) above shall cease to accrue, and said shares shall no longer be deemed to be outstanding, and all rights of the holders thereof as stockholders of the Corporation (except the right to receive from the Corporation the redemption price) shall cease. Upon surrender in accordance with said notice of the certificates for any shares so redeemed (properly endorsed or assigned for transfer, if the Board shall so require and the notice shall so state), such shares shall be redeemed by the Corporation at the applicable redemption price. In case fewer than all the shares represented by any such certificate are redeemed, a new certificate shall be issued representing the unredeemed shares without cost to the holder thereof.
- (5) Notwithstanding the foregoing provisions of this Section (c), if any dividends on this Series are in arrears, no shares of this Series shall be redeemed unless all outstanding shares of this Series are simultaneously redeemed, and the Corporation shall not purchase or otherwise acquire any shares of this Series; provided, however, that the foregoing shall not prevent the purchase or acquisition of shares of this Series pursuant to a purchase or exchange offer made on the same terms to holders of all outstanding shares of this Series.

(d) Liquidation Rights.

(1) Upon the dissolution, liquidation or winding up of the Corporation, the holders of the shares of this Series shall be entitled to receive and be paid out of the assets of the Corporation available for distribution to its stockholders, before any payment or

distribution shall be made on the Common Stock or on any other class of stock ranking junior to the shares of this Series upon liquidation, the amount of \$250 per share, plus a sum equal to all dividends (whether or not earned or declared) on such shares accrued and unpaid thereon to the date of final distribution.

- (2) Neither the sale of all or substantially all the property or business of the Corporation nor the merger or consolidation of the Corporation into or with any other corporation or the merger or consolidation of any other corporation into or with the Corporation, shall be deemed to be a dissolution, liquidation or winding up, voluntary or involuntary, for the purposes of this Section (d).
- (3) After the payment to the holders of the shares of this Series of the full preferential amounts provided for in this Section (d), the holders of this Series as such shall have no right or claim to any of the remaining assets of the Corporation.
- (4) In the event the assets of the Corporation available for distribution to the holders of shares of this Series upon any dissolution, liquidation or winding up of the Corporation, whether voluntary or involuntary, shall be insufficient to pay in full all amounts to which such holders are entitled pursuant to paragraph (1) of this Section (d), no such distribution shall be made on account of any shares of any other class or series of Preferred Stock ranking on a parity with the shares of this Series upon such dissolution, liquidation or winding up unless proportionate distributive amounts shall be paid on account of the shares of this Series, ratably, in proportion to the full distributable amounts for which holders of all such parity shares are respectively entitled upon such dissolution, liquidation or winding up.
- (e) <u>Conversion or Exchange</u>. The holders of shares of this Series shall not have any rights herein to convert such shares into or exchange such shares for shares of any other class or classes or of any other series of any class or classes of capital stock of the Corporation.
 - (f) Voting. The shares of this Series shall not have any voting powers, either general or special, except that:
 - (1) Unless the vote or consent of the holders of a greater number of shares shall then be required by law, the consent of the holders of at least 66 2/3% of all of the shares of this Series at the time outstanding, given in person or by proxy, either in writing or by a vote at a meeting called for the purpose at which the holders of shares of this Series shall vote together as a separate class, shall be necessary for authorizing, effecting or validating the amendment, alteration or repeal of any of the provisions of the Corporation's Amended and Restated Certificate of Incorporation or of any certificate amendatory thereof or supplemental thereto (including any Certificate of the Voting Powers, Designations, Preferences and Relative, Participating, Optional or Other Special Rights, and the Qualifications, Limitations or Restrictions thereof, or any similar document relating to any series of Preferred Stock) which would adversely affect the preferences, rights, powers or privileges of this Series;

(2) Unless the vote or consent of the holders of a greater number of shares shall then be required by law, the consent of the holders of at least 66 2/3% of all of the shares of this Series and all other series of Preferred Stock ranking on a parity with shares of this Series, either as to dividends or upon liquidation, at the time outstanding, given in person or by proxy, either in writing or by a vote at a meeting called for the purpose at which the holders of shares of this Series and such other series of Preferred Stock shall vote together as a single class without regard to series, shall be necessary for authorizing, effecting, increasing or validating the creation, authorization or issue of any shares of any class of stock of the Corporation ranking prior to the shares of this Series as to dividends or upon liquidation, or the reclassification of any authorized stock of the Corporation into any such prior shares, or the creation, authorization or issue of any obligation or security convertible into or evidencing the right to purchase any such prior shares.

(3) If, at the time of any annual meeting of stockholders for the election of directors, a default in preference dividends on any series of the Preferred Stock or any other class or series of preferred stock of the Corporation (other than any other class or series of the Corporation's preferred stock expressly entitled to elect additional directors to the Board by a vote separate and distinct from the vote provided for in this paragraph (3) ("Voting Preferred")) shall exist, the number of directors constituting the Board shall be increased by two (without duplication of any increase made pursuant to the terms of any other class or series of the Corporation's preferred stock other than any Voting Preferred) and the holders of the Corporation's preferred stock of all classes and series (other than any such Voting Preferred) shall have the right at such meeting, voting together as a single class without regard to class or series, to the exclusion of the holders of Common Stock and the Voting Preferred, to elect two directors of the Corporation to fill such newly created directorships. Such right shall continue until there are no dividends in arrears upon shares of any class or series of the Corporation's preferred stock ranking prior to or on a parity with shares of this Series as to dividends (other than any Voting Preferred). Each director elected by the holders of shares of any series of the Preferred Stock or any other class or series of the Corporation's preferred stock in an election provided for by this paragraph (3) (herein called a "Preferred Director") shall continue to serve as such director for the full term for which he shall have been elected, notwithstanding that prior to the end of such term a default in preference dividends shall cease to exist. Any Preferred Director may be removed by, and shall not be removed except by, the vote of the holders of record of the outstanding shares of the Corporation's preferred stock entitled to have originally voted for such director's election, voting together as a single class without regard to class or series, at a meeting of the stockholders, or of the holders of shares of the Corporation's preferred stock, called for that purpose. So long as a default in any preference dividends on any series of the Preferred Stock or any other class or series of preferred stock of the Corporation shall exist (other than any Voting Preferred) (A) any vacancy in the office of a Preferred Director may be filled (except as provided in the following clause (B)) by an instrument in writing signed by the remaining Preferred Director and filed with the Corporation and (B) in the case of the removal of any Preferred Director, the vacancy may be filled by the vote of the holders of the outstanding shares of the Corporation's preferred stock entitled to have originally voted for the removed director's election, voting together as a single

class without regard to class or series, at the same meeting at which such removal shall be voted. Each director appointed as aforesaid shall be deemed for all purposes hereto to be a Preferred Director.

Whenever the term of office of the Preferred Directors shall end and a default in preference dividends shall no longer exist, the number of directors constituting the Board shall be reduced by two. For purposes hereof, a "default in preference dividends" on any series of the Preferred Stock or any other class or series of preferred stock of the Corporation shall be deemed to have occurred whenever the amount of accrued dividends upon such class or series of the Corporation's preferred stock shall be equivalent to six full quarterly dividends or more, and, having so occurred, such default shall be deemed to exist thereafter until, but only until, all accrued dividends on all such shares of the Corporation's preferred stock of each and every series then Outstanding (other than any Voting Preferred or shares of any class or series ranking junior to shares of this Series as to dividends) shall have been paid to the end of the last preceding quarterly dividend period.

- (4) Without limiting the foregoing, under any circumstances in which the Series would have additional rights under Rhode Island law if the Corporation were incorporated under the Rhode Island Business Corporation Act (rather than the Delaware General Corporation Law), holders of shares of the Series shall be entitled to such rights, including, without limitation, voting rights under Chapter 7-1.1-55, voting and notice rights under Chapter 7-1.1-67 and dissenters' rights under Chapters 7-1.1-73 and 7-1.1-74 of the Rhode Island Business Corporation Act (as such Chapters may be amended from time to time).
- (g) <u>Reacquired Shares</u>. Shares of this Series which have been issued and reacquired through redemption or purchase shall, upon compliance with an applicable provision of the Delaware General Corporation Law, have the status of authorized and unissued shares of Preferred Stock and may be reissued but only as part of a new series of Preferred Stock to be created by resolution or resolutions of the Board.
- (h) <u>Relation to Existing Preferred Classes of Stock</u>. Shares of this Series are equal in rank and preference with all other series of the Preferred Stock outstanding on the date of original issue of the shares of this Series and are senior in rank and preference to the Common Stock of the Corporation.
 - (i) Relation to Other Preferred Classes of Stock. For purposes of this resolution, any stock of any class or classes of the Corporation shall be deemed to rank:
 - (1) prior to the shares of this Series, either as to dividends or upon liquidation, if the holders of such class or classes shall be entitled to the receipt of dividends or of amounts distributable upon dissolution, liquidation or winding up of the Corporation, as the case may be, in preference or priority to the holders of shares of this Series;
 - (2) on a parity with shares of this Series, either as to dividends or upon liquidation, whether or not the dividend rates, dividend payment dates or redemption or

liquidation prices per share or sinking fund provisions, if any, be different from those of this Series, if the holders of such stock shall be entitled to the receipt of dividends or of amounts distributable upon dissolution, liquidation or winding up of the Corporation, as the case may be, in proportion to their respective dividend rates or liquidation prices, without preference or priority, one over the other, as between the holders of such stock and the holders of shares of this Series; and

(3) junior to the shares of this Series, either as to dividends or upon liquidation, if such class shall be Common Stock or if the holders of shares of this Series shall be entitled to receipt of dividends or of amounts distributable upon dissolution, liquidation or winding up of the Corporation, as the case may be, in preference or priority to the holders of shares of such class or classes.

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Designation to be duly executed by Kenneth D. Lewis, its President, Chief Executive Officer
and Chairman of the Board, and attested to by Rachel R. Cummings, its Corporate Secretary, and has caused the corporate seal to be affixed hereto, this 28th day of January,
2004

BANK OF AMERICA CORPORATION

By: /s/ Kenneth D. Lewis

Chairman, President and Chief Executive Officer

ATTEST:

/S/ RACHEL R. CUMMINGS

Corporate Secretary

(Corporate Seal)

January 30, 2004

Bank of America Corporation Bank of America Corporate Center 100 North Tryon Street Charlotte, North Carolina 28255

Re: Registration Statement on Form S-4

Ladies and Gentlemen:

I and other members of my staff have acted as counsel to Bank of America Corporation, a Delaware corporation (the "Company"), in connection with the preparation and filing of a Registration Statement on Form S-4 (the "Registration Statement") relating to the issuance of up to 650,974,529 shares of the Company's common stock, 1,912,250 depositary shares, each representing a one-fifth interest in a share of 6.75% Perpetual Preferred Stock, and 3,500,000 depositary shares, each representing a one-fifth interest in a share of Fixed/Adjustable Rate Cumulative Preferred Stock (collectively, the "Shares"). The Shares are to be issued by the Company in connection with the merger of FleetBoston Financial Corporation with and into the Company (the "Merger").

In rendering this opinion, I have examined such corporate records and other documents, and I have reviewed such matters of law, as I have deemed necessary or appropriate. Based on the foregoing, I am of the opinion that all necessary corporate action on the part of the Company has been taken to authorize the issuance of the Shares in connection with the Merger, and when the Registration Statement has been declared effective by order of the Securities and Exchange Commission and the Shares have been issued in accordance with the terms and conditions set forth in the Registration Statement, the Shares will be legally and validly issued, fully paid and nonassessable.

I hereby consent to be named in the Registration Statement and in the related joint proxy statement-prospectus contained therein as the attorney who passed upon the legality of the Shares, and to the filing of a copy of this opinion as Exhibit 5(a) to the Registration Statement.

Very truly yours,

/s/ Paul J. Polking
Paul J. Polking
Executive Vice President and General Counsel
Bank of America Corporation

[LETTERHEAD OF CLEARY, GOTTLIEB, STEEN & HAMILTON]

January 30, 2004

Bank of America Corporation 100 N. Tryon St. Charlotte, North Carolina 28255

Ladies and Gentlemen:

We have acted as special counsel to Bank of America Corporation, a Delaware corporation ("Bank of America"), in connection with the proposed merger (the "Merger") of FleetBoston Financial Corporation, a Rhode Island corporation ("FleetBoston") with and into Bank of America, pursuant to the Agreement and Plan of Merger (the "Agreement"), dated as of October 27, 2003, by and between Bank of America and FleetBoston. At your request, and in connection with the Registration Statement on Form S-4 filed with the Securities and Exchange Commission in connection with the Merger (as amended through the date hereof, the "Registration Statement"), we are rendering our opinion concerning the material federal income tax consequences of the Merger. Any capitalized term used and not defined herein has the meaning given to it in the Agreement.

For purposes of the opinion set forth below, we have relied, with the consent of Bank of America and the consent of FleetBoston, upon the accuracy and completeness of the factual statements and representations (which statements and representations we have neither investigated nor verified) contained, respectively, in the certificates of the officers of Bank of America and FleetBoston dated the date hereof, and have assumed that such factual statements and representations will be accurate and complete as of the Effective Time (as if made as of such time) and that all such factual statements and representations made to the knowledge of any person or entity or with similar qualification are and will be true and correct as if made without such qualification. We have also relied upon the accuracy of the Registration Statement and the joint proxy statement/prospectus (the "Proxy Statement/Prospectus") contained therein, each as amended or supplemented through the date hereof.

We have also assumed that (i) the transactions contemplated by the Agreement will be consummated in accordance therewith and as described in the Proxy Statement/Prospectus (and no transaction or condition described therein and affecting this opinion will be waived by any party) and (ii) the Merger will be reported by Bank of America and FleetBoston on their respective federal income tax returns in a manner consistent with the opinion set forth below.

Bank of America Corporation Page 2

Based upon and subject to the foregoing, it is our opinion, under currently applicable U.S. federal income tax law, that

- (i) the Merger will constitute a reorganization within the meaning of Section 368(a) of the Code;
- (ii) no gain or loss will be recognized by FleetBoston stockholders who receive shares of Bank of America stock in exchange for shares of FleetBoston Stock, except with respect to cash received instead of fractional share interests in Bank of America common stock;
- (iii) the aggregate basis of the Bank of America stock received in the Merger will be the same as the aggregate basis of the FleetBoston stock for which it is exchanged, less any basis attributable to fractional share interests in Bank of America common stock for which cash is received;
- (iv) the holding period of Bank of America stock received in exchange for shares of FleetBoston stock will include the holding period of the FleetBoston stock for which it is exchanged; and
 - (v) no gain or loss will be recognized by Bank of America or FleetBoston in the Merger.

In addition, we hereby confirm our opinion set forth under the caption "MATERIAL FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER" in the Proxy Statement/Prospectus, subject to the limitations and qualifications stated therein.

We express no opinion on any issue relating to the tax consequences of the Merger other than those set forth above. Our opinion is based upon the Code, published judicial decisions, administrative regulations and published rulings and procedures as in existence on the date hereof. Future legislative, judicial or administrative changes, on either a prospective or retroactive basis, could affect our opinion. Further, our opinion is not binding upon the Internal Revenue Service or the courts, and there is no assurance that the Internal Revenue Service or a court will not take a contrary position. We undertake no responsibility to advise you of any future change in the matters stated herein or in the federal income tax laws or the application or interpretation thereof.

Furthermore, this opinion only addresses the tax treatment of only those FleetBoston stockholders that hold their FleetBoston stock as a capital asset within the meaning of Section 1221 of the Code, and does not address all the U.S. federal income tax consequences that may be relevant to particular FleetBoston stockholders in light of their individual circumstances or to FleetBoston stockholders that are subject to special rules, such as financial institutions, insurance companies, tax-exempt organizations, dealers in securities or currencies, traders in securities that elect to use a mark to market method of accounting, persons that hold FleetBoston stock as part of a straddle, hedge, constructive sale or conversion transaction, persons who are not citizens or residents of the United States, and stockholders who acquired

Bank of America Corporation Page 3

their shares of FleetBoston stock through the exercise of an employee stock option or otherwise as compensation.

We hereby consent to the filing of this opinion with the Securities and Exchange Commission as an exhibit to the Registration Statement, and to the references therein to us. In giving such consent, we do not thereby admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended.

We are furnishing this opinion solely in connection with the filing of the Registration Statement and this opinion is not to be relied upon for any other purpose.

Very truly yours,

CLEARY, GOTTLIEB, STEEN & HAMILTON

By: /S/ LESLIE B. SAMUELS

Leslie B. Samuels, a partner

[LETTERHEAD OF WACHTELL, LIPTON, ROSEN & KATZ]

January 30, 2004

FleetBoston Financial Corporation 100 Federal Street Boston, Massachusetts 02110

Ladies and Gentlemen:

We have acted as special counsel to FleetBoston Financial Corporation, a Rhode Island corporation ('FleetBoston''), in connection with the proposed merger (the "Merger") of FleetBoston with and into Bank of America Corporation, a Delaware corporation ('Bank of America''), pursuant to the Agreement and Plan of Merger (the "Agreement"), dated as of October 27, 2003, by and between FleetBoston and Bank of America. At your request, and in connection with the Registration Statement on Form S-4 filed with the Securities and Exchange Commission in connection with the Merger (as amended through the date hereof, the "Registration Statement"), we are rendering our opinion concerning the material federal income tax consequences of the Merger. Any capitalized term used and not defined herein has the meaning given to it in the Agreement.

For purposes of the opinion set forth below, we have relied, with the consent of FleetBoston and the consent of Bank of America, upon the accuracy and completeness of the factual statements and representations (which statements and representations we have neither investigated nor verified) contained, respectively, in the certificates of the officers of FleetBoston and Bank of America dated the date hereof, and have assumed that such factual statements and representations will be accurate and complete as of the Effective Time (as if made as of such time) and that all such factual statements and representations made to the knowledge of any person or entity or with similar qualification are and will be true and correct as if made without such qualification. We have also relied upon the accuracy of the Registration Statement and the joint proxy statement/prospectus (the "Proxy Statement/Prospectus") contained therein, each as amended or supplemented through the date hereof.

We have also assumed that (i) the transactions contemplated by the Agreement will be consummated in accordance therewith and as described in the Proxy Statement/Prospectus (and no transaction or condition described therein and affecting this opinion will be waived by any party) and (ii) the Merger will be reported by FleetBoston and Bank of America on their respective federal income tax returns in a manner consistent with the opinion set forth below.

Based upon and subject to the foregoing, it is our opinion, under currently applicable United States federal income tax law, that (i) the Merger will qualify as a "reorganization" within the meaning of Section 368(a) of the Code, (ii) no gain or loss will be recognized by FleetBoston stockholders who receive shares of Bank of America stock in exchange for shares of FleetBoston stock, except with respect to cash received instead of fractional share interests in Bank of America common stock, (iii) the aggregate basis of the Bank of America stock received in the Merger will be the same as the aggregate basis of the FleetBoston stock for which it is exchanged, less any basis attributable to fractional share interests in Bank of America common stock for which cash is received, (iv) the holding period of Bank of America stock received in exchange for shares of FleetBoston stock will include the holding period of the FleetBoston stock for which it is exchanged and (v) no gain or loss will be recognized by Bank of America or FleetBoston in the Merger. In addition, we hereby confirm our opinion set forth under the caption "MATERIAL FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER" in the Proxy Statement/Prospectus, subject to the limitations and qualifications stated therein.

We express no opinion on any issue relating to the tax consequences of the Merger other than those set forth above. Our opinion is based upon the Code, published judicial decisions, administrative regulations and published rulings and procedures as in existence on the date hereof. Future legislative, judicial or administrative changes, on either a prospective or retroactive basis, could affect our opinion. Further, our opinion is not binding upon the Internal Revenue Service or the courts, and there is no assurance that the Internal Revenue Service or a court will not take a contrary position. We undertake no responsibility to advise you of any future change in the matters stated herein or in the federal income tax laws or the application or interpretation thereof.

Furthermore, this opinion only addresses the tax treatment of only those FleetBoston stockholders that hold their FleetBoston stock as a capital asset within the meaning of Section 1221 of the Code, and does not address all the U.S. federal income tax consequences that may be relevant to particular FleetBoston stockholders in light of their individual circumstances or to FleetBoston stockholders that are subject to special rules, such as financial institutions, insurance companies, tax-exempt organizations, dealers in securities or currencies, traders in securities that elect to use a mark to market method of accounting, persons that hold FleetBoston stock as part of a straddle, hedge, constructive sale or conversion transaction, persons who are not citizens or residents of the United States, and stockholders who acquired their shares of FleetBoston stock through the exercise of an employee stock option or otherwise as compensation.

We hereby consent to the filing of this opinion with the Securities and Exchange Commission as an exhibit to the Registration Statement, and to the references therein to us. In giving such consent, we do not thereby admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended.

We are furnishing this opinion solely in connection with the filing of the Registration Statement and this opinion is not to be relied upon for any other purpose.

Very truly yours,

Wachtell, Lipton, Rosen & Katz

CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the incorporation by reference in this Registration Statement on Amendment No. 2 to Form S-4 of Bank of America Corporation of our report dated January 15, 2003 relating to the financial statements of Bank of America Corporation, which appears in Bank of America Corporation's 2002 Annual Report to Shareholders, which is incorporated by reference in its Annual Report on Form 10-K for the year ended December 31, 2002. We also consent to the reference to us under the heading "Experts" in such Registration Statement.

PricewaterhouseCoopers LLP

Charlotte, North Carolina January 30, 2004

CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the incorporation by reference in this Registration Statement on Amendment No. 2 to Form S-4 of Bank of America Corporation of our report dated January 16, 2003 relating to the financial statements of FleetBoston Financial Corporation, which appears in FleetBoston Financial Corporation's Annual Report on Form 10-K for the year ended December 31, 2002. We also consent to the reference to us under the heading "Experts" in such Registration Statement.

PricewaterhouseCoopers LLP

Boston, Massachusetts January 30, 2004 [Goldman, Sachs & Co. Letterhead]

PERSONAL AND CONFIDENTIAL

January 30, 2004

Board of Directors Bank of America Corporation Bank of America Corporate Center 100 North Tryon Street Charlotte, North Carolina 28255

Re: Amendment No. 2 to Registration Statement on Form S-4 of Bank of America Corporation (File No. 333-110924)

Ladies and Gentlemen:

Reference is made to our opinion letter, dated October 27, 2003, with respect to the fairness from a financial point of view to Bank of America Corporation (the "Company") of the exchange ratio of 0.5553 of a share of common stock, par value \$0.01 per share, of the Company to be issued in respect of each share of common stock, par value \$0.01 per share, of FleetBoston Financial Corporation ("FleetBoston") pursuant to the Agreement and Plan of Merger, dated as of October 27, 2003, between the Company and FleetBoston.

The foregoing opinion letter is provided for the information and assistance of the Board of Directors of the Company in connection with its consideration of the transaction contemplated therein and is not to be used, circulated, quoted or otherwise referred to for any other purpose, nor is it to be filed with, included in or referred to in whole or in part in any registration statement, proxy statement or any other document, except in accordance with our prior written consent. We understand that the Company has determined to include our opinion in the above-referenced Registration Statement, as amended.

Board of Directors Bank of America Corporation January 30, 2004 Page Two

In that regard, we hereby consent to the reference to our opinion under the captions "Summary," "The Merger—Background of the Merger," "The Merger—Bank of America's Reasons for the Merger; Recommendation of the Merger by the Bank of America Board of Directors," "The Merger—Analyses of Financial Advisors" and "The Merger—Opinion of Goldman Sachs" and to the inclusion of the foregoing opinion in the Joint Proxy Statement/Prospectus included in the above-mentioned Registration Statement, as amended. Notwithstanding the foregoing, it is understood that our consent is being delivered solely in connection with the filing of the above-mentioned version of the Registration Statement and that our opinion is not to be used, circulated, quoted or otherwise referred to for any other purpose, nor is it to be filed with, included in or referred to in whole or in part in any registration statement (including any subsequent amendments to the above-mentioned Registration Statement), proxy statement or any other document, except in accordance with our prior written consent. In giving such consent, we do not thereby admit that we come within the category of persons whose consent is required under Section 7 of the Securities Act of 1933 or the rules and regulations of the Securities and Exchange Commission thereunder.

Very truly yours,

s/ Goldman, Sachs & Co.

(GOLDMAN, SACHS & CO.)

Consent of Morgan Stanley & Co. Incorporated

We hereby consent to the use in the Amendment No. 2 to the Registration Statement of Bank of America Corporation on Form S-4 and in the Joint Proxy Statement/Prospectus of Bank of America Corporation and FleetBoston Financial Corporation, which is part of the Registration Statement, of our opinion dated October 26, 2003 appearing as Appendix E to such Joint Proxy Statement/Prospectus, and to the description of such opinion and to the references to our name contained therein under the headings "Summary—Morgan Stanley Provided an Opinion to the FleetBoston Board of Directors that the Exchange Ratio was Fair from a Financial Point of View to Holders of FleetBoston Common Stock," "The Merger—Background of the Merger," "The Merger—FleetBoston's Reasons for the Merger; Recommendation of the Merger by the FleetBoston Board of Directors," and "The Merger—Opinion of Morgan Stanley." In giving the foregoing consent, we do not admit that we come within the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended (the "Securities Act"), or the rules and regulations promulgated thereunder, nor do we admit that we are experts with respect to any part of such Registration Statement within the meaning of the term "experts" as used in the Securities Act or the rules and regulations promulgated thereunder.

MORGAN STANLEY	& CO. INCORPORATED
By:	/s/ Kirk R. Wilson

New York, New York January 30, 2004

								here if chang	e mark f address es are below.	
		The Boa	rd of Directors	recommend	ls a vote "	FOR" Items 1, 2, 3	and 4.			
tem 1 —	ADOPTION OF THE MERGER AGREEMENT	FOR	AGAINST	ABSTAIN	Item 4 –	BANK OF AM	OF PROPOSAL TO ADJOURN THE SERVICE SPECIAL MEETING, IF	FOR	AGAINST	ABSTAIN
Item 2 —	ADOPTION OF BANK OF AMERICA'S AMENDED STOCK PLAN					PROXIES	, TO SOLICIT ADDITIONAL			
tem 3 —	ADOPTION OF AN AMENDMENT TO INCREASE THE NUMBER OF AUTHORIZED SHARES OF BANK OF AMERICA COMMON STOCK									
						bring photo identifi	pecial Meeting. If "YES," please ication for admission and, if your street name, please bring proof		YES	NO
						or ourrorner.				CONSENT
						statements and no America Corporati Corporation may n for any future stock	Internet access of the annual rep ptices of stockholder meetings of I ion. I understand that Bank of Am no longer distribute printed materia kholder meeting until such conser and further that I may revoke my	Bank of erica als to me nt is		

Exhibit 99(g)

Date:

NOTE: Please sign your name as it appears above. When signing as attorney, executor, administrator, trustee, guardian or officer of a company or other entity, please provide your full title.

↑ FOLD AND DETACH HERE

PLEASE READ THE VOTING INSTRUCTIONS BELOW.

Bank of America Corporation encourages you to vote your shares. Please take the opportunity to use one of the three voting methods outlined below to cast your ballot.

VOTE BY INTERNET – http://www.eproxy.com/bac/

Use the Internet to vote your proxy 24 hours a day, 7 days a week. Have your proxy card in hand when you access the site. You will be prompted to enter your 11-digit Control Number, which is located below, to obtain your records and create an electronic ballot.

VOTE BY TELEPHONE – 1.800.435.6710

Use any touch-tone telephone to vote your proxy 24 hours a day, 7 days a week. Have your proxy card in hand when you call. You will be prompted to enter your 11-digit Control Number, which is located below, and then follow the simple instructions provided.

VOTE BY MAIL

Signature(s)

Mark, sign and date your proxy card and return it promptly in the enclosed, postage paid envelope.

Proxies submitted by Internet or telephone must be received by 11:59 p.m., Eastern time, on March 16, 2004. If you vote by Internet or telephone, please do not return your proxy card.

THANK YOU FOR VOTING.



This Proxy is Solicited on behalf of the Board of Directors Special Meeting of Stockholders to be held on March 17, 2004

You, the undersigned stockholder, appoint each of Katherine L. McLeod, Richard J. Parsons and Belva A. Wallace, your attorney and proxy, with full power of substitution, on your behalf and with all powers you would possess if personally present, to vote all shares of Common Stock or 7% Cumulative Redeemable Preferred Stock, Series B, of Bank of America Corporation that you would be entitled to vote at the Special Meeting of Stockholders to be held in the Palmetto Ballroom of the International Trade Center, 200 North College Street, Charlotte, North Carolina, at 10:00 a.m. (local time) on March 17, 2004. The shares represented by this proxy will be voted as instructed by you. If not otherwise specified, shares will be voted in accordance with the recommendations of the Board of Directors.

Voting by Internet or telephone. If you wish to vote by Internet or telephone, please follow the instructions on the lower reverse side of this proxy card. Voting by mail. If you wish to vote by mail, please sign your name exactly as it appears on this proxy card and mark, date and return it in the enclosed envelope. When signing as attorney, executor, administrator, trustee, guardian or officer of a company or other entity, please provide your full title.

Bank of America associates. If you have shares of Common Stock or ESOP Convertible Preferred Stock, Series C, in your Bank of America 401(k) Plan account, you must provide voting instructions to the plan trustees with this proxy card or by Internet or telephone in order for these shares to be voted. Your voting instructions will be held in strict confidence.

THIS PROXY IS CONTINUED ON THE REVERSE SIDE.

PLEASE SIGN ON THE REVERSE SIDE AND RETURN PROMPTLY OR VOTE BY INTERNET OR TELEPHONE.

↑ FOLD AND DETACH HERE **↑**

Attention Internet Users!

You can now access your stockholder information on the following secured Internet site:



http://www.bankofamerica.com/shareholder

Once you establish a personal identification number (PIN), you can view your account details and perform multiple transactions such as:

View account status
View recent account activity
View tax information

Print duplicate tax forms Change your address Request dividend check replacement Change your dividend election Sell book-entry shares View stock price information

Step 1: ESTABLISH A PIN

(1st time users only)
Enter your social security number
Click on the Establish PIN button
Select a personal identification number
Follow the instructions

Step 2: LOG IN

· (Returning users)
Enter your social security number
Enter your PIN
Follow the instructions

Step 3: SELECT INFORMATION

Your Account Status screen allows you to view basic information. To view details, click on any of the buttons at the top of the page.

If you are not an Internet user and wish to contact Bank of America Shareholder Relations, you may use one of the following methods:



Call

1.800.642.9855



Mrito:

Bank of America Shareholder Relations P. O. Box 3315 South Hackensack, NJ 07606-1915

(a) Fleet Boston Financial C/O EQUISERVE TRUST COMPANY N.A. P.O. BOX 8205 EDISON, NJ 08818-9016

The EquiServe Vote by Telephone and Vote by Internet systems can be accessed 24-hours a day, seven days a week until 11:59 p.m. EST, March 16, 2004.

Voter Control Number

	Your vote is impor	tant. Pleas	se vote imm	nediately.		
http://v 2. Enter y	Vote-by-Internet to the Internet and go to www.eproxyvote.com/fbf/ our Voter Control Number listed above and follow the easy steps outlined on cured website.	OR	OR Vote-by-Telephone 1. Call toll-free 1-877-PRX-VOTE (1-877-779-8683) 2. Enter your Voter Control Number listed above and follow the easy recorded instructions.			
	If you vote over the Internet or b	oy telepho	•	do not mail	your card.	
X	Please mark votes as in this example.					2373
	The Board of Directors rec	ommends	s a vote "FO	R" Items 1 a	and 2	
Item 1 — A	APPROVAL OF THE MERGER AGREEMENT	FOR	AGAINST	ABSTAIN	Please mark here if you will be attending the Special Meeting.	
Item 2 — APPROVAL OF PROPOSAL TO ADJOURN THE FLEETBOSTON SPECIAL MEETING, IF NECESSARY, TO SOLICIT ADDITIONAL PROXIES					g.	
					Please mark here if address changes are noted on reverse side.	

Signature NOTE: Please sign your name as it appears above. When signing as attorney, executor, administrator, trustee, guardian or officer of a company or other entity, please provide your full title. If shares are held jointly, each joint owner should sign.



IF YOU WISH TO ATTEND THE SPECIAL MEETING YOU MUST BRING A FORM OF PERSONAL PHOTO IDENTIFICATION WITH YOU. IN ADDITION, IF YOU HOLD YOUR SHARES THROUGH A BANK, BROKER, OR OTHER HOLDER OF RECORD, YOU MUST BRING A LETTER FROM THE HOLDER OF RECORD THAT CONFIRMS YOUR OWNERSHIP OF THOSE SHARES OR A BROKER FORM OF LEGAL PROXY. WE RESERVE THE RIGHT TO REFUSE ADMITTANCE TO ANYONE WITHOUT PROPER PROOF OF SHARE OWNERSHIP AND PROPER PHOTO IDENTIFICATION.

FOLD AND DETACH HERE								
P	Fleet Boston Financial This Proxy is Solicited on behalf of FleetBoston Board of Directors for the Special Meeting of Stockholders to be held on March 17, 2004							
R O X Y	You, the undersigned stockholder, hereby acknowledge receipt of the Notice and Joint Proxy Statement/Prospectus dated [], and hereby appoint Marian L. Heard, Francene S. Rodgers, and Paul J. Choquette, Jr., jointly and severally, with full power of substitution to each, as proxies for you and on your behalf to attend the Special Meeting of Stockholders of FleetBoston Financial Corporation, to be held in the auditorium of the Federal Reserve Bank of Boston, 600 Atlantic Avenue, Boston, Massachusetts, on Wednesday, March 17, 2004, at 10 a.m. (local time) or any adjournment thereof, and to vote as directed hereby all stock of the Corporation which you would be entitled to vote if personally present. The shares represented by this proxy will be voted as instructed by you and in the discretion of the proxies on all other matters. If not otherwise specified, shares will be voted in accordance with the recommendations of the Board of Directors. This proxy, if properly executed and delivered, will revoke all other prior-dated proxies related to these shares.							
	Voting by Internet or telephone. If you wish to vote by Internet or telephone, please follow the instructions on the lower reverse side of this proxy card.							
	Voting by mail. If you wish to vote by mail, please sign your name exactly as it appears on this proxy card and mark, date and return it in the enclosed envelope. When signing as attorney, executor, administrator, trustee, guardian or officer of a company or other entity, please provide your full title.							
	THIS PROXY IS CONTINUED ON THE REVERSE SIDE.							
	PLEASE SIGN ON THE REVERSE	SIDE AND RETURN PR	ROMPTLY OR VOTE BY INTERNET OR TELEPH	ONE.				
HAS YOU	R ADDRESS CHANGED?		DO YOU HAVE ANY COMMENTS?					
		-						
		•						
SEE DE	EVERSE			SEE DEVEDSE				

(Continued and to be dated and signed on reverse side.)

SIDE

SIDE